SYSTEM 80+

SHUTDOWN RISK EVALUATION REPORT

DCTR 10

JULY 31, 1992

ABB-COMPUSTION ENGINEERING NUCLEAR POWER SYSTEMS WINDSOR, CONNECTICUT

9208070093 920731 PDR ADDCK 05200002 A PDR

LEGAL NOTICE

THIS REPORT WAS PREPARED AS AN ACCOUNT OF WORK SPONSORED BY ABB COMBUSTION ENGINEERING. NEITHER ABB COMBUSTION ENGINEERING NOR ANY PERSON ACTING ON ITS BEHALF:

A. MAKES ANY WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED INCLUDING THE WARRANTIES OF FITNESS FOR A PARTICULAR PURPOSE OR MERCHANTABILITY, WITH RESPECT TO THE ACCURACY, COMPLETENESS, OR USEFULNESS OF THE INFORMATION CONTAINED IN THIS REPORT, OR THAT THE USE OF ANY INFORMATION, APPARATUS, METHOD, OR PROCESS DISCLOSED IN THIS REPORT MAY NOT INFRINGE PRIVATELY OWNED RIGHTS; OR

B. ASSUMES ANY LIABILITIES WITH RESPECT TO THE USE OR FOR DAMAGES RESULTING FROM THE USE OF, ANY INFORMATION, APPARATUS, METHOD OR PROCESS DISCLOSED IN THIS REPORT.

ABSTRACT

In engineering the System 80+ Standard Plant Design, ABB-Combustion Engineering recognized the significance of addressing safety during shutdown operations. System 80+ is engineered with features that enhance shutdown safety: 1) by deliberate system engineering, equipment specification and plant arrangements for shutdown operation, 2) by mode dependent control logic that assists and limits operations, 3) by instrumentation, displays and alarms that clearly portray plant status in each mode and 4? by thorough procedural guidance and Technical Specifications that address important shutdown evolutions. This report presents these features and evaluates them in the context of the specific shutdown issues identified by the NRC. The report fulfills the ABB-Combustion Engineering commitments to the NRC to 1) provide shutdown information in support of the System 80+ Design Certification and 2) provide responses to specific RAI's on shutdown operations.

DEFINITIONS

The following definitions of terms are employed throughout this report.

DEFINITIONS

AVAILABLE (AVAILABILITY): The status of a system, structure or component that is in service or can be placed in service in a FUNCTIONAL or OPERABLE state by immediate manual or automatic actuation.

CONTAINMENT CLOSURE: The action to secure primary (PWR) or secondary (BWR; containment and its associated structures, systems, and components as a FUNCTIONAL barrier to fission product release under existing plant conditions.

CONTINGENCY PLAN: An approved plan of compensatory actions:

- To maintain DEFENSE IN DEPTH by alternate means when preoutage planning reveals that specified systems, structures or components will be unavailable;
- To restore DEFENSE IN DEPTH when system AVAILABILITY drops below the planned DEFENSE IN DEPTH during the outage;
- To minimize likelihood of a lcss of KEY SAFETY FUNCTIONS during HIGHER RISK EVOLUTIONS.

DECAY HEAT REMOVAL CAPABILITY: The ability to maintain reactor coolant system (RCS) temperature and pressure, and spent fuel pool (SFP) temperature below specified limits following a shutdown.

DEFENSE IN DEPTH: For the purpose of managing risk during shutdown, defense in depth is the concept of.

- Providing systems, structures and components to ensure backup of KEY SAFETY FUNCTIONS using redundant, alternate or diverse methods;
- Planning and scheduling outage activities in a manner that optimizes safety system AVAILABILITY;
- Providing administrative controls that support and/or supplement the above elements.

DEFUELED: All fuel assemblies have been removed from the reactor vessel and placed in the spent fuel pool or other storage facility.

DIVISION: One or more trains that share a common component, e.g. AC power. Divisions are the highest level of separation and independence.

ESTIMATED CRITICAL POSITION (ECP): A calculated set of reactor conditions and/or parameters that define a critical reactor state. (Keff = 1.0)

FUNCTIONAL (FUNCTIONALITY): The ability of a system or component to perform its intended service with considerations that applicable technical specification requirements or licensing/design basis assumptions may not be maintained.

HIGHER RISK EVOLUTIONS: Outage activities, plant configurations or conditions during shutdown where the plant is more susceptible to an event causing the loss of a key safety function.

INVENTORY CONTROL: Measures established to ensure that irradiated fuel remains covered with coolant to maintain heat transfer and shielding requirements.

KEY SAFETY FUNCTIONS: During shutdown, they are decay heat removal, inventory control, power availability, reactivity control, and containment.

MID-LOOP: PWR condition with fuel in the reactor vessel and level below the top of the hot legs at their junction to the reactor vessel.

MODE: The reactor operating state defined by reactivity, power level and coolant temperature, as follows:

MODE	REACTIVITY CONDITION, K _{eff}	<pre>% RATED THERMAL POWER</pre>	COOLANT TEMPERATURE, °F
1	≥ 0.99	> 5	≥ 350
2	≥ 0,99	≤ 5	≥ 350
3	< 0.99	NA	≥ 350
4	< 0.99	NA	$350 > T_{ave} > 210$
5	< 0.99	NA	≤ 210
6*	≤ 0.95	NA	≤ 135

OPERABLE: The ability of a system to perform its specified function with all applicable technical specifications requirements satisfied.

REACTIVITY CONTROL: Measures established to precluded inadvertent dilutions, criticalities, power excursions or losses of shutdown margin, and to predict and monitor core behavior.

REDUCED INVENTORY: PWR condition with fuel in the reactor vessel and level lower than three feet below the reactor vessel flange.

RISK MANAGEMENT: Integrated process of assessing and reducing the likelihood and/or consequences of an adverse event.

SEUTDOWN: Plant status when the reactor core is subcritical and a startup is not in progress.

STARTUP: Plant status commencing with activities to neatup the RCS above 200 degrees F and to bring the reactor core to a critical condition and up to 5% of rated thermal power.

TRAIN: A set of safety related components that perform a safety function. Trains performing redundant functions are physically, electrically, and mechanically separated to the extent necessary to insure independent performance of its safety function.

LIST OF ACRONYMS

ACC	-	Advanced Control Complex
ACC	-	Advanced Control Complex
ADV	-	Atmosphere Dump Valve
ALWR	-	Advanced Light Water Reactor
ARO	-	All Rods Out
BAST	-	Boric Acid Storage Tank
CC	-	Component Cooling
CCW	1 m 1	Component Cooling Water System
CD		Condensate
CDF	-	Core Damage Frequency
CEA	-	Control Element Assembly
CET		Core Exit Thermocouple
CI	-	Containment Isolation
CPC		Core Protection Calculator
CRO		Control Room Operator
CS	1.4	Containment Spray
CSAS	-	Containment Spray Actuation Signal
CSS	_	Containment Spray System
CVCS	-	Chemcial and Volume Control System
CW	_	Circulating Water
DG	1.0	Diesel Cenerator
DRA	1.2.1	Design Basis Accident
DRF		Design Basis Fuent
DEHLS		Double Ended Hot Log Clot
DESIS	12111	Double Ended Sustian Log Clat
DF	12.2	Decontainment Eactor
DHR	112-11-1	Decay Heat Demoval
DTAS		Discrete Indication Alarm System
DLS	1.2.1.4.	Discrete indication Alarm System
AP	100	Dreseure Differential
DUT		Direct Vessel Triestion
FAR		Direct vesser injection
EAD		Exclusion Area Boundary
ECUS	0.702.02	Emergency core cooling System
LUS		Electrical Distribution System
EF		Emergency reedwater
EFAS	-	Emergency Feedwater Actuation Signal
EFW		Emergency Fledwater
EPRI	NT ST	Electric Power Research Institute
FW		Feed Water
GIS	-	Generated Iodine Spike
HACT		Head Area Cable Tray Assembly
HCR		Human Congnitive Reliability
HJTC		Heated Junction Thermocouple
HPSI	-	High Pressure Safety Injection
HVT	-	Holdup Volume Tank
I&C	-	Instrumentation & Control
IA	-	Instrument Air
IBD	**	Inadvertent Boron dilution
ICI	-	In-Core Instrument

.

INPO	-	Institute for Nuclear Power Operations
IPSO	-	Integrated Process Status Overview
IRWST	-	In-containment Refueling Water Storage Tank
LCO	-	Limiting Condition for Operation
LER	-	Licensee Event Report
LOCA	-	Loss of Coolant Accidents
LOCV	-	Loss of Condeser Vacuum
LOP	-	Loss of Offsite Power
LTOP	-	Low Temperature Overpressure Protection
MFW		Main Feedwater
MMI		Man-Machine Interface
MPC	-	Maximum Permissible Concentration
MSIV	-	Main Steam Isolation Valve
NPSH	-	Net Positive Suction Head
NPSHA	-	Net Positive Suction Head Available
OSI	-	Operational Support Information
P&ID	-	Piping and Instrumentation Diagram
PCT	-	Peak Clad Temperature
PNS-Bus		Normal Permanent Non-Safety Bus
RAI	-	Requests for Additional Information
RC	-	Reactor Coolant
RCP	-	Reactor Coolant Pump
RCS		Reactor Coolant System
REM	-	Realistic Evaluation Model
RHR	-	Residual Heat Removal
RTD	-	Hot Leg Resistance Temperature Detector
SC	-	Shutdown Cooling
SCS	-	Shutdown Cooling System
SD	-	Safety Depressurization
SDS	-	Safety Depressurization System
SG	-	Steam Generator
SGTR	-	Steam Generator Tube Rupture
SI	-	Safety Injection
SIS	-	Safety Injection System
SIT	-	Safety Injection Tank
SRP	-	Standard Review Plan
SUFW	-	Start Up Feedwater
SW		Service Water
TB		1 ine Bypass
TS	-	Te ni al Specification
URD	-	Utility Requirements Document

-vi-

TABLE OF CONTENTS

SECTION	SUBJECT	PAGE NO.
	ABSTRACT	i
	DEFINITIONS	ii
	LIST OF ACRONYMS	v
	TABLE OF CONTENTS	vii
	LIST OF TABLES	xx
	LIST OF FIGURES	xxiii
1.0	INTRODUCTION	1-1
1.1	PURPOSE	1-1
1.2	SCOPE	1~1
1.3	BACKGROUND	1-1
2 4	SYSTEM 80+ FEATURES	1-2
2.0	SHUTDOWN RISK ISSUES	2-1
2.1	PROCEDURES	2-1
2.1.1	ISSUE	2.1-1
2.1.2	ACCEPTANCE CRITERIA	2.1-1
2.1.3	DISCUSSION	2.1-1
2.1.4	RESOLUTION	2.1-2
2.2	TECHNICAL SPECIFICATION IMPROVEMENTS	2.2-1
2.2.1	ISSUE	2.2-1
2.2.2	ACCEPTANCE CRITERIA	2.2-1
2.2.3	DISCUSSION	2.2-1
2.2.4	RESOLUTION	2.2-2

SECTION	SUBJECT	PAGE NO.
2.3	REDUCED INVENTORY OPERATION AND GL 88-17 FIXES	2.3-1
2.3.1	ISSUE	2.3-1
2.3.2	ACCEPTANCE CRITERIA	2.3-1
2.3.3	DISCUSSION	2.3-2
2.3.3.1	Instrumentation for Shutdown Operations	2.3-2
2.3.3.2	SCS Design	2.3-3
2.3.3.3	<u>Steam Generator Nozzle Dam</u> <u>Integrity</u>	2.3-4
2.3.3.4	ALTERNATE INVENTORY ADDITIONS AND DHR METHODS	2,3-6
2.3.3.5	<u>OPERATIONS</u>	2.3-8
2.3.4	RESOLUTION	2.3-9
2.4	LOSS OF DECAY HEAT REMOVAL CAPABILITY	2.4-1
2.4.1	ISSUE	2.4-1
2.4.2	ACCEPTANCE CRITERIA	2.4-1
2.4.3	DISCUSSION	2.4-1
2.4.3.1	Shutdown Event Initiation and Analyses	2.4-2
2.4.3.1.1 2.4.3.1.2 2.4.3.1.3 2.4.3.1.3.1 2.4.3.1.3.1 2.4.3.1.3.1.1 2.4.3.1.3.1.2 2.4.3.1.3.2 2.4.3.1.3.2 2.4.3.1.3.2.1 2.4.3.1.3.2.1	Introduction Resistance to Initiators Recovery from Initiators Group I Initiators Recovery During Mode 5 Recovery During Mode 6 Group II Initiators Recovery During Mode 5 Recovery During Mode 5	2.4-2 2.4-3 2.4-5 2.4-6 2.4-6 2.4-7 2.4-7 2.4-7 2.4-8

SECTION	SUBJECT	PAGE NO.
2.4.3.1.3.2.3 2.4.3.1.3.2.4 2.4.3.1.4 2.4.3.1.5	Group III Initiators Group IV Initiators Recovery Based on Plant Configuration Conclusions	2.4-8 2.4-8 2.4-8
2.4.3.2	System 80+ AC Power Availability	2.4~25
2.4.3.2.1 2.4.3.2.2 2.4.3.2.3	Introduction Discussion Conclusion	2.4-25 2.4-25 2.4-26
2.4.3.3	System 80+ Diesel Generator Availability	2.4-27
2.4.3.3.1 2.4.3.3.2 2.4.3.3.3	Introduction Discussion Conclusion	2.4-27 2.4-27 2.4-29
2.4.4	RESOLUTION	2.4-29
2.5	PRIMARY/SECONDARY CONTAINMENT CAPABILITY AND SOURCE TERM	2.5-1
2.5.1	ISSUE	2.5-1
2.5.2	ACCEPTANCE CRITERIA	2.5-2
2.5.3	DISCUSSION	2.5-2
2.5.3.1	Problem Formulation	2.5-2
2.5.3.2	Containment Integrity	2.5-2
2.5.3.2.1 2.5.3.2.1.1 2.5.3.2.1.2 2.5.3.2.1.2.1 2.5.3.2.1.2.2 2.5.3.2.1.2.2 2.5.3.2.1.3 2.5.3.2.2 2.5.3.2.2 2.5.3.2.2.1 2.5.3.2.2.1 2.5.3.2.2.3 2.5.3.2.2.4	Integrity Requirements Modes 1-4 Mode 5 RCS Level Above Reduced Inventory RCS Level Below Reduced Inventory Mode 6 System 80+ Containment Features Building Arrangement and Ventilation Personnel Locks Equipment Hatch Penetrations	2.5-3 2.5-3 2.5-3 2.5-3 2.5-3 2.5-4 2.5-4 2.5-4 2.5-4 2.5-4 2.5-4 2.5-5 2.5-6
2.5.3.3	Events Analyzed	2.5-6

SECTION	SUBJECT	PAGE NO.
2.5.3.4	Acceptance Criteria	2.5-7
2.5.3.4.1	Radiation Limits	2.5-7
2.5.3.4.1.1	Loss of Shutdown Cooling; Site	
	Boundary Limits	2.5-7
2.2.3.4.1.2	Loss of Coolant Accidents; Site	
2.5.3.4.1.3	Boundary Limits	2.5-8
2.5.3.4.1.3.1	Air Borne Radiation	2.5-8
2.5.3.4.1.3.2	Whole Body Radiation	2.5-0
2.5.3.4.2	Temperature Limits	2.5-9
2.5.3.5	Analysis	2.5-10
2.5.3.5.1	Thermodynamic Conditions	2.5-10
2.5.3.5.2	Radiation Release	2.5-11
2.5.3.5.3	Results	2.5-12
2.5.3.5.3.1	Mode 5: Loss-of-Shutdown Cooling	
2525211	at Reduced Inventory	2.5-13
2.5.3.5.3.1.2	Dite Boundaries	2.5-13
2.5.3.5.3.2	Mode 5: LOCA	2.5-13
2.5.3.5.3.2.1	Site Boundaries	2.0-14
2.5.3.5.3.2.2	Utility Personnel	2.5-14
2.5.3.5.3.3	Mode 6: Refueling; Inventory	1109.74
	Boll-Off	2.5-14
2.2.2.2.2.3.1.1	Site Boundary Limits	2.5-15
6.0.0.0.0.0.2	Utility Personnel	2.5-15
2.5.4	RESOLUTION	2.5-15
2.5.4.1	Mode 5; Reduced Inventory (Loss of	
	Shutdown Cooling)	2.5-16
2.5.4.2	Mode 5; Full Inventory (LOCA)	2.5-16
2.5.4.3	Mode 6; Refueling Configuration (Inventory Boil Off)	2.5=16
2.6	RAPID BORON DILUTION	2.6-1
2.6.1	ISSUE	2.6-1
2.6.2	ACCEPTANCE CRITERIA	2.6-1

ECTION	SUBJECT	PAGE NO.
2.6.3	DISCUSSION	2.6-2
2.6.3.1	Identification of Dilution Sources	2.6-2
2.6.3.2	Event Analyzed	2.6-2
2.6.3.3	Mathematical Model	2.6-4
2.6.3.4	Results	2.6-4
2.6.3.5	Conclusion	2.6-4
2.6.4	RESOLUTION	2.6-4
2.7	FIRE PROTECTION	2.7-1
2.7.1	ISSUE	2.7-1
2.7.2	ACCEPTANCE CRITERIA	2.7-1
2.7.3	DISCUSSION	2.7-1
2.7.3.1	Mitigation of Fire Consequences	2.7-1
2.7.3.2	Detection and Suppression of Fires	2.7-2
2.7.3.3	Prevention of Fires	2.7-3
2.7.4	RESOLUTION	2.7-4
2.8	INSTRUMENTATION	2.8-1
2.8.1	ISSUE	2.8-1
2.8.2	ACCEPTANCE CRITERIA	2.8-2
2.8.3	DISCUSSION	2.8-3
2.8.3.1	Instrumentation Design Basis	2.8-3
2.8.3.2	Instrumentation Description	2.8-4
2.8.3.2.1	Level Temperature	2.8-4

SECTION	SUBJECT	PAGE NO.
2.8.3.2.3	Shutdown Cooling System	
	Performance	2.8-6
2.8.3.2.4	Quality Assurance	2.8-7
2.0.3.2.0	Display and Monitoring	
	Capability	2.8-8
2.8.3.2.5.1	IPSO	2.8-8
2.8.3.2.5.2	Alarm Tiles and Associated	
	Alarm Messages	2.8-10
2.0.3.2.3.3	Discrete Indicators	2.8-11
2.0.2.2.2.4	CRT Display Pages	2.8-12
61012121212	Indicators	
2.8.3.2.5.6	NUPLEY 80+ Alarm Currectoristics	2.8-12
	NOT DEA 607 ATATM CHATACLETISCICS	2.8-13
2.8.4	RESOLUTION	2.8-14
2.9	ECCS RECIRCULATION CAPABILITY	2.9-1
2.9.1	ISSUE	2 0-1
		6.3-1
2.9.2	ACCEPTANCE CRITERIA	2.9-1
2.9.3	DISCUSSION	2.9-2
2.9.4	RESOLUTION	
	RESOLUTION	2.9-4
2.10	EFFECTS OF PWR UPPER INTERNALS	2.10-1
2.10.1	ISSUE	2.10-1
2.10.2	ACCEPTANCE CRITERIA	2,10-1
2 10 3	DICOURCION	
2+10+2	DISCOSSION	2.10-1
2.10.4	RESOLUTION	2.10-3
2.11	FUEL HANDLING AND HEAVY LOADS	2.11-1
2.11.1	ISSUE	2 11-1
		2.11-1
2.11.2	ACCEPTANCE CRITERIA	2.11-1

SECTION	SUBJECT	PAGE NO.
2.11.3	DISCUSSION	2.11-1
2.11.4	RESOLUTION	2.11-4
2.12	POTENTIAL FOR DRAINING THE REACTOR COOLANT SYSTEM	2.12-1
2.12.1	ISSUE	2.12-1
2.12.2	ACCEPTANCE CRITERIA	2.12-1
2.12.2.1	Prevention Criteria	2.12-1
2.12.2.2	Detection Criteria	2.12-1
2.12.2.3	Mitigation Criteria	2.12-2
2.12.3	DISCUSSION	2.12-2
2.12.3.1	Potential Drain Paths Directly from the Reactor Coolant System	2.12-4
2.12.3.2	Potential Drainage Paths Through Interfacing Systems	2,12-7
2.12.3.2.1	Potential Drainage Paths from the RCS Through the SCS	2.12.7
2.12.3.2.2	Potential Drainage Paths from the RCS to the SIS/SCS	2.12.10
2.12.3.2.3	Potential Drainage Paths from the RCS to CVCS	2 12.10
2.12.3.2.4	Potential Drainage Paths from the RCS to the SS	2.12-10
2.12.4	RESOLUTION	2.12-11
2.13	FLOODING AND SPILLS	2.13-1
2.13.1	ISSUE	2.13-1
2.13.2	ACCEPTANCE CRITERIA	2.13-1
2.13.3	DISCUSSION	2.13-1

SECTION	SUBJECT	PAGE NO.
2.13.4	RESOLUTION	2.13-2
3.0	PROBABILISTIC RISK ASSESSMENT	3-1
3.1	PRA SUMMARY AND CONCLUSION	3-1
3.2	PRA INTRODUCTION	3-4
3.3	INITIATING EVENT FREQUENCIES	3-9
3.4	ACCIDENT SEQUENCES	3-12
3.4.1	LOSS OF DHR, MODE 4	3-12
3.4.2	LOSS OF DHR, MODE 5,6, REFUELING CAVITY EMPTY	3-19
3.4.3	LOSS OF DHR, MODE 5, REDUCED INVENTORY	3-22
3.4.4	LOSS OF DHR, MODE 6, REFUELING CAVITY FLOODED	3-25
3.4.5	LOCA, MODE 4, PRESSURE ABOVE 500 PSIG	3-27
3,4,6	LOCA, MODES 5, 6 (IRWST FULL)	3-31
3.4.7	LOCA OUTSIDE CONTAINMENT, MODES 6	3-33
3.4.8	LOCA INCIDE CONTAINMENT, MODE 6 (REFUELING CAVITY FULL)	3-35
3.4.9	LOSS OF OFFCITE POWER	3-36
3.4.10	CRITICALITY EVENTS	3-36
3.5	RADIOLOGICAL CONSEQUENCES	3-57
4.0	APPLICABILITY OF CHAPTER 15 ANALYSES	4-1
4.0.1	FORMAT AND CONTENT	4-1
4.1	INCREASE IN HEAT REMOVAL BY THE SECONDARY SYSTEM	4.1-1
4.1.0	INTRODUCTION	4.1-1

SECTION	SUBJECT	FAGE NO.
4.1.1	DECREASE IN FEEDWATER TEMPERATURE	4.1-1
4.1.2	INCREASE IN FEEDWATER FLOW	4.1-1
4.1.3	INCREASE MAIN STEAM FLOW	4.1-4
4.1.4	INADVERTENT OPENING OF A STEAM GENERATOR RELIEF OR SAFETY VALVE	4.1-4
4.1.5	STEAM SYSTEM PIPING FAILURES INSIDE AND OUTSIDE CONTAINMENT	4.1-5
4.2	DECREASE IN HEAT REMOVAL BY SE ONDARY SYSTEM	4.2-1
4.2.0	INTRODUCTION	4.2-1
4.2.1	LOSS OF EXTERNAL LOAD	4.2-1
4.2.2	TURBINE TRIP	4.2-1
4.2.3	LOSS OF CONDENSER VACUUM (LOCV)	4.2-1
4.2.4	MAIN STEAM ISOLATION VALVE CLOSURE	4.2-2
4.2.5	STEAM PRESSURE REGULATOR FAILURE	4.2-2
4.2.6	LOSS OF HON-EMERGENCY AC POWER TO THE STATION	4.2-3
4.2.7	LOSS OF NORMAL FEEDWATER FLCW	4.2-3
4.2.8	FEEDWATER SYSTEM PIPE BREAKS	4.2-3
4.3	DECREASE IN REACTOR COOLANT	4.3-1
4.3.0	INTRODUCTION	4.3
4.3.1	TOTAL LOSS OF REACTOR COOLANT FLOW	4.3-1
4.3.2	FLOW CONTROLLER MALFUNCTION CAUSING FLOW COASTDOWN	4.3-2

SECTION	SUBJECT	PAGE NO.
4.3.3	SINGLE REACTOR COOLANT PUMP ROTOR SEIZURE WITH LOSS CF OFFSITE POWER	4.3~2
4.3.4	REACTOR COOLANT PUMP SHAFT BREAK WIT'I LOSS OF OFFSITE POWER	4.3-2
4.4	REACTOR COOLANT PUMP SHAFT BREAK WITH LOSS OF OFFSITE POWER	4.4-1
4.4.0	INTRODUCTION	4.4-1
4.4.1	UNCONTROLLED CONTROL ELEMENT ASSEMBLY WITHDRAWAL FROM SUBCRITICAL OR LOW POWER CONDITIONS	4.4-1
4.4.2	UNCONTROLLED CONTROL ELEMENT ASSEMBLY WITHDRAWAL AT POWER	4.4-3
4.4.3	SINGLE CONTROL ELEMENT ASSEMBLY DROP	4.4-3
4.4.4	STARTUP OF AN INACTIVE REACTOR COOLANT PUMP	4.4-3
4.4.5	FLOW CONTROLLER MALFUNCTION CAUSING AN INCREASE IN BWR CORE FLOW RATE	4.4-3
4.4.6	INADVERTENT DEBORATION	4.4-3
4.4.7	INADVERTENT LOADING OF A FUEL ASSEMBLY INTO THE IMPROPER POSITION	4.4~3
4.4.3	CONTROL ELEMENT ASSEMBLY (CEA) EJECTION	4.4-3
4.5	INCREASE IN RCS INVENTORY	4.5-1
4.5.0	INTRODUCTION	4.5-1
4.5.1	INADVERTENT OPERATION OF THE ECCS	4.5-1
4.5.2	CVCS MALFUNCTION-PRESSURIZER LEVEL CONTROL SYSTEM MALFUNCTION WITH LOSS OF OFFSITE POWER	4.5~1
4.6	DECREASE IN REACTOR COOLANT SYSTEM	1.6-1

i tan is

SECTION	SUBJECT	PAGE NO.
4.6.0	INTRODUCTION	4.6-1
4.6.1	INADVERTENT OPENING OF A PRESSURIZER SAFETY/RELIEF VALVE	4.6~1
4.6.2	DOUBLE-ENDED BREAK OF A LETDOWN LINE OUTSIDE CONTAINMENT	4.6-1
4.6.3	STEAM GENERATOR TUBE RUPTURE	4.6-1
4.6.4	RADIOLOGICAL CONSEQUENCE OF MAIN STEAM LINE FAILURE OUTSIDE CONTAINMENT (BWR)	4.6-5
4.6.5	LOSS-OF-COOLANT ACCIDENT	4.6-5
4.7	RADIOACTIVE MATERIAL RELEASE FROM A SUBSYSTEM OR COMPONENT	4.7-1
4.7.0	INTRODUCTION	4.7.1
4.7.1	RADIOACTIVE GAS WASTE SYSTEM FAILURE	4.7.1
4.7.2	RADYOACTIVE LIQUID WASTE SYSTEM LEAK OR FAILURE	4.7.1
4.7.3	POSTULATED RADIOACTIVE RELEASES DUE TO LIQUID CONTAINING TANK FAILURES	4.7.1
4.7.4	FUEL HANDLING ACCIDENT	4.7.1
4.7.5	SPENT FUEL CASK DROP ACCIDENTS	4.7.1
5.0	APPLICABILITY OF CHAPTER 6 LOCA ANALYSES TO LOWER MODES OF OPERATION	5-1
5.1	ISSUE	5-1
5.2	ACCEPTANCE CRITERIA	5-1
5.3	DISCUSSION	5-1
5.3.1	DESCRIPTION OF LOCA SCENARIO	5-4

•

Ċ,

41

45

TABLE OF CONTENTS (Cont'd)

٩.

SECTION	SUBJECT	PAGE NO.
5.3.2	SELECTION OF REVERSE PLANT PARAMETERS AND CONDITIONS FOR MODE 4 ANALYSIS	5-5
5.3.3	ANALYSIS COMPUTER CODES	5-5
5.3.4	LOC: ANALYSIS FOR MODE 4	5-6
5.3.5	RESULTS OF LOCA ANALYSIS FOR MODE 4	5-6
5.3.5.1	Results of LOCA Cases With No SI Delivery	5-7
5.3.5.2	Initiation of SI Delivery Within Ten Minutes After Break	5~8
5.4	RESOLUTION	5-9
6.0	APPLICABILITY OF CHAPTER 6 CONTAINMENT ANALYSES	6-1
6.1	INTRODUCTION	6-1
6 2	LOSS OF COOLANT ACCIDENTS (LOCAS)	6-1
6.3	MAIN STEAM LINE BREAKS (MSLBs)	6-2
6.4	INADVERTENT OPFRATION OF CONTAINMENT HEAT REMOVAL SYSTEMS	6+3
6.5	CONCLUSION	6-3
7.0	SYSTEM 804 DESIGN FEATURES FOR SIMPLICITY OF SHUTDOWN OPERATIONS	7-1
7.1	INTRODUCTION	7-1
7.2	DISCUSSION	7-1
7.2.1	TECHNICAL SPECIFICATIONS FOR REDUCED INVENTORY	7-1
7.2.2	SHUTDOWN COOLING SYSTEM	7-1
7.2.3	CONTAINMENT SPRAY SYSTEM	7-2
7.2.4	COMPONENT COOLING WATER SYSTEM	7-2

SECTION	SUBJECT	PAGE NO.
7.2.5	STATION SERVICE WATER SYSTEM	7-3
7.2.6	ELECTRICAL DISTRIBUTION SYSTEM	7-3
7.2.7	NUPLEX 80+ ADVANCED CONTROL COMPLEX	7-3
7.2.8	REDUCED INVENTORY INSTRUMENTATION	7-5
7.2.9	CONTAINMENT	7-6
7.3	CONCLUSION	7-7
8.0	CONCLUSIONS	8-1
9.0	REFERENCES	9-1
	APPENDICES	
APPENDIX A	RESPONSES TO REQUESTS FOR ADDITIONAL INFORMATION	A-1
APPENDIX B	CESSAR-DC CHANGES	B-1
APPENDIX C	PROCEDURE GUIDANCE TO SUPPORT REDUCED RCS INVENTORY OPERATIONS	C-1
APPENDIX D	TECHNICAL SPECIFICATION MARKUPS OF LCO AND APPLICABILITY FOR SHUTDOWN OPERATIONS	D-1

APPENDIX E	TECHNICAL	SPECIFICATIONS FOR	
	REDUCED R	CS INVENTORY OPERATIONS	E-1

LIST OF TABLES

TABLE	TITLE	PAGE NO.
1-1	SHUTDOWN EVENT CATEGORIES AND SYSTEM 80+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION	1-4
1-2	SHUTDOWN EVENTS AND SYSTEM 80+ PREVENTION, DETECTION AND MITIGATION FEATURES	1-9
2.1-1	SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS	2.1-3
2.2-1	SYSTEM 80+ TECHNICAL SPECIFICATIONS MODIFICATIONS RELATED TO IMPROVED SHUTDOWN OPERATIONS	2.2-3
2.4-1	SUMMARY OF SYSTEM 80+ SCS DESIGN FEATURES THAT INCREASE IMMUNITY AGAINST INITIATORS	2.4-3
2.4-2	SCS INSTRUMENTATION	2.4-11
2.4-3	TERMINATION POINT (11 SHEETS, POINTS 1 THROUGH 11)	2.4-14
2.5-1	CONTAINMENT OPENINGS	2.5-17
2.5~2	MASS-ENERGY RELEASE FOR MODE 5 LOSS OF SHUTDOWN COOLING	2.5-17
2.5-3	MASS-ENERGY RELEASE FOR MODE 5 LOCA	2.5-18
2.5-4	MASS-ENERGY RELEASE FOR MODE 6 INVENTORY EOIL OFF	2.5-19
2.5-5	PROTECTION FACTORS AND EQUIVALENT MPC	2.5-19
2.6-1	POSSIBLE FLOW PATHS OF NON-BORATED WATER	2.6-5
2.6-2	RAPID BORON DILUTION ANALYSIS - ASSUMPTIONS AND INITIAL CONDITIONS	2.6-7
2.8-1	REDUCED INVENTORY INSTRUMENTATION PACKAGE	2.8-14
2.10-1	HYDRAULIC FLOW RESISTANCE DATA FOR SYSTEM 80+	2.10-3

LIST OF TABLES

TABLE	TITLE	PAGE NO.
2.12-1	FACTORS WHICH AFFECT THE RISK ASSOCIATED WITH AN INDICATOR	2.12-3
2.12-2	POTENTIAL DRAIN PATHS DIRECTLY FROM THE REACTOR COOLANT SYSTEM	2.12-5
2.12-3	GROUPING OF PRIMARY COOLANT DRAINAGE PATHS FROM THE SCS	2.12-9
3.1-1	SUMMARY OF BRANCH POINT FAILURE RATES	3-2
5.1-2	FREQUENCY OF CORE DAMAGE FROM SHUTDOWN EV S	3-3
3.1-3	COMPARISON OF CORE DAMAGE FREQUENCIES	3-3
3.2-1	PLANT STATES AND INITIATING EVENTS	3 - 8
3.3-1	OBSERVED FREQUENCIES OF DHR EVENTS IN PWRS	3-11
3.3-2	INITIATING EVENT FREQUENCY FOR PLANT STATES	3-11
3.5-1	FRACTIONAL CORE RADIOACTIVITY FOLLOWING SHUTDOWN	3-58
3.5-2	FRACTIONAL DELAY OF SELECT RADIOACTIVE NUCLIDE GROUPS FOLLOWING SHUTDOWN	3-58
4.0-1	INITIAL CONDITIONS	4-2
4.0-2	REACTOR VESSEL INLET COOLANT FLOW RATE	4-3
5-1	SYSTEM 80+ LOCA ANALYSIS, MODE 4 INITIAL CONDITIONS	5-11
6-1	ESFAS INSTRUMENTATION	6-5
6-2	CASES ANALYZED	6-6
6-3	INITIAL CONDITIONS FOR LOCA INITIATED FROM ZERO POWER	6-7

63

LIST OF TABLES

TABLE	TITLE	PAGE NO.
6-4	ACCIDENT CHRONOLOGY FOR LOCA INITIATED FROM ZERO POWER	6-8
6-5	INITIAL CONDITIONS FOR MSLB INITIATED FROM MODE 5	6-9
6-6	ACCIDENT CHRONOLOGY FOR MSLB INITIATED FROM MODE 5	6-10

.

1

FIGURE	SUBJECT	PAGE NO.
2.4-1	PLANT STATES AND TERMINATION POINTS FOR RESTORATION OF DHR	2.4-10
2.4~2	SAFETY INJECTION PIPING AND INSTRUMENTATION DIAGRAMS (SHEET 1 OF 3)	2.4-31
2.4-3	SAFETY INJECTION PIPING AND INSTRUMENTATION DIAGRAMS (SHEET 2 OF 3)	2.4-32
2.4-4	SAFETY INJECTION PIPING AND INSTRUMENTATION DIAGRAMS (SHEET 3 OF 3)	2.4.33
2.4-5	SYSTEM 80+ ELECTRICAL DISTRIBUTION	2.4-34
2.4-6	DIESEL LOAD SEQUENCER - SIMPLIFIED LOGIC DIAGRAM	2.4-35
2.5-1A	PERSONNEL LOCK; LEVEL 115	2.5-20
2.5-1B	EQUIPMENT HATCH; LEVEL 146	2.5-21
2.5-2	EVENT TIMES FOR CONTAINMENT CLOSURE	2.5-22
2.5~3	LIMITS OF CONTAINMENT CLOSURE TIME	2.5-23
2.5-4A	MODE 5 SDC, CONTAINMENT TEMP VS TIME	2.5-24
2.5-48	MODE 5 SDC, WHOLE BODY DOSE VS TIME	2.5-25
2.5-4C	MODE 5 SDC, THYROID DOSE VS TIME	2.5-26
2.5-4D	MODE 5 SDC, MPC VS TIME	2.5-27
2.5-4E	MODE 5 SDC, UTILITY PERSONNEL WHOLE BODY DOSE VS TIME	2.5-28
2.5-5A	MODE 5 LOCA, CONTAINMENT TEMPERATURE VS TIME	2.5-29
2.5-5B	MODE 5 LOCA, WHOLE BODY DOSE VS TIME	2.5-30
2.5-5C	MODE 5 LOCA, THYROID DOSE VS TIME	2.5-31
2.5-5D	MODE 5 LO , MPC VS TIME	2.5-32

LIST OF FIGURES

12

FIGURE	SUBJECT	PAGE NO.
2.5-5E	MODE 5 LOCA, UTILITY PERSONNEL WHOLE BODY DOSE VS TIME	2.5-33
2.5-6A	MODE 6 BOIL OFF, CONTAINMENT TEMP VS TIME	2.5-34
2.5-6B	MODE 6 BOIL OFF, WHOLE BODY DOSE VS TIME	2.5-35
2.5-6C	MODE 6 BOIL OFF, THYROID DOSE VS TIME	2.5-36
2.5-6D	MODE 6 BOIL OFF, MPC VS TIME	2.5-37
2.5-6E	MODE 6 BOIL OFF, UTILITY PERSONNEL WHOLE BODY DOSE VS TIME	2.5-38
2.7-1	NUCLEAR ISLAND FIRE BARRIER LOCATIONS PLANT AT ELEVATION 50+0	2.7-6
2.7-2	NUCLEAR ISLAND FIRE BARRIER LOCATIONS PLANT AT ELEVATION 70+0	2.7-7
2.8-1	REACTOR COOLANT SYSTEM ELEVATIONS RELATED TO SHUTDOWN COOLING OPERATIONS	2.8-18
2.8-2	DIFFERENTIAL PRESSURE INSTRUMENT TAP LOCATIONS - SCHEMATIC	2.8-19
2.8-3	SCHEMATIC REPRESENTATION OF THE INADEQUATE CORE COOLING, HJTC PROBES	2.8-20
2.8-4	SCHEMATIC REPRESENTATION OF THE NARROW-RANGE HEATED JUNCTION THERMOCOUPLE PROBES	2.8-21
2.9-1	LOCATIONS OF SAFETY INJECTION SYSTEM SUCTION IN IRWST	2.9-6
2.9-2	LOCATION OF WING WALL DEBRIS SCREEN ASSEMBLIES	2.9-7
2.9-3	LOCATIONS OF TRASH RACK AND SPILLWAY FOR IRWST AND HVT	2.9-8

П

FIGURE	SUBJECT	PAGE NO.
2.10-1	RESULTS OF MODE 6 NATURAL C) CULATION ANALYSIS OF SYSTEM 80+	2.10-4
2.11-1	CONTAINMENT BUILDING LOAD HANDLING PATHS	2.11-5
2.11-2	FUEL HANDLING BUILDING LOAD HANDLING PAINS	2.11-6
2.12-1	POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS FROM SCS TRAIN NO. 1 2	2.12-13
2.12-2	POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS FROM SIS/SCS DVI NOZZLE 1A 2	.12.18
2.12-3	POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS FROM CVCS 2	.12-20
2.12-4	POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS FROM PSS 2	.12-24
2.13-1	NUCLEAR ISLAND DETAILED ARRANGEMENT PLAN AT EL. 50+0	2.13-3
2.13-2	NUCLEAR ISLAND DETAILED ARRANGEMENT PLAN AT EL. 70+0	2.13-4
3.4-1	EVENT TREE FOR LOSS OF DHR, MODE 4	3-38
3.4-2	OPERATOR FAILS TO RESTORE THE OPERATIN TRAIN OF THE SHUTDOWN COOLING SYSTEM	G 3-39
3.4-3	OPERATOR FAILS TO START THE REDUNDANT TRAIN OF THE SHUTDOWN COOLING SYSTEM	3-40
3.4-4	OPERATOR FAILS TO MAKE USE OF THE STEA GENERATOR AS EMERGENCY DECAY HEAT REMOVA	M 3-41
3.4-5	OPERATO S ALS TO INITIATE FEED AND BLEED COOLING	3-42
3.4-6	OPERATOR FAILS TO INJECT INVENTORY FROM THE BORIC ACID STORAGE TANK TO THE RCS	M . 3-43

FIGURE	SUBJECT	PAGE NO.
C.4+7	EVENT THREE FOR LOSS OF DHR, MODE 5, REFUELING CAN EMPTY	6 3-44
3.4-8	OPERATOR FAILS TO INITIATE FEED AND BLEED COOLING	3-45
3.4-9	EVENT TREE FOR LOSS OF DHR, MODE 5 REDUCED INVENTORY	3-46
3.4-10	OPERATOR FAILS TO INITIATE MAKEUP UTILIZING THE SHJTDOWN COOLING SYSTEM	3-47
3.4-11	EVENT TREE FOR LOSS OF DHR, MODE 6 REFUELING CAVITY FULL	3-48
3.4-12	EVENT TREE FOR LOCA, MODE 4	3-49
3.4-13	OPERATOR FAILS TO CONTINUE TO MAKE USE OF STEAM GENERATORS AS EMERGENCY DECAY HEAT REMOVAL	3-50
3.4-14	OPERATOR FAILS TO INITIATL SAFETY DEPRESSURIZATION	3-51
3.4-15	EVENT TREE FOR LOCA MODES 5, 6 (IN CONTAINMENT LOCA)	3~52
3.4-16	OPERATOR FAILS TO MANUALLY INITIATE SAFETY INJECTION AFTER THE EVACUATION OF CONTAINMENT	3-53
3.4-17	EVENT TREE FOI. OCA, MODE 6 LOCA OUTSIDE CONTAINMENT	3-54
3.4-18	OPERATOR FAILS TO ISOLATE THE LEAK AT THE CONTAINMENT BOUNDARY	3-55
3.4-19	EVENT TREE FOR LOCA, MODE 6, LOCA IN CONTAINTMENT	3-56
4.0-1	INITIAL CONDITIONS FOR PRESSURE AND TEMPERATURE IN MODES 3 THROUGH 6	4-4
4.1-'	MAXIMUM POWER VS STEAM LINE BREAK AREA	4.1-8

FIGURE	SUBJECT	PAGE NO.
5-1	CLAD SURFACE TEMPERATURE, SAFETY INJECTION, INITIATED AT 10 MINUTES, DVI LINE BREAK	5-12
5-2	PRESSURIZER PRESSURE, NO SAFETY INJECTION, DVI LINE BREAK	5-13
5-3	REACTOR VESSEL MIXTURE HEIGHT, NO SAFETY INJECTION, DVI LINF BREAK	5-14
5-4	CLAD SURFACE TEMPERATURE, NO SAFETY INJECTIONS, DVI LINE BREAK	5-15
5-5	REACTOR VESSEL MIXTURE HEIGHT, THREE EREAK LOCATIONS	5-16
5-6	REACTOR VESSEL MIXTURE HEIGHT, THREE BREAK LOCATIONS COMPARED	5-17
5-7	COMPARISON OF HOT SPOT CLADDING TEMPERATURE, DVI LINE BREAK, WITH AND WITHOUT SAFETY INJECTION	5-18
6-1	CONTAINMENT PRESSURE VS. TIME FOR LOCA FROM ZERO POWER	6-11
6-2	CONTAINMENT ATMOSPHERE TEMPERATURE VS. TIME FOR LOCA FROM ZERO POWER	6-12
6-3	CONTAINMENT PRESSURE VS. TIME FOR MSLE FROM MODE 5	6-13
5-4	CONTAINMENT ATMOSPHERE VS. TIME FOR MSLB FROM MODE 5	6-14

THIS PAGE INTENTIONALLY BLANK

1.0 INTRODUCTION

1.1 PURPOSE

This report presents features of the System 80+ design which address the issues of shutdown risk. It further evaluates these features with respect to their ability to reduce and/or mitigate the consequences of this risk. It fulfills the commitment made to the NRC by ABB-Combustion Engineering (ABB-CE) in Reference 1 to submit shutdown risk information in support of the System 80+ Design Certification.

1.2 SCOPE

Sections 2.1 through 2.13 present detailed discussions on the specific shutdown issues. Following the detailed discussions of these shutdown risk issues, the report provides a probabilistic risk assessment in Section 3.0. This is followed in Sections 4.0, 5.0 and 6.0 by an evaluation of the applicability of the analyses in CESSAR-DC Chapters 6 and 15 to LOCA and accident events that are initiated from shutdown modes. Section 7.0 evaluates the features of System 80+ that simplify shutdown operations and thereby reduce the potential for initiating shutdown events. Conclusions of this report are provided in Section 8.0. The scope of the information presented was discussed with the NRC at a presentation by ABB-CE in Rockville, Maryland on December 18, 1991 and is outlined by the ABB-CE slides enclosed with the NRC minutes of the meeting in Reference 2.

The report also addresses the RAI's from the NRC staff on CESSAR-DC that pertain to shutdown risk. Appendix A of this report lists the RAI's and provides either the response or a referral to sections of the report which encompass the response to each RAI.

1.3 BACKGROUND

In Generic Letter No. 88-17 (Reference 4) the NRC issued recommendations to all holders of licenses for PWR's to implement certain "expeditious actions" before operating their plants in a reduced inventory condition and to implement, as soon as practical, "program enhancements" concerning operations during shutdown cooling. The objective was to prevent the reoccurrence of events that had occurred and that had the potential for core damage and/or release of radiation. In NUREG-1449, NRC staff evaluations of shutdown operations indicate that recommendations have been implemented and/or are underway at operating plants.

1.4 SYSTEM 804 FEATURES

In this section, a comparison is made between the characteristics of past events and the System 80+ design features. The categories

of shutdown events at operating plants are those used by the NRC in Chapter 2 of NUREG-1449 with little modification. These categories are mostly the same as the issues identified by Secy-91-283 and presented by ABB-CE at the December 18, 1991 meeting with the NRC (Reference 2). Each category encompasses a group of similar events that have in common the type of event initiator. Ultimately, if left unmitigated by automatic or manual actions, all events might eventually lead to over heating and/or physical damage to fuel with consequent radiation release, but each scenario sequence may differ. Depending upon the importance placed on each step in a sequence, the same events could be grouped differently. For example, the NUREG-1449 category, "Loss of Shutdown Cooling", includes the issues listed in Reference 2 as 1) Mid-Loop Operation, 2) Loss of Decay Heat Removal Capability and 3) Effect of PWR Upper Internals.

The categories employed in this section to group past events encompass (and for some categories are identical to) the issues which are listed by Reference 2 and which are presented in detail in this report with a few exceptions. The exceptions apply to postulated LOCA events initiated at high pressure and other significant events initiated at high pressure for which we do not have actual experience because they have not occurred in operating plants. They exist only as analyses for use as guidance to avoid the physical event and therefore are not included in the categories of past events.

Past events are grouped into the following ten categories:

- Loss of shutdown cooling
- Loss of electrical power
- Loss of reactor coolant
- Containment integrity
- Overpressurization
- Flooding and spills
- Boron and reactivity events
- Fire protection
- Heavy loads and fuel handling
- Mode change events

For each past event placed into a category an initiator is identified. The plant design objective is to prevent the occurrence of the event initiator, but realistically, absolute prevention is impracticable and may be impossible. A combination of prevention and mitigatic is employed in the System 80+ design.

Table 1-1 provides an overview of the System 80+ features that avoid core damage during shutdown operating modes. It lists the ten shutdown event categories and for each category it lists event initiators for past events. These initiators are presented in a generic fashion; each initiator representing many specific events thave occurred. For each initiator, the features of the System set lesign that are available to prevent occurrence of an initiator sta/or to mitigate the consequences of an initiator are listed.

Table 1-2 provides a list of specific past events initiated from shutdow. modes. Events were selected to include all ten event categories and all types of event initiators, but not all similar significant events that have occurred. Several information sources were utilized to compile this list. They include events listed in NUREC-1449 which were taken from the 1990 AEOD report (Reference 5) and which occurred mostly between January 1988 and July 1990 with some additional events. For events since July 1990, ABB-CE searched the INPO database for LER's using a selection of keywords pertinent to the ten event categories and to shutdown operation. Various other INPO and NSAC documents were also reviewed for significant event reports dating from 1976 to 1990.

Events in Table 1-2 are grouped into the ten categories given above. For each specific event, the features of the System 80+ design that apply to prevent and mitigate the event are indicated. A review of this table serves as a design review of the System 80+ capabilities to avoid core damage and/or significant radiation release during shutdown modes. The design features are discussed in more detail in the following Sections of this report. 1.)

TABLE 1-1

SHUTDOWN EVENT CATEGORIES AND SYSTEM 80+ FEALURES FOR PREVENTION, DETECTION AND MITIGATION

EVENT CATEGORY	EVENT INITIATOR		SYSTEM 8J+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION
Loss of Shutdown Cooling	SCS flow loss by pumo suction vortex.	(A)	Mid-loop level meximized by locating SCS suction piping at the bottom of the hot leg.
		6)	Hard piped venting for SCS pumps relieves gas binding more quickly and conveniently. There are loop seals in the suction lines.
		C)	One SCS suction line from each hot leg provides SCS redundancy with separation of pump suction sources.
		D)	Containment spray pumps interchangeable with SCS pumps provide redundant capacity and may take suction from IRWST to refull RCS and to mitigate gas binding.
	Inaccurate mid-loop level leading to suction vortex.	E)	With head on, reactor vessel level monitoring system level indications from vessel head to a 'evel below that required for SUS operation. Level indication is accurate for intended use.
		F)	Core exit thenacouples monitor coolant temperature down to 100°F prior to withdrawal of CETs prior to fuel shuffling. The RTDs and SCS temperatures are accurate during SCS operation.
		6)	With head off, level indication near hot leg elevation is provided by high resolution instruments.
		H)	SCS performance monitored on each of 2 SCS pumps by pump motor current, flow rate, discharge pressure and suction pressure. Possible SCS flow variance with decay heat to minimize potential for vortexing during mid-loop.
	Loss of flow while head off, upper internals in vessel and cavity flooded leads to core heatup.	13	Internals design limits coolant flow from cavity to core, however, high availability of SCS system and/or backups assures forced convection.
	Various low level and loss of RHR events.	33	Non-shared SCS system allows SCS maintenance and testing during Modes 1-4 prior to cold shutdown, increasing availability in Modes 5 and 6.
		K)	All SCS valves are motor operated, preventing failures on loss of mir if electro-pneumatic operators were used.

TABLE 1-1 (Continued)

SHUTDOWN EVENT CATEGORIES AND SYSTEM 80+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION

EVENT CATEGORY

EVENT INITIATOR

SYSTEM 80+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION

- L) Shutdown specific control room displays, tech specs, and procedural guidance reduces likelihood of personnel errors.
 - N) Inadvertent errors are reduced and early operator evaluation of failures is improved by 1.) IPSD overview display with critical function and system status specific to shutdown modes, 2.1 CRT displays with system lineups and component status and 3.) alarms that are dependent on plant mode and equipment status.
 - N) Prevention of inappropriate automatic actions from personnel errors by shutdown specific control logic (e.g., remove autoclosure interlocks from SCS suction valves.)
 - CCW availability is increased by 2 redundant Divisions, each with two pumps and heat exchangers.
 - P) Service water availability is increased by 2 Redundant Divisions, each with two pumps.
 - Q) Each SCS Division has four potential sources of AC power for increased availability.
 - A: Aiternate AC gas turbine provides third on-site power source.
 - B) Two switchyard interfaces provide flexibility.
 - C) Shutdown specific tech specs and procedural guidance reduce likelihood of personnel errors.
 - D) A reserve transformer provides an alternate supply to the safety bus if the normal source (unit auxiliary transformer) is de-energized.
 - E) Each safety division has a dedicated diesel generator.
 - F) No equipment is shared between diesels.
 - G) No equipment is shared with another unit.

1.) Loss of Shutdown Cooling (Continued)

2.) Loss of Electric Power

Equipment failure and/or inadvertent personnel error leading to loss of power and shutdown cooling

TABLE 1-1 (Continued)

SHUTDOWN EVENT CATEGORIES AND SYSTEM 90+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION

	EVENT CATEGORY	EVENT INITIATOR		SYSTEM 20+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION
3.2	Loss of Reactor Coolant	From shutdown more, equipment failure and/or personnel error leads to loss of coolant, usually through systems connected to RCS.	A2	Inadvertent errors are reduced and early operator evaluation of failures is improved by 1.) IPSO overview display with critical function and system status specific to shutdown modes 2.) CRT displays with system lineups and component status and 3.) alarms that are dependent on plant mode and equipment status.
		Inadvertent RPV pressurization while connected systems are open causing coolant level drop in vessel.	8)	Removal of repressurizer manway will not allow significant RV head pressurization. Thus, instruments are not affected.
			C)	In-core instrument seal table evolutions are prohibited by procedural guidance while vessel head is on and mid-loop evolutions are in progress, preventing seal leaks.
			03	Coulant loss via RCP during seal maintenance reduced by pump impeller weight creating seal.
		Cavity draining exposes fuel being transferred.	E)	Cavity draining limited by reinforced pool seal between vessel flange and cavity floor.
			£χ	Containment layout prevents total draining if seal fails.
4.)	Containment Integrity	Loss of shutdown cooling and/or loss of reactor coolant results in core boiling requiring rapid containment closure to prevent radiological release.	A)	Tech spec requires hatch and all penetrations closed during mid-loop evolutions. Containment configuration and size allow more outage activities within containment, resulting in less time without containment integrity.
			83	Redundancy in SCS system, electric power supply and support systems together with increased instrumentation reduce likelihood of an initiating event progressing to boiling.
		Personnel errors result in opening pathways from containment to atmosphere during shutdown evolutions.	0	Shutdown specific control room displays, tech specs, and procedural guidance reduce likelihood of personnel error.
5.)	Overpressurization	Inadvertent high pressure safety injection acturtion at low temperature pressurizes RCS and SCS system.	A2	SCS system relief values sized for maximum safety injection liquid flow.
5.7	>.) Overpressurization (Continued)		8)	RCS is vented through the pressurizer menway.
TABLE 1-1 (Continued)

SHUTDOWN EVENT CATEGORIES AND SYSTEM 80+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION

	EVENT CATEGORY	EVENT INITIATOR		SYSTEM 80+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION
			C2	Ring forged reactor vessel beitline and vessel material provide additional margin to pressurized thermal shock.
6.)	Flooding and Spills	Uncontrolled coolent flow from opened systems, typically combined with other inadvertent and/or poorly planned evolutions, floods essential equipment.	A)	Inadvertent errors are reduced and early op-rator evaluation of failures is improved by 1.) IPSO overview display with critical function and system status specific to shutdown modes 2.) CRT displays with system lineups ind component status and 3.) alarms that are dependent on plant mode and equipment status.
			8)	Shutdown specific control room displays, tech specs, and procedural guidance reduce likelihood of personnel errors.
			C)	Plant layout, including separation of redundant divisions, limits damage that may occur to affected division. No communication between divisions, including piping, electrical, HVAC, floor drains, etc.
7.3	Boron and Reactivity Events	Various CVCS misoperations and uncalibrated source range neutron monitors cause approach to criticality.	A)	Shutdown specific control room displays, tech specs, and procedural guidance reduce likelihood of improper operation.
		CVCS misoperation causes boron dilution or potential boron precipitation.	8)	Precipitation prevented by design that limits brown concentration to below cold precipitation concentration in most borated coolant lines, eliminating reed for most heat tracing.
			C3	Boron dilution alarm provides advanced warning.
1.)	Fire Protection	During shutdown evolutions, use of combustible materials plus ignition sources such as temporary power lines increases potential for fire damage to essential systems.	A)	Plant layout and fire berriers separate redundant divisions and systems to limit potential fire damage.
			8)	Combustible materials are limited in specific fire control areas.
2.3	Heavy Loads and Fuel Handling	Inadequate design and/or surveillance of lifting devices causes potential damage to fuel or essential equipment.	A)	Shutdown specific guidance limits pathways for heavy lifts.

10.3

TABLE 1-1 (Continued)

SHUTDOWN EVENT CATEGORIES AND SYSTEM BD+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION

EVENT CATEGORY	EVENT INITIATOR		SYSTEM 80+ FEATURES FOR PREVENTION, DETECTION AND MITIGATION
		8)	Plant arrangement minimizes potential for damaging drops.
		a	Proven design for fuel, core arrangement and fuel handling machine minimizes potential fuel drop.
Mode Change Events	Operator and/or procedural errors allow mode changes without satisfying entry requirements.	A2	Shutdown specific control room displays, tech specs, and procedural guidance reduce likelihood of personnel errors.
		83	Inadverient errors are reduced and early operator evaluation of failures is improved by 1.) IPSO overview display with critical function and system status specific to shutdown modes 2.) CRT displays with system lineups and component

equipment status.

2-1 1000

Sapitone poters and System Bo-Paperation, Britts-10m, and Writsaring, Flangers

system son versenies, beterlien, and mitgarise startes 1585 table (-1)	M, Ll, PA	M, 10, 17, 13, 28	ж. Ж	8. 15. 19	14. 15. 19. 19. 19. 19. 19. 19. 19.	24, 13, 11, 13, 18, 18	14, 13, 11, 11, 11, 19, 19
DATE EVENT SUMMARY	<pre>i2/07/81 IN MODE 5, SDC PLMP INIPPED DUE ID TESTIME 3M0 PCS TEMPERATURE ADDE 118 RG F TO 20% DES F 18 A0007 20 MIMUTES. MEALUP AMD COOLDOMM LIMITS WERE ETCEEDED. MODE 4 MAS EMIEDED WITHOUT SATISFING MEDE 4 LCD 5.</pre>	01/16/87 IN MORE 5, VITAL ROS MAS DE FENELIED CARSING LOSS NE FINANG TO I COM PRIME AND 2 SERVICE MATER FUMA-S. THE DIMAR CCS AND SERVICE ANTER FUMA-S MERE DUT OF SERVICE FOR MAINTENANCE. TAUS, A COMPLETE LOSS OF CCM AND SERVICE ANTER RESULTED. IN TOWALSTELEDS OF CCM AND SERVICE ANTER RESULTED. IN TOWALSTELEDS EFFECT MAS TO CARSE ALL CHARGENE PUMPE, ADDR INTERTION VATING, ROM TONING, AND DUG 5 TO BE INCRETAND ANTING, RUE MAS RESULTED ANTER, ANTER REFECT ANS TO CARSE ALL CHARGENE PUMPE, ADDR INTERTION VATING, RUE MAS RESULATED ANTING IN RESULTION VATING, RUE MAS RESULATED ANTING IN RESULATION VATING, RUE MAS RESULATED ANTING IN RESULATION FOR COMPLEX VIEW RESULTING ANTING AND RUE SERVICES ANTING RESULATION VATING, RUE MAS RESULATED ANTING INFORMATION RESULATED AND RUE AND RUE AND RUE AND RUE STORED ANTING IN RUESTION RUE SAL. COM RESIDER 2 AND RUES TARE.	09/15/02 DUDTAG COLD SAULDOWN ADVILE TE3-1960 DUCAPPOSSURE 940715/12090 SYSTEMS FLOW AND FROM THE DUCAPPOSSURE 1841641.00715 CANSED THE BANA SUCTION ISOLUTION VALUE 10 INDUCETEMILY (LOSE RECARSE OF INVOLUDET TEST 9407550.062, VESULTING IN LUSSS OF BANA FLOW.	nk/29/85 IMADEQUATE COMMUNICATION AND MONE PMOREDUMES DOMENUE REFUELING ONTAGE CANOGED LINGS OF ELECTRICAL POMER 10 THE RUMAINS DAM PUMP.	01/00/34 SF-FUELED, with BOIN THE MAIR TRANSFORMER AND DWE SF THE DAYS BUT OF SERVICE, A PHORLEM INVELOPED DM AN INCOMING ELECTRICAL FEEDER. SINCE DME DAS WAS DNI OF SERVICE, INE FLAM SHORLEN DMIT AND DISTORMEETED FROM THE PROBLEM FEEDER. DMENNER, INE DISCOMMEETED FROM THE PROBLEM FOR THE FLAM THE PROBLEM. DISCOMMEETED FROM THE PROBLEM FOR THE PLANT WAS FORMER, INE DISCOMMEETED FROM THE FLAM SHOLD THE PLANT WAS FORMERD, INE DISCOMMEETED FROM THE PLANT WAS FORMERD, DISC. LACK DF SA CANSED LOSS OF RUMMERD DAE AND LOSS OF ALL AC.	05/21/84 IMADESHATE PROTEDURE ALLOWED STEAHING SECOND RMS Power MEFONE STOPFIME IME ProMMIAME PLAN , LEADING 10 VOMTETING AMD ALM BINDING OF SHAE PLANFS.	<pre>chv1av3a LOSS OF BNBP DCLUBDDERCE WHILE MATER LEVEL was REINE LONGFED. PERSONNEL Enable with withOFER Funds Lass VALVE CAUSED INSCRUDINTE LEVEL INDICATION LEADING</pre>
EVENT CATEGORY	LIDSS OF SARTDOWN CADLING	LOSS OF SHITTDOWN COOLJING	LOSS OF SHOTFOWM CONLINE	1 LODS OF SHUTPOWN CODLING	1961 100 Michael 100 196	LOSS OF SHUTDOWE COSLING	LIDES OF SHIFTDOWN CODLINES
FLANT NUM	ALLSTONE 2	Kene	N.C. SUMER	BEAVER VALLEY	Pair Isabits	8.5. 0998 2	1 10012
(D. KC.	(1) HOM	(1) 3100	395/82008	74/83029	(E) 310e	536/84014	202/94032

(1) MD LICEWSE 2-24M MEPORT LIER: MEMBER AVAILABLE. THIS SUGNET SUMMARY MAD DBFAINED FROM ANDITHER SUBSEE OF INSTRUMETION.

The state

26/ 52/100

tant 1-2

5

4

SAUTOPAN ENENTS AND STSTEM BD+ PAGTEMITING, DETECTION, AND MITTEATION FEATURES Editaria E

SUSTEM ROM PARCHEMIJON, DETECTION, AND MITIGACING (SEE TARKE 1-1)		14. 18. 15. 19. 16. 19. 1M	94, 18, 1C, 13, 16, 16, 18, 18, 18	IC, 10, 19, 19	14, 18, 15, 10, 15, 16, 14, 1#	14, 15, 10, 15, 18, 18, 1#	18. 12, 11. 2*	14, 18, 15, 15, 18, 19, 18
AREAN SUMMERY	T3 CANITATION IN ANA POWE.	04/22/VS INAGCUBATE (EVEL IMDICATION DUSING CARINING IN MUTE S LAUGED LOSS OF ANN PURST SUCTION. PENSUMMER EXECT ALLONED SECOND RHN THAIN TO RE DUMM FOR MALING MELINGE WHILE DADINING, MAS IN PROSMESS. CHARACING RESIDNES LEVEL, THE OPERATE FUND WAS VEWIED AND ROM RESIDNED.	12/14/95 SHIFDOWN CODLING ANS LOST WHILE THE FLAMT MAS IN COLD SHIFDOWN. IT MAS ATTAIROTER TO VODFLETNG. THE CAUGH OF THE EVENT MAS IDENTIFIES TO RE INADECUALF PRODUCTORE CONFLED WITH LEVEL REASONEMENT AND THE LACH OF PROMULEDEE AND/OT MANT COMPILIONS. DS AT MANT POINT, UDBILING COM ACCED.	02/02/86 AT MIR-100F, RHA PUMP SHAFT SADEF AFTER (CUNTREDUC) DEFENDING FOR ARGUIT OF SANE, PLACING SECHND RHP PUMP IN OFFICIEN WAS DELATED RECAUSE A TRIPPES PROFERE WAS FOWERING THE SECTION VALUE. TENPERATING REGIS 13 DES 7 70 131 DEF F.	07/14/755 MEAR 5/3-400P, & SKAIM PATH WAS MET CLUSED AMD 15/961 BODPFED, DFEBATINE RAM PANN RESAM TE CANTIALE AMD WAS SHUTDOWN, AND 1051 FEVERAL TIMES DVER A 5 AMD 1/2 HOUR FEATOD, ACS TENVERATORE ADDE VA DEG F 10 222 DEG F.	04/10/PT AF MID-162P, 7 DAYS AFTER SAUTODAM WITH HEAD VEWTED, A LEAN CAUSED LOSS W ACS LEVEL AND 146 AND PLANE DECAME ALLS AFTER SAUTOPL AND AND AND VOLA, 28 MINUTES, ACS TENERATION FORSE OF DEG F TO 220 DEG F AND PRESSURE INDERAGED TO 10 9510.	09/11/90 LOSS OF SHOTPOWW CODULWE OCCUPATED WHILE IN MODE 5 WHEN NOW-LICENSED DEFENDION ECONE CONTOLED WHILE RACKING RECAVERS TREPERATOR ECONE TAKENES WILE POWER TO THE LOW PRESSNE TALECTION DUMPS, CODLANT TERMERATION LATERATOR TO DEFREES.	09/19/09 SMUTDOWN CONLING MAS LOST MNILE THE PLANF WAS IN THE REFUELTHE MODE. IT WAS ATTENUTED TO HHAP THE CANTENTIAL AUGUST BY EMTALMWENT OF ALMS THE PLANF DAVITATION DAL TO VENTELING MALEN ADMITTED ATR WHEN THE REACTOR VESSEL MALED LEVEL LONGRED MELOW THE TOP OF THE REACTOR COOLANT HOT LESS. MALEN
EVENT CATEDRA		LISS OF SHITTOWN COLLING	LIDES OF SHIFTIOWN CODILING	1 LOSS OF SAUTDOWN COOL ING	LOSS OF SAUTODAK CODLING	2 LOSS OF SHOTDOWN COR ING	LOSS OF SHUTDOWN COOLING	LIDES OF SHRITTOOMN COOLING
FLART NAME		Catewee 1	2 1001	CRYSTAL BINER	1930) 340 343 34 34	DIARLE LAWEDN	COMEL 3	1 80844
(£9, MC.		52058/CT#	RZU-SR/ MUS	342/86003	ND15 111	325/87905	\$00887282	454/88007

100

(1) MO LICENSEE EVENT REPORT (LER) MEMBER AVAILABLE. THIS EVENT STREMMEN MAS DETAINED FROM ANOTHER SUBJEE OF INFORMATION.

5

F

2

TARLE 1-2

SAUTODAM EVENTS AND STSTEM BOA PAEVENTION, DETETTION, AND MITTGATTON FEATURES

SYSTEM U.P. PREVENCION, DETECTION, AND MITIGATING FEATURES (SEE TEALE 1-1)		12 14	18, 10, 18, 19, 19, 19 Ja	13, 11, 13, 19, 19	16, 13, 19, 19, 18	15. 13	R. JA, 18
DATE EVENT SUMMARY	LEVEL MOS MCMUTIDRED BY VISUBALE BASERVING LEVEL IN REFILTING CANTTY, WALCH WASS & INCOMES AND/OF THE SLOWE WITH INCOMESED.	10/25/99 Practiones Energe Causers Loss of Power To the Decommentation of the Technic Montal Removal-Context Outlet watures Resultine IN Meal Remova-Context Outlet Watures Resultine In Loss of SAMIRONM CONTEXT ON TALLINGS OF POWER, THE VARIES DE SAMIRONM EDRO, MAG. INVESTIGATION REVEALS TO THE VALVES MER RESISEND TO TALLINGS OF POWER, WEET DITHON THE VALVES MERCINIC POEDMITIC POETINOWER DITHON THE VALVES MERCINIC POEDMITIC POETINOWER DITHON THE VALVES WEEKDERDED CAUSING THE VALVES TO THAT LINES MAD BEEN REVEADED CAUSING THE VALVES TO THAT CLOSED* ON LOSS OF POWER. CONLANT TIMOFEADINGE THEOREAGED TO REGREES.	<pre>11/221-58 (# MORE &, and Pump Lost its SUCTION PRESSURE NECARSE IT AECAME AIR ROMAD WHILE STRONE ITSTING UNLUE IN SPART SSITEM AIR SUMM DELTERING ALCONING UNLUE IN SPART SSITEM AIR SUMM SUCTION ALCONING AIR FRANTED IN MERSEOMATE VERSEO MONIZIMENT PIPTING TO AE FRUEED INTO PANE STETEM. THE PUMP MAS MONUMLLY STOPPED LAUSSING A LOSS OF AND. THE REALTER CORLAWT SYSTEM TEMP EMOREASED FROM NO DEG F TO 116 DEG F IM 30 MINUTES.</pre>	12/19/18/88 APLAY FARTALEM CANSER ANN ENVILON WALVE IN SIMELE "VETION LINE TO CLOSE. ANN LAST FOR 17 MARTES- ARCS TENPERATORE ROSE 12 DER F TO LAP AFF F.	02/23/89 DURING AM ESF RESPONSE TIME SUBVETLIANCE. INADEQUARE PRODEEDNEE CAUGED THE AUG PERM TESKATION VALUE. TO INADVERTENCIA CLOSE RESULTING IN LOSS OF RAME.	03/47/495 DAMAGE TO SCS PTPTM6 AND SUPPORTS RESULTING FROM THE SLAMMING SWITT DE CHECK VALVES WHEN DRE SCS4/FST PLON IS STOPPED WHILE ANDTHER IS REWRING. SUPPORTS HERE STIFFFOND, IT IS NOT CERTOIN IS THIS OCCURRENCE IS UNLINE TO MODES 2,3,4,5,444, 17 COULD BE DUE TO LONGE POESSUBLE.	04/11/90 L055 DF SMRIPDUM CTOLING COCUMPRES, MATLE IN MOBE 5, WHEN A SUCTION ISSEATION VALUE AND INADVERTENTLY CLOSED ANDER POWER MAS OFFENDED FROM AN ANTOLOGOUNG LAISED DX. THE OFFENDED FALLED TO DEAVIET THE EFFECTS OF OFFENDED FALLED TO DEAVIET THE EFFECTS OF OFFENDED FALLED TO SERVICE.
EVENT CATEGORY		DHE 1 LIDSS OF SHIFTNAM CONLINE	LOSS OF SAUTOWN CORLING	LOSS OF SWITDOWN CODI 1465	LOSS OF SAUTONAM CODLING	FS I LINSS DF SHITTOWN CODI, INS	LIDSS OF SHEETDOWN COOL (NG)
PLANT MONE		JON STEATINE	2 300100	1 SESWARE	2 400941986	Cal vert Cr. 199	2. eeskildidi 35
((8, %).		*1088/11	99088/299	113/88024	10068/15#	200687/2 T2	21006/821

28/62/89

sagnapaan quears awb system mov signapaan quears awb system mov

1886 1-2

ø

.

SYSTEM NOA PARTERNIA SETECTION, AND WILLSELING FEATURES CEL THE TO! 14, 18, 10, 10, 10, 18, 14, 18, 39 IC, ID, IR, 18 1H. JR. 1J 18, 18, 18 IM. 10 R. 18 18, 10 8 THE PERMIT WAS IN WORK 5. SHITTOTAM COOLING TRAIN IN 18355 AD SSOT GMW SOMPA BALOW SAFETUTA LARDWOAWCD DI NDMGA AD SSOT GLU BET STUDEINGC SABETUTAL LARDWOAWCD DI is and presenting, put the pump stopped as a pessel OF A MOUTITIEE OF NAMONE AND ADDREATIC ACTIONS AND 07/22/91 IMPROPERTY ADJISTED BELIEF VALUES IN THE COMPONENT realization and the contrained model a prese model 5. She train SUPERBERT FALLER IN MERAT MEDIL FINE IN LEMINE INADEGIDATE PRESSURITER VENT CAUERD INADCOURATE ATS LEVEL INDICATION WHILE DUALDING PERTELICS CAUTHY SACETARY CONTRACT AND A CONTRACT AND MAS (DEPARTING MACH THE PART RELATE ALERGIMO. SPL with RESTORED USING TRAIN A. THE PUMP LOST SECTION COLLING A., TER STSTER OFFICE OFFICE DOOR & PRESSING SPIRE ATSIDNAL MAI NEWBORL SYSTEM DELINE COLD SHITTERE MUNEWIGHT WIGH MCS PRESSUE SIGNAL THEI INITIATES as a work of maintenance and and control while PALTING LOSS OF JWSTRUMERT IMPERTER ONIPUL FOWER CAUSES 自己证 译 整体 分析目的 中國 任 解系统目标 词 化的名 好 REACTIONS IN THE ELECTRICAL FIMER SUPPLY DIRING PELDM FLAMME LEVEL LEADING IN UNDEFEIME AND LOSS PERSIMARI, EDHOR (MUSED CAUTATION 18 COMPONENT THE MATER PLANS WITH POSTWILL. (PSS OF NAME DARKELLANCE RESTANC. SHE AND RESTRICT NOTWO FIEL FIEL COLLINS MHILE COME MAS SFREEDED. CAUGED BY STARTUP OF & SUITAB EUM PUMP AND OF COM PANEWINGS AND POSEWILD, LOSS OF MAD. THURSDAY THE REPORT of the public flow 1% & REARCH LINE. DUFING MURE &. FURNIT SUMMERS 722.1% 1E. (#10E 5). 19/25/91 10/30/80 DATE (PLS IF SWITTING TOB. AN LINS OF SAUCEDAM COOLINE AND AND MANAGEMENT OF AND LOSS OF SHUTSING CORUNE LOSS OF SHOTPHER COLDER LOSS OF SAUTORS COLUMN 1055 DF SHITTOWN LODI TWO LOSS OF SHIFTED AN COOL 745 ETENT CATEGORY TIPRO CANNE 1 TURKES PERMIT & THERE PERMIT 1. GUOMGIESS An light days 「「日日」に Tanglaw 1 1 1017 \$28/91-009 20019-612 *54/2100H 10016/285 \$9915/502 20019-122 20016-052 124, 101, MUT 111

÷,

«Ib we liceweft futwir strong illes womste publiebut, THIS Eucad summary was neighbory express source of influencing...

5

C

Child and

Parts No.

SHEELEN RELEASE AND STSTEN BOARD FRANKES

SUSTER SOM PRESERVING AND DE RECTOR. AND MITISAUTURS FEATONES (SEE 1486.6 1-1) it, 湖、湖、西、湖、湖、 15 五、元、四、四、四、四 Ŕ 10 前,於,然 法"法"法 竹田 四 四 1 11/16/DM ANALE 18 COLD SHUTDOWN FOR MORE THAN 2 MONTH, LOSS POMER MAS LEGT FOR AN MEMORY AND THE DWE UPDARES DREATE TO THE VIIAL BIS 910 MDI (LOSE RELAISE THE OTHER DAG MAS INSPENDALE. DIESE MUNEP NAME ARTIFIE HAD OPERATORS NOT FOUND FROMELEW. DAY TANK add Made. Himcurd, THIS IIM MAS EXCEDED AND THE UTHEN AFTERNING THE PARPOLITE THAN THE FALLED DIE TO DUE ... PERSONNEL EARCH, ALL AC GFFSITE SEECRICAL BECOMMED DUE TO SERVICED BATTERY VOLTAGE. THE AMO MOLT AL MUS MAILA POMERS IN MATTREE CHARGERS MAS 10/11/87 ONE INCOMING LINE TO DUE OF SHE SHEETY MIGES MAS 10/15/87 A STREES IF ENGINEENED SAFETY FEATURE ALTIMITENES these and part laws first the part of the and A Date ACTISS FRAGES OF ACCESSA. 1 D/C (011 OF AN ANDIALISSA OL ING ANDIALISSA .340. AL SAM HALLAN istact fin worked (stallt dungs and wit untwed, Digit the figure in this but simply and figure and WHET OF THE WIRE, & RATTERY LIFE 11WE ESTIMATE MANUALY STARTED AND SHUTDOWN COT JMC MAS THEM armolwing PwCOMPAG LINE AREARER TRIPPED DAE 10 ALTIRENTAL JAGRING DE PROTECTION SELATS, THE DIESEL SLAEshing bits wet staat Michaele contact PRODUCT AT MID-LINE, MANY RENEW DIFFE. FUEL STREAM PSESSARATIUM FOR MAINTENAME. UNE OF 240 VITAL POWER RESIDENTION. FOWER 12 THE CONTROL ACCOM-AMMUNICIATIONS AND FUTNITIES AND LITST. AM REMOVED FROM SERVICE FOR MAIN/EMANCE. AT 146 RISES LOST FOMER, DA EST ACTUALISM COMPLICATES SERVICE, PINES 2 PAG 5 STARTED MAY AN MUTPUT F BFSITE POMEN BELIANED DEE TO SAFET METAL SCRFFM.SHY ARGINE BLEWL. DAE BLEWE WE BUT DF SERVICE FOR MODIFICATION. THE DME 10/16/87 MMELE SHUTTERME FOR MERTELING. INE STATTUR TRANSCORMEN MAS ALLIGENTALLY SHORTED TH (2013)934 PUBER FUSES 新工い, 15,1288. 125 MM 1 BC SUMMER MAS LITST. RESTORED AFTON 5 MUNICIS. 机化物 藏马 除口 和低色。 重动行 為外路11,為約1,長 in wateratery. FUERT SUBMERY 02/21/81 たち LOSS OF ELECTRECK, PUMER CESS OF RECENTCH PUMP LOSS OF ALEETRIFICAL POINTS LOSS OF LEEURICAL POWER LOSS OF REECTAICAL POWER EVENT CASESSINY CRYSTAL FLUER 2 PARTON FORMER FIRT EALMONE MUT COLT PLANT MAY MCIGALPE I STRUMB 3001287582 SUPARTERS \$0078704\$ 101 HOM 111 310M 111 310W 150, MD.

(1) MC LICENSES FUENT REPORT JERA MONERA MONITARIE, 2015 EURAT SUMMARY MAC URIANCE FORM JMCDUER SCHOLE OF INCOMMUNICAL

Page No.

THE REAL

THE NEWLYON. ATTENDING, AND MITIGATION SEATORES contribute Futures and System non

BENETATING DIMM FOR MAINTENANCE AND MAINS ALL ESS BATTERY CARLE WERE CAUSED LODS OF INSTRUMENT BUSICS WHICH COMPLICATED EVENT, NUMBER OFFSITE POWER MUS stanten but Mitause du a sebremen saulune sig mor 1000. Drevatum activum and beneven to be februite 11/16/95 PLANT MAS IN MORE 5 MITH GENERATIN CIRCUIT RESTING 网络白银 (11) 包括44、 61/64 9 氯化 11% 包括地名 1244 184 1845 1841 radgeb dir Dr Openariow, & Hechwellaw Swirtares a PERSONANCE FARENCENESS OF POMER PESCE. THE IN AUTOMOTIC ACTION OF AN ENERGENCY DILSEL IF STATES FOR MUNICAMEL, 100 LEVEL BUNISH 以来和自我 就们 19年 [1955 第一批曲 FGW 部分和自主体的1974 年 COLUMEN OFFICE FOR DEPARTMENT REACHED. THE OFFICERS, E IN PREPARATION FOR & FEST, THE MILTINGY COMPACT SECARGE THE DEFICES MEET TADGED BUT, ADDITIONAL SHALE OF DESITE POMER AND DALIEL REMIATION 14 fallsafess. Bit to a majulitudelt public. Power STAINMENT EQUIPMENT MATCH MAS (PEN, POMCA 10 MARE UNMARKED TO THE ADDRESS AND ADDRESS ADDRES THE & PRECIMENT AND EMPRILIES CANSING 1995 DF 100 MAS LOST TO TWO EMEMIENCY 480 VOLT NUSES. DW DETSEL REMEMBER STATED. MCMDE DF & MCSIGM the set is not set and the set a mark (1900) ONE OF THE DEFINE RENEW . "YS AND DAE DE THE SAURDES OF UNTSITE PONSER AFRE DUT DE SERVITE. FARTINE SIS CARSING 1205 OF ABOU VITAL POWER MEMDITES BULLME WHICH TIME INE REACTION COOLDWIT als tractif billing statistics, a fill potation WORLEW DELARCE RESTORAGE PORCE TO DRE DE THE 11/05/87 WHILE IN COLD SHUTDOWN, WITH MLL 2 DIESEL THE BUS AND REESTAN LOW SAUTIONN LODE (NG. STORE TEMPERATINE INCIDENT 17 DESERTS. RESTRIED WITHIN I MINUTES. NUMBER OF STREET 107100100 春時のいってにい ないのでいいない 11.80 LOTS OF PLECIRICAL POWER ABSS OF SLEETSAICAL POWER LOSS OF ELECTRICAL PORTS appriary and the 1 loss of Electratical Power LESS OF FLECTRICAL PIDER E-WHI CHIEFERN - 「 教師市 7月」 にある 2 INDIAL WEIDNE #11.070mm 7 MARKET ROME FEANT SATE 128/180105 「「日のほっ」「書」 010887620 112/84049/ 300047597 120. 101.

胞

"何"四"武

FFETEM 904 FREVENTING, DETECTION, AND MITIGATING FEATURES

(1-1 Jient 135)

鹅

H. B. H.

-

H

四、四

"我"我"我 Sirgeran at MIS-(03P, & LOSS OF DEFERING REPARTING

出

約,約,約,約,死

H

語

FEATONE MILED MIT LOAD IN THE MIS LIMIT. THE ROAD

ALE ANY MEDIED. A 2 MED F WES REAMERABILIES INCOMENT

ACCURACE.

PLAN MULLE COMMECTAT TO THE RUS MAS TRUPPED.

四 元 四

ASS BF ELECTRICAL POWER VORTLE 1 42819006

Fund No.

State and

contract within the termine the statistic and the statistic statistics and the statistic statistics and within the statistic statistics and s

SISTER BOA PREVENTION, RETECTION, AND NETIGATING FEATURES FIRE TABLE 1-11 拍 n) R. "出"出""出""出""出" 法"法"出"法"法" 拍 "王二二、王二二、王二 於 15 四 萬 行,近,四 14 02/20/81 [102] R. MRAWI RE-2111 FORES WHIE 2 W 2 \$15213 THE DE SERVICE ALTA CONCESSION LOSS IN DAR RE BUS. ANTLE CORE SHIT LANNED. LANSS OF CONCERNENT CONTINUES ANTER PERMIS, AUT SERVI FUEL POOL CONTINUES CONTINUESD ame Prime And The '-Juges of all States all States and the '-Just all states and the '-Just all states' and the '-Just all states' and the states and the states all states all states all states and the states all states LINSS OF POWER TO THE TURNALMENT ANDIATION MUNICATION CANCES LANDNERRING V. AS ACTUALIEN OF CONTAINING MU Oligibre Purs Bursidieries and a cost cost and costs and accel anterial to some burst and supple supple size costants of cost of supple rest root cost and s sizes the cost of supple rist. 拍 141 22/142 DICERT 1810 MMALE DICELLS DICERED GREW CON 1221 DEFINE CALL SHUTDERME BEST 201 14 1422 DICERED 00/0//41 PERSONNEL LAPOR DUTS/** CONJENTED FOR LOSS owilling wells in which to minimum and an and and PURSE ISRUATION SYSTEM DERIVE REFORTING COMPARENT CONTINUE FORM FIRMER MADE LOCAL FARM AN ALL DAMAGE (DF-517E SAMACE. FUEL POIDS CONCERNS FUE 26 MEMORIES. 三時期の 福田山 100 i. 3195 LINS OF LIEURICA POWER tiple believes at soil a 1.055 NF LECTRICAL PUBLIC LOGS OF ELECTRICAL PONCE LTTS BF ELECTRICAL POWER LOUS OF ELECTRICAL POWER EVENT CATEGORY Distrib Cantolin 1 there: polisi a 2 INCOME AND COME SHEETH SHEET S WILLSTONE 7 BRAEDWOOT 1 71281 102 10 101012-140 Tri-olume. 10014-152 141-91006 #28-31005 109, 40.

「御書」

١.

ではなので

cupitiona post"s and costs also enpoyentie. Motolical, and Altightiga Heatures

11/27/27 建黄属的 鱼 和目的公寓 的口能好。 希望 的口 彩 和目的 一個 LOUT SULTIME PRESSURE BELIEV WALKE MATHE STRATES SISA MARE ALLERS MARE TELEVISION LEXASTAND OF THE TELEVISION 研 (1842)、第41 31章 95111章 485,453 149434 (18518) regioner was lets week a sum rune forrige willer value tritter. Apolit 2 and 1/2 sound was been delited ALL COMMUTING TO THE SPENT FART FART STRATE FROM AND or with proper cutofies the line first proceeding values LL/21/BC DEE 10 PRESURFIER PORV RESIDE DAGAGERENSTICS. MARY 10: 220V 8130Y VALUE DEFMED. 192 P09V 4150 0719429 AND MCG PRESSONS DAGAFER FROM 2194 10 1345 CONTACT MATER INCENTION, THE PLAN CONTACTION, AND LODD OF NEW CORE (MG. DAME AND CONTRACT AN UNITY a MCORAL IN THE MACTOR COULMNT CASES PRODUCT 和此 课时 医 系统行动 (如日叶、游戏的 教室) 现在 WINIT & GUILDR & FIFE NUMBER IN CONTRACTOR STATE (STATE) (SAME) AND A STATE AND A STA CONTROL IN COLD SUBJECT AND THE REAL PARTIALLY DARRENT. Witholds and All The Long ball and and Links at 55 105 1054 03/18/700 FRUM SHUTDERS of 150 PS16, BHR SUCTION WALVE MAS 10 15年 1月11 南北行体 出於 常長年 PEE 111110 1 1966萬位 18 PECTECTER FOR REAL STATE (46. WITH THE RM 建立处理る 操 自己可能的 产油烧 网络 研究能量 2 物质 District room the station coulty due to the law law APPROPRIATE VOID GALLONG FROM THE SPEAR FUEL 127/01/189 AT 1204 PS15, APPRIT 67,000 GALLOWS OF NEWLID and survey routering and and an and an and COMMENTATIONS BESILTED IN A LIDT OF MERTIN 05/27/2時 44/811 25,806 201,895 66 85,5199 (201,887 業成 MEALTER CAULTY FINE DE THE THE THE ALLER OF THE THERE AN AL INSTRUCT A TRUET BE FIRE THISTOCHER ALR C.R. MAINTEMANCE. AND PUP SUBSIDIES LEVEL. DEG ALL IL SHE ADD TH ISALATE THE LEW. FIE(18年 作用國 6월51、 「「「「「「「「「「「「」」」」 ini. 1055 th PEALTRY LINE WIT 1855 05 REACTOR COLUMN (105) 如 推進相關 行風(186) LEGS OF MARTINE CPUE AND 山的5 好 或或口徑 口能 ## 上的时候 机构的 化硫酸丁 LOSS OF REACTOR CONLANT こうほどうない いまたい こ 記法語 書い 2 DECEMBER 2 1 GOOMCORDS い、読んれの読 THE PART NAME N ABIERS Pal 154955 1640187098 120481112 「たいなのない」たの湯 "TORE/ITE \$1068/PSP 123. 45. NOTE 222

prints dot popularities unitaria date matricenties filalest

23

-

唐, 正, 佳.

出,位,比

約一次

10, 10

-

携

10 THE PAGE.

() WE LEFRICE EVENT REPORT (LER) MUMPER MARILARE, THIS EVENT

SUMMARY ARS DETACATE FOR AND THE SUMPLY OF AN INCIDENT.

6

25

Page No.

1.4 Plant

Sanifikam Eagers and Eastik 204 Properties, Scittige, and Milladien Gainers

LER. NG. PLANT NAME EVENT CATEGORY

North Townell I LOSS OF REACTOR CERLARD

12 81

ANTIMATIC INGRE

SATE

SUSTEM Non PERSUNDAL DENGLING, AND WITHATING READINES (SSEE TANK F-1)

Perificio Eduluade de Sea, la ale Mencele IMESTA "UN TUDE al 196 dea table deste feste is la chessia Learante ches entretila. Pos consideran

21

Faue No.

2-1 31401

23021231 MÜLLENIS GMS SAMEAL MOLLING HOR WELLENIS MALENIST MOLLING

STSTEM BD+ PREVENTION, DETECTION, AND MICLOSITHG REATURES TANK 1-13	46, 42, 34, 39	4		*		5	
RATE EVENT SUMMURY	<pre>c5/22/35 3% PEFJIELING. CONTACHNEWT INTEGRATIN CODED NOT BE E578PEISEED WITHIN THE REDUCTOR B ADDRES WITH THE E.ECTAICAL POWER AUDILARIE. CONDITION LASTED GROUT 9 HOLMS.</pre>	42/13/85 1% MODE 3, A SECURITY OFFICER REQUIRED TO UMLOFF ROTH WORKL AND EMERGENCY MATCHES, MAMPINIATED THE EMERGENCY MATCH SO THAIT ROTH DODAS WENG DOEM. COMPUTION EXISTED ANGUT 21 NOVES.	04/15/04 DURING REFUELING, POIN DURES US REASONNEL AIR LUCX MERE INCONSTRATE LEFT DER FUR AUG ABOUT A NUMES.	CS/25/PS IN REFUELING. BIRING CORE ALTERATIONS, CONTAINNEWT UEWTLATION COULING MATER STSTEM YENT VALUE WAS FOUND LUCYED DEFL. THIS CONTSINUEDTON CREATER A PATH FROM THE UNVER THIS CONTAINNEWT TO THE AUSTLIARP RELIAINNEWT. THE OPEN VALUE THESTOF CONTAINNEWT.	09/30/99 DN THE STRADATE DECASIONS CORE ALTERATIONS WORE PERFORMED WITHOUT CONTALENDER INTERSTIT OF DUDIEED BY TECHNICAL SPECIFICATIONS, A PATHWAY EXISTED FROM THE STEAM SIMEALTOR SECONDARY SIME NAMADY IN CONTALMARKET INNEDDOL THE STEAM SEMELARIS ZIMESPHERIE RELIEF VALUE.	08/29/90 EMEJMEREAUSE EVALUATION =METICATES THAT THE AMA PSMMTS MAN MUT PARTURE ADERDATE AMPLATT THE EINMIDIATEST STAAN PIRMTS ADEAL THE ACTIONS IN THE REFERENCE ALAM MODEL. CODARCITIVE ACTIONS (MELLING PROJECTIVEL CHARGES.	02/15/90 MAIER TURBANENCE BURING THE BEAETON CANITY FILL DECEMBING FARANCE AN INTELACE IN FUNCTION
EVENT CATEGORY	ALTO SHALM INNER LINE (MILMO)	CONTRIMENT INTERIO	CONTAINMENT INTEGRITY	CONTRAINMENT INTEGRATIV	ALL DO SIN' SHEMALAD NO.	CONTRAINMENT INTEGRETTY	CONTRONMENT INVESTORY
21.0kt 40%	2 withs	T Indee est	1.11111	M (Billy: 1	1 A J Intel State	PDIMT SEACH 1	RANES UNLEY 2
158°, 40',	(i) lun	(1) 3105	111 MILE	(1) III N	117 330M	11066/992	11096/219

14° 14 ATNERNEDE.

(1) NO (TELENGE EVENT REPORT LEED) NUMBER AUGULARE. THIS EVENT SUMMARY MOST (RELATED FORM ANOTHER SUMMER OF INFORMATION.

CONTALMENT INTEGRITY BILLSTONE 2 191405-911

CONTRACTOR THREE DATES 7 JULISTONE 7 334-49038

AIRBORNE ACTIVITY RESULTING IN AUTOMATIC CONTAINNEWS PORTE TERMS IN AUTOMATIC

10/02/90 PERSONNEL EARLY ANILE DRAINER STEAN GENERATUR DUALNE REFUELING CHICAGE CRUSES OPEN PATHWAY FROM CONTAINMENT THROUGH NEMANY AND ADV TO THE OUTSIDE

19-104/TO DURING WIDE & CORE ALTERATIONS, MAINTENNALE PERCONNEL ESDID AFOID (ED IN DADADLEDELITY OF

4h. 1C

Page No., 54 72/72

14111-1

SAUTEDWAR EVERYS AMD SYSTEM BO+ POEUEWEIDM, DETECTION, AMD MICIEATION FEATURES

67 3		1. 19 . 19						
ESTE EVENT SUMMARY	COMPANIENT PUBLIE "NU-UT AUTOMATIC CEOLATION EAPAFULITY.	10/20/90 A DESIGN DEFIENCY IN THE CONTAINERS SQUATEDN UR.N. POSITION INDICATION SYSTER CAUSE INFRUEER LLICHTANTION OF SCHERAL ISONATION C.LUE POSITION SMDICATOR LIGHTS DURING AND SUPERAM.	10/11/10 EVENING SIDVERLARYE TESTING WERN FLANT IN EDUE SHETERNMA (MERE SI A RESIDN DEFICIENCY IN THE EXAMINENT TERATION VALUE WHICH SIMPLIES EXPONENT (DRING WEATER TO THE WEITHON SHEETE TANK FOREER CAUSED THE WALVE TO NE WEITHON SHEETE TANK	DI POD-PUT FANDERDART PRESERVATIVE MAINTENNANCE PROCEPOSTS RESULTS 3M ELLESSIVE LEANAGE THORMAN VIEWIANG FORT BASALT DW INNET PODA DE COMMINIMMENT ENERGENY ATRUCK, LEANAGE ANG DETENTING POMPLETEN DESING COLD SULTEDNM MUDE WITH REACTOR CONDUCTERY DESING COLD	01/10/91 EQUIPMENT FAILURE-MALFUMETION SECURITS IN UNLATIONING OF ENERGENCY PERSONNEL NATON DURING NOT STANDBY (ANDE 3).	02/06/91 INNORGIANTE FROMTENDERS NESSA TED IM BREACH DF LANTAINNEWT DGTLDING PENETAMITONS DURING REFORLING.	02/27/91 EVENT OCCURRED DN 2716, 2720, AND 2727. RANDEDIARE ADMINISTRATIVE CONTROLS FLUE EDWIPPENT FAILUNG JA THE ADDIATION MONITORIAL SYLEY ACSULTS IN A JUATION OF FOR SIGNLS CONTROLATION PERIO DY JOINE - UNDUCK PELIEFI FOR THE CONTRACTOR PENE 7 JUAT N SYSTEM DIBLNG COLD SHOTODAN (MUDE 5) AND C	93/95/51 DURY AG COLD SHOFDOWN IMODE 5), AN ENGINEERING EV-LUNTION OF CONTAINMENT INSTACHMENT210N 7.40 TRATTONS REVEALED THAT INNOFEDUALE PROCEDURES FARTHONE TO CALIPARATION OF PRESSURE TRANSMITTERS RESULTED THA A VIDILATION OF CONTAINMENT INTERSTIT. CORRECTIVE ACTION CONSISTER OF ACTIGING THE FERTIMENT PROCEDURES.
EVENT CATEGORY		CONTRIMMENT INTERATIV	E A CONTAINNENT INTERFITY	CONTAINNENT INTEGRITY	CONTRIMENT INTEGRAT	CONTRINNENT JATESETTE	CONTAINMENT INTEGRITY	2 CONTAINMENT JATESAUTY
PLANT NUME		T JEINER BARBA	CloseC11C02	1 530055 164	Catempo 1	NCRTH ANNA 1	1 4005	CONNECT FOUR VANNE
108 ° 40		63001/600	1212/1002/14	255.23,004	12/42/02	\$00b1/8.1	Vosila/EL	9000-a/117

19. A.

넕

¥

03/IS/PE FERSENEEL EFFERE CAURES A MARINE CONTAINFERD ISOUATION VALVE TO REMAIN OFFA DURING MOT STANDEY

212/91010 CONNECTION YANNER 1 CONTAINMENT INTEGRITY

1. 1.

¥

SYSTEM BOA PREVENTION, DETECTION, AND WITTONTING FEATURES (STE TARKE 5-1)

-

ų

いる 大福田

FRENEWARDOW, DETECTION, AND MILISADOW SEATURES SHITTER EVENTS AND STS 24 BOAR

> FURNT CATEGORY SHITH IND'IS [[Pb. 80].

CONTRIMMENT INTEGRITY 1148L0 CANTON 1 90014/612

.

CONTRACTOR INTERNET I HIGH LHIGH 266/101004

CONTRACTOR INTEGRATIV N(80195 1 21010-090

CONSECUTER INTERVIEW F01WT REACH 2 10015-100

CONTA! WHEN! INTEERETY SI, BEEF 1 315/21008

COMPETISON ADMILE | CONTAINMENT |WIEGELSY 213/91025

CONTAINMENT INIEGRITY T Main Dine 10016/025

AND DESCRIPTION OF A DE No.

THERE IT FRUNTSTAGE & RECACH OF CONTAINING I 1916396179. AUTUANT SHARLE PINE IN RADISTICK MUNICIPALMS SECTOR STICLE. CUMPADAMENT VENT TUDUATION SYSTEM DECTMS COLS NEGHTERS IN ACTIMUTOR OF USE STORAL FUR THE Contribute.

We a set out and the setter and we set the set APPROFIL IN MINUTES REFINE METHOD CODEED TO LITES BURING STEAM SEWERSTON CARVIE RUDATION, THE THAN 200 DES F USING THE RMD SYSTEM. 「白い後日にいた

¥

Participate Professiones allowed a Pressory Romanav DECR TO THE GRANDLUS HEWITLATION SYSTEM TO REMAIN UFEN DUPING MUDER 1, 2, 5, AND 4. 14512340

¥

40, 40 10-10-11-568808480 658084 8570462 0564 COMPUTAMAGNAL MOLICH Parties sturies and parties.

¥ THE REAL PROPERTY OF THE PARTY PARTY PARTY PARTY OF THE PARTY PART STATION AND LARY BUILDING THERE COMPARED MAN TO ALTON OF NULLESS AND DAMIN THE WE WERE 11/10/41 FALLINE OF MAINING NUMBER PERSONNEL PLANUE SPRINED FROMEDURE RESIDIED IN INCORPER CONTRAMMENT INTERSTIT DURING NEFTICALINE.

10° 10 ALTERNET POLESURES RESULTE IN PROBATA PEPER INC. 10/21/11

40. 40 12/05/91 IMANUMMIE PROFEMBLES RESULTS IN UPEN SUNTRIANENT SOLAT, DR 200, 95 SIMING CONF 40 TERATIONS, MITCH 13 CONTAINANCERT ATMOSPHAGED. TO THE CONTSIDE ATMOSPHAGED CONTRACT TO TECH SPEC REDUIDENFRIS THAT EACH PERCENTION PROVIDERS FIRECT ACCESS FROM THE Same of County Pr an ISOLATION ME. un

SYSTEM BOA PREVENTION, BETECTION, AND MITISALING FEATURES. 12-1 3381 3351

TABLE 1-7

Page Nu.

SAUTONA EVENTS AND SYSTEM ROM FREVENTIME, DETERTION, AND MITTARTION FEATURES

光 い、沢 M. N. ň 21. 12 Ŕ and a state of the state of a second state of the second state of ANTHAN MAIL STATUT THIS NEWDOW, WORLD MANE DW SHE 05/19/30 PM 199005 ABS 物印刷 599106-10000 944519 20145 485 NAME (E LOUNARE PATH CONSTER & PORTEGORY E LOUNSING, 144 TENERALING DEEXEMPTINE PROPERTIES SYSTEM OF DEEXEMPTINE 10 ADV PSIS (150 PSIS APPONE INE DESISM). THE EVENT DECOMPED RELAYEE THE RHM COLD LES INJECTION VALUE DID NOT SEAT PROPERLY AND ALLONED RES AS 15 REQUIRED LITER RCS REPRESENTING LETS THAT #20 PERM RES VARIABLE REALVED MES 526 FILM. THE LPH THENTIE AND THE STREAMENT WITH THEME AND THE WEAK 10/27/39 DME 1941# 3ME 2001110# BILLINE WAS SME230011125 OFFICIALE IN CILD SAUTONER AND DR SAUTONER COR THE TALLS NO BITH PRESSERVER PORT 5 FALLED TO DEA MARGALT. 150LA710% 0F 10F INTERMITTENT, REALINES TO ME OFFICIE FOR LIDE. LINDHLIM GINGWIN SIGNI SIT COM NULIHOSING WILSAS TARM THE SAFETH VALUE. THERE MADE REGIM VALUES IPSTREAM OF THE POINT S. THE POINT'S MENU MIT PORCE-OPSTALED RELIES WALVES (PORV) MAILY ARE THE PORT IS SET AT A LONG PRESSING 01.19.30 FAILORE OF DRE LOW LEWERARDONE OFFENDESCORE PARTECTION SYSTEM TO DRESAUSE. THE FORM MAS WITIGATED DY OREDATED ACTION. ISOLATION OF 1 NOT SPECIFIES AND ACTUATOR DIFFARTION MAS CODE AND TO PRESSALIT THE LINE. EVENT SUPPLIER PARALLEL. SPETCH. 100 2011/01/1005338483940 0161796E55UR12011006 的任用的任务结婚了及110% 第211年1日第2日第2日第三日第 EVENT LATEGORY MILLSTONE 2 PRI STREET THAN THAT I'V Sussiv 1 東京の近日 235/99017 10086/1234 12058/112 NCIE (1) HER. MD.

(1) NO LICENSEE SURVE REPORT SEEN NUMBER ANALORUE. TRUS ESENT SUMMARY MIS TRATANO FROM AND/DEC SURVER OF TWO SAMATTOM.

OF THE PENERATURE IMPOST RESULTING TH LUGS OF LOW

transportune presentioner phonecritem pik the net.

DES F. THE PORU 5 WOULD WOT DPEN WITH & FAILINE

SYSDER New PREVENTION. DETECTION. DAD WITSCATING FEATURES.

Page No.

7-1 3160

PERSONAL NETTING. AND PUTCHING PERSONS California Evenus AND SYSTEM SOL

S. 37. 87 妆, 环, 肥 题, 読, 死 Fail (NES ALIMED LATE MARE 10 FLM BE DAVETE UNI THE DESK NAMEDIC. FLORENS THE TURSTAE RULLDING TO novilities the series int cavity stat saturb and apoint 200,000 A DETH IF TH PROPER AND THE REPORT OF MICH 1881 SUBSCIENCED F. BERTHER INTEREDUCTION CLOSED VIE VES BRICH BPEN DIFF FLUENCE INF REAM, CENTRING ALL THREE COM 10/12/94 DURING MAIN/FRAME, & VALUE HAD BEEN NEMDED FROM CONTAINMENT IN AROUT 20 MINUTES. CONTAINMENT SIMP STATED. THE FLEL TRANSFER DAM MAS WIT TH USE AND THE FLET FORL WES ISOLATED FAUNT THE CANETY AT THE Cast Streamin, red net sitt Pan, that whe mere mere THE MARTER LEVEL MAS APPLIE THE SERVICE CONTRACT AND THE CAPDOS POIM MUSICE THE PAY MEN COLATED IN THE COMPONENT CODE THE MOTER PLAN PROM. DUFTER (BALS DANS LEVEL IN CONTAINNENT REACHED AROUS CRUSS-COMMECT VALUE RETWEEN THE OFFENDING TRAIN LANGENED ANTER ANTERAN INCIDALATION ID PROVIDE and the data de service that . Loadag some the FLANDS NUTLED IN STRUCT WITH PUPING - DW 18 HMCHES. 11 MAS POSTICATED THAT FOR A WORST WELESSARY LEAN PATH. DIE TO JOU'N FAILURE AND MUTER FIMP MUTIPS IN THE RAY. SIME ELECTRICAL NULLING CONTRINCT SAFE, V ALLATED SOUTHWAY, the static with that takes an out the pick and UF THE EVENT. NOT BULLETIN BA-05 MBS 155989. FLANDES HETHE LEDGE, A LABOR ANDUT DE SERVICE MATERIAL MANAGES (FFEM FIR MATERIAM), UNIVE silvas: and these sat, but sat the distant with sailes of stairs (souther stilled man the where a varue was newered and the at amother LOCATEDN. 2 LEAN UTCLEMED. FLAMMETS MAD NEED. 10/09/75 WHELE 2W COLF SHUTTOWN, WITH FUE CONDENSE 12/22/87 IN MUNE 5 FOR REFIEL DISARE, WITH TWO REAM BENERIED MAILS INCOURSE & PORTINGLY OFEN Plant and and causing Lines of AL COM Figm. THERE DISCHARTS "小田田 北北江国南 #1115 Date: FLOOTON AND SPILLS F1000066 488 55115 COMMETTICHT VANHEE FLOODING AND SPILLS FL000186 440 571115 EVENT CATESOCY 2 18004 RD1081 PLIANT NAME DEDWEE 1 一西西 2124048/232 B1/24/2/82 247784/111 128. 40. 「「二」 たい

SYSTEM NOA ENERGYETTIN, DETECTION, AND MILIDATING PERIODS (SEE TABLE 1-1)

「「「「」」 00/05/80 APRUT 10.000 GALIONS 32 MATER ENTERED THE REFILLENCE POILDING MAGE ONE TRAIN 35 CONTRIMNENT When we print supplies and the method when GASKET OF & MEAT PRINKED FAMILED.

FLEDDING AND SPILLS ACGUIDE 7 570/88010

Fage 46.

2-6 3/882

Sanityam 2-44/5 AND SYSTEM DIA PSETENTIAL METERICAL AND MITIGATIM FEDUDAES

EVENT CATESORY BAUN LINU'LE NOTE FLI 242, 80.

FLICTING AND SPILLS SUITH TUTAS

FLOODING and SPILLS FARLEY 1 348/71004

EVENT SIMMORY

1190

SYSTEM BOA PRE-DEPUTION, BED-LTION, AND MITISATING FEATURES (SEE TABLE 1-1)

04/00/99 MAILE TH- NEALTOR CAVITY MIS REAMS FILLED, DWE DE THOU FORD STONE PAILESS HAD NOT REEM INSTALLED. FULS D'ARTER & USDUN FILLED HAD NOT REEL UNDER EUROPPENT STORM'E AREA TO THE CONTANNERST ARE 17,000 SAU DF MARTER ENTERED FORMANNERST.

10, 15

語 "思 /AV/22/91 PERSTANEL EARDAR CAUSER SONG GALLERS DF MATER TO SOLPT FREM SHOT TO CONTAINNERT FLORE WARK FLORE UNLT SS MURE INVERTER PERSENT OF CONTAINNERT SPENT SYS FR DUPTIME VALUE INSTIMUL.

> sto wo coccase event memory seen women available. This future SUMMARY AND DETAILED FAUN ANDINES SOUNCE OF INFORMATION.

Fage Wo.

THE BUT

FOR CENTRAL STRETCHIN. AND MICHAELING FEADINES. AND AND AND CHE STREET THOUSAND

SPERTER Store PRESENTION, INCOME, MAD WITHOUTSMO STATUSSING 78. 27 * ń 2 10 损 OI FOMER, ONE CHANGING FUMP SKIIED. METHL FILTWES SIMILAR MATERIAL AND FILMS IN THE SUCTION LINES TO CADDS TIL RETWEEK THEREY POINT 3 MMD 4 ALLWEE DAS RESULTED IN LOSS OF THE MORMAL AND ENVIRONMENT REPORT PUMPED 1810 14E MIS. 1916, 18E 48B. 78W RESULTED of/realing matter wit standar, and analysis in Daaming mars. CRITICAL POSITING. REACTOR REDAK CRITICAL REDAK COPECTED. THALINE MAS PERIODATES STATUS AND SAU CANTERS PF INNOPSERIAL EMETING OF VET, DOG TO EPADPHEOUS OF LEVEL INDUCATION, RESALTED TO LOSS and functionality, suppressing sailed the approximity of MORTH. REALTHE TRUE OCCURATED DIS TO MIGH SOMACE Ear Print Stat. Statute in UNIT & 10 bissing Print ALTROOM SAS PINITHS OF ALL MURIC ALLO INMESSES NALIFICATION IN CITS NEWLY IN THE LIFE AND NATIONAL BURING THE RECTIONATION, EFST AL CHARGING PUNCS. FOREWOOD, FILTERS IN CAUCH DUCTION AND ALTONED TO THE NACT. BUE TO ADDRESS CONCEMPORITOR IN THE TANK MIS HIGHER THAN THAT PETCHEING SAS REARD, AND MES RESIDER ISHED IN DIRING STRATTOR, DIG TO STRATE IN THE ESTIMATED NUCCE NO 131555 5. JUNITERANIE NELTHIOUN ITIS. ACID FLOW PATHS TO INE VOLUME COMPALE TANKS. PUECS, CAUGES BY FAILURE OF MEDAMMICK, SER, . STRAITFILMINE IN THE NMOL, MEADINED BORDW AND RESTA PARTICLES WERE FOUND IN THE STARD. INTERPER VALUE AL DAMENT DISTING MUTADAT 部 和5 金属制的 有效目的 计数数转换行 的现在分词像 1055 OF ALL CEMERIFICAL CHARGEME PERCS. Fight aparts and still STHI SPEach IDNAL 北京 法代 著語 招言 記 第四日日日 思 ABIRIT NO MIMORIS. - THERE COMMENTS BURNE FLUE. M 1811 5. 18/#1-10 11115/281 18/12/150 18/99/190 DATE BURCH AND PEACITURY ENEMIS STREET STOLLING IN SHE WORDS PORCH AND PERCITCITY EVENTS RURDA AND REALTINGTY . JEWITS REACH AND REACTIVETY EVENTS FUEMT CATEGORY THREE PERMIT 3 2 34 URDERS NOS 1 1080180 FLANT MARE THE REAL 710181002 Boudd/890 ŝ MULE VILL ALL SIDE SOLE III 411 31GM 12

焼ぼ

1) WE LICENSE FUENT REPORT 1. BY MEMOR AUGULE. THIS EVENT CHANNER AND THIS FORM AND THE SUBJECT OF THE SHORE THE

REACH AND REACTIVITY EVENUS 1 43365 3 NM 527687264

Canner, ITY. PRACHEDUS (EVE) INDICATION WAS CANSED BY LEAK IN THREADED FITTING OF LEAEL TEAMCHITTED

(1-1 THUE 1-1)

Fame No.

Taple 2-2

FREDEWILLE TELTIN, ME WITISHIER SEATORS AND POST OF AND AND AND POST OF AND

NUMBER AND THE RAD SALLING OF UNDERAFES MATTER MATTER CURLART BY 17 PPM. PHK MUNIMER SAUDDER MANSIN OF #1 FOMFR, HEAT THARED SECTION LINES IF BORIC RCID IT FOLLOWING REFUELING, PERSONNEL ERROR RESULTS IN T THERE WILL THE S WILL A 18 MUR 5 MAR THERE WILLS SECONTAPENDIDES OF REACTON REFORTING CAULTY MITE 和自己的 146 的短端 (186)和自然的 第二次 如此136 APPROACHES. EVENT OCCURACE WRILE SOURCE RANGE CHARGENE AND MURATION MAS REMUNED FROM SERVICE Stat alter interiment in the same same same INAMUSTIC PUMPS SHARED PF THE IND UNITS MUNI-MUMICE STREET, THE 24 NO. 1002045 STREET, STRE 10 FIGOD PTAK REMOM CONCENTRATION, MAS MOT INDEEDAPLE RELATES OF LOW SEMPERATINE. MUNITURS MERE IMPERANCE. 「「「「「「「「」」」」 in the second 110 BERETH AND REALTIVITY EVENTS SURGE AND SEACTIVITY EVENTS BURDW AND REACTIVITY ENERTS REMEM AND REACTIVIST EVENTS EVENT CATEGORY 1 INFORT FEMALE 1 32 5 M 100 C Addis C LANS 281/84012 200a8/902 91068/087 Soundarios. 150° - #0.

臣

SHSTER BOA PARTERN, DETECTION, AND MITTUREIME SEATURES 12-1 3MH 131

11.12

法、法 12.12 CONCENTRATION. NUMBER CONCENTRATION NASS REDUCTS BY VALUE LEARING MIG COURTS BY DAMAGER ADJUSTACK ow laught building could succeed a team in the act standarded MARENT VALUE CAUSED AN INCREASE IN RUS SWEWIGHT LESS THAN BU PPM DUCK & PERIOD OF MINE SAYS. OPERATOR TRANEL STOPS DIPING MAINTEMANCE. RESULTING IN & DESCREASE IN MES BONDM JEOPARDIZINE REGATION CAPABILITY.

10/25/89 by 18 PROMERS WITH THE BORIC ACTO REMARK IN THE CONCENTRATION TO DECREASE HELEW THE RESIDES 2000 編訂 1 1 月 9 8 東京 1 東京和島市 1 長時前 1 月11 1 第 CVCS. THE MIDE OF OPERATION FOR THE READER MAS CHANNERS IN MANNAL, ALS FILLING MAS IN PRINTINGS OF THE MES. THIS RESULTED IN DETREMED MURIN CONCENTRATION. ADJUSTNERS CANDED THE RENOM

97/22/29 COME ALTERATIONS WIRE IN FROMESS WITHING AND/RE CUMPATINERS STATE STATE BANKS W'USED FIST CONNET SAFIE SADIELATIONS IN CONTROL SOUND IN ADMETHON. AUTHORITY IN UNIVERSITY EXCENT.

BORTHE AND REACTIVITY EVENTS

こ 市出業住にない

414/190904

法

Face Sec.

(1) 111

FRE VEWIICHM. DE VELTIBON, AND MITIGATEON FEATURES THE WILLDRY STRING WEDLING

> ARIPACAS MUC ONE 2 & 186 PROFESION DNE 7 FIRE PROTECTION F3RE PROTECTION #011331044 3814 FIRE PROTECTION FIRT PROTECTION EVENT CATEGORY 1 83838 18103 APP DWGAS THE. C. NOSNINGA THEN THE IT TRUJAN] OCPAGE 1 20014-192 12075-990 70012-265 10010-892 80018-692 138, 40.

FIRE FROTECTION 1 3M02533EH - 62016-CCF

FUERI SURRAUE

加加

SYSTEM SIM FREVENTION, DETECTION, AND MILYSGATIME FEATURES /SEE TABLE 1-1

1413 JANUSALANI NU NI SI WESH HENDE THANGENTS ON BILLO MARTIES PERFISEITEN 324.

四,四

50° 35 - Die COMPASSION MATERIALS SAMPLED REFUELTED NUTLES. A2/14/95 UNERA BARRED ELECTRIC EXTENSION CORDS ICHICED

SAFETY PELATED EGUIPMENT IN FIRE ARES.

100

DNE ESF TEATH. REDUMBANT SAFETE AFLATED EDUTPHENT CENTER DURING HER STANDEY DISARLED EDUIPMENT IN 02725/91 \$18E 1% EMERGENEY SAFETY FEATURES MOTOR CONTROL MAINIAINED SAFE SHUTTOWN CONDITION. UNADLE TU VERSEY STAFF PLANS CORF. MF PLACED 1810 AHE CONDITION FOLLOWING MORE: CASE FIRE RECOVER WELESSARY DEPRESSIBILATION CONTROL SYSTEMS CONLD NDT RE ENEDITED. 14/142/52

10

10 06/12/91 FULLDWING CRAFT WINN AREA. FIRE RANALER BECKS MERE 12FT DREW. IN VIOLATION OF PRODUCTORAL. -ENCREMENTY FEATURES RIGHE MUT SECURE 14 WIGH ATTERN OF PROCEDURE, PEREWILAL FOR LOSS OF AGAZGAPE FOLEOWENCE WORK IN FREM. FERE RAMATER SEALS IN 153

11/21/91 1MADEDIATE PROCEDUARS DURING CONSTRUCTION RESULTED TH ENGENCED FIRE PORTECTION BARRIES RETWEEN AND AND CONTAINNEWE RECERCIDATION SPRAN CUBICLES IN ENCINECIED SAFETY FEATURES ROLLETNG.

다.

1

14915 1-2

-

1.2016 No.

+08 W31515 GND SIM315 MEGGLINDS

FREGENTION. DETERION. AND MITIGATION FEATURES	DATE EVENI SUMMARY	1 / PRIOR DANAGE TO INTERNALS WHILE LIFTING LED TO STUCK FUEL ACCEMPLIES DURING SCTURTING AND POTENTIAL RADIOACTIVE RELEATE DURING CONSERVENT RETRIEDAL HAMADLING.	1.7 IMADEDIDIE FREICEDIDIE ALLONED MISPLACEMENT OF SPEWT FLEE. CAUSING FREITIAL FOG LOCAL DUCEMERTING IN SPEWT FUEL, FUEL, FUEL.	09/29/90 HABREBUATE PARTEBUARS LED TO A CHEL AGGEMERY DROP DMID 10P DF FUEL NACK IN SPENT FLEL PORL.	10/05/99 DESIGN ERSON UN POLAR CRAME COM.D LEAD TO FAILURE AT MUCH LOWER LOADS 19AM SPECIFIES DESIGN.	10/26/90 SHAREBUATE FROCEDURE LED TO LACK OF SUBPRESILANCE FOR SPENT FUEL BUILDING VENTLATION SYSTEM AND POTENTIAL FOR RADIATION RELEASE SHOULD EVENT DOCUR.	02/12/91 SIMBLE OVERIBAD EVENT OR CORROSION FATISNE/SIMESS CIRCUTS FOR FALAXIME CANSES FAILURE OF HELD DOWN POLIS FOR FOLAR CHARC MALL, SUFFICIENT MARGIN EXISTED TO ACCEPT RESUCED ROL. CONFISIONATION.	02/13/91 FWADERDIATE FMOCEDURE ALLOWED MEDREMENT OF REAVV DIEDED EVEN FIEL STOBAGE FODL MATE ALL TABLE DIEDELS WERE IMPREMANEE CAUSEND FOTEWING, FOR BADDATEDW FELEASE UPDW DROV EVENT MAEN VENTILATION UNMOUNTABLE.	02/18/91 INDECIDIE FROCEDURE CONTROLS ALIBNED OFER DOOD IN FDEL HANDLING BUILDING DURING FIEL MOVENENT PROVIDING FOTENTIAL FOR SADIATION ALLEASE.	CARPAINE PER-JONNEL ERROR OMITTED REDUZEED SIRVERLANCE DF FIEL RANDLING MOCHERE FRIA.: 10 USE DUGING REFEELING.	02//29/72 DURING REFUELING. 3MILE LIFTIME THE UPPER SUIDE STRUCTURE, A FIEL ASSEMBLY STACK THE BOTTOM OF THE DULINE STRUCTURE ASSEMBLY SUSPEMBLE ABOVE THE COPE. CONTA6%1 NITERPITY WAS STRIFTED AND NOW-ESSEMITA. PERSONAGL MERE TYP WAS STRIFTED.	
	EVENT LATESORY	SALTHORES TALL GAR SCHOLL LORD	HEAVY LOADS AND FIRT HARPELINE	Shu Yakon Tafu awa Sadati kucah	HEAVY LOADS AND FIST HANNI, ING	DHE I HEAVY LOADS AND FUEL HONDLING	HEAVY LOADS AND FIEL HANDLING	I HEAVY LOADS AND FIEL HANDING	HEAVY LOGDS AND FUEL HANDLONG	HEAVY LOADS OND FIRM HANDLING	HEAVY LOADS AND FIRE HANDERING	IT HEERS WEIMERS ANAL AREE. THIS S VENT
	PLONE NAME	a and an and and	NC GUTHE 1	Z NOBAR	2 3001	DRY ANSAS WIN.	THETON POINT 2	DIANCO CAN'ON	500/TH TETAS 1	2 100303199	530951 Ted	MEEL EVENT REPORT
	175. Wh.	Busua-982	91016-692	80898-558	318-90010	51066-512	247/93654	8100407.5°5	souie Baa	¥9016-282	NDTE (1)	43317 00 015

98. 98

98° 88

¥

104

14

104

86 '45

58, 98, 9C

3

2

SUMMARY MAS BUIRINED FROM ANDINER SOURCE OF TWEDRMATTICH.

SYSTEM KOM PREVENTION. DETECTION, JAND WITTIGHTING FEATURES ISSE TABLE 1-11

Face Ve. 1 04.074-97

1-1 1Mai

SHOTOPAK EVENTS AND SVSTEM BOA RECENTION. DETECTION. AND WITTERTID'S FELTORES

SYSTEM DOA FREUENIEDN, DETECTION, AND MITTERTANS REATHRE	PA. 106. 108	14, 134, 145	19. 108. 109	10a. 10p
ANTHAN I AND ANTHAN	(9)/2/201 BOTH TREARS OF SI WERE ELECTED FOR ABOUT AT HOURS HOULE IN MODES J AND 4. MALLE IN MODE 5. BOTH TRAINS WERE RECOVED DUE TO TESTENG. THE PLANT SIGNLES FAMEL IMDUCATER FULS CONDITION, HOWEVER, IT WES MOT LOGGED AND FROMER FORDERDES WERE MOT FOLLOWED FOR FEMOLEJ OF THE STOTER WORK FAON SERVICE.	12/96/32 ZM INAPUENTENT VITEMATIC 51 DCCARRED WHICH BLOTVED ADDITIONAL SI ACTUMITURS SWILL RESET BY ELOSIME INF BEACTOR TRIP BREARIES, THE PROJEMBE USED TO REACTOR TRIP BREARIES, THE PROJECTION INFORMATION TO RECOVER "90M THE SI ACTUMITUM", TWILED IN WAS ADDITED FOR THE IC LEPUE THE SI ALGOVED FOR BY TO 36 HOURS WHICH IS MOT COMMENT. THE SI WAS UNRUICHED ADDIT 22 BOLDS LATE.	04/10/91 HEATING UP. BHEALERS FOR HET SUCTIBU VALVES FROM SMST WERE LEFT TAGEED DUT. THIS IS A TELOWICAL SPECIFICATION VIOLATION RELAUES TWO FLEW PATHS TO HAT ARE BEDRIDED. CAUSE MAS FAILING TO REVIEW ISLOUT LOGGOON AND FOUR MAS FAILING TO REVER	10/27/91 PERSONNEL SOROR MESOR F5 IN PLANT ENTERING NOT SHUTDOWN (MODE A) FROM EQLE SHUTDOWN (MODE 5) WITH INDEREADELE EDNEQIMENT SPEAK SYSTEM AND FECS.
EVERT CATEGORY	MORE CHANNER EVENTS	NORE CHANNE EVEN.S	KIDH CHANES EVENTS	NUTE CHANGE EVENTS
Bern Intil	TROSEN.	401.1% como 1	1 330000	Bistoria II
116. ND.	411 June 1	CTF BLOW	MOTE (1)	569167858

ern wir interwere formt serbist siefen wumber ander ander huls event siewenen was deringen ameriken sindere of twendemartism.

2.0 SHUTDOWN RISK ISSUES

Sections 2.1 through 2.13 present detailed evaluations of specific shutdown issues that were identified at the December 18, 1991 meeting with the NRC and that are listed in Reference 2. Each section is subdivided into four subsections. The first subsection states the issue consistent with the interpretation and evaluation in NUREG-1449 and the appropriate RAIs. The second subsection lists the acceptance criteria that are employed to evaluate the System 80+ design to prevent and/or mitigate unacceptable consequences related to each shutdown issue. The third subsection discusses the postulated plant scenarios, the analyses and the evaluations considered by to assure that the shutdown issue is adequately addressed. Finally, the fourth subsection states how the issue is resolved by the System 80+ design. Depending upon each issue and its significance in evaluating System 80+, the content of these subsections varies. Where appropriate, reference is made to RAI's on the issue, both outstanding and previously submitted. Appendix A contains the responses to all the RAI's.

2.1 PROCEDURES

2.1.1 ISSUE

The operational guidance provided by the plant designer to the owner/operator might not be sufficient to insure that procedures to avoid, detect, mitigate, and/or recover from abnormal events initiated from shutdown operations can be developed by the plant owner/operator.

2.1.2 ACCEPTANCE CRITERIA

The operational guidance provided by the plant designer to the owner/operator shall be sufficient to properly utilize design features that are available to detect, mitigate and assist recovery from abnormal events initiated during shutdown operations.

2.1.3 DISCUSSION

The System 80+ design incorporates advanced features which promote safer and simpler plant operation. The features include redundancy and diversity of components and systems, dedicated and/or permanently aligned systems, and an advanced informa on stem which better informs the operations staff of plant status, potential adverse system interactions, and available recovery paths if an abnormal event occurs. These features also contribute to improved operability and maintainability that should significantly reduce the initiating situations that have contributed to increased shutdown risk.

The plant owner/operator is responsible for preparing detailed procedures for normal, abnormal, and emergency operations using guidance developed by the plant designer and plant site specific information. The plant designer's guidance generally is in the form of suggested operational sequences that preserve the safety bases of the design. Since shutdown operations must be intimately connected to an outage strategy, specific procedures cannot be imposed by the plant designer to cover the array of possible shutdown events. However, the plant designer can provide guides which instruct the plant owner/operator in the use of design features which can detect, mitigate, and assist recovery from abnormal events that can occur during shutdown operations.

The operational guidance contained in this report will be provided to the plant owner/operator through the plant designer's Operational Support Information (OSI) program. The intent of the OSI program is to insure that features of the System 80+ design are effectively utilized in the operation of the plant as specified in the plant owner/operator's operations program. The OSI program also provides a formal means to transfer design related bases for operations, regulatory operational commitments and related information that is typically provided to and required by a plant license applicant. The OSI program is staged to develop more detailed information as the plant design matures and is expected to be completed during the construction and pre-operational phases of a plant project. The OSI program integrates information from various interrelated areas, e.g., the Maintenance Plan, the Reliability Assurance Program and as-procured equipment characteristics, to insure the owner/operator can efficiently operate the plant within the design bases.

A summary of general operational guidance related to shutdown operations is provided in Table 2.1-1. Specific details are contained in the appropriate sections of this report.

An outline of the operational guidance developed to support RCS reduced inventory operations is provided in Appendix C. This guidance together with supporting information from this report is sufficient for a plant owner/operator to develop an operational guideline for reduced RCS inventory operations. The development of a detailed procedure by the plant owner/operator would require specific equipment characteristics of procured components and results of the pre-operational testing (CESSAR DC Chapter 14) to determine system performance values. As an example to support midloop operation, the shutdown coolant system flows, suction line vortexing characteristics, level instrumentation calibration, among others, would be measured during plant startup. This data would be used to verify performance as well as to provide operational data. Specific testing requirements for shutdown oriented instrumentation will be added to CESSAR DC Chapter 14 in an update amendment.

2.1.4 RESOLUTION

The issue of procedures for shutdown operation is resolved for System 80+ by providing operational guidance to address use of advanced design features to detect, mitigate and assist recovery from abnormal events initiated from shutdown operations.

TABLE 2.1-1

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS

TOPIC	PROCEDURAL GUIDANCE	REFERENCE TO REPORT SECTION
Unplanned Draining of the Reactor Coolant	Prevention Administratively Control Major Potential Draindown Paths Identified for Shutdown Modes	2.12.1, 2.12.2.1(3), 2.12.3, 2.12.3.2.1, 2.12.4
	 Monitor Instrumentation for RCS Level, Inventory and Temperature Controls a. Refueling pool level. b. Containment and subsphere sump levels. c. Level indicators and alarms: EDT, RDT, IRWST, HVT, VCT. d. RCS operational leakage (Tech. Spec. Surveillance). e. RCS level indicators and alarms. 1) Pressurizer level 2) RV Delta-P level instruments f. Pressurizer pressure 	2.3.3.1, 2.12.4, 2.8 Table 2.8-1

TOPIC

TABLE 2.1-1

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS

	PROCEDURAL GUIDANCE
nned Draining of the or Coolant	Prevention
	Administratively Control Major Potential Draindown Paths Identified for Shutdown Modes
	Identification
	Monitor Instrumentation for RCS Level, Inventory and Temperature Controls
	 a. Refueling pool level. b. Containment and subsphere sump levels.
	c. Level indicators and alarms: EDT, RDT, IRWST, HVT, VCT.
	d. RCS operational leakage (Tech. Spec. Surveillance).
	e. RCS level indicators and alarms.
	1) Pressurizer level

- 2) RV Delta-P level instruments
- f. Pressurizer pressure

2.3.3.1, 2.12.4, 2.8 Table 2.8-1

2.12.1, 2.12.2.1(3), 2.12.3, 2.12.3.2.1, 2.12.4

REFERENCE TO REPORT SECTION

Unplan Reacto information that is typically provided to and required by a plant license applicant. The OSI program is staged to develop more detailed information as the plant design matures and is expected to be completed during the construction and pre-operational phases of a plant project. The OSI program integrates information from various interrelated areas, e.g., the Maintenance Plan, the Reliability Assurance Program and as-procured equipment characteristics, to insure the owner/operator can efficiently operate the plant within the design bases.

A summary of general operational guidance related to shutdown operations is provided in Table 2.1-1. Specific details are contained in the appropriate sections of this report.

An outline of the operational guidance developed to support RCS reduced inventory operations is provided in Appendix C. This guidance together with supporting information from this report is sufficient for a plant owner/operator to develop an operational guideline for reduced RCS inventory operations. The development of a detailed procedure by the plant owner/operator would require specific equipment characteristics of procured components and results of the pre-operational testing (CESSAR DC Chapter 14) to determine system performance values. As an example to support midloop operation, the shutdown coolant system flows, suction line vortexing characteristics, level instrumentation calibration, among others, would be measured during plant startup. This data would be used to verify performance as well as to provide operational data. Specific testing requirements for shutdown priented instrumentation will be added to CESSAR DC Chapter 14 in an update amendment.

2.1.4 RESOLUTION

The issue of procedures for shutdown operation is resolved for System 80+ by providing operational guidance to address use of advanced design features to detect, mitigate, and assist recovery from abnormal events initiated from shutdown operations.

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS

TOPIC	PROCEDURAL GUIDANCE	REFERENCE TO REPORT SECTION		
	g. RCS temperature 1) Core Exit Thermocouples 2) Resistance Temperature Detectors 3) Shutdown Cooling a) SG Parameters b) Shutdown Cooling System	2.8, 2.3.3.1		
	MITIGATION (Immediate Operator Action) Identify leakage path. Isolate leakage path. Make up losses. a. Safety Injection b. SDC via IRWST C. Containment spray from IRWST via SDC lines d. Charging pumps e. BAST f. Safety Injection Tanks	2.3.3.4, 2.12.3, 2.12.4, 2.12.2.3(2) Appendix C		

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS

TOPIC PROCEDURAL GUIDANCE		REFERENCE TO REPORT SECTION
Heavy Loads	Restrictions specified for:	2.11.3
 Drop of transported equipment. Drop of fuel bundle Refueling pool seal integrity. Loads over ICI table. 	 a. Lift Height b. Travel Directions c. Systems Lineup (Specified in CESSAR DC, Chapter 9 and Plant Designer's "Heavy Load Guides") 	
<u>Outage Maintenance</u>	Strategy for Shutdown Operations	2.4.3.2.2
	 a. Define operating and operational divisions. b. Limit maintenance activities to components and systems not included in a). 	Appendix C

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN GPERATIONS

TOPIC	PROCEDURAL GUIDALCE	REFERENCE TO REPORT SECTION
Fire Protection	Administratively require fire protection systems to remain operable in shutdown modes.	2.7.3.2, and 2.7.3.3
	Procedurally Control: a. Combustible materials b. Housekeeping c. Hot work	
	Pre-Fire Plan a. Outline fire fighting strategies b. Monitor status of fire barriers	2.7.3.2

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS

TOPIC	PROCEDURAL GUIDANCE	REFERENCE TO REPORT SECTION
RCS Cooling Using Feed and Bleed (other systems not	RCS Pressurized	
available)	1. Start SI pump.	2.4.3.1.3.1.1
	 Reduce pressure through Safety Depressurization System (SDS) venting to IRWST. (Maintain subcooled temperatures in KCS). 	2.4.3.1.3.2.1
	 Secure operating RCPs (if applicable) 	
	 Cycle SI feed and SDS bleed to reduce RCS pressure and temperature. 	
	 When depressurized, open SDS and Run SI continuously. 	
	 Align SDC heat exchanger for IRWST cooling. 	
	 Restore Normal SDC systems. 	

SUMMARY OF PROCEDURAL GUIDANCE RELATED TO SHUTDOWN OPERATIONS

TOPIC	PROCEDURAL GUIDANCE	REFERENCE TO REPORT SECTION	
	RCS Depressurized		
	 Start SI Open SDS Secure RCP's (if RCS not vented) Align SDC heat exchanger for IRWST Cooling. Restore normal SDC Systems 		
SG Tube Rupture	Include in Emergency Procedure Guides a requirement to maintain a positive primary to secondary pressure differential.	Table 2.6-1 Section C(a)	
Lockout of main feedwater pumps in shutdown modes with RTCBs closed.	Administratively lockout main feedwater pumps if subcritical.	4.1.1 and 4.1.2	

2.2 TECHNICAL SPECIFICATION IMPROVEMENTS

2.2.1 ISSUE

When a plant is operated within the limiting conditions for operation provided by the technical specifications, the consequences of design basis events should be bounded by the results of the safety analyses. However, limiting conditions for operation developed for power operation might not be sufficient to insure that the consequences of events initiated from shutdown modes are bounded by the analyses. Technical specification should (include the necessary limiting conditions for operation that are applicable to shutdown modes.

2.2.2 ACCEPTANCE CRITERIA

Technical specification shall insure that when the plant is operated within the limiting conditions for operation applicable to the mode of operation, consequences of design basis events shall be bounded by the results of safety analyses for that mode.

2.2.3 DISCUSSION

The System 80+ design incorporates advanced features which promote safer operation and greater margins to operating limits. The features include redundancy and diversity of components and systems and an advanced information system which better informs the operations staff of plant status, potential adverse system interactions, and the recovery paths if an abnormal event occurs. These features also contribute to improved operability and maintainability that should significantly reduce the initiating situations that have contributed to increased shutdown risk.

One objective of the plant designer is to reduce the operational constraints that limit the plant owner's flexibility to operate the plant as efficiently as possible. Another objective is to formally impose the operational constraints required to insure the plant remains within analyzed bounds for operation through the initial set of technical specifications. Overly restrictive technical specifications especially for shutdown modes may unnecessarily complicate operations and may increase risks by prolonging the shutdown period and adding to staff stress. The objective of this assessment of shutdown risk for the System 80+ relative to technical specifications is to modify existing technical specifications to the extent necessary to address event initiators not fully covered by analysis of the traditional design basis events.

A summary of the proposed technical specification modifications is provided in Table 2.2-1, with a markup of the LCO and APPLICABILITY sections of the present technical specifications, as applicable, in Appendix D. In addition, a new technical specification section applicable to reduced RCS inventory operations in Modes 5 and 6 (with fuel in the core) is presented in Appendix E.

The technical specification modifications and additions reflect:

- a. the added redundancy and diversity of the System 80+ design that allows these modifications without affecting operational flexibility;
- b. analysis of events initiated during shutdown operations;
- c. assessment of the risk of operating in these plant configurations for extended periods, e.g., refueling, unplanned maintenance.

The rationale for the technical specification modifications is contained in the appropriate sections of this report.

2.2.4 RESOLUTION

The issue of shutdown specific technical specifications is resolved for System 80+ by modifications and additions to the technical specifications based upon safety analyses performed for Modes 2 through 6. These modifications and additions provide additional ensurance that the consequences of transients and accidents which might occur during shut down modes of operation are less limiting than those given in Chapter 6 and 15 of CESSAR-DC. The proposed technical specification modifications and additions will be reviewed against the shutdown PRA findings and a change package will be proposed for a future CESSAR-DC amendment.

TABLE 2.2-1

SYSTEM 80+ TECHNICAL SPECIFICATIONS MODIFICATIONS

RELATED TO IMPROVED SHUTDOWN OPERATIONS

Item	Tech Spec	Number and Title	Type of Change	LCO	Bases
1.	3.1.1	Shutdown Margin (SDM) >210°F	Revision	Change mode applicability ≤500°F	Extended applicability.
2.	3.1.2	SDM ≤ 210°F	Replace	Add Kn-1 and ECP requirements.	Provide protection for ejected CEA and CEA group withdrawl in shutdown modes.
3.	3.1.10	SDM Test Exemption for CEDMs Testing	New	Allow CEDMs Testing in Modes 4 and 5	Provide exceptions to Test Operability of CEDMs. Movement of only one CEA at a time is allowed.
4.	3.3.1	RPS Instrumen- tation	Revision	Specify the modes of applicability in Table 3.3.1-1. Extend SG Pressure-Low to Mode 3 and RC Flow-Low to Modes 3, 4, and 5 when the CEAs can be moved.	Provide Reactor Trip function for Steam Line Break (SLB) in Shutdown Modes.
5.	3.3.5	Core Protection Calculators	Revision	Extend Operability to Modes 3, 4, and 5 when CEAs can be moved.	Provides Reactor Trip Function for unplanned CEA Group withdrawal.
TABLE 2.2-1 (Cont'd)

EYSTEM 80+ TECHNICAL SPECIFICATIONS MODIFICATIONS

RELATED TO IMPROVED SHUTDOWN OPERATIONS

Item	Tech Spec	Nurber and Title	Type of Change	LCO	Bases
б.	3.3.10	ESFAS Instru- mentation- Automatic Actuation	Revision	Add Mode 4 to CSAS Mode applicability in Table 3.3.12-1	Insures availability of automatic CSAS for mitigation of LOCA event in Shutdown Mode 4.
7.	3.4.11	RCS P/T Limits	Revision	Add minimum Pressure Restriction RCS Temperatures between 483°F and 543°F.	Provide a SIAS for SLB and other increased heat removal events initiated in this temperature regime.
8.	3.5.3	LTOP	Revision	Change restriction on number of SI pumps operable to 2.	Two SI divisions required operable in applicable modes.
9.	3.5.4	Safety Injection System	Revision	Extend requirements for 2 SI divisions to all of Modes 4, 5, and 6.	Required for RCS inventory makeup for LOCA events in lower operating modes.
10.	3.5.4	IRWST	Revision	 a. Extend operability requirements to Modes 5 and 6. b. Specify maximum water temperature at 110°F. 	a. For compatibility with 3.5.3.b. Presently stated in Bases.

TABLE 2.2-1 (Cont'd)

SYSTEM 804 TECHNICAL SPECIFICATIONS MODIFICATIONS

RELATED TO IMPROVED SHUTDOWN OPERATIONS

Item	Tech Spec	Number and Title	Type of Change	LCO	Bases
11.	3.8.2	AC Sources (Shutdown)	Revision	Require 1 circuit between the offsite transmission network to each onsite Class 1E distribution system in Modes 5 and 6.	Provide additional backup AC power source.
12.	3.3.15	Boron Dilution Alarm	New	Both boron dilution alarms shall be operable in Modes 3, 4, 5, and 6.	Provide additional protection for prevention of an inadvertent boron dilution of the RCS.
13.	3.3.14	Accident Monitoring Instrumen- tation (AMI)		Add Radiation Monitoring Instrumentation for a. SG Liquid Blowdown b. Steam Line c. Air Fjectors d. Stack to Table 3.3.14-1	Required for SG tube rupture detection in shutdown modes.
14.	3.1.6	Shutdown CEA Insertion Limits		Add special test exceptions and applicability to only critical conditions.	Clarify applicability and STE's.

TABLE 2.2-1 (Cont'd)

SYSTEM 80+ TECHNICAL SPECIFICATIONS MODIFICATIONS

RELATED TO IMPROVED SHUTHOWN OPERATIONS

Item	Tech Spec	Number and Title	Type of Change	LCO	Bases
15.	3.1.7	Regulating CEA Insertion Limits		Same as Item 14.	Sames as Item 14.
16.	3.3.12	ESFAS Instrumentati on Manual Actuatior		Add Mode 4 to CSAS Mode Applicability in Table 3.3.12-1	Insures availability of manual CSAS for mitigation of LOCA event in shutdown mode 4.
17.	3.4.11	LTOP		Delete requirement for SI pumps.	2 required for Shutdown, see LCO 2.5.3. LTOP sizing increased to avoid PTS.
18.	3.8.5	DC Sources (Shutdown)		Clarify LCO to provide most reliable line up.	Prevents loss of operable D/G due to maintenance.
19.	3.8.8	Distribution Systems (Shutdown;		Clarify LCO to provide most reliable line up.	Prevents loss of operable D/G due to maintenance.
20,	3.9.4	Shutdown Cooling (Refueling Operations)		Require Additional SDC division to be operable.	Allows increased reliability for decay heat removal.

2.3 REDUCED INVENTORY OPERATION AND GL 88-17 FIXES

2.3.1 ISSUE

The NRC has voiced increasing concern over the safety of operations during plant shutdowns. Plant events which have occurred in the industry have highlighted the need for 2 close examination of operations during reduced inventory conditions in the reactor coolant system. Following the Diablo Canyon incident, the NRC published Generic Letter 88-17, which required that holders of cherating licenses or construction permits address a number of deficiencies in order to enhance the safety of shutdown operations and reduce the risk to the public. Specific areas of concern include:

- instrumentation which would greatly improve the operator's monitoring capability during reduced inventory operations,
- the availability of existing equipment for use in mitigating a loss of SCS or loss of RCS inventory,
- 3. nozzle dam installation procedures which would ensure a vent pathway is available so that RCS pressurization can be minimized if shutdown cooling is lost.
- alternate ways to add inventory to keep the core covered should SCS be lost,
- 5. administrative procedures that would avoid RCS perturbations during reduced inventory operations, and
- 6. containment closure issues.

The NRC has specified that programmed enhancements should accomplish a comprehensive improvement in the plant's at lity to cope with shutdown operations. The NRC asserted that plants are not well designed for reduced inventory operations, that procedures are incomplete for shutdown cooling recovery or alternate actions and that mitigating features may not be available under shutdown conditions. Therefore, the NRC has recommended that licensees implement means to prevent accident initiation, to monitor a progression that may lead to core damage and to evaluate consequences and, where needed, to provide mitigation.

2.3.2 ACCEPTANCE CRITERIA

The System 80+ design shall reflect a comprehensive consideration of shutdown and lower power risk, by adequately addressing all GL 88-17 recommendations and other issues relevant to reduced inventory, especially in the areas of instrumentation, technical specifications, procedures, equipment availability and analyses.

2.3-1

2.3.3 DISCUSSION

During plant shutdowns, certain maintenance and testing activities require a draindown of the RCS to a partially filled condition. Normal maintenance activities include the replacement of RCP seals and journal bearings. A testing activity requiring RCS draindown is the Technical Specification for inservice inspection of the steam generator tubes. The use of nozzle dams during maintenance and testing activities minimizes the time during which the RLS must be operated in a partially filled condition. To minimize operating time at mid-loop level, nozzle dams are installed on the st am generators and the RCS is reflooded to continue maintenance ind testing.

While the RCS coolant level is lowered to within the hot leg, the risk of loosing shutdown cooling is increased due to the possibility of vortexing at the SCS suction line interface with the hot leg. In the worst scenario, subsequent to vortexing in the SCS suction line, a large percentage of air is entrained into the SCS suction piping and the SCS pump performance is degraded or recrupted. If SCS operation is not reestablished, core boiling and pressurization can produce very rapid core uncovery, sometimes in as little as 15 to 20 minutes. This phenomenon, and the high probability of it occurring, prompted the NRC to issue the recommendations of GL 88-17.

System 80+ design features result in practical and significant benefits during reduced inventory operations. These design features are outlined in Sections 2.3.3.1 through 2.3.3.5 which follow. Details of the capabilities of these System 80+ design features to enhance safety during reduced inventory conditions and of the analytical bases for changes to Technical Specifications and procedure guidance to the owner/operator are presented in other sections of this report.

2.3.3.1 Instrumentation for Shutdown Operations

Diverse, accurate, and redundant instrumentation (including control room CRT displays) give continuous system status and provide the operations staff with precise information to monitor reduced inventory operations and to respond to loss of shutdown cooling events, should they occur. Detailed information on reduced inventory instrumentation is included in Section 2.8 of this Report.

Analyses form the basis for instrument design and calibration so as to assure correct instrument operability during reduced inventory states. Phenomena which can affect instrumentation operation are considered in the recommended use of instrument types for various scenarios. Instrumentation availability during reduced inventory is assured via the plant Technical Specifications that are provided in Section 2.2 of this report. A general description of the types of instrumentation which are used for reduced inventory is outlined below.

1. Redundant and independent wide and narrow range level sensors are provided for continuous monitoring of RCS level during draindown operations. The level indicators provide monitoring capability from the pre-drain down normal level in the pressurizer to a point lower than that required for SCS operation. The level indicators are calibrated for low temperature operation and they provide a high degree of accuracy. Indication in the main control room and low-low, low, high and high-high level alarms are provided.

The wide range level instruments cover draining from the pressurizer to below the bottom of the hot leg and are available with the head on and off the vessel. The narrow range level instruments cover reduced inventory operations, and are also available with head on and off the vessel. The narrow range instruments are accurate for measuring level within the hot leg. During a draindown, level monitoring would be transitioned from the wide range level instruments to the narrow range instruments when the greatest degree of accuracy is required during operations with level within the hot leg region.

- 2. Several independent diverse temperature measurements representative of core exit temperature are provided during reduced inventory operations. Temperature indication is available when the head is located both on and off the vessel. Since temperature is valuable in guiding SCS restoration actions and in monitoring the success of recovery actions, alarm setpoints are based on integrated response times necessary to support SCS recovery, event mitigation, time to boil, and containment closure.
- SCS operation monitoring instrumentation is provided that assures precise knowledge of the status of the operating SCS loop; including pressure, temperature, flow and pump performance indicators.

2.3.3.2 SCS Design

The functional design of the Shutdown Cooling System (SCS) is substantially complete for System 80+. Design features that improve SCS performance during shutdown operation are detailed below.

 The System 80+ SCS suction lines do not contain any loop seals. An improved suction piping layout allows self venting. Entrained air travels back up to the hot leg without the possibility of being trapped anywhere in the SCS suction line. This feature allows the SCS pumps to be restarted without requiring complicated venting procedures, assuring an expedited reflood of the shortdown cooling pump suction.

- The two SCS suction lines are independent and redundant to each other. Problems associated with a specific suctio. Line would not limit the other shutdown cooling train from being operated, after level recovery (if necessary), for continued decay heat removal.
- 3. The two containment spray system pumps are interchangeable with the SCS pumps and are designed to back up the SCS pumps in the event of a non-electrical pump failure. Thus, there are four pumps available for shutdown cooling, provided support systems are available. Plant Technical Specifications will assure pump availability during shutdown operations.
- 4. There are no interlocks on the shutdown cooling suction piping which have the potential for disturbing shutdown cooling. Although previous designs (e.g., System 80) included interlocks to isolate the SCS in the event of an unanticipated RCS pressurization during shutdown cooling, this interlock has been deleted from the System 80+ design per the EPRI ALWR Utility Requirements Document. This reasonably reduces the likelihood of losses of SCS.

2.3.3.3 Steam Generator Nozzle Dam Integrity

The System 80+ design addresses the NRC concern for preventing significant pressurization in the upper plenum of the reactor vessel during core boiling scenarios. System 80+ procedural guidance recommend, a nozzle dam installation and removal sequence by which the nozzle dams are installed in the cold legs first. After the cold leg nozzle dams are installed, the hot leg nozzle dams can be installed. Likewise, when removing the nozzle dams, the hot leg nozzle dams are removed first, subsequently cold leg nozzle dams can be removed. This installation and removal procedure will maximize the time that the steam generators are available for reflux boiling in the case of a loss of decay heat removal, and minimize the time that both hot legs are simultaneously blocked by nozzle dams. This installation procedure requires that the pressurizer manway is opened so that a hot side vent pathway exists prior to blocking both RCS hot legs with nozzle dams.

In the System 80+ design, the ability of the RCS to withstand pressurization during reduced inventory operations with the nozzle dams installed is limited by the design pressure of the steam generator hot and cold leg nozzle dams. Based on field hydrostatic tests performed on nozzle dams, a conservative value of 40 psia is assumed for this pressure limit. In order to assure that the nozzle dam design pressure is not exceeded during reduced inventory operations with boiling conditions in the reactor vessel, the System 80+ design includes a requirement will be imposed to establish a mid loop vent pathway via the pressurizer manway before operating in reduced inventory. When the manway is opened to the containment atmosphere, it provides sufficient venting capacity to prevent RCS pressurization and subsequent nozzle dam failure. The pressurizer manway vent pathway is of sufficient capacity to prevent core uncovery due solely to pressurization of the hot side resulting from boiling in the core ccolant.

The pressurizer manway will be closed except during normal RCS draindown activities (see Section 2.1). During a normal draindown to reduced inventory operations, when pressurizer level decreases to a preestablished setpoint, the RCS vent pathway will be aligned by opening the pressurizer manway. Following refueling operations, RCS integrity will not be reestablished until the RCS coolant level reaches the pressurizer. Only at that point will the manway be reinstalled. This mid loop vent alignment allows sufficient venting of the RCS to the pressurizer cubicle should SCS be lost, resulting in onset of core boiling.

Analyses have indicated that the pressurizer manway opened and relieving to the pressurizer cubicle will be sufficient for venting the RCS during RCS boiling and preventing steam generator nozzle dam failure. An acceptable, conservative RCS equilibrium pressure which is below the assumed steam generator nozzle dam design pressure has been calculated to occur 4 days post shutdown. Therefore, the earliest time after shutdown (from full power) for operating at mid loop level is recommended as 4 days. Based on industry operational data, a reasonable minimum RCS cooldown from Mode 1, followed by a draindown from normal RCS level to mid loop, can be performed in approximately 4.5 days. Therefore, the 4 day requirement does not impact the achievable start time for nozzle dam installation. Additionally, it provides the necessary degree of protection required for loss of decay heat removal scenarios. This data is to be incorporated into guidance for the owner/operator to employ when planning outage evolutions. Additionally, procedural guidance regarding the earliest time after shutdown for entry to reduced inventory operations is provided in Section 2.2 of this report. Such restrictions are implemented to minimize the consequences of a loss of shutdown cooling event during reduced inventory operations.

The specified time after shutdown is based on the following analytical results:

- decay heat vs. time after shutdown

- the resultant RCS heat up rate assuming a total loss of decay heat removal
- the consequential maximum RCS steam pressure for Mode 5, reached by boiling RCS inventory.

Power history (unit specific, cycle specific) decay heat loads vs. time plots could be used to calculate lower decay heat loads at 4 days, thus allowing entry into mid loop operation and nozzle dam installation earlier during shutdown. In fact, the use of the pressurizer manway as the vent pathway yields an acceptable equilibrium pressure (below steam generator nozzle dam design pressure) at 2 days post shutdown. This capability would be further supported by higher nozzle dam pressure limits than assumed in the analysis. Although this meets EPRI ALWR requirements for refueling scheduling, it is not recommended that the plant enter mid loop level operations before 4 days. Complying with procedural guidance which requires mid loop operation no earlier than 4 days adds margin to the relationship between RCS pressure and nozzle dam design pressure. Furthermore, the entry time for mid loop level reasonably limits the makeup flow required to match boil-off to within several makeup schemes available in the System 80+ design (see Section 2.3.3.4). It fixes the time to boil (assuming an initial RCS temperature of 150 deg F) to greater than 15 minutes, which lengthens the time available for loss of shutdown cooling system (SCS) mitigation actions.

2.3.3.4 Alternate Inventory Additions and DHR Methods

The effective management of time and efforts is crucial to coping with a loss of shutdown cooling. Awareness of time constraints provides information that is useful in deciding how to allocate effort. If shutdown cooling cannot be restored within the time to core uncovery, getting a source of water lined up to keep the core covered becomes a first priority. Inventory makeup directly extends the margin of safety prior to uncovering the core.

Successful coping with a loss of shutdown cooling would include performing the steps outlined in Section 2.4 of this report. One of the last measures specified in that section includes adding makeup to the RCS to replenish beil off. This is thought of as a last resort measure in the System 80+ design due to the multiple success paths available to restore decay heat removal, and the time available to take corrective actions.

However, as required by Generic Letter 88-17, sufficient existing equipment should be maintained in an operable or available status so as to mitigate a loss of RCS inventory should core boiling or an uncontrolled and significant inventory loss occur. Generic Letter 88-17 also recommends that the water addition rate capable of being provided by each of the means should be at least sufficient to keep the _ore covered. Finally, Generic Letter 88-17 states that the path of water addition must assure that akeup flow does not bypass the reactor vessel before exiting any opening in RCS.

For the System 80+ design, at least 2 available means of adding inventory to the RCS will be available whenever the RCS is in a reduced inventory condition. Operating guidance is provided to specify the source of makeup water, the means of providing makeup to the RCS, and the recommended implementation strategy. The guidance will designate makeup pathways that ensure that makeup water does not bypass the reactor vessel.

The water addition rate capable of being provided should be at least sufficient to makeup for the boil off rate. This keeps the core covered and provides an adequate degree of protection for loss of decay heat removal scenarios. With the earliest nozzle dam installation occurring at 4 days post trip, the decay heat present would require approximately 135 gpm of makeup flow to compensate for boil off. This value is based on the pressurizer manway being opened, venting steam to the pressurizer cubicle. The steaming rate, and therefore, the required makeup rate, will reduce beyond 4 days post trip. The exact makeup rate may require adjustment based on actual, as-built conditions.

For Mode 5, reduced inventory operations, a shutdown cooling pump, the containment spray pumps, or safety injection pumps will be utilized as described in Section 2.4 to provide pumped makeup (for boil off) should SCS be lost. Procedural guidance will caution operators on using a containment spray pump in the same loop as the affected shutdown cooling pump, especially if the shutdown cooling pump has been lost due to air entrainment/pump cavitation. The makeup pump (shutdown cooling, containment spray or safety injection) will be aligned to the in-containment refueling water storage tank (IRWST) as the preferred source for makeup. An alternate source for borated water is the boric acid storage tank (BAST).

For Modes 5 and 6 the charging system via a charging pump (or alternatively, a boric cid makeup pump) will be utilized to provide pumped makeup should all methods of decay heat and inventory replenishment delineated above be lost. The pump chosen will be aligned to the BAST.

If no method of pumped inventory addition is available, a source for gravity feed inventory addition is via the safety injection tanks (SITs). This is applicable in Modes 5 and 6, and is considered only as a last resort. This method of gravity feed will only be implemented if SCS is lost along with all other means of supplying water to the RCS, and RCS boiling is occurring. If at least 2 tanks available provides assurance that over 20,000 gallons of water are available for discharge. Inventory addition from 2 SITs would assure approximately a 5 1/2 ft. rise in RCS water level and would make up for approximately 3 hours of core boil off.

Use of the shutdown cooling pumps, containment spray pumps, or safety injection pumps ensures that makeup flow will not bypass the core regardless of postulated openings in the RCS. Flow delivery will be through the direct vessel injection (DVI) nozzles. Even in the case of a DVI line break, sufficient inventory will be added to make up for core boil off. Use of the charging pump or boric acid makeup pump also ensures that makeup flow will not bypass the core, since it will be injected via the cold leg and enter the core via the normal charging path.

2.3.3.5 Operations

Procedures and Technical Specifications necessary to support the program are identified and will be implemented into the plant design. Procedural guidance for the conduct of mid-loop draindowns is provided to assure that no testing or maintenance activity adversely affects the NSSS during mid-loop operations. Guidance will be provided that assures that testing and maintenance activities performed during reduced inventory avoid operations that deliberately lead to perturbations in the RCS and all supporting systems necessary to maintain the RCS in a stable condition. These operations include (but are not limited to):

- RCS drain operations
- shutdown cooling testing and maintainance activities
- reactor coolant gas vent system testing and maintenance
- component cooling water testing and maintenance
- withdrawal of the incore instrumentation for refueling
- safety injection system testing and maintenance
- personnel communications system perturbations
- in-core instrument seal table evolutions while the reactor vessel head is on and mid loop operations are in progress.

Avoiding RCS and support system perturbations assures that adequate operating, operable, and/or available equipment of high reliability is provided for cooling the RCS and for avoiding a loss of RCS cooling. These actions also maintain sufficient existing equipment in an operable or available status so as to mitigate a loss of SCS or a loss of RCS inventory, should either occur. Adequate communications are essential to activities related to the RCS or systems necessary to maintain the RCS in a stable, controlled condition.

Due to the Diablo Canyon incident and other industry events, the requirements for evacuating personnel from the containment building, closing of the containment building e ment hatch and containment air lock doors, and isolation of personnel seding outside containment were evaluated based on time to core uncovery criteria. A description of the containment closure conditions referred to, along with a description of containment closure design features, is contained in Section 6.0 of this report.

2.3.4 RESOLUTION

The resolution of the reduced inventory issue on System 80+ is complete. Resolution is comprised of the results of the analyses outlined above, related evaluations in Section 2.4 on availability of decay heat removal, Technical Specifications in Section 2.2 and procedural guidance in Section 2.1.

The System 80+ design reflects a comprehensive consideration of shutdown and low power risk by adequately addressing all Generic Letter 88-17 recommendations and other issues relevant to reduced inventory.

77a.wp(9212)bh

2.4 LOSE OF DECAY HEAT REMOVAL CAPABILITY

2.4.1 ISSUE

Events that have occuired at operating plants der nstrate the vulnerability during shutdown Modes to loss of decay heat removal. The variety of maintenance activities taking place at shutdown combined with the possible system and equipment interactions that may occur lead to many conceivable scenarios for experiencing a loss of decay heat removal. Three dominant design objectives have evolved from the emphasis placed on prevention of shutdown events:

- 1. Provide reductant Shutdown Cooling System capacity and identify alternate decay heat removal capability.
- Provide instrumentation to effectively monitor shutdown operations, including critical plant configurations such as mid-loop.
- 3. Provide flexible redundancy in AC power.

The System 80+ features that address these issues are presented below in the context of demonstrating an integrated design capable of avoiding unacceptable consequences from the entire spectrum of potential event scenarios.

2.4.2 ACCEPTANCE CRITERIA

All e.ent scenarios may be characterized by initiation, detection, mitigation and consequence. To measure the success of the integrated response of System 80+ to events initiated from Modes 2 through 6, two criteria related to the potential for radiological release are adopted here. Significant release can only occur from fuel cladding rupture resulting from heatup after the coolant level drops below the top of the active core. Therefore, the first acceptance criterion is that there shall be no fuel cladding failure resulting from postulated events, excluding LOCA, initiated from Modes 2 through 6. The second criterion is that the radiological exposure of the public to events resulting in the loss of decay heat removal shall be limited to a fraction of the loCFR100 limits that is specified in sections of this report where applicable.

2.4.3 DISCUSSION

In this section, an evaluation is presented of the System 80+ features that are designed to prevent violation of the above uriterion. Section 2.4.3.1 examines events and event initiators which potentially result in the loss of shutdown cooling leading to boiling. Causes of past events considered include mid-loop operation, power failure and operator error. Appropriate Technical

2.4-1

Specification limitations and procedural guidance are identified by the analyses and are provided in Sections 2.1 and 2.2.

Section 2.4.3.2 presents the features of System 80+ which help prevent a loss of decay heat removal due to the loss of AC power. This is one of the specific concerns identified in NUREG-1410. The discussion in this section is directly related to the means of coping with a loss of decay heat removal. This concern is also identified in NUREG-1410 and evaluated in Section 2.4.3.1.

Section 2.4.3.3 presents the features of System 80+ that help assure the availability of the diesel generator. This issue was also identified in NUREG-1410. Availability of the diesel generator has been a significant factor in numerous past events.

Taken together, these sections demonstrate the integrated capability of the System 80+ to prevent and mitigate a loss of decay heat removal to ensure that the acceptance criteria are not violated.

2.4.3.1 Shutdown Event Initiation and Analyses

4.3.1.1 Introduction

This section examines events which could result in a loss of the Shutdown Cooling System (SCS) due to various initiators (events which challenge the SCS, such as a loss of power, inadvertent closure of a valve in the pump suction line and air ingestion in the pump suction) under various plant configurations and modes of operation and the ways these events can be prevented, detected and mitigated.

The discussion is structured into three parts. The first part focuses on design features which improve the SCS's resistance to initiators. The emphasis here is on hardware design. The second part assumes a loss of the SCS, regardless of the initiator, and discusses the ability of System 80+ to recover from the event. Here too, the emphasis is primarily on hardware design. The third part recognizes the limitations of hardware design as a response to initiator; and the need to demonstrate that adequate redundancy is provided to cover all possible plant configurations. This will include the plant's ability to cope with a loss of DHR. The emphasis here is on operator actions, operating procedures and technical specifications in the context of the various plant configurations which can exist in Modes 4, 5 and 6.

2.4.3.1.2 Resistance to Initiators

Design improvements have been made to the System 80+ SCS that reduce the likelihood for a loss of DHR. This is, in part, the result of applying a "beyond single failure criteria" design philosophy to improve the SCS's ability to withstand a wide range of initiators, including a loss of power, equipment failure, control system failure and operator error. The major design features attributed to the SCS's increased resistance are summarized in Table 2.4-1. The SCS Piping and Instrumentation Diagrams are shown in Figure 2.4-2 through 2.4-4.

TABLE 2.4-1

SUMMARY OF SYSTEM 80+ SCS DESIGN FEATURES THAT INCREASE RESISTANCE AGAINST INITIATORS

SCS Nozzle at bottom of hot leg. Dedicated DHR function. Independent suction lines for each train. Elimination of auto-closure interlocks in suction valves. Elimination of cross train communication. Increased system design pressure. Improved flow control. Improved flow control. Improved protection against pump excessive flow conditions. Flexibility to reduce flowrates to maximize NPSHA. Improved RCS level instrumentation at mid-loop. Instrumentation to indicate incipient pump cavitation. Elimination of loop seals in suction lines. Improved AC power reliability.

The most important feature that was added for the System 80+ SCS is its dedication to the DHR function. No portions of the SCS are included in the Emergency Core Cooling System (ECCS) as has been done with past designs. This means the SCS components do not double for components credited in other safety systems during Modes 1 through 4. This single design change allows various SCS improvements and simplifications including the ability to perform routine maintenance outside of shutdown cooling modes (5 and 6).

The system is comprised of two dentical, redundant, and totally separate trains each capable of performing the required DHR function. Dedicated heat exchangers have been provided in each SCS train. Previous designs used a single heat exchanger for both the SCS and the conta nment spray system (CSS), and as a result this required system realignment as the plant moved through modes 4 to 5. However, the System 80+ design eliminates these manual actions which decreas, the potential for operator error during Modes 5 and 6.

Each SCS train has independent suction lines from the RCS hot-legs. There are no cross-connections between SCS trains. Direct vessel injection (DVI) introduced for System 80+ and the dedicated DHR function of the SCS have anabled each train to be separate. This allowed simplifications in the arrangements resulting in greater protection for each pump from suction line failures due to air ingestion and discharge line failures resulting from pump to pump interaction.

Interlocks are provided on the SCS suction isolation values to prevent these values from being opened when RCS pressure is above the SCS entry pressure. These interlocks are enabled when RCS pressure is slightly above the shutdown cooling initiation pressure. All interlocks to close these values, such as closure on high pressure, have been eliminated. During an overpressurization transient, the SCS will be maintained active to continue DHR. The SCS is designed to mitigate these events with low temperature overpressure protection (LTOP) using spring-loaded relief values and by an increased SCS design pressure.

System flow control and protection from pump overspeed has been improved. System flow control is accomplished using valving and fixed resistance orifices in each train (orifices are not shown in Figure 2.4-1). The orifices are sized to limit the maximum flowrate from the SCS pumps and adjustments to shutdown cooling flowrate to match decay heat levels is accomplished by modulating valves. This design philosophy not only minimizes seat wear due to high fluid velocities resulting from throttling, but also prevents pump excessive flow conditions.

Operating procedures for the SCS during reduced inventory operation provide minimum flowrates necessary to perform DHR as a function of time after shutdown. SCS flowrate will be decreased as the cooling requirements decrease from lower decay heat levels. The lower shutdown cooling flowrate increases the net positive suction head available (NPSHA) to the SCS pumps. This provides greater operational margin for the RCS during midloop when SCS NPSHA is at a minimum and the potential for cavitation is at a maximum.

The ability to accurately measure and provide the RCS fluid levels has been the cause of many incidents resulting in the loss of DHR. System 80+ has made many improvements in the instrumentation for measuring the liquid level in the RCS and data display to the operator in the control room. Further discussion on this topic appears in sections 2.3 and 2.8 of this report.

Improvements have been made in the instrumentation of the SCS to provide the operator with more information about critical points in



the system. The intent is to provide the operator with detailed system parameters so appropriate actions can be taken before the loss of DHR occurs. If a loss of the SCS does occur these parameters will aid in the correct and timely evaluation of the initiator thus decreasing SCS recovery time. Major new instruments which have been included in the System 80+ design are suction and discharge pressure indicators and SCS pump motor current indication. These instruments are all indicated in the main control room.

Suction piping arrangements have been simplified and improved. Several incidents have been attributed to the presence of loop seals in the suction lines that allow air to collect and lead to the reduction of NPSHA and air binding. System 80+ arrangements for the suction lines do not have loop seals and thereby enhance the ability of the pumps to survive low NPSHA conditions.

Improvements in AC power reliability are discussed in sub-section 2.4.3.2.

The System 80+ SCS design features presented above, and summarized in Table 2.4-1, provide a way to minimize a loss of the SCS. These design features also address initiators which are known to have defeated DHR systems in currently operating plants. A summary of these initiators and corresponding SCS deign features are provided in Tables 1-1 and 1-2.

2.4.3.1.3 Recovery from Initiators

Recognizing that some initiators may defeat the SCS, the System 80+ design will require that both SCS trains and one division of AC power be operable during Modes 5 and 6. This allows safety injection and containment spray equipment in the redundant division to undergo maintenance activities as necessary.

Table 1-2 provides a detailed listing of events that have resulted in the loss of shutdown cooling. Table 1-1 summarizes design features incorporated into System 80+ to prevent, detect and mitigate the effects of the events listed in Table 1-2. Consequently, a detailed listing of all potential initiators will not be provided in this section. Instead, initiators that result in the loss of DHR are categorized into four groups. This categorization is structured primarily to simplify the discussion but may also aid in constructing diagnostic loss of SCS procedures. These groups relate the initiator to a location in the system with respect to the SC pump. The instrumentation provided for monitoring the pump's performance identify whether the failure is in the suction line, discharge line, the pump itself, or a power failure. With proper diagnostic information from these groups, the operator can perform appropriate recovery actions to restore DHR. Table 2.4-2 identifies the groups, some representative initiators

in each group, a brief description of the event and the instrumentation available to detect the event. The discussion that follows examines how DHR can be recovered using this information assuming a loss of a SCS train.

2.4.3.1.3.1 Group I Initiators

Group I initiators include a failure in the suction side of the SC pump. Suction line initiators are the most common during the midloop operation. These would include air ingestion, inadvertent closure of a valve in the suction line, failure of a relief valve to close, leakage from the system and procedural errors. The result of any of these initiators is to reduce the NPSHA for the SC pump.

Information provided to the operator in the control room for detecting and diagnosing these events include various alarms and indicators. The SCS includes an alarm for a low flow condition during shutdown cooling. This will be the initial indication of a possible suction line initiator since its set point is above the onset of cavitation. The typical alarm used for SCS flow is set to indicate a drop in flow from the design value of 5000 gpm to approximately 3000 gpm. This, in conjunction with a low suction pressure, fluctuating motor current and near normal discharge pressure (during the onset of cavitation), will confirm a Group I initiator (SC pump suction).

2.4.3.1.3.1.1 Recovery During Node 5

The equipment available to recover from this initiator depends on the mode of operation and includes the redundant SCS train, one of two containment spray (CS) and two of four safety injection (SI) pumps. (Technical Specifications will require that two AC sources will be available to each division of class 1E AC power during reduced inventory operations in Modes 5 and 6. See section 2.2 of this report.) The CS pumps are identical to the SC pumps and provide a redundant source for DHR flow during Modes 5 and 6. The SI pumps provide a viable source of DHR flow in Mode 5 as their capacity will match the reduced decay heat generation rate.

If the redundant SCS train cannot be used to recover from a Group I initiator during Mode 5, the CS pumps can be used to re-establish inventory control and DHR. The CS pumps can be aligned to take suction from either the RCS hot-legs or the In-containment Refueling Water Storage Tank (IRWST). During a Group I initiator, however, the CS pumps, which are normally aligned to the IRWST, can be used to re-establish inventory control by injecting IRWST water into the reactor vessel through the DVI nozzles (see Figure 2.4-1). This alignment can also provide DHR using the SCS heat exchanger. This response requires operator action to open an SDS valve RCS to allow water to circulate through the RCS to effect DHR (by opening a Safety Depressurization System (SDS) valve), the manual opening of one (normally locked closed) cross-connect valve and the actuation of the CS pump from the control room. Once the event is terminated, either the original SCS train or the redundant train can be activated to resume DHR. If a group I initiator produces a loss of coolant accident outside of containment, the operator has over 24 hours to identify and mitigate the event based on CS pump flow to match boiloff rate assuming the source of water is the IRWST.

If the CS pump is not functional, the SI pumps can be used to reestablish inventory control by injecting IRWST water into the reactor vessel. DHR would then be performed by either the redundant SCS train after RCS level is recovered, by "break" flow, if the initiator provided an opening in the system, or by feed-andbleed by opening a SDS valve. In the extreme, a single SI pump can provide sufficient flow to match boil-off thereby extending operator response time to identify the initiator and terminate the event.

2.4.3.1.3.1.2 Recovery During Mode 6

The equipment available to recover from this initiator during Mode 6 includes the redundant SCS train and possibly the opposite division's CS pump.

In this mode, the primary recovery system will be the redundant SCS train. However, if it is not available, then a CS pump may be aligned to take suction from the RCS hot-leg and discharge into the DVI nozzles. The success of this action is dependant on the particular Group I initiator since the operable CS pump must use the same RCS suction as the defeated SCS train (see Figure 2.4-1) and the opposite division's CS pump may be inoperable due to maintenance.

2.4.3.1.3.2 Group II Initiators

Group II initiators include a failure in the discharge side of the SC pump. Discharge line initiators include inadvertent closure or opening of a valve and the inadvertent actuation or leakage from a relief valve. The result of Group II initiators is to change the SCS system resistance curve. The pump will respond in accordance with its characteristic curve. Specifically, for the closure of a valve in the discharge line, the system resistance will increase resulting in a decrease in DHR flow and power consumption with a concurrent increase in discharge head. Pump by-pass lines prevent pump operation at shutoff. For the inadvertent opening of a valve, there will be a reduction in system resistance which will produce an increase in flow and a decrease in power consumption at a lower pump head. Information provided to the operator in the control room to detect and diagnose these events include the same alarms and indicators discussed in connection with Group I initiators (sub-section 2.4.3.1.3.1). The instrumentation critical to identifying a Group II initiator includes the SC pump discharge pressure, flowrate and motor current indication (see Table 2.4-2).

2.4.3.1.3.2.1 Recovery During Mode 5

The equipment available to recover from this initiator during Mode 5 include the redundant SCS train, one CS pump and two SI pumps.

In this mode, the primary recovery system will be the redundant SCS train as the potential to lose the other SCS pump is unlikely. For the low probability case where the redundant SCS train cannot be used, the CS pump may be used to re-establish inventory control and DHR. If the CS pump is not functional, the SI pumps can be used to re-establish inventory control. DHR then would be performed by either the redundant SCS train after RCS level has been recovered, by break flow or by feed-and-bleed.

2.4.3.1.3.2.2 Recovery During Mode 6

The equipment available to recover from this initiator during Mode 6 include the redundant SCS train and possibly the CS pump. The success of using the CS pump is dependent on the specific Group II initiator since the CS pump shares injection lines with the defeated SCS train.

2.4.3.1.3.3 Group III Initiators

Group JIJ initiators include a mechanical failure of the SC pump. Table 2.4-2 shows some examples. Recovery from Group III initiators include activating the redundant SCS train or aligning the CS pump.

2.4.3.1.3.4 Group IV Initiators

Group IV initiators are due to a loss of AC power. Recovery from Group IV initiators include automatic actuation of battery power, actuation of the combustion turbine or, if the loss of power is local to the train, activating the redundant SCS train or CS pump. System 80+ AC power availability is discussed in section 2.4.3.2.

2.4.3.1.4 Recovery Based on Plant Configuration

Section 2.4.3.1.3 provided a general discussion regarding the recovery from initiators. This section examines the recovery from initiators for several specific plant configurations and modes. This analysis illustrates the capability of System 80+ to recover from losses of DHR and identifies new procedural requirements and

technical specifications to minimize such losses and facilitate mitigative actions.

Figure 2.4-1 facilitates the identification of several major plant configurations of interest for shutdown risk. The termination points shown in the figure relate to Table 2.4-3 which provides an analysis of the configuration identifying possible initiators, current applicable technical specifications, new technical specifications, additional procedural requirements, alternative support equipment and systems available to mitigate losses of DHR and recovery actions from initiators.

2.4.3.1.5 Conclusions

The System 80+ SCS design features provide the necessary redundancy, flexibility, and diversity to reduce the likelihood of losing decay heat removal due to a loss of the SCS. The features of the design, the Technical Specifications, and the procedure guidance allow shutdown activities within certain limits and provide operational guidance for system flexibility and assurance that a loss of decay heat removal is unlikely.



 PLANT STATES AND TERMINATION POINTS
 Figure

 EVETEN
 PLANT STATES AND TERMINATION POINTS
 2.4-1

TABLE 2.4-2					
SCS_ENSTRUMENTATION					
INITIATOR	ITEN	RESULT	INDICATORS/ALARMS		
I - FAILURE IN THE SUCTION LINE					
INADVERTENT SIGNAL CLOSES MOTOR OPERATED VALVE	SI-651, 6533, 655 OR SI-652, 654, 666	LOSS OF COOLING FLOW	LOW FLOW ALERM FLUCTUATING DISCHARGE PRESSURE CURRENT FLUCTUATIONS LOW SUCTION PRESSURE POSITION INDICATION ON VALVE CPERATORS	F1-302 & F1-305	
OPERATOR ERROR IN CLOSING SCS SUCTION ISOLATION VALVE	SI-106 & SI-107	LOSS OF COOLING FLOW	LOW FLOW ARM FLUCTUATING DISCHARGE PRESSURE CURRENT FLUCTUATIONS LOW SUCTION PRESSURE POSITION INDICATION ON VALVE OPERATORS	FI-302 & FI-305 P-302 & P-305 I-306 & I-307 P-300 & P-301	
LOW RES LEVEL RESULTING IN VORTEX FORMATION AND AIR ENTRAINMENT		PUMPS WILL CAVITATE RESULTING IN LOSS OF COOLING FLOW	LOW FLOW ARM FLUCTUATING DISCHARGE PRESSURE CURRENT FLUCTUATIONS LOW SUCTION PRESSURE RCS LEVEL		
11 - FAILURE IN THE 31	11 - FAILURE IN THE DISCHARGE LINE				
INADVERTENT SIGNAL CLOSES MOTOR OPERATED VALVE	s1-310 & 312, 601 OR s1-311 & 313, 600	SYSTEM RESISTANCE INCREASES CAUSING THE PUMP TO OPERATE NEAR SHUTOFF	LOW FLOW ARM F.UCTUATING DISCHARGE PRESSURE CURRENT FLUCTUATIONS LOW SUCTION PRESSURE RCS LEVEL	F1-302 & F1-305 P-302 & P-305 P-300 & P-301 1-306 & 1-307	
OPERATOR ERROR IN CLOSING SCS DISCHARGE ISOLATION VALVE	81-579, 578	SYSTEM RESISTANCE INCREASES CAUSING THE PUMP TO OPERATE NEAR SHUTDEF	LOW FLOW ARM FLUCTUATING DISCHARGE PRESSURE CURRENT FLUCTUATIONS LOW SUCTION PRESSURE RCS LEVEL	F1-302 & F1-305 P-302 & P-305 P-300 & P-301 I-366 & I-307	

		TABLE	2.4-2	
		SCS INSTRU	PREMIATION	
INITIATOR	ITEM	RESULT	INDICATORS/ALARMS	
11 - FAILURE IN THE DI	SCHARGE LINE (CONT)			
INADVERTENT SIGNAL OPENS MOTOR OPERATED VALVE	SI-690, 691	SYSTEM RESISTANCE DECREASES CAUSING PUMP TO EXCEED ITS RUNOUT FLOW. ALSO, FLOW SPLIT MAY CAUSE RCS TO HEAT UP AS LESS FLOW WILL BE DELIVERED.	LOW FLOW INDICATION LOW DISCHARGE PRESSURE SUCTION PRESSURE DECREASES INCREASED POWER CONSUMPTION POSITION INDICATION ON VALVE OPERATORS	F1-302 & F1-305 P-302 & P-305 P-300 & P-301 1-306 & 1-307
VALVES IN THE IRWST TEST PATH NOT CLOSED FOLLOWING COMPLETION OF SC FULL FLOW TEST	\$1-315,693 & 301 OR \$1-304,686 & 300	PUMPS WILL DRAIN THE RCS INVENTORY INTO THE IRWST THROUGH THE TEST PATH THEN LOSE SUCTION AS THE FLUID IN THE HOT LEG DROPS	LIQUID LEVEL INSTR FOR MIDLOOP OPERATION RAPID DECREASE IN RCS PRESSURE RAPID DECREASE IN RC EVEL LIQUID LEVEL ALARMS IN THE IRWST TEMP INDICATION IN THE IRWST AFTER LEVEL JECREASES BELOW MIDLOOP LOW FLOW ALARM NORMAL DISCHARGE PRESSURE CURRENT FLUCTUATIONS	SEE SECTION 2.3 L-350, L-351 T-350, T-351 F1-302 & F1-305 P-302 & P-305 1-306 & 1-307 P-302 & P-305
SCS AC. 'E TRAIN IS USED TO FILL REFUELING POOL	SI-450 & 458 SI-454 & 455	THE SCS HAS BEEN DESIGLED TO SUPPORT EPRI REQUIREMENT 6.3.1.2 FOR REFUELING POOL (RFP) FILL. IF THE RFP IS FILLED USING THE TRAIN PERFORMING SC, THEN THE RCS INVENTORY COULD BE TRANSFERED TO THE RFP.	WO DETECTION UNTIL RCS INVENTORY DECREASES DECREASES BELOW THE MIDLOOP. LOW FLOW ALARM NORMAL DISCHARGE PRESSURE CURRENT FLUCTUATIONS LOW SUCTION PRESSURE	F1-302 & F1-305 P-302 & P-305 P-300 & P-301 I-306 & I-307
INADVERTENT CROSS CONKECT TO THE CONTAINMENT SPRAY SYSTEM	si-430, 431	LOSS OF COOLANT FLOW	LOW RCS LEVEL	

			TABLE 2.4-	-2 TATION	
INITIATOR	1169	RESULT		IND!CATORS/ALARMS	
III - FAILED PUMP					
SHAFT FAILURE		LOSS CF COOL	ANT FLOW LO	M FLOW ARM DISCHARGE PRESSURE	F1-302 & F1-305 P-305 & P-306

77a.wp(9212)bh

TABLE 2.4-3

Plant Configuration	Modes 4, 5 or 6
Initiators	Loss of power
Current Applicable Technical Specifications	LCO 3.8.1 - 3.8.8
New Technical Specification Requirements	See section 2.2
New Procedural Requirements	See section 2.1
Recovery From Initiators	See section 2.4.3.2 and 2.4.3.3.

Plant Configuration	Mode 4. RCS cold leg temperature less then 317°F. IRWST Full.
Initiators	Group I-IV. RCS line break.
Current Applicable Technical Specifications	LCO 3.4.6
	Two RCS loops or two SCS trains or any combination of these to be operable. One RCS loop or SCS train to be in operation. LCO 3.5.1
	Four SIT's operable when pressurizer pressure is greater than 900 psia. LCO 3.5.4 IRWST operable.
	LCO 3.6.6 Two CSS trains operable. LCO 3.8.2 AC Power (Shutdown)
New Technical Specification Requirements	Two SI pumps operable.
New Procedural Requirements	None
Alternative Support Equipment/Systems	None required.
Recovery From Initiators	DHR will be provided by sources other than the SCS when the RCS pressure is above [500] psig. During these conditions, the ECCS will be operable. The SIS will be available by automatic actuation down to RCS pressures of 400 psig (SIAS cutout pressure) and manually to 317° F. The CSS is operable through out Mode 4. Below [500] psig Group I - IV initiators can be mitigated per section 2.4.3.1.

Plant Configuration	Mode 4. RCS temperature greater than 317°F. IRWST full.
Initiators	Group I - III (for RCS pressure less then [500] psig) Group IV RCS line break.
Current Applicable Technical Specifications	<pre>LCO 3.4.6 Two RCS loops or two SCS trains or any combination of these to be operable. One RCS loop or SCS train to be in operation. LCO 3.5.1 Four SIT's operable when pressurizer pressure is greater than 900 psia. LCO 3.5.3 Two SIS trains operable. LCO 3.5.4 IRWST operable. LCO 3.6.6 Two CSS trains operable.</pre>
New Technical Specification Requirements	None
New Procedural Requirements	None
Alternative Support Equipment/Systems	None required.
Recovery From Initiators	DHR will be provided by sources other than the SCS when the RCS pressure is above [500] psig. During these conditions, the ECCS will be operable. The SIS will be available by automatic actuation down to RCS pressures of 400 psig (SIAS cutout pressure) and manually to 317° F. The CSS is operable through out Mode 4. Below [500] psig Group I - IV initiators can be mitigated per section 2.4.3.1.

Plant Configuration	Mode 5 RCS in reduced inventory. Nozzle dams installed. IRWST full.
Initiators Current Applicable Technical Specifications	Group I-IV. RCS line break. LCO 3.4.8 Two SCS trains operable. One SCS train operating. LCO 3.4.11
New Technical Specification Requirements New Procedural	LTOP operable with a maximum of two SIS train operable. LCO 3.8.2 AC Power (Shutdown) One CS pump operable. Two SI pumps operable. Midloop vent operable.
Requirements Alternative Support Equipment/Systems	Maintain shutdown cooling (SC) flow rate near the minimum required for DHR. Pumps Boric Acid Make-up pump. Tanks Safety Injection Tanks (SIT)
Recovery From Initiators	<pre>Boric Acid Storage Tank (BAST) Regain inventory control SC, CS or SI pumps can be used to inject IRWST water into the RCS to regain water level. If these pumps are not functional inventory control can be established using a charging pump or boric acid make- up pump by injecting BAST water into the RCS. Regain DHR capability DHR can be regained by using the redundant SCS train once level is recovered. If the redundant SC pump is not functional DHR can be established using the CS pump. If the CS pump is not functional, DHR can be established by feed and bleed using SI pumps and opening the SDS valves.</pre>
Time To Boil	Approximately 10-15 minutes.

Plant Configuration	Mode 5. RCS in reduced inventory. Nozzle dams not installed. RCS closed (mid loop vent or RCP seals). IRWST full.
Initiators	Group I-IV. RCS line break.
Current Applicable Technical Specifications	LCO 3.4.8 Two SCS trains operable. One SCS train operating. LCO 3.4.11 LTOP operable with a maximum of one SIS
New Technical Specification Requirements	One CS pump operable. Two SI pumps operable. Midloop vent operable.
Requirements	Maintain shutdown cooling (SC) flow rate near the minimum required for DHP
Alternative Support Equipment/Systems	Pumps Pumps Boric Acid Make-up pump. Tanks Safety Injection Tarks (SIT) Boric Acid Storage Tank (BAST) Steam Generators
Recovery From Initiators	<pre>Regain inventory control SC, CS or SI pumps can be used to i ject IRWST water into the RCS to regain water level. If these pumps are not functional inventory control can be established using charging pump or boric acid make-up pump by injecting BAST water into the RCS. Regain DHR capability DHR can be regained by using the redundant SCS train. If the redundant SC pump is not operable DHR can be regained using the CS pump. If the CS pump is not functional DHR can be established initially by reflux boiling, then by feed and bleed using SI pumps and opening the SDS valves.</pre>

Plant Configuration	Mode 5. RCS in reduced inventory. Nozzle dams not installed. RCS open (manway)
Initiators	Group I-IV.
Current Applicable	KCS line break.
Specifications	LCO 3.4.8 Two SCS trains operable. One SCS train
	LCO 3.4.11 LTOP operable with a maximum of two SIS
	LCO 3.8.2 AC Power (Shutdown)
New Technical Specification	
Réquirements	One CS pump operable. Two SI pumps operable. Midloop vent operable.
New Procedural Requirements	Maintain shutdown cooling (SC) flow rate near the minimum required for DHR
Alternative Support Equipment/Systems	Pumps Charging pump. Boric Acid Make-up pump.
	Tanks Safety Injection Tanks (SIT) Boric Acid Storage Tank (BAST) Steam Generators
Recovery	Devisin inventery senturi
FION INICIALOUD	SC, CS, SC or SI pumps can be used to inject IRWST water into the RCS to regain water level. If these pumps are not functional inventory control can be established using charging pump or boric acid make-up pump by injecting BAST water into the RCS. SITs can also be used. Regain DHR capability DHR can be regained by using the redundant SCS train. If the redundant SC pump is not operable DHR can be regained using the CS pump. IF the CS pump is not
	functional, DHR can be established by reflux boiling, or feed and bleed using CS or SI pumps and utilizing the open pressurizer manway.

77a.wp(9212)bh

TABLE 2,4-3

Plant Configuration	Mode 5. RCS not in reduced inventory. Nozzle dams installed. IRWST full.
Initiators	Group I-IV. RCS line break.
Current Applicable Technical Specifications	LCO 3.4.8 Two SCS trains operable. One SCS train operating. LCO 3.4.11 LTOP operable with a maximum of one SIS train operable.
New Technical Specification Requirements	Two SI pumps operable.
New Procedural Requirements	One CS pump available. Midloop vent operable.
A: ternative Support Equipment/Systems	Pumps CS pump. SI pump. Charging pump. Boric Acid Make-up pump. Tanks Safety Injection Tanks (SIT) Boric Acid Storage Tank (BAST)
Recovery From Initiators	Regain DHR capability DHR can be regained by using the redundant SCS train. If the redundant SC pump is not functional DHR can be regained using the CS pump. If the CS pump is not functional, DHR can be established by feed and bleed using SI pumps or utilizing the open pressurizer manway.

0

	TABLE 2.4-3 TERMINATION POINT 8
Plant Configuration	Mode 5. RCS water level above reduced inventory. Nozzle dams not installed. IRWST full.
Initiators	Group I-IV. RCS line break.
Current Applicable Technical Specifications	LCO 3.4.8 Two SCS trains operable. One SCS train operating. LCO 3.4.11 LTOP operable with a maximum of one SIS
New Technical Specification Requirements	Two SI Pumps operable.
New Procedural Requirements	One CS pump available.
Alternative Support Equipment/Systems	Pumps CS pump. SI pump. Charging pump. Boric Acid Make-up pump. Tanks Safety Injection Tanks (SIT) Boric Acid Storage Tank (BAST)
Recovery From Initiators	Regain DHR capability DHR can be regained by using the redundant SCS train. If the redundant SC pump is not functional, DHR can be regained using the CS pump. If the CS pump is not functional, DHR can be established by reflux boiling, or feed and bleed using SI pumps and opening the SDS valves.

ð

Plant Configuration	Mode 6 Refueling pool empty IRWST Full
Initiators	Group 1-IV LOCA
Current Applicable Technical Specifications	LCO 3.9.5 Two SCS trains operable. One SCS train operating.
New Technical Specification Requirements	Two SI pumps operable.
New Procedural Requirements	None.
Alternative Support Equipment/Systems	One CS pump available.
Recovery From Initiators	Regain DHR capability DHR can be regained using the redundant SCS train. If the SCS pump is not functional, DHR can be regained using the CS pump and SCS heat exchanger. If the CS pump is not functional, DHR can be established using feed and bleed since the IRWST is not fully drained.

1

.

TABLE 2.4-3

a Historica (H

TERMINATION POINT 10

A DE LA DESCRIPTION OF A DESCRIPTION OF	
Plant Configuration	Mode 6. Refueling pool filled. Reactor vessel head off. Upper internals in place. IRWST empty.
Initiators	Group I-IV
	RCS line break.
Current Applicable Technical Specifications	LCO 3.9.4 For high water level, one SCS train operable and in operation. LCO 3.9.5 For low water level, two SCS trains operable and one in operation.
New Technical Specification Requirements	Two SCS pumps operable.
New Procedural Requirements	One CS pump available.
Alternative Support Equipment/Systems	Instrumentation Refueling Pool water level indication in addition to high and low level alarm. Pumps Charging pumps. Boric acid make-up pumps. Tanks Boric Acid Storage Tank (BAST)
Recovery From Initiators	Regain DHR capability DHR can be regained by using the redundant SCS train. If the redundant SCS pump is not functional, DHR can be established by either passive or active means as described in section 2.10.3.
	If DHR has been defeated due to an inter- system LOCA, DHR can be regained by matching boil-off using the charging pumps or boric acid make-up pumps injecting BAST water.

.

Plant Configuration	Mode 6. Refueling pool filled. Reactor vessel head off. Upper internals removed. IRWST empty.
Initiators	Group I-IV
	RCS line break.
Current Applicable Technical Specifications	LCO 3.9.4 For high water level, one SCS train operable and in operation. LCO 3.9.5 For low water level, two SCS trains operable and one in operation.
New Technics' Specificati Requirements	Two SCS pumps operable.
New Procedural Requirements	One CS pump available.
Altornative Support Equipment/Systems	Instrumentation Refueling Pool water level indication in addition to high and low level alarm. Pumps Charging pumps. Boric acid make-up pumps. Tanks Boric Acid Storage Tank (BAST)
Recovery From Initiators	Regain DHR capability DHR can be regained by using the redundant SCS train. If the redundant SCS pump is not functional, DHR can be established by feed and bleed.
	If DHR has been defeated due to an inter- system LOCA, DHR can be regained by matching boil-off using the charging pumps or boric acid make-up pumps injecting BAST water.
2.4.3.2 System 80+ AC Power Reliability

2.4.3.2.1 Introduction

This section presents the System 80+ features that increase the availability of electrical power to supply the Class 1E buses and the capability to restore power if the electrical source is interrupted. The electrical distribution system provides redundant and diverse sources of power to the Class 1E buses during shutdown modes and reduced in antory in the reactor coolant system and provides redundancy a 1 flexibility to insure re-energizing the Class 1E buses is post ble if power is interrupted.

2.4.3.2.2 Discussion

Electrical power sources need to be carefully managed during shutdown operations to maintain a desired level of safety. This is especially true during reduced inventory operations. Reduced inventory requires heightened awareness to manage the risks of maintaining an electrical source to the Class 1E buses and of insuring an alternate source is available. The potential for a complete loss of decay heat removal due to the loss of electrical power is lowered when the electrical supply requirements for shutdown modes and reduced inventory are managed properly.

The management and operation of these electrical sources will be guided by Technical Specifications for shutdown operations and reduced inventory. Technical Specifications will be written to identify the minimum acceptable electrical distribution system alignments for operating in shutdown modes and reduced inventory. The operation of the electrical distribution system during shutdown modes and reduced inventory can be guided by procedures for normal alignments and for aligning alternate electrical sources if normal sources are interrupted.

The electrical distribution system design will provide flexibility and redundancy to allow for the management of competing priorities during shutdown. These competing priorities include the need to perform maintenance on electrical system equipment versus the need to have electrical sources available to provide power to the Class 1E buses.

The System 80+ electrical system design (see Figure 2.4-5) provides the redundancy and flexibility to insure the risks associated with shutdown modes and re red inventory operations are lowered to acceptable levels. P is is accomplished by providing two independent divisions of AC Electrical Power. Each division has two 4.16 KV Safety Buses with three sources of electrical power. These three sources are:

- 1. Normal-Permanent Non-Safety Bus (PNS-Bus),
- 2. Alternate-Reserve Transformer and
- 3. Emergency-Diesel Generator.

The normal source (PNS-Bus) of power to the Safety Bus has three sources of electrical power. The three sources are: (1) Normal -The division related Unit Auxiliary Transformer (UAT) being powered from Switchyard Interface I through the Unit Main Transformer (UMT), (2) Alternate - The division related Reserve Transformer being served from Switchyard Interface II, and (3) Backup - the Combustion Turbine.

Therefore, the Class 1E Safety Buses have the potential to be fed from four different ultimate sources during shutdown modes and reduced inventory operations. These sources are:

- 1. Switchyard Interface I,
- 2. Switchyard Interface II,
- 3. Diesel Generator, and
- 4. Combustion Turbine.

This distribution system provides the shutdown manageme team with the flexibility to perform shutdown activities on source of power to a division 4.16 KV Safety Bus and still me tain other diverse sources of reliable electrical power to the 4.1 KV Safety Bus.

Along with the electrical system design features, the System 80+ Technical Specifications include shutdown modes and reduced inventory operation Limiting Conditions for Operations (LCOs). The LCOs provide minimum acceptable electrical distribution alignments. Guidance is also provided by procedure to the operation staff to insure available source alignments are identified whenever shutdown activities are in progress. Add tional procedural guidance is provided for aligning any available source(s) to the Safety Bus(es) if power to the bus(es) is interrupted. The procedure guidance and Technical Specifications are provided in Sections 2.1 and 2.2 of this report.

2.4.3.2.3 Conclusion

The System 80+ electrical distribution system design features provide the necessary reduce ency, flexibility, and diversity to reduce the likelihood of losi decay heat removal due to a loss of electrical power. The features of the design, the Technical Specifications, and the procedure guidance allow shutdown activities within certain limits and provide operational guidance for system flexibility and assurance that a loss of the decay heat removal is extremely unlikely.

2.4.3.3 Bystem 80+ Diesel Generator Availability

2.4.3.3.1 Introduction

The availability of the Diesel Generator and the Diesel Loading Sequencer to automatically start and load during shutdown modes of operation is one of the issues identified in NUREG-1410. The availability of the Diesel Generator instrumentation and control system to provide reliable indications and automatic trip signals for Diesel Generator protection during emergency operation (e.g. automatic start while in shutdown modes); and the availability of adequate information and indications to identify, diagnose, and correct Diesel Generator operational problems are significant to the overall maintenance of decay heat removal as presented in Section 2.4.3.1.

The Diesel Generator (DG) and Diesel Load Sequencer (DLS) provide emergency power to the Class 1E buses during shutdown modes of operation with the same methods used during power modes of operation. The Instrumentation and Control (I&C) system for the DG provides signals to start the diesel for emergency operation, applicable protective trips to prevent or limit damage to the DG at all times and DG status to the Control Room and to the local control panel. This status includes trip signals (alarms, indications and recordings), parameter indications, and alarms for abnormal parameters. Also, controls for starting, stopping, synchronizing, and loading the DG are provided in the Control Room and at the local control panel.

2.4.3.3.2 Discussion

The Diesel Generator (DG) and Diesel Load Sequencer (DLS) need to maintain a consistent means of operation independent of the plant operation condition. This ensures the operating staff is not required to learn different operating schemes and therefore reduces potential human error.

The System 80+ DG and DLS provides this simplicity of operation. The DG is the emergency source of power to the Class 1E bus. The DG and the DLS are available for operation during shutdown conditions unless undergoing maintenance. The Class 1E buses are monitored for undervoltage and degraded voltage conditions. If either condition is sensed, the DG is started and the DLS is initiated (see Figure 2.4-6 copied from CESSAR-DC Figure 7.3-5). For a loss of power to the Class 1E bus, the response of the DG and DLS is not dependent on plant operational modes. Therefore, the response of the System 80+ equipment provides the operator with the same parameters and indication to be monitored whether shutdown or operated at power. This design characteristic provides a basis for consistency in operating procedures and operator training. This eliminates the necessity of two sets of procedures dependent on plant operating conditions. It also eliminates extra required training for the operation staff. (Detail on the Emergency Diesel Generators can be found in CESSAR-DC Section 8.3.1.1.4).

The DG I&C system needs to ensure the diesel is protecter during all modes of operation. However, certain protective trip: sed to be bypassed during emergency operation.

The System 80+ Diesel Generator protection system provides automatic trips to prevent or limit damage to the DG. The protection trips provided during emergency operation are:

- 1. Engine Overspeed,
- 2. Generator Differential Protection,
- 3. Low-Low Lube Oil Pressure, and
- 4. Generator Voltage Controlled Overcurrent.

These trips are provided in accordance with Reg. Guide 1.9 Position 7. All other trips are bypassed during emergency operation. (See CESSAR-DC Section 8.3.1.1.4.4 for a complete description of trips bypassed during emergency operation). The protection circuitry is dependent on the initiating signal and not dependent on plant operational modes. The sensing of an undervoltage or degraded voltage condition during shutdown causes an automatic DG start, activates the protective circuitry, and bypasses all non-emergency trips. This circuitry allows for consistency in the operational response to an emergency start of the DG independent of plant operating mode.

The I&C system needs to ensure the operator is informed of the DG's operational status. This status includes paramater indications and alarms. The I&C systems need to provide controls to allow the operator to start and load the diesel to provide power to the Class IE buses. This status and control scheme needs to be provided locally and in the control room.

The System 80+ cori, room is designated as the Nuplex 80+ Advanced Control Complex (ACC). The Nuplex 80+ ACC presents the operator with the information and controls necessary to complete any tasks identified in a task analysis process. The task analysis for DG operation identit ... the parameters, alarms, and controls required to operate DG from the Nuplex 80+ ACC. This identified status and introl scheme is presented to the Control nm Operator on the ectrical Distribution Auxiliary Console. presentation of this information is accomplished in accordance a structural and hierarchial format discussed in CESSAR-DC Section 18.7.1. This formatting provides the operator with parameter displays, alarm status, alarm categorization, and alarm rior ty. This meth. I of information presentation provides the .int.ol Room Operator (CRO) with the tools necessary to monitor and/or liagnose DG status.

The System 80+ local control panels for the DG provides the Plant Equipment Operator with the same information and controls as is available to the CRO. The DG status information and control scheme on the local control panel utilizes the same Man-Machine Interface (MMI) features used i he Nuplex 80+ ACC. These features meet the System 80+ human factors standards and guidelines.

2.4.3.3.3 Conclusion

The System 80+ Diesel Generator instrumentation and control systems design features provide starting signals for the DG and DLS initiation and protective trip signals for DG emergency operation and provide DG status information to the control room and local control panel which allows the operator to operate, monitor, and diagnose DG and DLS operation. These features of the System 80+ design enhance the operator's interface with the emergency equipment and reduces the potential of human error.

2.4.4 RESOLUTION

The issue regarding vulnerability during shutdown modes to a loss of decay heat removal (DHR) is resolved for System 80+ by the design features for the Shutdown Cooling System (SCS), instrumentation and controls, electrical power distribution system, new technical specifications and procedure guidance described in this and other sections of this report. These features demonstrate the control of this report. These features demonstrate the control of this report. These features demonstrate the control of the section of this report. These features demonstrate the control of the section of the provide the necessary redundancy, flexibility and diversity to sign icantly reduce the likelihood of losing DHR.

THIS PAGE INTENTIONALLY BLANK

9

- 4.

.

1.0

OVERSIZE DOCUMENT PAGE PULLED

SEE APERTURE CARDS

NUMBER OF OVERSIZE PAGES FILMED ON APERTURE CARDS _____

9208070093-01-03

APERTURE CARD/HARD COPY AVAILABLE FROM

RFCORDS AND REPORTS MANAGEMENT BRANCH



SYSTEM 80+ ELECTRICAL DISTRIBUTION

SYSTEM

Figure

2.4-5



LOAD SHED SIGNALS

2.5 PRIMARY/SECONDARY CONTAINMENT CAPABILITY AND SOURCE TERM

2.5.1 ISSUE

This section addresses the ability of the containment to protect the public from the consequences of a release of radiation during the time the containment is oper.

This issue is related to events initiated in Mode 5 or 6 which have the potential for radiological release. The events which will be considered are the loss of decay heat removal capability initiated by either a loss of shutdown cooling or by a loss of coolant caused by either operator error or a pipe break.

Following a loss of decay heat removal not the result of a pipe break, a radiological release from an open containment can occur when the time for the core to reach saturation is less than the time to restore RCS cooling and, failing this, the additional time to evacuate, close and isolate the containment. The time for the coolant to reach saturation is a function of plant conditions at the time the event is initiated.

Time to restore includes the time to detect that decay heat removal has been lost plus the time to restore either shutdown cooling or initiate alternate means of cooling. Time to detect depends on the instrumentation available to detect that Primary System cooling has been lost. The time to restore decay heat removal depends on the available systems and procedures.

Once Primary System cooling has been lost measures must be taken to evacuate and seal the containment before the system begins to boil. The time to close and isolate the containment depends on:

Design, operation, condition and status of equipment to close penetrations, equipment hatches and personnel air-locks,

Procedures for routing material and lines through these openings,

Training of personnel

Conditions of pressure, temperature and radiation within the containment as the core uncovers.

77a.wp(9212)bh

2.5.2 ACCEPTANCE CRITERIA

The following acceptance criteria apply to the issue addressed in this section:

- Radiological exposure of the public to any event resulting in a loss of decay heat removal shall be limited to a small fraction of the limits stated in 10CFR100.
- Radiological exposure to the public to any event resulting in a pipe break shall be limited to the limit stated in 10CFR100.

2.5.3 DISCUSSION

2.5.3.1 Problem Formulation

At issue is the radiological exposure of the public, for events leading to a release of radiation, while the containment is open. The release will depend on the events considered, the containment integrity; e.g. access areas open, and technical specifications and procedures for closing the containment.

Conservative assumptions are made in the analysis that leads to the time in which the containment must be closed such that radiation levels for personnel inside the containment and at the site boundaries are within limits of applicable acceptance criteria. Containment closure depends on conditions of pressure and temperature within the containment following the start of the event. These will influence the time to achieve containment closure: pressure through the dose release rate outside containment, and temperature through the limits on work time within the containment.

The results are used to support recommended changes to technical specifications and/or procedures.

2.5.3.2 Containment Integrity

The integrity of the containment vessel is to ensure that the release of any radioactivity does not exceed the limits established in 10CFR100. Containment integrity is maintained in accordance with Technical Specifications (TS) in Modes 1, 2, 3, 4 and 5, with reduced inventory, and Mode 6, with reduced inventory or core alterations.

In Modes 1, 2, 3 and 4 the containment is required to be operable per TS 3.6.1 (Containment). Integrity exists when the items defined in the Definitions Section of the Technical Specification are satisfied. Additional Technical Specifications for containment personnel locks and containment isolation valves provide specific

2.5-2

actions and surveillance requirements to ensure containment integrity is not compromised.

In Mode 5, with the Reactor Coolant System (RCS) in reduced inventory, and Mode 6 during core alteration, or reduced inventory, containment integrity is maintained in accordance with TS 3.9.3 and 3.10.5 (Containment Penetrations).

2.5.3.2.1 Integrity Requirements

2.5.3.2.1.1 Modes 1-4

Maintaining containment integrity in Modes 1-4 is accomplished by ensuring compliance with Technical Specifications. Prior to entry into Mode 4 from Mode 5, all surveillance requirements of TS 3.6.1 (Containment), TS 3.6.2 (Containment Personnel Locks) and TS 3.6.3 (Containment Isolation Valves) are verified complete, in accordance with the applicable procedures.

2.5.3.2.1.2 Node 5

Mode 5 is divided into two operational conditions:

- 1. RCS Level above reduced inventory
- 2. RCS Level below reduced inventory

Entry into or out of these operational conditions is controlled by procedures and Technical Specifications and require verification by the Senior Reactor Operator.

2.5.3.2.1.2.1 RCS Level Above Reduced Inventory

There are no Technical Specification requirements on containment integrity in Mode 5 when not in reduced inventory. Therefore proceeding from Mode 4 to Mode 5 does not require compliance with Technical Specifications dealing with containment integrity. It is during this mode of operation that equipment for maintenance and refueling outages, and support personnel, are moved into and out of containment through the one equipment hatch and two personnel locks. Also during this mode the surveillance testing of containment penetrations is completed and verified in accordance with site specific procedures.

2.5.3.2.1.2.2 RCS Level Below Reduced Inventory

In Mode 5 the RCS may be drained to facilitate installation of the steam generator nozzle dams as well as other maintenance. Draining the RCS to a reduced inventory level (>3 feet below the reactor flange) requires (through the Containment Penetrations Technical Specifications) monitoring for any leakage of radiation through the penetrations. To maintain containment integrity, the equipment hatch and one of the two doors on each of the personnel locks must be closed during core alterations.

2.5.3.2.1.3 Mode 6

The potential for fuel handling accidents in Mode 6 establish the requirement that containment integrity be maintained. Thus the equipment hatch and one of the two doors on each of the personnel locks must be closed during core alterations.

Entry from Mode 5 to Mode 6 may require verification of containment penetration status. Since Mode 5 has two possible operational states, containment configuration must be satisfied and verified by the Senior Reactor Operator. Entry into Mode 6 from Mode 5 reduced inventory operation requires monitoring of containment penetrations for radiation leakage. Entry into Mode 6 from Mode 5, not at reduced inventory, requires verification of penetration status.

2.5.3.2.2 System 804 Containment Features

2.5.3.2.2.1 Building Arrangement and Ventilation

The containment openings are surrounded by the Nuclear Annex Building. Therefore, there are no direct openings to the outside environment, and all leakage and air flow from containment openings (personnel locks, equipment hatch, open penetrations) is exhausted into the Nuclear Annex.

The Nuclear Annex Ventilation System (a non-safety grade system) draws air from various points in the Nuclear Annex and exhausts to the unit vent. If high radiation levels are detected by the system radiation monitor, the exhaust flow automatically aligns to a filter train. The filter train consists of particulate filters and carbon absorbers to remove radioactive material prior to exhausting into the unit vent.

2.5.3.2.2.2 Personnel Locks

The personnel locks allow passage of the work force into and out of the containment during all modes of operation. System 80+ has two personnel locks; one at elevation 115+6 (Figure 2.5-1A), one at elevation 146.

Each personnel lock is a right circular cylinder approximately 10 feet in diameter with a door at both ends. The locks form part of the containment pressure boundary. Therefore, closure and sealing of the locks prevents leakage of radioactive material.

The design and testing of the personnel locks ensures its ability to withstand pressures in excess of the maximum pressure following

containment DBA. Closure of a single door assures containment integrity.

Each of the doors contains double seals and local leakage rate testing capability to provide pressure integrity. To effect a leak tight seal, the personnel lock design uses pressure seated doors. Any leakage passes into the Nuclear Annex.

Each personnel lock is provided with limit switches on both doora that provide control room indication of door position. The doors are interlocked, to prevent simultaneous opening thus compromising containment integrity during Modes 1-4.

The normal alignment of the personnel locks during the various modes of operation is listed in Table 2.5-1.

In Mode 5 with inventory greater than the reduced level (<3 feet below the flange) both personnel locks can be only opened during an outage when it is necessary to transfer equipment into and out of containment. Closure can be initiated by dispatching personnel from the control room if containment integrity needs to be restored. Closure of both doors can be accomplished within 10 minutes.

2.5.3.2.2.3 Equipment Hatch

The containment equipment hatch provides a means for moving large equipment and components into and out of containment. On System 80+ the hatch is 22 feet in diameter and located on the 146 elevation (Figure 2.5.-1B). Normal alignment of the equipment hatch during the modes of operation is listed in Table 2.5-1.

The hatch, when closed, is part of the containment pressure boundary. Sealing is by means of a double seal which is Type B leak rate tested in accordance with 10CFR50, Appendix J, prior to entry into Mode 4.

The equipment hatch is removed following cleanup of containment atmosphere and entry into Mode 5, at full inventory. The hatch moves horizontally on a rail systom. This design allows the hatch to be moved, allowing equipment to be transferred in and out of containment without interference. The rail system is designed to minimize hatch movement thus reduce closure time.

The rail system utilizes a AC powered trolley. This AC power is from a 1E bus which is normally supplied from offsite power through the Unit Auxiliary Transformers. On loss of offsite power, power can be supplied from the Reserve Transformer, Emergency Diesel Generator, or the Combustion Turbine. In the event of the failure of all power sources the trolley system is designed to be operated manually. Before proceeding to Mode 5, at reduced inventory, or Mode 6, reduced inventory, or core alterations, the equipment hatch must be closed. With or without AC power closure time is less than one hour.

After being set in place, the hatch is bolted. Technical Specifications require that in Modes 1-4 all bolts be in place and tightened. In Mode 5, with reduced inventory, and Mode 6, Tech Specs require that [four] bolts be in place and tightened. This minimum number of bolts is sufficient to secure the hatch so that no visible gap can be seen between the seals and sealing surface.

The hatch is designed to be pressure seated. Thus any increase in pressure inside the containment will act to seal the hatch. In addition any radiation leakage will be into the Nuclear Annex.

2.5.3.2.2.4 Penetrations

There are [100] fluid system penetrations in the containment vessel. Each penetration is provided with a means of isolation by the Containment Isolation System (CESSAR-DC, Section 6.2.4).

Procedures, to meet Technical Specification surveillance requirements, are provided for maintaining proper valve alignment to ensure containment integrity, prior to entry into Mode 4, Mode 5 (at reduced inventory) or Mode 6, reduced inventory of core alterations. In Mode 5, with the RCS level greater than reduced inventory, these penetrations are leak tested in accordance with 10CFR50, Appendix J. Mis-alignment of the valves can result in leakage paths limited by size of these normally small diameter (<.75 inches) valves.

2.5.3.3 Events Analyzed

The radiological release is a function of the mass of coolant entering the containment either as subcooled liquid or steam. In mode 5 this release can be the result of either a loss of shutdown cooling (Section 2.4) or a LOCA (Section 5.0).

The release of radiation in Mode 5, at reduced 'nventory, can be the result of the events discussed in Section 2.4 leading to a loss of shutdown cooling. The most conservative assumption is events in which the system is in a reduced inventory condition resulting in the minimum time for the coolant to reach saturation. Per EPRI ALWR outage guidelines, the earliest a plant will enter Mode 5, at reduced inventory, is 50 hours after shutdown. For events leading to loss of shutdown cooling the minimum time to reach saturation was 10.5 minutes. Core uncovery was reached 55 minutes after saturation. The mass-energy releases for this event are listed in Table 2.5-2.

For the LOCAs initiated from full inventory, discussed in Section 5.3.5.1, a break in the DVI line of 0.4 ft² or less results in a time to reach saturation of 1.7 minutes, a time to the initiate core uncovery of 7.17 minutes and an additional time of 16 minutes for the peak clad temperature to reach 2200°F. The mass-energy releases are listed in Table 2.5-3.

Mode 6 events are all assumed to result from loss of shutdown cooling and subsequent heating of the coolant to saturation. Per the EPRI guidelines for ALWR outages the earliest a plant will start refueling is 85 hours after shutdown. Decay heat consistent with this time was assumed in calculating mass-energy releases in Mode 6 (Table 2.5-4). The time to reach saturation, assuming an initial temperature of 135°F, with this decay heat rate is 14.6 minutes with an additional 125 minutes to core uncovery.

2.5.3.4 Acceptance Criteria

Per Section 2.5.2 Acceptance Criteria are based on limits to radiological exposure to the public stated in 10CFR100. However, radiation exposure and containment temperature will affect the ability of utility personnel to close containment within an acceptable time. Acceptance criteria are stated to meet site boundary limits of 10CFR100 and for utility personnel, based on utility guidelines.

2.5.3.4.1 Radiation Limits

Two limits on radiation levels will be considered; limits on exposure to utility personnel working in the containment and site boundary limits for release from the containment. These limits are used to establish the time at which the containment must be closed to prevent exceeding either the off-site or in-containment acceptance limits.

2.5.3.4.1.1 Loss of Shutdown Cooling; Site Boundary Limits

The acceptance criteria in Section 2.5.2 for Loss of Shutdown Cooling refer to limits based on a fraction of the whole body dose (25 rem) and thyroid dose (300 rem) mandated in 10CFR100. The fraction selected is related to the event probability per reactor year, or the event frequency.

The fraction is taken as 10% of the integrated whole body and thyroid doses stated in 10CFR100 for two hour exposure;

Whole Body dose < .10 (25 rem) = 2.5 rem Thyroid dose < .10 (300 rem) = 30 rem

2.5.3.4.1.2 Loss of Coolant Accidents; Site Boundary Limits

Per the acceptance criteria in Section 2.5.2 exposure to the public to any event resulting in a pipe break are the limits stated in 10CFR100;

Whole Body dose < 25 rem

Thyroid dose < 300 rem

2.5.3.4.1.3 Limits on Utility Personnel

Routine containment closure does not require special precautions outside of routine radiation work permits which provides workers with instructions on clothing required and the radiation exposure of the work areas. In addition the atmosphere both inside and outside containment is continually monitored for radiation levels. Maximum levels are established to limit exposure to workers both inside and outside containment and yet permit work to continue to mitigate the effects of any accident and close the containment.

2.5.3.4.1.3.1 Air Borne Radiation

Equipment hatch installation during an accident situation requires added precautions to protect workers from both internal and external exposure while work is being performed and the containment is being closed. Workers are protected from external contamination by being required to wear anti-contamination clothing. If airborne radiation levels in the work area reach or exceed .25 MPC workers are required to don one of the following types of breathing apparatus; forced flow respirator supplied from a breathing air system, self-contained breathing apparatus, full face cartridge respirator.

Work times for workers varies, depending on prior exposure history. Quarterly exposure of 520 MPCs have been established as the limit at which a worker will be required to exit containment. Each type of self contained breathing unit has associated with it a protection factor which reduces internal exposure, per the following relationship;

Received MPC * Hour = (Measured MPC*time)/Protection Factor

Protection factors and the equivalent measured MPCs used for determining the maximum time for utility personnel in containment are listed in Table 2.5-5.

Radiation levels are measured locally. Thus times at which exposure reaches unacceptable limits will vary with location within containment. The acceptance limit serves as an indication of the time limits based on radiation exposure.

2.5.3.4.1.3.2 Whole Body Radiation

Areas outside and inside containment are continually monitored for radiation. Maximum whole body dose of 2.5 rems, 10% the 10CFR100 limits, is permitted to allow work to continue to mitigate consequences of the accident.

2.5.3.4.2 Temperature Limits

A combination of environmental (temperature, relative humidity), and work related (type of clothing, type of work) factors influence the work time duration within containment, in addition to radiation exposure. The temperature, as radiation exposure, varies with location in containment. Thus, work times based on temperature limits will also vary. The acceptance criteria is not absolute but serves as an indicator of the work time based on average containment conditions.

Work times will also be influenced by the type of protective clothing. NUREG 1449 notes an upper limit on temperature of 160°F to avoid burning the lungs. However, a self contained breathing pack will provide air at a breathable temperature for a longer period of time.

Guidelines for the limits on the time in which work can be performed in high temperature humid environments are established in EPRI-NP4453 LRI. These "Stay Times" are based on an average, or global, wet bulb temperature, adjusted for type of clothing (work clothing without or with vapor barrier) and type of work (light, moderate or heavy). At the containment initial conditions of 100°F and 50% relative humidity, assuming no protective clothing, maximum work times for moderate work is longer than two hours. Where is with protective clothing the time is reduced to about 60 minutes.

Acceptance limits on temperature will be assumed based on the maximum times needed to closed containment in each of the modes and events considered;

For Mode 5 loss of shutdown cooling at reduced inventory, and Mode 6 with loss of inventory due to boil-off, a minimum time needed to close the personnel hatches of 10 minutes of moderate work is assumed.

For Mode 5 LOCA, a minimum time of 60 minutes of moderate work is assumed.

77a.wp (9212)bh

2.5.3.5 Analysis

Calculations were done to predict pressure, temperature and activity within the containment for the Mode 5 and 6 events discussed above. These calculations were done in two steps.

2.5.3.5.1 Thermodynamic Conditions

A lumped parameter nodal model is used to predict pressure, temperature versus time for a given rate of mass and energy flow into the containment. The model includes provisions for varying, as a function of time, flow areas open to the ambient.

Mass and energy are conserved in the vapor space for both condensible and non-condensibles components. The vapor space model assumes:

- 1. Complete and instantaneous mixing of the flows,
- 2. Quasi-static equilibrium for temperature,
- 3. All constituents are uniformly distributed,
- Dalton's law applicable to find total pressure; containment pressures is the sum of the partial pressures of the non-condensibles and vapor pressure of the steam.
- Ideal Gas law applicable for determining partial pressures of the non-condensibles (air),
- Steam partial pressures determined from steam tables, for both saturated and superheated conditions.

The code models a sump for the accumulation of condensed water, allowing for flashing of the water based on the temperature of the water exceeding the saturation temperature, based on containment pressure.

Heat losses to passive (walls) and active (fan-coolers) heat sinks are included as options.

The method has been validated by comparison with the more detailed, multi-node code, CONTRANS used for the detailed containment analysis shown in CESSAR-DC Chapter 6. The main limitation, as compared to CONTRANS, is in the representation of passive heat sinks by a slab geometry, rather than the more representative, multi-node model. This results in an over-prediction of heat removal resulting in lower containment temperatures. However, the present analysis assumes no passive or active heat sinks, resulting in the maximum values and rates of change of temperature and pressure with time inside the containment.

The following nominal initial conditions are assumed in the calculations;

Containment volume = 3,377,000 ft³

Pressure = 14.7 psia

Temperature = 100°F

Relative humidity = 50%

Based on values for the open areas, the analysis predicts thermodynamic conditions inside the containment versus time for the mass flow rate and enthalpy release for the Modes 5 and 6 events (Section 2.5.3.3). The calculation starts (calculation time =0) at the time at which the coolant reaches saturation.

2.5.3.5.2 Radiation Release

The amounts of activity inside and leaving the containment as a function of time are computed using results of the analysis for the thermodynamic variables in the containment; e.g., containment steam mass, pressure, temperature, integrated mass flow into the containment and integrated steam and air flow leaving the containment. The procedure calculates, for a given RCS activity:

- 1. Curies present in the containment,
- 2. Curies input to the containment,
- 3. Integrated Curies into the containment,
- 4. Curies discharged out of the containment,
- 5. Integrated Curies discharged out of the containment.

The model assumes perfect mixing in the containment of the incoming RCS activity without assuming any benefit from decontamination factors (DF=1). The amount of exiting activity is based on the volumetric discharge of the air-steam mixture.

The assumption of no heat removal in the thermodynamic calculation results in high values of pressure versus time, resulting in a conservatively high mass rate of flow though the open areas.

In determining atmospheric concentrations a two hour EAB dispersion factor of 4.97x10⁻⁴ was used (CESSAR-DC). Atmospheric releases were calculated assuming no mitigating effect of the filters in the Annex building surrounding the containment.

In calculating radiation limits for utility personnel a maximum control room atmospheric dispersion factor of 2.0 x10⁻⁴ was used (CESSAR-DC).

RCS specific activity for the events analyzed were taken as the Technical Specification limits of a gross activity of (100/E)

microcurie/gram and a dose equivalent I-131 specific activity of 1 microcurie/gram.

2.5.3.5.3 Results

The calculation for activity, pressure and temperature starts at the time the coolant seaches saturation and continues, assuming no recovery, through core uncovery.

The time limit of when containment must be closed is a function of when acceptance criteria for radiation exposure to the public are met.

The time available for closing the containment (closure window) is the difference between the acceptance criteria time limit and the time when the event is detected.

The earliest time for detection (see Section 2.8 for detection methods), resulting in the maximum closure window, is at the initiation of the event (Figure 2.5-2). The latest time the event will be detected is assumed to be when the coolant reaches saturation. This results in the minimum closure window for closing the containment. Closure times herein are conservatively based on the minimum closure window.

The dose rates and containment temperature are functions of the open area of the equipment and personnel penetrations and how long they are open. Per Section 2.5.3.2, in Mode 5, with a full inventory, both equipment and personnel penetrations are open, while in Mode 5, with reduced inventory, and for Mode 6 reduced inventory or core alterations, one door in each of the two personnel locks must be closed but not sealed.

For Mode 5 analysis for the LOCA, assumes full inventory, thus equipment hatches (380 ft^2) and personnel locks (60 ft^2) are assumed open, to maximize release to ambient.

Mode 5, Loss of Shutdown Cooling, at reduced inventory fluid level is at the mid-plane of the hot legs. For Mode 6 the fluid level is assumed at the level of the upper flange. This analysis conservatively assumes that both personnel locks (60 ft²) are open.

The thermodynamic and radiation analyses were done assuming that the equipment hatch and personnel lock areas remain open throughout the calculation.

Times at which the open areas must be closed, based on the exposure limits in Sections 2.5.3.4.1.1 and 2.5.3.4.1.2, are shown schematically in Figure 2.5-3:

- For LOCA, limiting site boundary doses to; < 25 rem whole body, < 300 rem thyroid for a two hour period following start of the release.
- For loss shutdown cooling, limiting site boundary doses to; <
 2.5 rem whole ' 'iy, < 30 rem thyroid for a two hour period following the lease.

The time available for utility personnel to complete containment closure are based on the limits in Sections 2.5.3.4.1.3 and 2.5.3.4.2;

- Limiting dose to utility personnel within containment of 520 MPCs,
- 2. Limiting utility personnel to a whole body dose < 2.5 rems,
- Temperature limits inside containment based on minimum work time of 10 minutes of moderate work for Loss of Shutdown Cooling events and 60 minutes of moderate work for LOCA,

2.5.3.5.3.1 Mode 5: Loss-of-Shutdown Cooling at Reduced Inventory

The mass-energy release for this case is listed in Table 2.5-2 (Mode 5, Loss of Shutdown Cooling). Results are shown in Figure 2.5-4.

2.5.3.5.3.1.1 Site Boundaries

The integrated two hour doses for both whole body and tyroid exposures are below the acceptance levels;

Whole Body(rem) = .216 < 2.5 Thyroid _em) = 4.54 < 30

Thus the minimum closure window based on protection to the public is over two hours.

2.5.3.5.3.1.2 Utility Personnel

The MPC level at which breathing protection must be used is reached almost immediately. The minimum closure window, based on the time equivalent MPC level for full face cartridge protection, is about 89 minutes. With the next level of protection (forced air) the MPC level after two hours is, Maximum MPC after two hours =15.85 < 520

The whole body rem levels after two hours, for utility personnel, are below the acceptance levels.

Whole Body Dose (rems) = .087 < 2.5

Containment temperature after about two hours is about 180°F. However at the ten minute limit needed to close the personnel locks the temperature is 109°F. This temperature is lower than the EPRI temperature and humidity (120°F, 50% RH) limit needed for 10 minutes of moderate work, providing work time of about 30 minutes in which to close the personnel hatches.

2.5.3.5.3.2 Mode 5: LOCA

The mass-energy release for this case is listed in Table 2.5-3 (Mode 5, LOCA). Results are shown in Figures 2.5-5.

2.5.3.5.3.2.1 Site Boundaries

The whole body and tyroid integrated doses after two hours are both below the acceptance limits for a LOCA;

Whole Body(rem) = 1.35 < 25 Thyroid(rem) = 97.8 < 300

The minimum closure window, based on site boundary limits is greater than two hours.

2.5.3.5.3.2.2 Utility Personnel

The lower equivalent MPC level for full face cartridge is reached almost immediately. However the maximum MPC levels with either air supplied or self-contained unit are not attained during the two hour period. Thus the minimum closure time window is greater than two hours.

Maximum MPC after 2 hours = 173.5 < 520

The rem levels for utility personnel are below the acceptance levels.

Maximum whole Body Dose after 2 hours = .54 < 2.5 rems

Containment temperature within the first 10 minutes of the LOCA rises to about 170°F and then, due to the decrease in mass flow from the break and increased flow out of containment, decreases to an equilibrium value of 130°F. Based on the EPRI guidelines minimum closure time window to close the containment would be less than the time of one hour to close the equipment hatch.

2.5.3.5.3.3 Mode 6: Refueling; Inventory Boil-Off

The mass-energy releases for this case is listed in and Table 2.5-4 (Mode 6, Boil Off of Inventory). Results are shown in Figure 2.5-6.

2.5.3.5.3.3.1 Site Boundary Limits

The two hour integrated release is below the acceptance limits for both whole body and thyroid exposures.

Whole Body(rem) = .227 < 2.5 Thyroid(rem) = 4.8 < 30

Thus the minimum closure window based on exposure to the public is in excess of two hours.

2.5.3.5.3.3.2 Utility Personnel

The MPC level at which breathing protection must be used is reached almost immediately. The minimum closure window, based on the lowest level of protection (face cartridge), is reached in about 86 minutes. The exposure limit with the next level of protection (Forced Flow) is, after two hours,

Maximum MPC after two hours = 16.3 < 520

The whole body levels for utility personnel are below the acceptance level.

Maximum whole body dose after two hours = .091 < 2.5 rems

At the 10 minute point, the time set to close the personnel hatches, containment temperature is about 110°F. The EPRI guidelines allows for 30 minutes of moderate work at this temperature.

2.5.4 RESOLUTION

At issue is release of radiation to the public during the time the containment is open in Modes 5 and 6.

Events in Mode 5 or 6 considered in which radiation may be released include loss of shutdown cooling or loss of coolant.

The time in which the containment closure must be completed is based on dose limits set by 10CFR100 and the rate of release for the events considered.

The time to close the containment depends on: design, operation, condition and status of equipment; procedures and training of personnel in closing the containment. Conditions of pressure, temperature and radiation within the containment influence the time available for utility personnel to close the containment.

2.5.4.1 Mode 5; Reduced Inventory (Loss of Shutdown Cooling)

Results of the analysis show that the integrated two hour whole body and thyroid doses are below the acceptance limits of 2.5 rem (whole body) and 30 rem (thyroid).

Radiation and temperature within the containment are within limits to provide a 10 minute closure window needed by utility personnel to close the containment.

2.5.4.2 Mode 5; Full Inventory (LOCA)

The analysis indicates that the integrated two hour whole body and thyroid doses are below the acceptance limits of 25 rem (whole body) and 300 rem (thyroid).

Internal radiation levels for utility personnel are within acceptable limits provided either air supplied or self-contained breathing units are used. Whole body dose is also within acceptance limits.

Temperature within the containment are at levels that could limit the time available for utility personnel to close the containment within one hour. The use of multiple crews would help but the temperature levels could permit only short work intervals per crew.

The following should be considered to either decrease the time needed to close and seal the equipment hatch and/or lengthen the work time within containment:

- The use of fan coolers to decrease the maximum temperatures thus permitting longer work periods,
- Improved design of the equipment hatch to reduce the time needed to close the containment,
- A review of procedures and training with the objective of reducing the time needed to close the containment following a LOCA.

2.5.4.3 Mode 6: Refueling Configuration (Inventory Boil Off)

Results of the analysis show that the integrated two hour whole body and thyroid doses are below the acceptance limits of 2.5 rem (whole body) and 30 rem (thyroid).

Radiation and temperature within the containment are within the limits necessary for the 10 minutes needed by utility personnel to close the personnel hatches in the containment.

TABLE 2.5-1

CONTAINMENT OPENINGS

Opening	Area		Normal		
	ft²	Modes 1-4	Mode 5 >Reduce: Inventory	Mode 5 <reduced Inventory</reduced 	Mode 6
Equipment Hatch	380	Closed	Open	Closed	Closed
Personnel Locks (2)	30/Lock	Closed	Open	1 door closed per lock	1 door close per lock

TABLE 2.5-2

MASS-ENERGY	RELEASE FOR	MODE 5	LOSS OF	SHUTDOWN	COOLING
Time (sec)		Mass Flo	2) 2)	En (Bt	thalpy u/lbm)
0		0			0
.025		13.8			1162.8
1000		13.8			1162.8
3000		13.8			1162.8
3300	Т	ime to r	each cor	e uncover	У
10,000		13.8			1162.8

1

5

MASS-ENERGY	RELEASE FOR MODE	5 LOCA
Time (sec)	Mass Flow (lbm/sec)	Enthalpy (Btu/lbm)
0	0	0
6.2	2128.8	267.9
100.1	1896.8	278.5
200.1	1383.5	292.3
300.1	1284.3	294.4
400.1	841.3	308.9
430	Time to reach	core uncovery
500.1	151.2	551.6
600.1	119.9	639,6
700.1	96.5	707.8
1100.2	98.4	430.4
1190	Time for Fuel to	Exceed 2200 F
1500.2	129.3	380.5
1900	134	383.4
2300	133.9	385.6
2500	133.4	386.2
3000	132	386.1
3500	131	384.1
4000	130.5	381
10,000	130,5	381

•

TABLE 2.5-3

eps:	2.	231	T .:	127	2	82	4
4	n.	D.	<u>44</u>	8.2	6.9	44	

MASS-ENERGY	RELEASE FOR MODE 6 INVENTO	DRY BOIL-OFF
Time (sec)	Mass Flow (1bm/sec)	Enthalpy (Btu/lbm)
0	0	0
1000	14.39	1150.5
4000	14.39	1150.5
6000	14.39	1150.5
7500	Time to reach core	uncovery
10,000	14.39	1150.5

TABLE 2.5-5

PROTECTION FACTORS AND EQUIVALENT MPC

Type of Breathing Unit	Protection Factor	Equiv. MPC*
Full Face Cartridge	50	2.6 x 10^4
Air Supplied Forced Flow	2000	1.04 x 10^6
Self-Contained Breathing	10,000	5.2 x 10^6

* Based exposure limit of 520 MPC



PERSONNEL LOCK LEVEL 115





A. MAXIMUM CONTAINMENT CLOSURE WINDOW



B. MINIMUM CONTAINMENT CLOSURE WINDOW



Figure

2.5-2







SYSTEM 30 5

MODE 5 SDC WHOLE BODY DOSE vs TIME Figure

2.5-.'9



THYROND DOSE (REM)

SYSTEM BO S Figure MODE 5 SDC THYROID DOSE vs TIME 2.5-4C




WHOLE BODY DOSE

75 SYSTEM

MODE 5 SDC UTILITY PERSONNEL WHOLE BODY DOSE vs TIME Figure

2.5-4E





MHOLE BOOY DOSE (REM)

MODE S LOCA WHOLE BODY DOSE VS TIME

Figure









Ł

3



MODE 6 BOIL-OFF WHOLE BODY DOSE vs TIME

SYSTEM ID 5

Figure

2.5-6B



2.5-6C





77a.wp(9212)bh

2.6 RAPID BORON DILUTION

2.6.1 ISSUE

The issues of the rapid boron dilution can be broken down into three categories as follows:

- The introduction of deborated water into the RCS via Shotdown Cooling System (SCS), which flows into the RCS through the Direct Vessel Injection (DVI) lines, during maintenance of inline components.
- 2. Introduction of a water slug into the RCS during startup or refueling operations, including a specific example from NUREG-1449 (Reference 3). In that example, a loss of offsite power has occurred and the charging pumps are returned on line, powered by the Emergency Diesel Generators. If the plant were in startup mode - i.e., deboration in progress - the charging pumps could continue to operate causing a "slug" of unborated water to collect in the lower plenum of the reactor vessal. If it is then assumed that offsite power is restored and the RCP's are restarted, then a water slug of deborated water can be injected into the core.
- 3. A potential boron dilution resulting from inleakage from the secondary side of a steam generator during a SGTR event.

All the above issues will be addressed in the discussion and resolution sections of this report.

2.6.2 ACCEPTANCE CRITEKIA

The acceptance criteria for the rapid boron dilution event should be consistent with the acceptance criteria that are necessary to meet the relevant requirements of GDC 10, 15 and 26. Specifically, these criteria are as follows:

- Pressure in the reactor coolant and main steam systems should be naintained below the RCS P/T limits (see Figure 3.4.3-1 of Technical Specification 3.4.3) or below 110% of the design value, whichever is less.
- Fuel cladding integrity shall be maintained by ensuring that the minimum DNBR remains above the 95/95 DNBR limit for PWRs and CPR remains above the MCPR safety limit for BWRs based on acceptable correlations (see SRP Section 4.4).
- An incident of moderate frequency should not generate a more serious plant condition without other faults occurring independently.

4. An incident of moderate frequency in combination with any single active component failure, or single operator error, shall be considered and is an event for which an estimate of the number of potential fuel failures shall be provided for radiological dose calculations. For such accidents, the number of fuel failures must be assumed for all rods for which the DNBR or CPR falls below those values cited above for cladding integrity unless it can be shown, based on an acceptable fuel damage model (see SRP Section 4.2), that fewer failures occur. There shall be no loss of function of any fission product barrier other than the fuel cladding.

The above criteria are the same requirements as the acceptance criteria for the Inadvertent Boron Dilution (IBD) event as stated in NUREG-0800 Section 15.4.6, Reference 6, with the exception of Item 5. This criteria states that the available operator action time be 30 minutes for an IBD event during refusing conditions and 15 minutes for startup, cold shutdown and power operation. This requirement is not applicable to a "rapid" boron dilution event.

2.6.3 DISCUSSION

2.6.3.1 Identification of Dilution Sources

A study was performed to identify possible flow paths of nonborated water which could potentially result in a water slug being injected into the RCS which subsequently finds its way into the core. The results of this study are presented in Table 2.6-1. The study concluded that considering restrictions on operations (see Section 2.1), the only source of non-borated water is the DVI lines. The maximum slug volume was determined to be 60 FT³. The study also considered the issues identified in Section 2.6.1. The conclusion as shown in the resolutions of Table 2.6-1 is that for the System 80+ design, the scenarios defined by these issues do not result in a potential source of a non-borated water slug.

2.6.3.2 Event Analyzed

As mentioned in Section 2.6.3.1, the only credible source of an unborated water slug is the DVI lines, the volume of which is a maximum of 60ft³. This event was thus analyzed to determine the impact on the core and reactor coolant system. Table 2.6-2 list the assumptions and initial conditions used in the analysis.

The water slug was assumed to be injected into the reactor vessel via the DVI lines at the maximum flowrate of the 4 high pressure safety injection (HPSI) pumps. After the water slug was injected, a reactor coolant pump was assumed to start in order to instantaneously flush the water slug through the reactor vessel system and into the core.

Two cases were analyzed. The first case assumed the plant to be in Mode 5 reduced inventory with a boron concentration resulting in a K-effective of 0.99 with all rods out (ARO) of the core. This configuration results in the least amount of water with the highest boron concentration prior to the injection of the water slug. The second case likewise assumed reduced inventory, and a boron concentration associated with ARO conditions. However, Mode 3 conditions were used. This resulted in a reduced fluid density and increased boron concentration. The purpose of the second case was to bound conditions in Mode 3, 3 and 6 (without reduced inventory) by maximizing the change in reactivity resulting from a larger differential in temperature and boron levels between the DVI injection water and the RCS water. The change in the boron concentration, and RCS temperature, resulting from an unborated cold water slug being injected into ARO boron conditions was quantified in terms of reactivity units and compared with the margin to criticality available. In Modes 3, 4, and 5 this value will be equivalent to the required shutdown margin, since the reactor trip breakers will be open. This is required per the technical specification associated with the low flow trip, which requires at least two reactor coclant pumps be in operation if the reactor trip breakers are closed.

The above two cases bound the consequences in Modes 3 through 5, since reduced inventory conditions were assumed, horon concentrations much higher than would be expected to occur with 6.5% Ak/k shutdown margin were used to conservatively calculate changes in boron concentration; in addition, saturated RCS conditions were also used, thus resulting in the minimum initial RCS fluid mass. Although the Mode 6 boron concentrations would be larger than those utilized in the analyses, the slug which reaches the core will have a higher boron concentration than that which resulted from the mode 3 analysis due to the higher initial RCS boron concentration. Since the minimum value of the boron concentration for the Mode 3 case was considerably higher than the critical boron concentration with all rods out in Mode 6, the result of the Mode 3 analysis verifies the acceptability of the Mode 6 case.

2.6.3.3 Mathematical Model

The above scenarios were modeled utilizing a Computational Fluid Dynamics (CFD) software package utilizing a 2-D model.

2.6.3.4 Results

The results of the above analysis demonstrated that with a rapid injection of an unborated water slug of 60 FT³ into the reactor coolant system, in conjunction with the operational constraints as stated in the Technical Specification identified in section 2.6.3.2

that the maximum positive reactivity addition for both cases 1 and 2 is less than 2%. This is significantly less than the available shurdown margin of greater than or equal to 6.5%. For Mode 6 the maximum positive reactivity insertion will result in a K-effective from criticality.

2.6.3.5 Conclusion

The analysis confirmed that the Acceptance Criteria, as stated in Section 2.6.2 has been met. The core remained substantially subcritical, thus RCS pressure and DNBR Limits were not violated.

2.6.4 RESOLUTION

The design of the System 80+ plant minimizes the possibility of a Rapid Boron Dilution event. Analyses have shown that the core remains subcritical when the maximum credible water slug is "flushed" through the RCS. The concern of a slug being produced by the charging pumps following a loss of offsite power is not credible in the System 80+ design. The issue of a slug forming as a result of a SGTR event will be prohibited by the Emergency Procedure Guides.

In summary, the issue of a rapid boron dilution event for System 80+ can be considered resolved.

-	
Sec.	
100	
ehek.	
100	
-	
r - 4	
-	
-	
12	
-	
(CP)	
100	
100	
100	
100	
- 25	
1	
-	
- FÜ	
27	
1.1	
1.	

TABLE 2.6-1

POSSIBLE FLOW PATHS OF WOM-BORATED WATER

FLOW PATH

SAFETY INJECTION SYSTEM

à

STANDBY

-

SYSTEM

RESOLUTION

- . RCS LEARAGE THROUGH 1st ISOLATION CHECK VALVE (S1-217,-227,-237,-267) REF: PFS-91-044 é
- LEAKAGE THROUGH SIS HOT LEG INJECTION ISOLATION VALVE (SI-522,-532) DILUTES SIS HOT LEG INJECTION LINE à

å.

à INADVERTENT REFILL OF SIS SECTIONS WITH NON-BORATED WATER POST-MAINTENANCE ú

NONE

SHUTDOWN COOLING SYSTEM SIS OPERATION

-

N

(ISOLATED) STANDBY

é

LEAKAGE OF RCS FLUID THROWGH 1st ISOLATION VALVE (SI-651,-652) ė

ź

LEAKAGE OF COMPONENT CODLING WATER THRONGH & RUPTURED SCHX TUBE ó

à

- RESULTS IN A DILUTED SLUG OF WATER (ASSUMED 0 ppm BOROW) WITH A VOLUME OF 15 CU. FT. PER DVI LINE (50 CU. FT. TOTAL)
- OPERATOR ACTION PREVENTS INJECTION OF SLUG PRESSURE INSTRUMENTATION PROVIDED IN LINE BY DIVERTING FLOW THROUGH SIT FILL AND TO DETECT LEAKAGE

DRAIN LINE

01 0350 OWNER/OPERATOR PROCEDURES MUST POR THAT REFILL THE SIS ... (NO -INCTICAL JOURCES NO NON-BORATED SOURCES OF WAT EXIST IN THE SYS BO+ DESIEW)

SIS PUMPS TAKE SUCTION FROM BORATED IRWST

- OPERATOR IS REGULARD TO WARM UP SCS LEAKAGE IS INTO A BORATED SCS, WILL WOT RESULT IN A SUUG OF PURE WATER INJECTING INTO RCS PER OPERATIONAL AND CHECK BORON CONC. BEFORE PROCEDURES
 - VALVES AND NORMALLY CLOSED GATE AND DILUTION WOULD BE BOUNDED BY CHECK LEAKAGE IS INTO A BORATED SCS GLOBE VALVES
- OPERATOR CHECKS BORON CONCENTRATION MAXIMUM OF IS 150 PSIG, PRESSURE WOULD GUICKLY STABILIZE BEFORE APPROXIMATE DILUTION OF 1 CALLON UPON SCS HEATUP, WILL DETECT DILUTION AND CORRECT BEFORE SIGNIFICANT DILUTION RESULTS INJECTION

- 1

1	c	1	
2	Ç	2	
1	1	1	5
1	2	4	
ï	ļ		
è	5	ļ	
ĥ	ļ		è
1	Ē	2	a
Ş	3	ł	
í,		8	
i	2	2	
i	Ļ	Į	

TABLE 2.6-1 (Continued)

POSSIBLE FLOW PATHS OF MIM-BURATED WATER

SYSTEM

FLOW PATH

- ú INADVERTENT REFILL OF SCS SECTIONS WITH NON-SORATED MATER POST-MAINTENANCE ú
- LEAKAGE OF CCW THROUGH RUPTURED SCHX ŝ
- (NON-1501 47ED) SCS OPERATION

rá.

- TUBE

INJECTION OF PURE WATER THROUGH CVCS PURIFICATION LINE INTO SCS á

ó

REACTOR COOLANT SYSTEM

ú

STEAM GENERATORS

STEAM GENERATOR TUBE RUPTURE RESULTING IN SECONDARY FLOW TO RCS ń

ń

ZUPTURED TUBE DURING HYDROSTATIC TEST LEAKAGE OF SECONDARY FLUID THROUGH WITH FUEL IN CORE, OF OF 800 PSIG (AFTER MAJOR STEAM GENERATOR MAINTENANCE) á

RESOLUTION

- OWNER/OPERATOR PROCEDURES MUST REQUIRE THAT NO NOM-BORATED SOURCES OF WATER ARE UPED TO REFILL THE SCS ... (NO PRACTICAL SOURCES EXIST IN THE STS. 80+ DESIGN)
- IF COUS PRESSURE > SCS PRESSURE, COM INFLOW WILL MIX WITH FLOW FROM HOT LEG, BURCH CONC. >0 ppm

k

ŝ

- PRESSURE FROM OPERATING SCS PUMP 15 LIKELY TO CREATE & dD SUCH THAT COM INFLOW IS PRECLUDED
- LOSS OF CCM INVERTORY INTO SCS WILL EVENTUALLY BE DETECTED BY CCM SURGE TANK LOW LEVEL ALARMS
 - COMPARED TO THE CESSAR-DC CMP, 15 ANALYSIS, INADVERT, BORCH DILUTION POSSIBLE VOLUME OF LEAKAGE AND RESULTING BORON CONC. IS SMALL EVENT
- THE CHARGING PUMP EVENT OF CESSAR-OC THIS EVENT IS NOT COINCIDENT WITH CHAPTER 15, SECTION 15.4.6
- CMLY SOURCE OF NON-BORATED WATER IN CVCS IS THE RMWST...DESIGN OF CVCS PREVENTS THIS SITUATION (I.E., MULTIPLE FAILURES ARE MECESSARY FOR THIS TO OCCUR)
- PROCEDURES WILL REGULARE THAT & POSITIVE OF EXIST BETWEEN PRIMARY AND SECONDARY SIDES REF: 841 440.109 OF 5.6.
- OPERATOR WILL DETECT PRESSURE CHANCE AND VERY UNLIKELY THAT FUEL WILL BE IN CORE CORRECT BEFORE STARTING RCPs DURING THIS TEST

å

[ABLE 2.6-1 (Continued)

Low and sector to provide the

SYSTEM

CHEMICAL AND VOLUME CONTROL SYSTEM a. (CVCS)

å

POSSIBLE FLOW PATHS OF WOM-BORNATED WATER

FLOW PATH

- UPON STARTUP FROM MODE 6, POWER IS LOST...RCPS AND CHARGING PUMPS SHUT OFF, DIESELS POWER UP AND CHARGING PUMPS CONTINUE TO PROVIDE FLOW TO VESSEL (PURE MATER). RCPS START ONCE OF STEL (PURE MATER). RCPS START ONCE OF DILUTED WATER INTO CORE REF: MRC IMFO. WOTICE 91-54
- b. UNABLE TO BORATE VCT DUE TO NITROGEN b. GAS BINDING OF BAMPS
- c. INJECTION OF PURE WATER INTO RCS FROM c. THE RMMST

RESOLUTION

ú

- CHARGING PUMPS ARE POWERED OFF AAC SOURCE (CAS TURBINE), PUMPS MUST BE MANUALLY ALIGNED TO AAC BUS BY OFERATOR, PROCEDURES DICTATE OPERATOR ACTION
- BASTS ARE NOT PRESSURIZED BY ANY GAS, VENTED TO GUMS., PRECLUDES GAS BINDING DF BAMPS
- DESIGN OF CVCS PREVENTS THIS EITUATION (1.E., MULTIPLE FAILURES ARE NECESSARY FOR THIS TO OCCUR) HOWEVER, CONSIDERED IN THE CESSAR-DC CHAPTER 15 AMALYSIS

Table 2.6-2

Rapid Boron Dilutions Analysis-Assumptions and Initial Conditions

Parameters

Conditions

		Case 1	Case 2
1.	RCS Liquid Volume	Mode 5, Reduced Inventory (mid-loop)	Mode 3, Reduced Inventory (mid-loop)
2.	RCS Temperature	210°F	572°F
3.	RCS Pressure	atmospheric	1250 psia
4.	CEA Configuration	N - 1	N-1
5.	RCS Boron Concentration	1% subcritical assuming ARO	1% subcritical assuming ARO
6.	Available Shutdown Margin	7/6.5% <u>A</u> K/K	7/6.5% <u>A</u> K/K
7.	Water Slug Volume, Boron Concentration	60 FT ³ , 0 ppm	60 FT^3 , 0 ppm
8.	Water Slug Injection Method	4 HPSI Pumps, Maximum Flow	4 HPSI Pumps, Maximum Flow
9.	Water Slug Temperature	40°F	40°F
10.	Single Failure	*	*

*No single failure will impact the event consequences.

2.7 FIRE PROTECTION

2.7.1 ISSUE

The risk of fire during shutdown operations is higher than when the plant is in power operation. This increase in risk is due to the presence of transient combustibles and ignition sources such as welding, grinding, and cutting operations necessary to support shutdown maintenance activities. Another risk is the reduced level of fire protection for systems such as the shutdown cooling and fuel pool cooling systems when the plant is in a shutdown mode, resulting in a higher susceptibility of failure due to fire.

2.7.2 ACCEPTANCE CRITERIA

A defense in depth philosophy shall be employed in the design of the fire protection system in order to reduce the overall shutdown risk due to fire. The elements in this defense in depth philosophy are:

- 1. Prevent a fire from occurring,
- 2. Promptly detect and suppress a fire,
- 3. Mitigate the consequences of a fire.

The fire protection features shall be independent from other features or systems which are routinely taken out of service during shutdown modes of operation.

2.7.3 DISCUSSION

For clarity the three elements of the defense in depth philosophy outlined above will be discussed in reverse order. Only Division 1 of a system is discussed; Division 2 is identical to Division 1.

2.7.3.1 Mitigation of Fire Consequences

DIVISIONAL SEPARATION

Shutdown Cooling System components for each division are completely separated from each other with 3-hour rated fire barriers with no communicating openings (see CESSAR-DC Figure 9.5.1-2 reproduced here as Figure 2.7-1). All penetrations within these barriers are sealed with assemblies that are qualified to maintain the integrity of the 3-hour rating. This assures that a fire involving one division of Shutdown Cooling System corponents will not affect the redundant division.

INTERDIVISIONAL SEPARATION

Within each division, the containment spray pump and the shutdown cooling pump can be interchanged with each other. These pumps can be used interchangeably with valve manipulations guided by approved procedures. For each division, the shatdown cooling pump is separated from the containment spray pump with 3-hour rated fire barriers and 3-hour rated fire doors for openings. The valve which allows switchover from one pump to the other is located in a This will enable operators to make the separate fire area. switchover without being exposed to a fire involving either the Containment Spray or Shutdown Cooling Systems. Finally, the containment spray pump is powered from a safety bus separate from the shutdown cooling pump. The safety buses are separated from each other with 3-hour rated fire walls. For example, the Division 1 Safety Bus A is located in Fire Area 65 and the Division 1 Safety Bus C is located in Fire Area 70 (see CESSAR-DC Figure 9.5.1-3 reproduced here as Figure 2.7-2).

This interdivisional mechanical and electrical separation assures the operating of shutdown cooling can be maintained if a fire occurs concurrent with the redundant division being out of service.

2.7.3.2 Detection and Suppression of Fires

DETECTION

Fire Area 38 contains the Division 1 containment spray pump and heat exchanger and Fire Area 41 contains the shutdown cooling pump and heat exchanger. These areas were evaluated during the recently completed System 80+ Fire Hazards Assessment. This assessment considered the fixed and transient combustible loads in these areas and the importance of the components to plant shutdown. Both areas will be equipped with full area coverage ceiling mounted ionization smoke detectors. These detectors provide an early warning alarm at the central fire alarm console in the event of a fire. Detector location and spacing is based on engineering analysis to optimize detector effectiveness. This analysis will be referenced in the System 80+ Fire Hazards Analysis to be completed later in the design process.

The detection system is highly reliable and will be kept in service at all times, even during shutdown modes of operation.

SUPPRESSION

The System 80+ Fire Hazards Assessment concludes that a fixed automatic suppression in the form of automatic sprinklers is not warranted. This is due to the minimal combustible loadings in these areas. This will be verified later by engineering analysis, which is similar to the analysis for detector layout and location, and will be referenced in the System 80+ Fire Hazards Analysis to be completed by the plant designer before operations.

Portable fire extinguishers and fixed manual fire hose stations provide manual fire fighting capability. The fire hoses are supplied from a dedicated fire protection water supply. Because of the fire barrier arrangement discussed previously, manual fire fighting activities can be accomplished without exposing either the redundant division equipment or interdivisional equipment to the effects of smoke or hot gases from a fire.

MANUAL FIRE FIGHTING

A fully trained and equipped on-site fire brigade would provide fire fighting activities for the System 80+. (See CESSAR-DC Section 9.5.1.9.3.) The brigade would be thoroughly familiar with the plant layout and will conduct sufficient fire drills and fire pre-planning to effectively control and suppress any credible fire. A documented pre-fire plan which outlines the necessary fire fighting strategies, will be prepared prior to plant start-up.

MAINTAINED LEVEL OF FIRE PROTECTION

The System 80+ fire protection system is not degraded or reduced during plant shutdown. There will be no reason to breach the fire boundaries, interrupt the detection system, or impair the fire hose (standpipe) system. All of these features are provided specifically for fire protection and are not shared with or dependent on any other systems or features.

2.7.3.3 Prevention of Fires

Prevention is the most important element in the defense in depth philosophy. When this element is successful there is no need to employ the other elements. To facilitate the implementation of this element, work place procedures and guidelines will be established by the owner-operator based on guidance provided by the plant designer. Procedural guidance would include control of combustibles, housekeeping, and control of hotwork. The preparation of these procedures will consider those areas in which a fire during shutdown modes of operation could pose a risk. The procedures will include requirements to reduce the risk of fire ignition during shutdown. For example, the control of combustibles procedure may establish a maximum amount and configuration of combustible materials that may be left unattended in any of these This will not be based solely on an arbitrary "good areas. engineering practice" approach, but will consider the amount of combustibles necessary to result in a fire that could cause unacceptable damage. The control of hotwork and housekeeping procedures will be developed by the owner-operat and implemented

so as to not place unnecessary restrictions on shutdown maintenance activities, yet will provide a high level of fire prevention.

2.7.4 RESOLUTION

The fire protection features provided by the System 80+ design are consistent with the acceptance criteria outlined in Section 2.7.2. These features will significantly reduce the risk due to fire during shutdown operation to an acceptable level. The combination of fire protection features resulting from employing the defense in depth philosophy will minimize the potential for fite damage to systems required for shutdown operations.

This issue has been resolved by the design features of System 80+.







all some



2.8 INSTRUMENTATION

2.8.1 ISSUE

Over the past several years, industry and regulatory concern with a loss of shutdown cooling has increased. Despite an emphasis on improved shutdown procedures, the frequency of some incidents has not been reduced, particularly for losses of shutdown cooling during mid-loop operations. Furthermore, the effects of a loss of shutdown cooling are more serious than originally realized. The Nuclear Regulatory Commission (NRC) has requested responses to several design issues related to Nuclear Steam Supply Systems (NSSS) operations while on shutdown cooling; specifically during reduced inventory operations.

Operators have, in many cases, had difficulty in determining plant parameters and equipment status during depressurized, shutdown conditions. This is due to the amount and quality of information available being marginally adequate or inadequate for prevention, recognition and mitigation of abnormal conditions in a timely manner. In particular, this information includes the reactor coolant system water level, reactor core exit temperature, and performance of decay heat removal systems.

Losses of shutdown cooling can be partially attributed to misleading, inaccurate, or erroneous vessel level indication, particularly when vessel coolant level is lowered to within the hot leg between the level required for steam generator nozzle dam installation and the level required to prevent vortexing in the shutdown cooling suction line. Refer to Figure 2.8-1. Providing an adequate fluid level in the hot leg above the level at which vortexing occurs will ensure that the shutdown cooling fluid will not entrain air. This scenario has been a contributor to the loss of shutdown cooling due to pump cavitation.

The NRC has recommended that advanced reactor designs include an enhanced instrumentation package which assures:

- that reduced inventory operations can be accurately and continuously measured. For example, accurate instrumentation can establish reactor coolant level anytime during the draindown process. Accurate level measurement can assist in differentiating between the anticipated dynamic effects of the draindown process and additional, unintended inventory losses; and
- 2. that a loss of decay heat removal event during reduced inventory operations can be readily detected. This ensures a timely response to a loss of shutdown cooling event. The instrumentation should "provide reliable indication of parameters that describe the state of the Reactor Coolant

System (RCS) and the performance of systems normally used to cool the RCS for both normal and accident conditions" (Reference 4).

The NRC has specified that instrumentation for reduced inventory conditions should provide both visible and audible indications of abnormal conditions in reactor vessel temperature and level, and decay heat removal system performance.

2.8.2 ACCEPTANCE CRITERIA

The instrumentation provided for reduced inventory operations in the System 80+ design will reduce the safety risks associated with shutdown modes of operation. Instrumentation will be provided to avoid causing or contributing to a loss of shutdown cooling at reduced inventory conditions, and to aid in correctly interpreting a loss of shutdown cooling, should one occur.

The following recommendations are taken from Enclosure 2 to Reference 4:

"At a minimum, provide the following in the Control Room (CR):

- 1. two independent RCS level indications when the reactor vessel (RV) head is on the vessel
- at least two independent temperature measurements representative of the core exit whenever the RV head is located on the top of the RV (we [NRC] suggest that temperature indications be provided at all times)
- 3. the capability of continuously monitoring decay heat removal (DHR) system performance whenever a DHR system is being used for cooling the RCS
- visible and audible indications of abnormal conditions in temperature, level and DHR system performance."

Also, Enclosure 2 of Reference 4 includes NRC concerns and suggestions on meeting these recommendations. These include, for example:

- "1. We suggest that licensees investigate ways to provide [accurate; temperature [measurements] even if the head is removed, particularly if a lowered RCS inventory condition exists.
- We expect sufficient information [be provided] to the operators that an approaching [DHR system] malfunction is clearly indicated.

- 3. We expect both audible alarms and a panel indication when conditions exist which jeopardize continued operation of a DHR system, as well as when DHR is lost.
- 4. The low limit of level indication must be below the level necessary for operation of the DHR system. Level information is necessary under loss of DHR conditions since it provides an indication of core coverage and ... of the time to core uncovery. It is also useful in mitigating the loss of DHR accident."

Section 2.8.3.2 of this report contains the description of the System 80+ instrumentation package for reduced inventory operations, including:

- the monitored parameters,
- instrumentation ranges and accuracies,
- alarm setpoints,
- instrument availability,
- display and monitoring capability, and
- quality assurance.

A summary of the System 80+ design features which meet each of the above mentioned NRC recommendations for instrumentation are provided in the following.

2.8.3 DISCUSSION

2.8.3.1 Instrumentation Design Basis

To effectively monitor the draindown process to mid-loop via System 80+ enhanced instrumentation, information obtained from plant analyses forms the basis for the instrument's design requirements.

Instrumentation specified for reduced inventory operations is based on analyses in the following areas:

- operations from a solid plant to mid-loop conditions (which define dynamic draindown characteristics);
- instrumentation features which will reduce the likelihood of operator error during shutdown operation;
- possible ways in which shutdown cooling can be lost while the plant is in a reduced inventory condition;

- flow dynamics of the shutdown cooling system (SCS), including those which contribute to vortexing;
- the plant response to losses of shutdown cooling, due to various initiators, including RCS thermal hydraulic effects and manometric effects; and
- mitigation planning aimed at the reinitiation of shutdown cooling, delaying the onset of boiling, and delaying core uncovery.

The design goals of the instrumentation package are to provide:

- prevention enhanced monitoring capabilities for prevention of a complete loss of SCS operation, and
- mitigation the timely response to a loss of SCS.

These goals have been achieved with the design features of the System 80+ instrumentation described in the following.

2.8.3.2 Instrumentation Description

Table 2.8-1 describes the instrumentation package for reduced inventory operations included in the System 80+ design. Additional describes are provided below.

2.8.3.2.1 Level

Four unique sets of instruments are provided for the measurement of level during RCS draindown and reduced inventory operations. These instruments make up the refueling water level indication system.

The first set of instruments is a pair of wide-range, dP-based level sensors. These sensors are provided to measure level between the pressurizer and the junction of each SCS suction line with the RCS during draindown operations. Another pair of dP-based level sensors is utilized to determine RCS water level once it is within the reactor vessel. These narrow-range level sensors function to measure level between the direct vessel injection (DVI) nozzle and the junction of the SCS suction lines with the RCS.

One wide-range and one narrow-range dP instrument are connected to each SCS suction line. Separate lower level taps are provided for each instrument. See Figure 2.8-2. Because of the location of the upper level taps, each of these dP instruments will operate with, or without, the reactor vessel head in place.

In addition to the dP-based instruments described above, two heatedjunction thermocouple (HJTC) systems will also be available for reactor vessel level measurement during Mode 5 reduced inventory operations. The first system displays the output from the two inadequate core cooling probes which are located inside the reactor vessel. The range of these probes extends from the reactor vessel head to the fuel alignment plate (See Figure 2.8-3). The measurement of RCS water level via these probes is limited only to those periods when the reactor vessel head is installed.

A second HJTC system provides narrow-range level indication for midloop operations via measurement of reactor vessel water level in the hot leg region. This system displays the output from two HJTC probes specifically designed with thermocouples clustered in the hot leg region (see Figu e 2.8-4). The benefit of this design is that it permits very accurate measurement when the reactor vessel is in the hot legs.

The HJTC systems compensate for the flow gradient across the core associated with the operation of only one SCS suction line. The HJTC instruments are located in areas of the core which minimize the effect of the core outlet nozzles. The HJTC sensors have an accuracy and response time consistent with the maximum draindown rate of the RCS. The HJTCs are designed so that instrument signal and power are transmitted on individual electrical conductors. Failure of one HJTC sensor will not result in a loss of signal from the remaining sensors. The measurement of RCS water level via these probes is limited to those periods when the reactor vessel head is installed.

The use of both wide-range and narrow-range dP instruments, and two pairs of HJTC probes for refueling water level monitoring provides highly reliable, redundant, and independent indication of reactor vessel water level. Overlapping instrument ranges provide continuous draindown measurement from the pressurizer to a level below that necessary for SCS operation. Since this level instrumentation is independent, common mode misoperation, or failures due to dynamic effects, will not be masked.

Each independent level instrument provides a suitable measurement, and is accurate, for its intended range of use. For mid-loop operations, the narrow-range HJTC probes provide accurate level measurement to within one inch of vessel level. This is critical since there is a very narrow margin between the RCS water level necessary for nozzle dam installation, and that required to prevent SCS pump cavitation. The refueling water level instrumentation is displayed and alarmed in the control room because of its importance to plant safety.

2.8.3.2.2 Temperature

Several instruments are available for continuous temperature measurements during reduced inventory operations with the reactor vessel head on. These include:

- core exit thermocouples (CETs),
- shutdown cooling heat exchanger inlet and return line temperature sensors,
- hot leg resistance temperature detectors (RTDs), and
- refueling water level instruments temperature sensor (HJTC probe only).

All provide representative indications of the core exit temperature when the shutdown cooling system is operational. If the shutdown cooling system is lost, the CETs, hot leg RTDs, and refueling water level instruments temperature sensors (HJTC) input are available to track the response to the loss of shutdown cooling or the approach to boiling.

Per Enclosure 2 to Reference 4, temperature measurement is provided with the reactor vessel head off. The temperature instruments operable during this mode are the hot leg resistance temperature detectors and, prior to fuel shuffle, the CETs. Core exit fluid temperature can be measured through the use of hot leg RTDs as long as the SCS is operable. Each RCS hot leg has a total of five RTDs which are located in the hot leg at the junction of the SCS suction nozzle. In relation to the hot leg horizontal centerline, two RTDs are located above the centerline, one is at the centerline, and two are below the centerline. Only the lowermost two in each hot leg will provide input to the temperature reading for mid-loop operations, since they will be the only ones in full contact with reactor coolant. The lowest probes penetrate the internal diameter of the hot leg pipe at approximately 10" below the midloop fluid level, thus assuring accurate readings are provided.

All temperature sensors will have associated alarms in the control room to be used as aids in determining the response to a loss of shutdown cooling and tracking the approach to boiling. Awareness of time constraints operator via training provides information that is useful for deciding how to allocate effort.

2.8.3.2.3 Shutdown Cooling System Performance

As stated in Enclosure 2 to Reference 4, sufficient information will be available to the control room operator to indicate an approaching shutdown cooling system malfunction. Indications of sufficient pump suction pressure and possible vortexing include unsteady pump current (as indicated by SCS/containment spray system (CSS) pump motor current), loss or reduction in shutdown cooling flow (as indicated by the shutdown cooling system flowrate), insufficient pump NPSH (as indicated by the pump suction pressure sensor), or indication of rising RCS level (as water is displaced by the air and vapor in the shutdown cooling system). If a pump gives indications of air ingestion or cavitation, alarms will prompt the operator to stop the pump immediately. As detailed in Section 2.8.3.2.5, shutdown cooling panel displays will include valve lineup information for critical shutdown cooling flowpaths.

2.8.3.2.4 Quality Assurance

The following instruments are designated as safety related and therefore within the scope of environmental qualification and quality assurance.

- core exit thermocouples
- hot leg resistance temperature detectors
- refueling water level temperature sensor (unheated thermocouple)
- refueling water level instrument (ICCI heated junction thermocouple based design)
- shutdown cooling flowmeter
- shutdown cooling heat exchanger inlet and return line temperature sensors
- shutdown cooling valve position indicators

The safety related designation of these instruments is a consequence of their required functions in other plant modes of operation, including for some, inadequate core cooling. The CENP Quality Assurance Program designates items which are safety-related as Quality Class 1 equipment, and therefore, are subject to the highest level of quality activity.

Enclosure 2 to Reference 4 states: "...we will accept the following for resolving the items identified in the letter: (2) reliable equipment in lieu of the comparable safety grade classification" The CENP Quality Assurance Program designates items which are not safety-related but nevertheless require a high level of quality activity, as Quality Class 2 equipment. In this case, where reliable and accurate instrumentation is required for reduced RCS inventory conditions, designating the instrument as Quality Class 2 requires that a quality program be implemented that assures that quality is commensurate with intended use. In the procurement of the instrumentation, appropriate technical requirements and quality requirements are specified in the purchase order to this end. The following list of Quality Class 2 instruments identified on Table 2.8-1 are classified as non safetyrelated:

- refueling water level indicator (wide and narrow range dP design),
- refueling water level indicator (clustered HJTC design),
- shutdown cooling pump suction and discharge pressure sensors, and
- SCS pump/CS pump ammeter.

2.8.3.2.5 Display and Monitoring Capability

Details of the NUPLEX 80+ Advanced Control Complex Information presentation and panel layout evaluation are described in CESSAR-DC Section 18.7. In addition to the following summary, refer to Section 18.7 for detailed or supplementary explanation of control room information presentation.

The operator obtains plant information from a number of sources in the NUPLEX 80+ control room, which include:

- A large plant overview status board known as the Integrated Process Status Overview (IPSO),
- 2. Alarm tiles and associated alarm messages,
- Discrete indicators which provide frequently used and important information,
- CRT display formats containing essentially all power plant information, and
- 5. Component and process control indicators.

There are a number of NUPLEX 80+ design features in 1 through 5 above that specifically implement indications, alarms, and displays applicable to depressurized, shutdown conditions. They are described in the following sections.

2.8.3.2.5.1 Integrated Process Status Overview (IPSO)

1PSO is used for quickly assessing overall plant status, organizing operational concerns, and establishing priorities for operator action. Information provided on the IPSO display includes:

- Major system and component statuses shown on an overview schematic which are representative of the current operating heat transport systems,
- Alarms to aid the operator in quickly identifying the location of important status information,
- Deviations from control setpoints and identification of improving or degrading trends to improve the operator's awareness of plant conditions, and
- Key representative parameters (e.g., RCS temperature and reactor vessel level).

Alarm windows are provided for plant critical functions:

-	Reactivity Control	-	Electrical Generation*
-	Core Heat Removal	-	Heat Rejection*
-	RCS Heat Removal	-	Containment Environment Control
-	RCS Inventory Cont.	-	Containment Isolation
-	RCS Pressure Cont.	-	Radiological Emissions Control
-	Steam/Feedwater Conversion*		

*For power production only

Nuplex 80+ alarms are mode-dependent and equipment dependent to ensure their validity for different operational conditions. For all modes, including shutdown and refueling conditions, individual sensed process parameter values and alarm states are used to determine critical function alarms, either directly or as processed by an algorithm that uses more than one (1) process parameter input. In either case, the operator quickly is made aware of the affected critical function(s). For example, a high core exit temperature alarm state would be used as an input to the Core Heat Removal critical safety function alarm during a loss of shutdown cooling.

The systems represented on IPSO are the major heat transport pathways and systems that are required to support the heat transport process. These systems include those that require availability monitoring per Regulatory Guide 1.47, and all major success paths that support the Plant Critical Functions.

The following systems have dynamic operating status representations on IPSO. Their identifying descriptors on the IPSO display are shown below:

- CC Component cooling water
- CD Condensate
- CI Containment isolation
- CS Containment spray
| CW | - | Circulating water |
|----|----|-------------------------|
| EF | | Emergency feedwater |
| FW | | Feedwater |
| IA | ** | Instrument air |
| SC | ** | Shutdown cooling |
| RC | - | Reactor coolant |
| SI | - | Safety injection |
| SW | - | Service water |
| TB | | Turbine bypass |
| SD | - | Safety Depressurization |

System information presented on IPSO includes system operational status, any change in operational status (i.e., active to inactive, or inactive to active) and the existence of alarms associated with the system. Alarm information on systems helps to directly inform an operator about possible underlying causes of critical function alarms. The IPSO display, as well as all display pages, is also available at any data processing system CRT, which includes control room panels, the control room supervisor's desk, assistant operator workstations, and the technical support station.

2.8.3.2.5.2 Alarm Tiles and Associated Alarm Messages

Alarm tiles are displayed on electroluminescent flat panel displays in the Discrete Indication Alarm System (DIAS). These tiles are functionally grouped and located on the appropriate control room panel. Shutdown cooling system alarm tiles are located on the Engineered Safety Features panel. This panel includes the controls for Safety Injection, the Safety Injection Tanks, Shutdown Cooling, Reactor Cavity Flood, Safety Depressurization, Emergency Feedwater, Containment Spray, IRWST, and Containment Isolation. Individual alarm inputs to the shutdown cooling alarm tiles include (for each train;:

- low shutdown cooling pump header pressure
- low shutdown cooling flow
- high shutdown cooling heat exchanger outlet temperature
- shutdown cooling pump motor current deviation

In addition, this panel will have a tile for RCS conditions, with individual inputs for shutdown, depressurized conditions:

- low RCS water level
- high core exit temperature
- low refueling cavity level

To ensure alarm validity, all NUPLEX 80+ alarms are mode and equipment status dependent, and signal validation of inputs is done where multiple signals of the same process parameter exist. These features eliminate nuisance alarms and help ensure a true "dark board" when alarms do not exist. These features enhance operator diagnosis of alarms when they do exist.

When alarm tiles in DIAS are acknowledged, the operator is presented with a DIAS display with alarm messages showing which of the alarm tile inputs caused the alarm.

2.8.3.2.5.3 Discrete Indicators

Discrete indicators are provided on the NUPLEX 80+ control room workstations to provide the operator with information that (1) is frequently used to assess system level performance, and (2) allows continued operation if the Data Processing System should become unavailable. Discrete indicators use validated process parameter inputs where multiple process parameter measurements exist, and include trend information for routine monitoring, and diagnosis of abnormal conditions. Where analog data is composed of different ranges of information, DIAS automatically shifts to the appropriate range, and indicates to the operator that a range change has occurred.

Discrete Indicator displays to support shutdown cooling for key parameters are on the Engineered Safety Features panel. These include:

Shutdown Cooling System (per train)

- inlet temperature
- outlet temperature
- pump header pressure
- flow
- heat exchanger inlet temperature
- heat exchanger outlet temperature
- pump motor current

Reactor Coolant System

- pressurizer level
- reactor coolant system level

77a.wp (9212)bh

- pressure
- core exit temperature

- refueling cavity level

2.8.3.2.5.4 CRT Display Fages

CRT display pages contain, in a structured hierarchy, all the System 80+ plant information that is available to the operator. The CRT pages are useful for information presentation because they allow graphic layouts of plant processes in formats that are consistent with the operator's visualization of the plant. In addition, CRT formats are designed to aid operational activities of the plant by providing trends, categorized listings, messages, operational prompts, as well as alert the operator to abnormal processes.

The IPSO display page forms the apex of the NUPLEX 80+ CRT display page hierarchy. Three levels exist below IPSO: general monitoring, system/component control, detail/diagnostic. Each level of the hierarchy provides an information content designed to satisfy particular operational needs.

The CRT displays are provided by the Data Processing System (DPS). Any display page is available at any CRT. Operator acknowledgement of CRT alarms also acknowledges the same alarm in DIAS (and vice versa). The CRT alarm actuation message indicates the cause of the alarm, similar to DIAS.

The shutdown cooling system will be shown on a Level 2 display, with more detailed information on two Level 3 displays, one per shutdown cooling train. These displays will include all necessary information to clearly describe the status and performance of the system. This includes system mimic, component activity (e.g., on/off or open/closed) component controllability (e.g., key valves locked open or closed), system parameters (e.g., temperature, level), and system/component alarms. The Level 2 display will include reactor coolant system level and core exit temperature to integrate the shutdown cooling and RCS status for this display. The RCS is also presented on a separate Level 2 display.

2.8.3.2.5.5 Component and Process Control Indicators

NUPLEX 80+ component control features (e.g., actuation/switches/ controls) provide the primary method by which the operator actuates equipment and systems. The shutdown cooling system controls are functionally grouped within a system mimic on the Engineered Safety Features panel. At that panel, shutdown cooling system control is integrated with DIAS alarm tiles important to shutdown cooling and with CRT display of the shutdown cooling system. Controls, alarms and CRT displays for other systems applicable to shutdown operations, such as component cooling water and safety injection, are available at that panel as well.

2.8.3.2.5.6 NUPLEX 80+ Alarm Characteristics

There are a number of special features in the design of the CUPLEX 80+ alarm system that support operator diagnosis of alarm conditions and that would be particularly supportive of depressurized, shutdown operations. These are:

- 1. Mode and Equipment Status Dependency
- 2. Audible Alarm Information
- 3. Stop Flash Feature
- 4. Operator Established Alarms
- 5. Operator Aids

In addition, the categorization of all alarms is considered in the bases for alarm display location. For instance, alarms that indicate approach to potential equipment damage, but do not affect critical function or success path status, are presented only on alarm tiles. These would not be included as input to alarms shown on IPSG.

A key feature to aid operator navigation in the CRT display page hierarchy also includes alarm categorization to assist the operator. This feature, the "display page menu", is on each CRT display page. The menu indicates alarms exist in various sectors of the hierarchy, and depending on the sector, the operator can distinguish between lower level alarms that are critical function or success path related, and those that are not (e.g., personnel hazard, or equipment damage).

By displaying critical function alarms, success path alarms, personnel hazard alarms and equipment damage alarms on unique display locations, the operator can rapidly determine the type and relative significance of alarms. For example, an Inventory Control critical function alarm, without a concurrent Volume Control success path performance alarm immediately suggests that inventory may be decreasing due to a non-success-path cause, such as a coolant leak, in which case indication for IRWST level, and containment temperature, pressure and humidity would be immediately checked by the operator. Similar distinctions can be made by the operator for single or multiple alarm conditions to assist the operator in quickly establishing needs and priorities for operator action.

2.8.4 RESOLUTION

The issue of instrumentation for shutdown operation is resolved on System 80+ by the instrumentation and control room displays described in the previous sections of this report. This instrumentation will meet or exceed the recommendations of Generic Letter 88-17, and will significantly reduce risk associated with operations during shutdown, particularly when the reactor is in a reduced inventory condition, as long as prior to the start of draining the reduced inventory instrumentation is placed into operation.

The NUPLEX 80+ Advanced Control Room Complex provides an overview display, indicators, CRT displays, and alarms that meet the acceptance criteria in Section 2.8.2. Indication and alarms are provided on discrete indicators, alarm tile windows and CRTs for RCS level and temperature. In addition, shutdown cooling system status and performance is monitored on CRTs. Shutdown cooling system performance is alarmed on IPSC, alarm tile windows and on CRTs. Also, all alarms are processed for their individual effect on plant critical functions such as reactivity control, core heat removal and RCS heat removal.

TABLE 2.8-1

REDUCED INVENTORY INSTRUMENTATION PACKAGE

Monitored Parameter	Instrument Type	Instrument Function	Range	Indication and Alarm Location	Comments
RCS Water level	Refueling Water Level Indication System (dP based design)	Continuous, redundant wide range RCS water level indication during draindown operations.	Wide Range: two instruments, each with a tap at hot leg/ SCS suction line interface, reference leg at top of pressurizer.	Control Room, with low and low-low level alarms.	Highly reliable. Meets NRC requirement for water level measurement to a point lower than that required for SCS operation.
RCS Water Level	Refueling Water Level Indication System (ICCI HJTC system)	Independent continuous level indication in the reactor vessel.	Top of the vessel down to the fuel alignment plate.	Control Room, with low and low-low level alarms.	Redundant exial strings of thermocouples from the vessel head to the fuel alignment plate. System provides excellent accuracy and continuous measurement.
RCS Water Level	Refueling Water Level Indicating System (clustered HJTC design)	Independent, continuous, narrow-range level indication in the réactor vessel.	Top of the vessel down to the fuel alignment plate.	Control Room, with low-low, low, high and high-high level alarms.	Redundant axial strings of thermocouples from the vessel head to the fuel alignment plate. This instrument is different from the ICCI HJTC system discussed above in that thermocouples are clustered in the hot leg regior to provice greater instrument accuracy (<1").
RCS Water Level	Refueling Water Level Indication (dP based design)	Continuous, redundant nerrox range level indication during reduced inventory operations.	Narrow range: two instruments, each with a tap at SCS suction line/ hot leg interface, reference leg at DVI nozzle.	Control Room, with low and low-low level alarms.	Highly reliable for mid loop operations. Meets NRC requirement for water level measurement to a point lower than required for SCS operation.

77a.wp (9212)bh

ò

TABLE 2.8-1 (Continued)

REDUCED INVENTORY INSTRUMENTATION PACKAGE

Monitored Parameter	Instrument Type	Instrument Function	Range	Indication and Alarm Location	Connects
RCS Temperature	CETs (thermocouple design)	Measures temperature of coolant exiting core.	Optimized for SCS and refueling modes. Will measure boiling. Approximate range 50-250 deg f.	Control Room, with alarms at high and high- high temperature.	Tracks approach to boiling. Temperature indication provided even when head is off vessel. Not evailable during fuel shuffling. Availability will be maximized.
RCS Temperature	Refueling water level probe (heated junction thermocouple based design)	Continuous, independent temperature measurement inside the vessel.	Optimized for SCS and refueling modes. Will measure boiling. Approximate range 100 - 250 deg F.	Control Room with alarms at high and high- high temperatures.	Indicates actual vessel water temperature. Tracks approach to boiling.
Hot Leg Temperature	Resistance Temperature Detectors (RTDs)	Measures core exit temperature in the hot leg at both SCS suction line regions. Redundant RTDs provided in each hot leg.	Optimized for shutdown operations. Approximate range 50 - 250 deg F.	Control Room, alarms at high temperature.	Temperature indication is affected by loss of shutdown cooling flow, since flow by the RTDs will not occur.
SCS Flowrate	Flowmeter	Decay heat removal system performance.	Bounds SCS pump flow range.	Control Room, includes low flow alarm.	One located in each SCS return line to the RCS. Can be used to measure CSP flow if CSPs are used for SCS.
SCS Pump/CS Pump Discharge Pressure	Pressure sensor	Measures individual pump discharge pressures.	0 to system design pressure.	Control Room with low pressure alarm.	One instrument located at the discharge to each pump. ldentifies individual pump status.
SCS Pump/CS Pump Motor Current	Ammeter	Measure current drawn by pump motor. Fluctuations show air entrainment.	0 to system design pressure	Control Room, Blarms with preset drop in current.	Confirms pump status (individual pump air entrainment) independent of pressure and flow indicators.

C

7

TABLE 2.8-1 (Continued)

REDUCED INVENTORY INSTRUMENTATION PACKAGE

Monitored Parameter	Instrument Type	Instrument Function	Range	Indication and Alarma Location	Comments
SCS Pump/CS Pump suction pressure	Pressure sensor	Measure pump suction pressure in each pump.	0 to system design pressure,	Control Room with low pressure alarm.	One instrument located at the suction of each pump. Identifies individual pump status.
SCHX Inlet and Leturn Line Temperature	Temperature sensor	Measures temperature in the suction and discharge lines of the shutdown cooling heat exchanger.	40 - 392 deg. F.	Control Room alarms at high temperature,	Temperature indication only available when SCS is operational,
SCS Valve Position Indication	Valve position indication open/closed or throttled.	Status of valve positions in the SCS.	Open/closed/thr ottled position ⊰ndication.	Control Room	Will provide information of system lineup status and available flowpaths.



2.8-1



SYSTEM DIFFERENTIAL PRESSURE INSTRUMENT TAP LOCATIONS SCHEMATIC 2.8-2



NOT TO SCALE (ONE OF TWO PROBES SHOWN)

SYSTEM 30 5

SCHEMATIC REPRESENTATION OF THE INADEQUATE CORE COOLING HJTC PROBES Figure

2.8-3



*EXAMPLE OF SENSOR POSITIONING. NOT INTENDED TO BE TO SCALE (ONE OF TWO INSTRUMENT PROBES SHOWN)

SYSTEM 30 5

SCHEMATIC REPRESENTATION OF THE NARROW-RANGE HEATED JUNCTION THERMOCOUPLE PROBES Figure

THIS PAGE INTENTIONALLY BLANK

2.9 ECCS RECIRCULATION CAPABILITY

2.9.1 153UE

The issue is the potential for loss of flow to the Containment Spray (CS) and Safety Injection (SI) pumps during accident conditions. System flow could be inhibited by a number of factors. These factors include:

- 1. Hydraulic effects, such as air ingestion and vortex formation.
- Debris in the IRWST resulting from maintenance activities or deterioration of insulation from actuation of containment sprays or from LOCA consequences, or
- 3. The combined effects of items (1) and (2).

2.9.2 ACCEPTANCE CRITERIA

The design of the System 80+ Incontainment Refueling Water Storage Tank (IRWST) and Holdup Volume Tank (HVT) and their associated debris-blocking devices shall comply with the requirements of General Design Critericn 35 of Title 10, Code of Federal Regulations, Part 50, Appendix A. Design Criterion 35 requires that "...suitable containment capabilities shall be provided to assure that ... the system's safety function can be accomplished." To satisfy this requirement, the IRWST and HVT are designed to provide a clean and reliable source of water to the SI pumps for long-term reci: sulation. The containment is designed to direct containment spray water and emergency core cooling water to the HVT and then to the IRWST.

The SIS shall meet the acceptance criteria specified in USNRC Standard Review Plan section 6.3, Emergency Core Cooling System, Revision 2. In particular, Section 6.3 of the SRP addresses the availability of an adequate source of water for the SIS. Acceptance criteria pertaining to the design of the containment emergency sumps are provided in SRP section 6.2.2, Containment Heat Removal Systems, Revision 4. These criteria address the drainage of containment spray water and emergency core cooling water to the recirculation suction points (sumps) and the screen assemblies surrounding these suction points. Regulatory Guide 1.82. Water Sources for Long-term Recirculation Cooling Following a Loss-of-Coolant Accident, Rev. 1, provides the guidelines for the design of the IRWST and the HVT, and the design of the screens associated with these tanks. Technical considerations related to this issue are detailed in NUKEG-0897, Containment Emergency Sump Performance, Revision 1.

2.9.3 DISCUSSION

Water introduced into the System 80+ containment from a RCS break or from containment sprays drains into the Holdup Volume Tank (HVT). This tank serves the purpose of the "containment sump." The Holdup Volume Tank is therefore the low collection point in containment. The contents of this tank are directed to the IRW3T through the two IRWST spillways (see Figure 2.9-3). The IRWST serves as the single water source of long-term recirculation for emergency core cooling and containment heat removal.

With the System 80+ design, it should be noted that the IRWST does not serve as the containment sump; this tank specifically serves as a storage tank for refueling water, a clean and reliable source of water for Safety Injection, and a heat sink for condensing steam discharged from the pressurizer.

The arrangement of the IRWST within containment meets the multisump requirement of Reg. Guide 1.82, Water Sources for Long-term Recirculation Cooling Following a LOCA. The general plant arrangement separates redundant trains of the SIS and the CSS. The divisional boundary provides complete separation between divisions and effectively creates two identical support buildings. The result is a plant arrangement with two SI pumps and one CS pump in each division. Within each division, the two SI trains (and each CS train) are separated by a quadrant wall to isolate the trains from each other to the maximum extent practical. Each of the four SI pumps has its own suction connection to the IRWST (see Figure 2.9-1) and each of the two CS pumps shares one of these four connections.

Following an accident, water introduced into containment drains to the Holdup Volume Tank. Debris that may exist in containment may be transported to the HVT with this fluid. Debris greater than 1.5 inches diameter is prevented from entering the HVT by a vertical trash rack, which is located at the entrance to the HVT (see Figure 2.9-3). The vertical trash rack is greater than six feet high and more than forty feet long. A debris curb exists at the base of this trash rack to prevent high density debris that may be swept along the floor by fluid flow toward the HVT from reaching the trash rack. The vertical orientation of the trash rack will help impede the deposition of debris buildup on the screen surface. Particles that are smaller than the trash rack mesh will enter the Holdup Volume Tank.

The Holdup Volume Tank is designed to function as a solids trap to help prevent debris from entering the IRWST. High density debris that makes its way through the trash rack will accumulate in the bottom of this tank. The IRWST spillways are located at a high enough elevation to assure that much of the higher-density debris (and debris that tends to sink slowly) wil' settle to the bottom of the HVT before spilling over into the IRWST. Debris that remains in suspension will make its way to the IRWST spillways. The spillways are shown in Figure 2.9-3. Screens are not present in these so llways to assure uninter. upted flow to the IRWST.

The fine debris that is introduced into the IRWS? is prevented from entering the SIS suction p.ping by a debris screen. These screens are located at each end of the four wing walls that serve as supports for the reactor coolant pumps (see Figures 2.9-1 and 2.9-These wing wall assemblies extend from the IRWST floor to the 2) . maximum IRWST water level, assuring that all debris will be filtered before reaching the SIS suction lines. The screen assemblies completely enclose the suction lin by running from the end of each wing wall (.) the side walls of the Holdup Volume Tank or the primary shield walls, as applicable. The wing-wall screens have the capability of removing particles greater than 0.09 inches diameter This screen size is consistent with the screens used on currently operating units. The wing-wall screens are the final barrier to debris before the SIS suction lines.

Blockage of the dobris screens is a major concern with respect to recirculation. The System 80+ screens have a vertical orientation to prevent debris from settling on the screen surfaces. This helps in keeping the screens clear. The design considered the types and quantities of insulation used for the System 80+ components, since post-LOCA deterioration of this insulation is the major potential source of debris in containment. The location of insulation with respect to a HVT and IRWST as well as the possible location of breaks have lso been considered. The effective areas of the screens have een determined according to the guidelines provided in Appendix . to Regulatory Guide 1.82, Guidelines for Review of Sump Design and Water Source for Emergency Core Cooling.

The debris screens have been designed to withstand the vibratory motion of a seismic event without loss of structural integrity. Each screen is capable of withstanding loads imposed by postulated missiles as well as loads due to pressure head differentials.

Consideration has also been given to the materials used for the debris screers. Materials have been selected to avoid degradation during periods of inactivity (i.e., no submergence), and during periods in which the screens are partially or fully submerged.

Each screen used in the System 80+ design is provided with an access opening to allow for inspection of the racks or screens. The screens will be visually examined periodically to detect any corrosion or structural degradation during refueling outage periods. As seen in figure 2.9-1, the suction lines are located within the confines of the wing wall away from the IRWST spargers. This wall design nelps isolate the suction lines from the open sections of the IRWST, where most of the maintenance activities will be performed. The fine wing-wall screen will filter any trash generated from this type of activity. In the event that maintenance is needed within the wing walls and near the ECCS suction inlets, permanent box-like screens over the suction miping will protect these lines.

Long-term return of spray water from upper level elevations is not dependent on individual piping runs or spillways. Multiple passive spillways are provided to route water back to the Holdup Volume Tank. Major openings such as hatches and stairwells are also available to return water to the screened entrance to the HVT.

Protection against air ingestion by SIS pumps is also a major concern with respect to recirculation and has been considered in the System 30+ design. The location and size of the suction lines in the IRWST have been chosen such that air entrainment is Pump air ingestion analysis is based on minimum minimized. submergence, maximum Froude number, and maximum pipe velocities. The available surface area used in determining the design coolant velocity has been calculated conservatively to account for blockage that may result as per Reg. Guide 1.82, Appendix A, Guidelines for Review of Sump Design and Water Sources for Emergency Core Cooling. The minimum water level in the IRWST has been conservatively calculated to be 75+6 (elevation). This water level allows for sufficient NPSH for the C stainment Spray pumps and Safety Injection pumps operating at runout flow. A conservative margin has been provided between the elevation of the suction piping opening and this minimum water level to minimize the possibility of air ingestion. Applying the parameters of the IRWST to the equations in Reg. Guide 1.82, Appendix A, yields zero air ingestion at normal pump flowrates and less than 2% air ingestion at pump runout flowrates. The IRWST suction lines are also provided with vortex suppressors to aid in minimizing air ingestion by the SIS pumps. The guidelines in Appendix A of Reg. Guide 1.82 regarding the design of these vortex suppressors have been considered.

During normal full power operation, it is possible to perform a full flow test of the SIS and CSS pumps while taking suction from the IRWST and returning to the IRWST via a recirculation line (see SIS P&IDs in CESSAR-DC, Figures 6.3.2-1 A, B, C). This testing can verify the satisfactory hydraulic performance of the IRWST by running the pumps at runout flow.

2.9.4 RESOLUTION

The design of the System 80+ IRWST and HVT assures that a clean and reliable source of borated water is available for ECCS recirculation. The arrangement of the IRWST within the System 80+ containment offers advantages over conventional sumps. Like sumps, the tank serves as the single source of water for SIS and CSS pump recirculation, but the protection afforded the SIS pumps against debris ingestion or blockage is significantly greater than in current designs. First, we ter in containment draining back to the IRWST must pass through a large trash rack before entering the HVT. The HVT serves as an effective solids trap for high density debris. Lower density debris that makes its way into the IRWST via the IRWST spillways encounters debris screens that filter fine particles from the SIS suction inlets. Each of the four SIS pumps have separate IRWST suction lines and each of the two CSS pumps takes suction from one of these four lines. Box screens at all four suction lines provide a final trap.

Multiple spillways are available to return water from the upper containment elevations to the IRWST. The drain pathways are fully redundant to assure recirculation capability. The location of the suction inlets within the IRWST provide additional protection against suction inlet damage and/or blockage.

Consideration has been given to IRWST hydraulic performance, the generation of potential debris and associated effects (including debris screen blockage), and the preservation of SIS pump NPSH during post-LOCA conditions in the overall design. The performance of the design is deemed acceptable with respect to these considerations.

This issue has been resolved by design features of System 80+.

THIS PAGE INTENTIONALLY BLANK







.



2.10 EFFECTS OF PWR UPPER INTERNALS

2.10.1 ISSUE

Events with the potential for loss of Decay Heat Removal (DHR) have initiated from plant configurations with the reactor vessel head removed, the refueling pool filled with water and the reactor upper internals still in place. Under these conditions, the reactor vessel upper internals may provide sufficient hydraulic resistance to natural circulation flow between the refueling pool and the reactor core to inhibit, or even prevent, the refueling pool water from cooling the core under circumstances wher forced convection DHR has been lost.

2.10.2 ACCEPTANCE CRITERIA

When the reactor vessel head is off and the core and upper internals are in the vessel, any one or more of the following conditions shall be satisfied:

- Demonstration by analysis that the time to boil exceeds the time required to evacuate and establish containment integrity; and/or
- Demonstration by analysis that either a natural or forced circulation flow path, with or without heat exchangers, can be established to perform DHR for a sufficiently long period of time to allow plant operators to terminate the event.

2.10.3 DISCUSSION

In mode 6 configurations with the vessel head removed, loss of shutdown cooling events could be of concern if the upper internals inhibit natural circulation cooling of the core via the heat sink in the upper refueling pool. An analysis to predict the extent of natural circulation flow through the Upper Guide Structure (UGS) is described below. Resolution of this issue is based on the results of that analysis.

Hydraulic flow resistance data are provided in Table 2.10-1. These were utilized in an analysis to ascertain the possibility of natural circulation flow communication through the upper internals between the core and refueling pool for a Mode 6 configuration with the vessel head removed.

The entry flow path through access holes in the core support barrel flange to the upper downcomer from the refueling pool was shown to admit negligible flow in view of its high hydraulic resistance. Equal flow areas for the down-flow and up-flow were assumed through the core and upper internals with turnaround at the base of the active core. An iterative algorithm, based on the principles of conservation of mass, momentum and energy, was used to determine the transient natural circulation flow, after loss of shutdown cooling, and the time to saturation at the core outlet. Sensitivity of the results to magnitude of the upper internals flow resistance was also determined.

Several key assumptions made in the analysis include one-dimensional flow with no transverse momentum and energy exchange, no conduction loss through the vessel, no convection from the surface of the refueling pool and no heat storage within the bounds of the upper internals. The decay heat generation was conservatively assumed constant, characteristic of two days after shutdown, with an additional 10 percent uncertainty. No credit was taken for the available need developed at the core exit due to the forty foot elevation difference. Other simplifications included a uniform refueling pool temperature together with equal metal and fluid temperatures in each core flow zone that actually exhibit a linear variation axially.

This analysis indicates that natural circulation flow makes the water in the upper refueling pool available for cooling the core and significantly extends the time required for the core outlet temperature to reach saturation under these circumstances. Fluid temperature versus time variation results as a function of initial coolant temperature are shown in Figure 2.10-1. The curves in the figure for the maximum Mode 6 initial coolant temperature of 135°F indicate that the core outlet temperature would not reach saturation until 35 minutes after the loss of decay heat removal. The calculated natural circulation flow rate increases throughout this time period with a value of about one percent of the rated design core flow rate at one minute after loss of DHR rising to 1.5 percent where the outlet temperature of 100°F, the time to saturation increases significantly to 125.5 minutes.

A sensitivity evaluation shows that it would take only a few minutes to reach saturation at the core outlet in the complete absence of flow communication with the refueling pool. Another analysis indicates an additional time period of at least one hour to core uncovery after saturation is reached throughout the entire core volume in addition to the core outlet.

These results make it possible to conclude that the presence of the upper internals does not prevent natural circulation communication with the refueling pool. Depending upon the initial fluid temperature, the time to boil will exceed 35 minutes and may be several hours in duration. Ample time, therefore, exists for the implementation of operating procedures to deal with the restoration of forced circulation decay heat removal and/or to begin the process of containment closure, which requires 10 minutes when in Mode 6 configurations. 77a.wp (9212)bh

TABLE 2.10-1

HYDRAULIC FLOW RESISTANCE DATA FOR SYSTEM 80+

Region	Dimensionless Flow Resistance K	Cross-sectional Flow Area - Ft ²	
Core Support Barrel Flange Entry	1823.0	2.949	
Upper Downcomer	0.17	40.53	
Lower Downcor er	2.644	33.53	
Inlet Plenum	14.0	115.6	
Core	12.77	60.8	
Fuel Alignment Plate	0.69	20.6	
Upper Guide Structure	1.246	3.34	

2.10.4 RESOLUTION

The results discussed above indicate the availability of a natural circulation flow path through the upper internals in the vessel in the event of a lcss of both SCS trains during Mode 6 when the refueling pool is full. The heat removal effected by transference of core-generated decay heat to the upper refueling pool would allow plant operators a minimum time period of at least 35 minutes to terminate the event before the core outlet temperature reached saturation. Substantially more time would be available before possible uncovery of the core. This time period is of sufficient duration that the issue can be considered resolved on the basis of both of the previously stated acceptance criteria. It is clearly longer than the minimum Mode 6 containment closure time of 10 minutes and sufficiently long enough to allow plant operators to restore shutdown cooling.



יה_הס	RESULTS OF MODE 6	Figure
SYSTEMI 57	NATURAL CIRCULATION ANALYSIS OF SYSTEM 80+ TEMPERATURE vs TIME	2.10-1

2.11 FUEL HANDLING AND HEAVY LOADS

2.11.1 ISSUE

Questions have been raised regarding the potential for damage to fuel and safety related equipment due to dropping of heavy loads during plant shutdown. Related issues involve the transport of heavy loads within the reactor containment building and the spent fuel building. These include dropping the reactor vessel closure head and internals, dropping the head area cable trays [HACTS], accidental release of a fuel assembly, and movement of the spent fuel storage cask. Drop accidents involving primary NSSS piping are not considered, since by design piping is routed beneath the reactor refueling pool.

2.11.2 ACCEPTANCE CRITERIA

Fuel and safety related equipment shall not be subject to damage that may adversely effect public health. Also, fuel assemblies located within the reactor or within storage racks shall remain subcritical during and following postulated load drop accidents.

2.11.3 DISCUSSION

The transport of heavy loads within the containment building and the spent fuel building is controlled by integrating relevant design characteristics for the building and the handling equipment. Plant layout, equipment design and handling procedures are chosen to insure that heavy loads are restricted to preassigned travel zones. Equipment interlocks and procedures are also used to insure that load transport is accomplished in a predictable manner.

Specific issues associated with the transport of heavy loads within the containment building include movement of the reactor vessel closure head, the reactor internals, the HACTS, and individual fuel assemblies. Special measures are taken to safeguard these operations and mitigate the consequences of postulated load drop accidents.

Procedural guidance for raising the reactor closure head, as provided in CESSAR-DC Section 9.1.4.2.3.3, specifies that the fuel transfer tube valve be closed and that the pool water level follow the vertical movement of the closure head as it is raised from the reactor. This insures that the containment building remains isolated from the spent fuel pool building during transport of the closure head. Also, by isolating the containment building from the spent fuel building the spent fuel pool is protected against drain down that might occur as a result of a postulated drop accident. Evaluations have been performed which demonstrate that a postulated head drop, from its specified maximum lift height, onto the reactor vessel will not result in a significant risk to public safety. Though the reactor vessel and internals may sustain damage, the reactor vessel will remain filled and the fuel will remain covered and in a subcritical configuration. Evaluations of the reactor internals demonstrate that a drop accident involving the internals would be less severe than the postulated head drop accident.

Travel paths for the closure head and the internals, leading from the reactor vessel to the respective storage stands, are arranged so that the transported loads do not pass directly over the ICI seal table (Refer to Figure 2.11-1). If it is postulated that portions of these structures do impact the seal table, seal housings and guide tubing above the seal table, the resulting damage would be localized to these components. Under these conditions, the water level within the vessel will remain at the flange level. In addition should the reactor cavity pool seal be damaged to the extent that there will be significant pool drainage, the reactor vessel will remain filled.

The refueling machine is structurally designed to withstand the effects of design basis seismic motions. In addition, this machine is provided with interlocks which restrict machine movements to permissible zones as well as lock the fuel grapple in place. The refueling machine is designed to transport one fuel assembly at a time between the reactor core and the fuel transfer system. It is also designed to transport CEA rod and ICI disposal containers between an intermediato storage rack and the fuel transfer system. The grapple for the disposal containers is the same design as the one used for fuel assemblies.

The refueling machine is designed so that it can not pass over the top of the ICI seal table. This precludes the possibility of load drop accident involving a fuel assembly falling onto the ICI seal table. Also, during normal refueling operations the travel path is restricted so that it passes over the reactor cavity pool seal at predetermined locations. As a minimum, the pool seal is designed so that it will withstand without leakage a postulated fuel drop accident in these zones. If, for other postulated reasons, there is significant drainage of the pool, it is possible to rapidly lower a fuel assembly on the refueling machine grapple to an elevation which insures that it remains submersed in water. The assembly may be inserted into the reactor vessel or lowered into the deep end of the refueling pool, adjacent to the fuel transfer system.

The head area cable tray assembly [HACTS], which is used to route power and signal lines away from the reactor vessel closure head, is designed to be handled by t - auxiliary hoist on the polar crane. The cable trays are rat - vertically by this hoist from the installed position over the reactor vessel and moved to a storage position on top of either of the two steam generator walls. The trays are handled by four separate slings that are fastened to specially designed lift fixtures on the scructural frame of the LACTS. All lifting components are designed in accordance with the criteria of NUREG-0612, Control of Heavy Loads at Nuclear Power plants. Prior to movement of the cable trays, the reactor must be in the shutdown mode and depressurized.

During a postulated load drop accident involving the impact of the HACTS onto the reactor vessel, the maximum impact energy is estimated to be about twenty percent of that associated with the reactor vessel closure head drop.

Though the reactor vessel may be damaged, the level of damage to the vessel and its support would be less severe than that associated with dropping the reactor vessel head. Furthermore, since the cable tray assembly is not a rigid structure, an appreciable fraction of the impact energy would be dissipated during plastic deformation of the HACTS itself.

For the postulated dropped cable tray accident, it is most probable that the HACTS would impact the closure head lift rig and the CEDM pressure housings. These structures are likely to be permanently deformed by bending a 4/or buckling. In some instances the pressure housings may also leak. The extent of damage, however, would not be sufficient to cause the reactor vessel to drain down, nor to adversely affect core criticality.

As with the containment building, special consideration is given to the transport of heavy loads and fuel assemblies in the fuel building. Also restrictions regarding the transport of heavy loads over fuel storage racks, and movement of the fuel shipping cask apply (Refer to Figure 2.11-2).

Transport of the fuel shipping cask within the spent fuel building is accomplished using a special high capacity hoist. The cask is transported using a staggered lift from the wash down area to the laydown area, where fuel loading takes place. This is done to limit the maximum drop height for the respective regions. In each case the floors and walls have been designed to withstand a postulated cask drop accident.

The spent fuel pool is connected to the cask laydown area by a gate, which will be closed to isolate the two zones during cask movement. The elevation of the gate is specified so that fuel located in the spent fuel storage racks would remain submerged following a postulated pool drain down through the gate.

The hoist used for transport of the fuel shipping cask is mechanically interlocked to prevent travel over the spent fuel

pool. This interlock prevents the possibility of inadvertent movement of heavy loads over the spent fuel storage racks.

New fuel enters the fuel building through a designated unloading area. It is handled and transported to new fuel storage racks by an intermediate capacity hoist. The lift height of the hoist is rescricted to limit the maximum drop height of the fuel and tool onto the new fuel storage racks. Like the fuel storage cask hoist, this hoist is also mechanically interlocked to prevent travel over the spent fuel pool.

The fuel handling machine used in the fuel building is similar in design to the refueling machine. It is structurally designed to withstand seismic excitations. Also, it is provided with interlocks to control the movement of fuel within the pool.

Both the new fuel and spent fuel storage racks are designed to withstand impact energies associated with postulated fuel drop accidents. They are designed to limit damage to the stored fuel and to maintain it in a subcritical configuration. Plant operating procedures also restrict the transport of loads over the fuel storage areas so that they do not exceed the design requirements for the storage racks. The consequences of dropping a spent fuel assembly in the spent fuel pool have been evaluated and the results presented in CESSAR-DC, Section 15.7.4. It has been shown that a postulated accident of this type would not present a risk to public health.

2.11.4 RESOLUTION

Ą

The issue of fuel handling and heavy loads is resolved for System -80+ by the equipment design and building layout which satisfy applicable criteria and provide physical limitations to movement and by administrative limitations in Chapter 9 of CESSAR-DC. The design and layout of the System 80+ plant incorporates features that have proven to be successful on other plants. Appropriate improvements have been introduced which provide an additional margin of safety during handling operations.





ה הה		Figure
SYSTEM IN 52	FUEL HANDLING BUILDING LOAD HANDLING PATHS	2.11-2

2.12 POTENTIAL FOR DRAINING THE REACTOR COOLANT SYSTEM

2.12.1 ISSUE

The issue is the risk of losing primary coolant from the reactor coolant system during Modes 2 through 6 (shutdown, hot standby, hot shutdown, cold shutdown, and refueling). The safety significance of draining the coolant from the reactor coolant system during shutdown is that such an event can directly lead to voiding in the core and eventual core damage. The draining of the reactor coolant system may also lead to a loss of decay heat removal cooling capability which in turn could lead to core uncovery.

2.12.2 ACCEPTANCE CRITERIA

The criteria utilized to evaluate the adequacy of the System 80+ design with respect to the potential for draining the RCS are prevention, detection and mitigation. Prevention is the preferred criteria but in some instances detection and mitigation are to be provided.

2.12.2.1 Prevention Criteria

- The design shall prevent or inhibit the draining through the use of isolation valves, interlocks, and system alignment restrictions during the various modes of plant operation.
- 2. The design shall minimize the potential for component failure, inadvertent action, or human/operator error to result in the rapid draining of the reactor vessel. Redundant components shall be provided as appropriate. The design shall provide instrumentation, overview displays, and alarms to clearly supply the operator with equipment status specific to shutdown modes.
- 3. Technical specifications and procedural guidance shall be provided to the plant owner/operator to assist in identifying plant conditions and configurations in Modes 2-6 that could result in a potential primary coolant drainage event.

2.12.2.2 Detection Criteria

The design shall have the capability to detect and monitor a drainage event. Drainage pathways that could lead to a loss of decay heat removal or core uncovery shall be considered. Appropriate instrumentation, displays and alarms shall be provided. The adequacy of System 80+ instrumentation to detect drainage events shall be confirmed and additional instrumentation shall be provided as necessary.

77a.wp(9212)bh

2.12.2.3 <u>Mitigation Criteria</u>

- The design shall have the capability to mitigate the loss of primary coolant from the reactor including isolation of a drain path and the ability to provide a source and path for sufficient make-up.
- Technical specifications and procedural guidance shall be provided to identify potential make-up water injection sources and paths in the event a drainage path does occur. Recovery actions shall be specified.

2.12.3 DISCUSSION

Primary coolant can drain from the reactor vessel due to a path directly from the Reactor Coolant System (RCS) or by way of paths through systems interfacing with the RCS. Plant modes of operation characterize the potential for draining the reactor coolant system (the alignments and conditions, such as pressure, temperature and flow which can exist within and between the RCS and interfacing systems) and the rate at which such draining can occur.

The causes of RCS draining (initiators) can be categorized into two groups. The first group includes components and equipment that fail to operate as intended. This could, for example, result from equipment malfunction (e.g., stuck open relief valve). The second group includes operator error such as misoperation of valves or pumps.

There are several key factors that affect the probability and the consequence (i.e., the risk) associated with an initiator. Consideration of these factors can aid in the development of procedures for prevention, detection, identification, and mitigation or termination of such events. These factors are the plant configuration, the ability to respond to the event, and the characterization of the initiator. These are summarized in Table 2.12-1.

TABLE 2.12-1

FACTORS WHICH AFFECT THE RISK ASSOCIATED WITH AN INITIATOR

Plant Configuration... System alignment with the RCS. Initial water level in the RCS. Availability of mitigating systems. Use of temporary seals (e.g., nozzle dams)

Ability to Respond... Detection of a draining event. Identification of the initiator. Termination or mitigation of the event.

Characterization of the Initiator... Probability of the initiator. Rate of drainage from the RCS.

Plant configuration is most notably characterized by system alignments with the RCS during various modes of shutdown operation. These alignments define the possible drain paths from the reactor vessel. The initial water level in the RCS influences operator response time. The use of temporary seals in the RCS (e.g., nozzle dams) and interfacing systems during maintenance and refueling activities can either be the source of an initiator (e.g., if the dam fails) or preclude mitigating actions by defeating injection paths (e.g., if an in place dam blocks circulation). Plant configuration is also characterized by the availability of mitigating systems. Mitigating systems must have a sufficient source of borated water and the ability to deliver this water to the reactor vessel at a rate greater than or equal to the rate at which water is draining from the RCS. Finally, maintenance activities must not preclude mitigating systems from being used to respond to the event.

Another factor that can affect the risk associated with an initiator is the ability to quickly determine that a loss of coolant is occurring (detection) and the source of the loss (identification). This defines instrumentation and control requirements to respond to draining events.

The analysis presented in this section was conducted ... examining design drawings and piping and instrumentation diagrams for System 80+ to define potential drain paths from the reactor vessel and to
consider these drain paths in the context of the factors shown in Table 2.12-1. A potential drain path is any opening in the RCS (seal, manway) or any interfacing system piping path that can take primary coolant away from the RCS. Many openings and interfacing systems are designed specifically to allow fluid to leave the RCS or interfacing system (e.g., relief valves) or to circulate primary coolant for normal letdown, purification, charging (CVCS), sampling (SS) activities or shutdown cooling (SCS).

A shutdown risk drage ath results in an unplanned loss of primary coolant from the R ith the accompanying lowering of the water level in the react vessel. The path could be short (seal leakage) or relatively long (via piping of an interfacing system). The driving head for the draining flow will depend on the relative conditions between the coolant in the reactor vessel and the conditions at the end of the drain path. This includes RCS pressure, elevation head, pump head, and backpressure.

For purposes of this analysis, a shutdown risk drain path is categorized as either major or minor. A major drain path is one that, based on analysis for certain specified plant configurations, could result in a rapid drain flowrate. The preferred recovery would be to isolate the drainage source before the RCS water level reaches the break level and to add makeup to the RCS. It is possible, however, that such a major path could drain the RCS to the bottom of the hot leg elevation too rapidly before the operator could take mitigative action. Such an occurrence could result in a loss of SCS due to insufficient water level. The identification of major drain path does not imply that such a path is likely or probable for System 80+, but only that it is possible and that its potential consequence requires special attention be directed to prevention and recovery procedures. Minor drain paths are those that result in a drain flowrate that can be compensated for using available make-up systems, or are otherwise insignificant.

The identification of a minor drain path requires no further action. The identification of a major drain path requires specific procedural guidance to aid the operator in avoiding this path. It also results in providing the operator the means and guidance to recover from such an event. The means to detect, mitigate and recover from such postulated major drainage events are those described in Sections 2.3, 2.4, and 2.8 of this report.

2.12.3.1 Potential Drain Paths Directly from the Reactor Coolant System

Potential drainage paths directly from the Reactor Coolant System are associated with the reactor coolant pumps, the steam generators, the In-Core Instrument Seals and the reactor cavity seal. Note that reactor cavity seal leakage is not an actual RCS leakage but it is a form of inventory loss during Mode 6. Table 2.12-2 identifies those paths which are major (i.e., have the potential to rapidly drain the RCS to a critical water level) and minor (i.e., can be controlled by available systems).

TABLE 2.12-2

POTENTIAL DRAIN PATHS DIRECTLY FROM THE REACTOR COOLANT SYSTEM

Major Drain Paths

Steam Generator Nozzle Dam Failure Steam Generator Manway Opening

Minor Drain Paths

Reactor Coolant Pump Seal Leakage ICI Seal Table/Housing Leakage Steam Generator Tube Rupture Reactor Cavity Seal Leakage

2.12.3.1.1 Major Drain Paths Directly from RCS

The failure of temporary steam generator dams and the opening or leakage through steam generator manways could lead to a rapid loss of primary coolant perhaps to an RCS water level at the bottom of the hot legs. A discussion of steam generator nozzle dam integrity is presented in Section 2.3.3.3 of this report. The System 80+ design includes a requirement to establish a mid loop vent pathway before operating in reduced inventory. When opened to the containment atmoshphere, it provides sufficient venting capacity to prevent RCS pressurizations and subsequent dam failure. Guidance to address the safety and risk aspects of nozzle dam installation timing is presented in Section 2.3.3.3 of this report. System 80+ procedural guidance will preclude manway openings that would lead to a loss of reactor coolant.

2.12.3.1.2 Minor Drain Paths Directly from the RCS

There are three stages of RCP seals for each System 80+ RCP and each of the three seals are capable of operating at full RCS pressure. However, the reactor coolant pump can potentially suffer some leakage from pump seals during any mode. Seal leakage is detected by a RCP bleed-off flow alarm (F-156, -166, -176, -186). The source of leakage can be identified to a specific pump. Drainage from RCP seals would be manageable and can be compensated by available systems.

The System 80+ In-Core Instrumentation (ICI) system design employs instrument tubes that terminate in the refueling cavity at an elevation several feet above the reactor vessel flange. The System 80+ design does not employ temporary thimble tube seals. Even so, there has been concern expressed in Section 6.7.2 of NUREG-1449, (DRAFT) that evolutions could exist that would provide a potentially significant flow path between the bottom of the reactor vessel and the top of the seal table, particularly if the RCS is Table evolutions are prohibited by procedural pressurized. guidance (see also Section 2.3.3.5 of this report) while the vessel head is on and mid-loop evolutions are in progress, thus, preventing seal leaks. An evaluation of heavy load handling for System 80+ relative to ICI seal table is discussed in Section 2.11 of this report. Travel paths for the closure head and internals from the reactor vessel to the respective storage stands are arranged so that the transported loads do not pass directly over the ICI seal table. This precludes the possibility of a load drop accident falling on the ICI table.

A steam generator can leak primary coolant through tube failure during any mode when the RCS is pressurized. An evaluation of steam generator tube rupture initiated in a shutdown mode is presented in Section 4.6.3 of this report. The leaks from a ruptured tube during shutdown is minor relative to reaching a RCS water level at the bottom of the hot legs.

Reactor cavity seal failure could be postulated during Mode 6 when the refueling pool is full. Although such a drainage could be a concern relative to any fuel being transported, the drainage would be self limiting to level of the reactor vessel flange and the SCS will be uninterrupted.

2.12.3.2 Potential Drainage Faths Through Interfacing Systems

Potential RCS interfacing system drainage paths from the Shutdown Cooling System (SCS), the Safety Injection System (SIS), the Chemical and Volume Control System (CVCS) and the Sampling System (SS) are illustrated in Figures 2.12-1, 2.12-2, 2.12-3 and 2.12-4. The originating points of the drainage paths are location on the Reactor Coolant System. The paths include piping, valves, pumps, heat exchangers and orifices. The greek symbol \$\$\$\$ represents the piping outer diameter. The terms IC and OC are used to show whether a path segment is inside containment or outside containment, respectively. The "letter-number" number designation in the boxes generally refers to system valves which are of the motor operated, manually operated, check or relief type. The letter "F" followed by a number referes to a flow measuring device. Those paths represented by the darker/bold lines identify major drain paths. The end points of these paths are various tanks or systems assumed to be open for maintenance.

An assessment of these paths was made relating potential flow rates and the time it would take for the RCS water level to reach the bottom of the hot-legs. An assessment of the relative probability, of single versus multiple valve misalignments or failures was not considered at this time. All paths were considered possible, even though it is recognized, multiple misalignments are less likely.

2.12.3.2.1 Potential Drainage Paths From TL RCS Through the SCS (SCS Trains: From RCS Hot Leg 1 Through DVI Nozzle 1A and From RCS Hot Leg 2 Through DVI Nozzle 1B)

This discussion is specific to train 1 but applies to train 2 as well.

This overall path represents one of the two normal shutdown cooling trains. As long as this path is isolated from the RCS during Modes 2, 3 and part of 4 loss of coolant from the RCS through the SCS is prevented. While the SCS is in operation during part of Mode 4 and during Modes 5 and 6 there exists potential paths from the SCS through which the primary coolant passing through the SCS could be drained. The main initiators to open a drain path would be an overpressurization leading to the lifting of a relief valve (which fails to reseat) and/or operator error opening a valve or series of valves to an open system. A valve failure is another possible initiator. An open system is a system which has been drained for maintenance activities.

The SCS suction piping contains two motor operated isolation valves (SI-651 and SI-653) in series inside the containment (see Figure 2.12-1). These valves are closed during Modes 1, 2 and 3. In addition, there is also a motor operated valve (SI-655) outside the containment. All three valves can be operated from the control room and have position indication in the control room. An alarm exists to notify the operator if the two motorized valves inside containment are not fully closed coincident with high RCS pressure. Operator actions require that the RCS be depressurized below the maximum pressure for SCS operation in order to clear the permissive SCS interlock. Therefore, drainage from the RCS through the SCS to interfacing systems in the normal flow path direction is prevented during Modes 2 and 3. Flow will occur through the SCS train during part of Mode 4 and throughout Modes 5 and 6.

The potential drain paths from the SCS as shown in Figure 2.12-1 are summarized in Table 2.12-3. The 30 paths of Figure 2.12-1 have been broken down into groups of similar drain path characteristics.

TABLE 2.12-3

GROUPING OF PRIMARY COOLANT DRAINAGE PATHS FROM THE SCS

Minor Drain Paths

Group

Paths

Α.	Thermal Relief Valve Discharge	1, 3, 18, 19, 21, 22, 25, 27, 28
в.	Paths to the Sampling System	6, 12
с.	Paths to the CVCS	5, 8, 9, 15, 17
D.	Paths to the CCWS	10, 11, 14
Ε.	Paths to SIT #1	23
F.	Paths to IRWST, RDT, and EDT Through Manual Valves and Small Piping	24, 26, 29, 30

Major Drain Paths

G .	Paths Through Motor Operated Valves and Large Piping to the IRWST	4, 16
н.	SCS Suction Relief Valve Discharge	2
	and the second	

- I. Paths Through large Manual Valves 7, 13 to CS Pump Suction and Discharge
- J. Path to the Refueling Pool 20

Group A through F paths are minor paths and include relief valves, paths through small piping (1/2 in - 3 in O.D.) to assumed "open" systems, SC pump seal leakage, SC mini-flow heat exchanger tube leakage, SC heat exchanger tube leakage and a postulated path to the SITs.

The paths of groups A through F, if they were to occur and lead to a loss of primary coolant, could be mitigated with available equipment and make up sources during Modes 4, 5 and 6. The discharge would be slow enough such that a loss of primary coolant to the bottom of the hot legs should not occur before detection and mitigation have been accomplished. No new design features, technical specifications or procedural guidance are identified for the paths in these groups. Groups G through J represent major paths that, if established, could result in a rapid loss of reactor coolant during Modes 5 and 6. Depending on the availability of equipment and systems to perform mitigative action during these modes such a rapid discharge may lead to an RCS water level at the bottom of the hot legs with a concurrent loss of both SCS trains. Procedural guidance to aid the operator in addressing these paths is specified in Section 2.1. The recovery action for this scenario is discussed in Section 2.4.3.1.

2.12.3.2.2 Potential Drainage Paths from the RCS to the SIS/SCS (From DV) Nozzles 1A, 2A, 1B and 2B)

The potential drain paths from the RCS through the SIS/SCS, originating at the DVI nozzles, are shown in Figure 2.12-2.

The potential drain paths presented in Figure 2.12-2 are "reverse" paths relative to the normal safety injection flow direction. As a result, in order to assume RCS drainage from the DVI nozzles, it is postulated in this analysis that multiple failures of check valves occur.

The DVI nozzles in System 80+ are located several (4-5) feet above the hot leg and cold legs. Therefore, loss of coolant through a DVI nozzle would be self limiting. The level of primary coolant would stabilize at the level of the DVI nozzle, and not result in a loss of shutdown cooling. No new design features, technical specifications or procedural guidance have been identified for these paths associated with flow from DVI nozzles.

2.12.3.2.3 Potential Drainage Paths from the RCS to CVCS

The potential drain paths from the RCS through the CVCS are shown in Figure 2.12-3. The potential drain paths presented in Figure 2.12-3 are the normal letdown, charging, RCP seal injection, RCP seal leak off, RCP seal bleed off and drain paths associated with the CVCS design. A major opening in the letdown or charging line needs to occur for any appreciable drainage to occur.

The paths defined for the CVCS, if established, would be manageable with available make up sources during Modes 2, 3, 4, 5 or 6. This discharge would be slow enough such that a loss of primary coolant to the hot leg level should not occur before detection and mitigation have been accomplished. No new design features, technical specification or procedural guidance are identified for paths associated with the CVCS.

2.12.3.2.4 Potential Drainage Paths from the RCS to the SS

The potential drain paths from the RCS through the SS are shown in Figure 2.12-4. The potential drain paths presented in Figure 2.12-

4 are normal sampling paths. A major opening in the sampling lines would need to occur for a net loss of primary coolant to occur.

The paths defined for the SS, if they were to occur, would be manageable with available make up sources during Modes 2, 3, 4, 5 or 6. The discharge would be slow enough such that a loss of primary coolant to the hot leg level should not occur before detection and mitigation have been accomplished. No new design features, technical specifications or procedural requirements are identified for paths accoriated with the SS.

2.12.4 RESOLUTION

The shutdown risk issue of the potential for draining the System 80+ RCS is resolved primarily by design features, technical specifications and procedural guidance to prevent a drainage event from occurring and to allow the operator to recover in a timely manner if such an event occurs.

The vast majority of potential paths reviewed were judged to be minor such that the drain flow rate can be compensated using available detection and mitigating systems or are otherwise insignificant. System 80+ design features, technical specifications and procedural guidance are sufficient for such paths.

An examination of the potential drainage paths for various System 80+ plant arrangements and operating configurations has provided candidate paths, that if assumed to be opened, could lead to a rapid loss of primary coolant. The candidate paths primarily involve those opened by misoperation or misalignment of one or multiple valves by the operator. The importance of such potential major drainage paths to a shutdown risk scenario has led to procedural guidance (see Section 2.1) being specified to aid the operator in addressing these paths

The issue of potential RCS drain paths is ultimately resolved, from a core uncovery prevention perspective, by the use of System 80 design features to detect, mitigate and recover from postulated loss of shutdown cooling events as described in Section 2.4 of this report.



•

1.1

Sheet 1 of 5

.



Sheet 3 of (n

....

SYSTEM JU 'J'	171111111	m clu
FROM SCS TRAIN NO. 1	POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS	
2.12-1		Figure

.

4



8-17

-

PATH

.

1



.

•

-...

Sheet 4 of 5

2.12-1

5":gure

-

* *

.



Sheet 5 of 5

Sheet 1 of



3

-

0



PATH

Sheet 2 of 2







POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS 2.12-3

Figure

1" ¢

RCP 28

10

SWEREN III CD

\$





OC - OUTSIDE CONTAINMENT



Figure

POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS FROM CVCS 2.12-3

SYSTEM III CS *



IC - INSIDE CONTAINMENT

PATH

Sheet 3 of

SYSTEM III CS POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS FROM CVCS 2.12-3 Figure



PATH

Sheet 4 of 4

SWEETEN III CS POTENTIAL RCS INTERFACING SYSTEM DRAINAGE PATHS 2.12-3 Fig' re





.....

-

IC - INSIDE CONTAINMENT OC - OUTSIDE CONTAINMENT

2.12-4

and a straight

Figure

**

• •

.

THIS PAGE INTENTIONALLY ELANK

2.13 FLOODING AND SPILLS

2.13.1 ISSUE

Essential systems may be at higher risk for failure due to flooding and spills during shutdown because of the varied and interrelated maintenance activities that may be in progress simultaneously. Past events have involved, for example, spills from the component cooling water system, service water system, condensers, and refueling pool seals. The issue addressed here is the potential for loss of decay heat removal as a consequence of spills and internal flooding that may disable components of the shutdown cooling system.

2.13.2 ACCEPTANCE CRITERIA

The flood protection design will provide separation of redundant equipment to ensure decay heat removal (DHR) systems availability and capability are not precluded due to flooding and spills.

2.13.3 DISCUSSION

The flood protection provided insures a boundary of separation between redundant DHR systems. The separation includes components and structures to prevent the migration of water. Preventing the migration of water eliminates the potential for rendering redundant DHR equipment inoperable.

The System 80+ design provides separation and flood barriers to prevent the flood of redundant equipment. The design features a divisional separation. This divisional separation is a wall in the Nuclear Annex and the Reactor Building Sub-Sphere. The wall forms a barrier between the Division 1 and the Division 2 mechanical and electrical equipment. This wall contains no unsealed penetrations below the 70' elevation level. This wall is along column line 17 (see CESSAR-DC Figures 1.2-4 and 1.2-5, reproduced here as Figures 2.13-1 and 2.13-2). Additional separation of the divisions is provided by the floor drain systems. The sumps and floor drains located in the Nuclear Annex and the Reactor Building Sub-Sphere are divisionally separated. This design feature prevents the migration of floodwater from one division to the other through the floor drains.

The Systems 80+ design utilizes flood doors to provide separation within the same division. In the Reactor Building Sub-Sphere, the flood doors provide quadrant separation, therefore equipment is protected from floods within the same division. Flood doors also provide protection for Reactor Building Sub-Sphere Quadrants A and B from flooding outside the sub-sphere. This protects the Shutdown Cooling Systems from floods that could occur in the Nuclear Arnex and migrate into the sub-sphere. Flood doors also provide protection for the Vital Electrical Equipment located in the Nuclear Annex on elevation 50' (see Figure 2.13-1).

The System 80+ does not have any raw water systems inside the Nuclear Annex or Reactor Building. This design provides a significant contribution to flood protection because the flood sources are finite. Two significant sources of water are the Component Cooling Water System and the Emergency Feedwater Storage Tanks. Emptying the entire volume of water contained in a division of either of these systems will not flood above the 70' elevation. Therefore, no migration of water to the other division or to other protected areas (e.g., electrical equipment) will occur due to the flood. This ensures the redundant systems and equipment located in the other division are available for decay heat removal.

2.13.4 RESOLUTION

The System 80+ flood protection design features are consistent with acceptance criteria outlined above in Section 2.13.2. These features resolve the issue of flooding and spills during shutdown operations on System 80+ by providing separation of redundant equipment required for decay heat removal. This separation provides the availability of DHR when a flood has occurred within the Nuclear Annex or Reactor Building Sub-Sphere.



Sand Stand of S

9





Arris assertion



3.0 PROBABILISTIC RISK ASSESSMENT

3.1 PRA SUMMARY AND CONCLUSIONS

An assessment of the risk associated with internal events during shutdown modes of operation was performed. The scope of this assessment included internal events during Mode 4 through 6 operation.

Event trees were developed and quantified for loss of decay heat removal (DHR) and loss of coolant inventory for Modes 4 through 6. Loss of offsite power and criticality events were not quantified after an initial review showed these events to be a small contributor to risk. In quantifying the core damage 'requency (CDF), emphasis was placed on the human error probabilities (HEPs) because earlier studies have shown that these dominate the shutdown risk. The system failure rates were either taken from the System 80+ PRA or estimated using simplified fault trees. Table 3.1-1 summarizes the operator error rates, mechanical failure rates and event tree branch failure rates that are developed in Section 3.4.

The results from the Human Reliability Analysis reflect engineering judgement and assumptions associated with current operations practice. As further design and operations detail become available, the analysis will be updated as part of the PRA maintenance program to reflect a more accurate assessment of the impact of human error at shutdown. It is believed, however, that the analysis presented here reflects the correct level of dominance in certain events with respect to risk and that further analysis will provide more accuracy, with respect to the models, but should not affect the results of the dominant sequences.

Table 3.1-2 summarizes the contributors to core damage frequency by modes and initiating event. Mode 6 LOCA represents 69% of the risk. The leading contributor is a LOCA outside containment with failure to isolate while the refueling cavity is full. Loss of DHR in Mode 6 while the refueling cavity is full is the second leading contributor to risk (17%). In both sequences, the IRWST is empty and feed and bleed is not available. If one of the two SCS trains is not restored, then core damage will result.

Table 3.1-3 compares the CDF for shutdown events with those for power operation. The CDF for internal events during shutdown modes is 15% larger than for internal events at power and smaller in size than at power external events. The EPRI goal is to have CDF less that 1.0E-5/rv. The CDF from shutdown is significantly less than this goal and the sum of the CDF is also less than the EPRI goal. This PRA contributed to technical specification and instrumentation changes. One example of such a change was a technical specification change requiring two SIS trains be available in Modes 5 and 6 when the IRWST is available.

Table 3.1-1

SUMMARY OF JRANCH POINT FAILURE RATES

	Operator Error	Mechanical Failure	Total	Description	
REP	INCL	INCL	1 1.0E-02	REPair during 81 hour coolant boil-off	
CS2	INCL	INCL	3.1E-03	Containment Spray system (2 of 2)	
DP	3.0E-03	8.0E-04	3.8E-03	Depressurize with SDS	
MUI	3.1E-03	1.0E-03	4.1E-03	Make Up Inventory	
OI	0.0E-02	4.0E-02	9.0E-02	Operat Isolates leak	
OR	0.84 *1.0E-03	0.16	1.6E-01	Operator Restarts SCS train	
0S1	2.8E-03	2.0E-02	2.3E-02	Operator Starts standby SCS train (1 of 1)	
052	2.8E-03	2.0E-03	4.8E-03	Operator Starts 1 of 2 SCS trains	
SGHR	3.4E-02	7.05-04	3.5E-02	Operator starts SG Heat Removal	
SGCOM	3.4E-05	INCL	3.4E-05	Commission error in continued use of SG	
SIP2	2.3E-3	7.0E-04	3.0E-03	Manually start 1 of 2 SIS	
SCSFB	3.1E-03	1.8E-03	4.9E-03	Manually use SCS for Feed and Bleed	
SIFB2	2.7E-03	1.5E-03	4.2E-03	Manually use SIS for Feed and Bleed(1 of 2)	
BOC	3.0E-03	0.1	1.0E-01	Boil-Off using CVCS	
OIC	8.2E-03	8.0E-04	9.0E-03	Operator Isolates leak at Containment	

FREQUENCY OF		RE DAMAGE FROM SHUTDOWN EVE		
LOSS OF DHR MODE	CDF	LOCA CDF	TOTAL	
4	2.3E-9	2.3E-8	2.5E-8	
5	4.1E-10	5.4E-9	5.8E-9	
5 REDUCED INVENTORY	6.5E-8	2.0E-8	8.5E-8	
6	1.3E-7	5.3E-7	6.6E+7	
TOTAL	2.0E-7	5.7E-7	7.7E-7	

TABLE 3.1-2

TABLE 3.1-3

COMPARISON OF CORE DAMAGE FREQUENCIES

POWER	OPERATION,	INTERNAL	EVENTS*	6.7E-7
POWER	OPERATION,	EXTERNAL	EVENTS*	1.2E-6
SHUTDO	OWN, INTERN	AL EVENTS		7.7E-7

* FROM SYSTEM 80+ PRA7

3.2 PRA INTRODUCTION

Until recently, emphasis has been on the safety of power plants during power operation. This was due to the fact that the plant is in this configuration most of the time and the core power, decay heat rate and fission product inventory are highest at this time. The System 80+ PRA⁷ documents the risk associated with the operation of this ALWR at normal power. More recently, people have been investigating the risk of plants during shutdown and refueling. During these modes of operation, the plant has lower decay heat generation rates and fission product inventory. The plant configuration is not as well defined as in full power operation because of the maintenance and testing that is being performed. This section estimates the risk of operation of this plant in Modes 4 through 6. Mode 1 through 3 are covered in Ref. 7.

The awareness of the risks of plant operation during refueling and maintenance outages developed slowly. Although to core damage has occurred during reactor outages, a few even is have occurred which were precursors to more severe accidents. One of the first events to increase the awareness of risks in outages was the loss of the refueling cavity seal at Connecticut Yankee on August 21, 1984. In this event, 200,000 gallons of water quickly spilled into the containment, draining the refueling cavity. This event would have been more difficult to handle if the refueling had actually started. The seal failure had not been considered in the safety analysis.

Another event which increased the awareness of the risks during outages was the Vcgtle 1 event on March 20, 1990. The event started with a truck backing into equipment in a switch yard, causing a loss of power to the first auxiliary transformer. The second auxiliary transformer was out for maintenance. One of the diesels was also out for maintenance. The second kept tripping due to an instrument failure. This combination of maintenance activities and failures led to a station blackout and loss of Decay Heat Removal (DHR). Under normal conditions, with the vessel filled or with the refueling cavity flooded, the operator would have many hours to restore DHR. In this incident, the plant was in mid-loop operation and the primary inventory was greatly reduced. When DHR was restored in 41 minutes, the primary coolant temperature had risen 46 deg F to 136 deg F. This incident demonstrates the unusual plant configurations that can exist during an outage and the risks associated with maintenance activities and mid-loop operation.

During 1991, there were a rash of incidents during shutdown. After four events occurred within six days in March, the NRC issued Information Notice No. 91-22 describing these events. One plant had two incidents during the same outage. There were at least seven events where loss of DHR occurred in 1991. All these events increased the awareness of the risks during outages.

The NRC requested additional information from the ALWR participants pertaining to shutdown risk. The EPRI ALWR Requirements Document then was modified to include a risk assessment for shutdown modes. This analysis satisfies that requirement. This analysis uses a simplified event tree approach to estimate the core damage frequency (level 1 part of a PRA), with only a qualitative evaluation of the radiological releases.

The first step in the analysis was the identification of the initiating events of potential interest. This was done by first defining the plant conditions (in terms of physical parameters such as temperature, pressure, and inventory for the RCS) that will exist for different plant evolutions. For each of these operating conditions, general category of initiating events were then defined. These initiating events were small LOCAs, Loss of Decay Heat Removal (DHR), Loss of AC, and Boron Dilution. The frequencies for these events were determined by operational history.

For each plant condition and initiating event, the plant and operator response was estimated based on the advanced instrumentation, procedures, technical specifications, and safety systems employed in the System 80+ design. The plant states and operator response were modeled and quantified using simplified event trees. The unavailability of each system was estimated using simplified assessments and adaptation of models developed for power operations. Care was taken in estimating the reliability of human actions. Earlier studies found that operator actions are one of the dominant factors in this analysis.

This analysis was performed with the insight obtained from the previous PRAs. In 1981, NSAC-84 looked at the shutdown risk at Zion. This study concluded that failures during reduced inventory operation accounted for 61% of the Core-Damage Frequency (CDF). Operator actions were required in almost all sequences. Operator failure to determine the proper actions to restore DHR accounted for 56% of the total CDF. Loss of DHR also accounted for 56% of the CDF. NUREG/CR-5015 tended to confirm the findings in NSAC-84. The Seabrook Shutdown PRA concluded that 82% of the CDF was due to loss of DHR and 71% was from reduced inventory operation. The study also showed that early health risks were dominated by LOCAs with the containment open. The NRC's Shutdown PRA for Surry (as summarized in NUREG-1449) showed the importance of plant specifics such as the controls of the ADVs, and the response to Generic Letter 88-17. The insights gained from reviewing these PRAs helped in analyzing the System 80+ plant.

This analysis is an extension of the CESSAR-DC System 80+ PRA. The System 80+ PRA used the standard "small event tree/large fault tree" approach used by most of the industry. A detailed discussion of the method employed is given in Section 2.0 of Reference 7. This section describes how that procedure was modified. The modifications were suggested by EPRI⁹. The scope of the Shutdown PRA is to evaluate the core damage frequency (level 1 PRA) using a simplified assessment. Only a qualitative evaluation of the radiological consequences are presented.

Event trees were developed for the accident sequences for each plant state in Modes 4 through 6 given in Table 3.2-1. The plant states were developed based on what equipment was available to mitigate the event. Mode 3 was not considered in this study because the plant configuration is very similar to Mode 1 and 2 and therefore, the effects are enveloped in the Mode 1 analysis.

The event trees are based on the transient studies for shutdown events described in Reference 4. Assumptions about operating or maintenance actions are based on current plant practices or improvements suggested in recent shutdown studies (e.g. NUREG-1449) and are listed in the event tree element descriptions. This will facilitate verification when the plant is complete and operating. The event trees are supported by fault trees and Human Reliability Analysis (HRA) models. This analysis is for a simplified assessment for the operating conditions other than power operation conditions to ensure that there are no unrecognized or unacceptable sources of risk associated with these conditions. This assessment is limited in scope because of the limited level of operational detail that is available at this time and because many requirements aimed at addressing problems at existing operating plants are currently being developed.

The first step in the analysis is the identification of the initiating events of potential interest and the estimation of the frequencies for these events. A search of the earlier shutdown risk assessments, LER data base, and NPRDS was performed. The initiating evont frequencies were proportioned to the plant states based on the fraction of the outage time spent in each state. The plant states represent plant modes, or part of a plant mode, which have certain physical parameters and equipment availability (both safety and operational).

For each plant state and initiating event, a functional assessment of the plant response is made. A team of engineers, including specialists in thermal-hydraulics, operator error analysis, plant design, and outage management, were assembled. The team developed the plant response to each transient, including the sequence of events, available instrumentation and alarms, human response, and system response. The times to core damage were also estimated. Emphases was on the human performance, which has been shown in past studies to be very important in satisfactorily assessing the shutdown risk. The event trees were developed with participation of the full team.

The branch points of the event trees were evaluated using simplified fault trees for the responding systems. Human error rates were evaluated using Human Reliability Analysis models. These models are similar to event trees and model the operators actions sequentiality, including recovery. The models also include operator interaction with the available instrumentation. This detailed analysis of operator actions is used because earlier studies showed that operator actions dominated the analyses. Best estimate assumptions were made about the operator's environment.

Also, little use has been made of such time based models as the human cognitive reliability model. This is because more data would be needed to evaluate the performance impact of the redesigned control room on the time lines associated with certain tasks. Consequently, consideration for the impact of time available to perform a task, and other factors utilized by the cognitive reliability model, have been dealt with by judicious application of certain performance shaping factors defined in the System 80+ PRA. These are:

Availability of necessary indication Accuracy of indication Training Workload Annunciated Stress Level of Experience Quality/availability of procedures Ergonomic design of display/control

Each of the above factors can contribute a factor by which the Human Error Probability (HEP) can be modified, given the judgement of the analyst. This modification provides a value that more accurately reflects the probability of human error.

The estimated unavailability of plant systems is based on an adaptation of models developed for power operation. Earlier studies have shown that the core damage frequencies were dominated by operator error. Therefore, less detail is provided in the system analysis. The assumptions used for each system are presented in the discussion of each event tree branch point. 77a.wp(9212)bh

TABLE 3.2-1

PLANT STATES AND INITIATING EVENTS

Plant States

MODE 4, normal inventory
MODE 5, normal inventory
MODE 5, Reduced inventory
MODE 6, Refueling cavity empty
MODE 6, Refueling cavity full

Initiating Events

LOCA, inside containment LOCA, outside containment Loss of DHR Loss of Offsite Power Boron Dilution

3.3 INITIATING EVENT FREQUENCIES

A search of relevant literature and databases was conducted in order to assess the frequency of occurrence for the specified events actually experienced at pressurized water reactors during plant shutdown periods.

The initiating events identified and examined consisted of:

- Boron and Reactivity Events;
- Loss of Electrical Power;
- Loss of Reactor Coolant;
- Loss of Decay Heat Removal

The INPO Licensee Event Reports database constitutes the most reliable source of documentation for licensee svents, and was searched for the above listed events.

The literature search included direct examinations of the following:

- A. NUREG-1449, "Shutdown and Low-Power Operation at Commercial Nuclear Power Plants in the United States (Draft)", published 2/92
- B. INPO 91-007, "Selected Significant Operating Experience Report Recommendations 1980-1990", published 9/91
- C. NSAC-52, "Residual Heat Removal Experience Review and Safety Analysis, Pressurized Water Reactors", published 1/83
- D. FI L=1344, "?WR Low Power and Shutdown Accident Frequencies Programs, Phase 1A - Coarse Screening Analysis, 11/13/91, Prepared for NRC"

E. NRC Information Notice " .= 55, issued 8/31/90

F. NRC Information Notice 91-22, issued 3/19/91

The documents listed above were based to a large extent on examinations of LERs, but also provided documentation for a number of events not documented in the 'LER database. Additionally, the documents above also drew 1 semation from other literature, including:

G. NSAC-144, "Loss of Offsite or At U.S. Nuclear Power Plants, All Years Through 1988", published 1989
H. NUREG/CR-4550, Volume 1, Rev 1, SAND86-2084, Jan 1990

J. Sequence Coding and Search System, Office of Analysis and Evaluation of Operational Data, U.S. Nuclear Regulatory Commission.

In total, the literature and database provide a reasonably complete source of data for calculating the event frequencies. Nevertheless in order to further ensure that such frequencies have been calculated based on the earliest date recorded for each event, as summarized in Table 3.3-1.

The frequency for each initiating event was distributed over all the plant states based on the fraction of time that the plant is in each configuration. A 23 day refueling schedule was developed for the System 80+. This schedule was divided into each plant state listed in Table 3.2-1. The number of hours in Modes 4 and 5 was then doubled to account for forced outages. The first column of Table 3.3-2 gives the fraction of time (given as %) that the plant is in each of these states. The frequencies for loss of DHR and LOCA are then distributed over the various plant states based on the percent of time in each plant state. This distribution assumes that the human errors and mechanical failures that cause the initial event are random in nature.

TABLE 3.3-1

OBSERVED FREQUENCIES OF DHR EVENTS IN PWRS

Event	Earliest Event Date	Number of Events	Reactor Years (see note *)	Frequency (per reactor-yr)
Boron/Reactivity	03/14/82	12	747.3	. 02
Loss of Electrical Power	06/07/73	52	1412.3	.04
Loss of Reactor Coolant	11/14/79	51	924.7	.06
Loss of Shutdown Cooling	09/02/76	202	1165.3	.20

*note: total reactor-years (PWR) between "Earliest Event Date" and 01/01/92

TABLE 3.3-2

INITIATING EVENT FREQUENCY FOR PLANT STATES

PLANT STATE	* F IN 8	IOURS	FREQUENCIES, LOSS OF DHR	/RY _OCA
MODE 4		21	0.042	0.013
MODE 5		27	0.054	0.016
MODE 5 REDUCED INVENTOR	RY	18	0.026	0,011
MODE 6 REFUELING CAVITY	NOT	16 FLOODED	0.032	0,009
MODE 6 REFUELING CAVITY	FLO	18 DED	0.036	0.011
SUM		100	0.20	0.06

3.4 ACCIDENT SEQUENCES

For each of the accident sequences considered in this study, an event tree was developed and evaluated. The event tree starts with the initiating event frequency developed in Section 3.3 and describes the sequence of events as discrete events which are completed either successfully or not. Each specific sequence leads to either core damage or a safe plant condition. The following subsections discuss each event tree and their branch points. Similar branch points are used in many of the event trees. Such event trees are described in some detail the first time they are used and referred to in later uses.

The event tree branches are composed of an operator error and a system mechanical failure. Since earlier shutdown risk studies have identified operator error as a significant contributor to risk, emphases has been placed on operator error. The Human Reliability Analysis (HRA) trees are also given and discussed in this section at the branch point where they are first used. System fault trees for hardware failures are not presented in this report, but are presented in the System 80+ PRA².

3.4.1 LOSS OF DHR, MODE 4

Figure 3.4-1 gives the event tree for loss of DHR during Mode 4 operation. The following paragraphs discuss each branch point of the event tree. The operator has approximately 2 hours to restore DHR²¹ before core damage, assuming that the loss of DHR occurs at the beginning of the outage (a conservative assumption). In this mode, two SIS trains are available and the IRWST is full. It is also assumed that the SGs are still available.

LDHRM4

The initiating event frequency for loss of SCS while in mode 4 is 0.042/ry. The derivation for this frequency is given in Section 3.3.

OR (Operator Restores)

In 16% of the DHR losses, equipment failures were identified as the cause of loss of DHR³. It is assumed no operator action will restore these mechanical failures in the short term in the operating SCS train.

In the remaining cases, the SCS train is lost because of some operator error and can be restored by the operator taking some manual action. The operator has the ability to detect loss of the SCS with flow alarms, CCW alarms, temperature readout, pump suction and discharge readout, and pump current readout. The operator will have procedures and training in the operation of the SCS system and will be using it at regular intervals.

The performance of the SCS operating train may be lost because of some operator error that causes the pumps to trip but does no significant damage to the system. In this case the operator may be able to restart the operating SCS train by restarting the pump. In order to achieve this, the operator must identify that there has been a loss of shutdown cooling (task A), restart the pump (task B), adjust the SDC flow control valve(task c) and adjust the SDC temperature controller as necessary (task D, see Figure 3.4-2).

The operator can identify the loss of SDC by SDC alarms, SDC flow alarms, Component cooling water alarms, pump suction and discharge indications, pump component indications and Decay heat removal critical function alarms. Once the identification has been achieved, the process of restart will proceed according to the available lower mode functional recovery procedures.

Task	HEP	PSF(specific)	Source (NUREG- CR-1278)
A	0.0001	/5 (control room ergonomics)	20-23
В	0.001 + 0.0005/5	/5 (control room ergonomics)	13-3
с	0.001 + 0.0005	/5 (control room ergonomics)	13-3
D	0.001 + 0.0005	/5 (control room ergonomics)	13-13

Quantification

HEP = 0.0001/5 + 0.0011/5 + 0.0015/5 + 0.0015/5= 1.0×10^{-3}

The total failure rate for this branch point is the sum of the mechanical failure rate (0.16) and recovery from operator error. It is dominated by the mechanical failures and is: OR = 0.16

OS1 (Operator Starts 1 of 1 trains)

If the operator can't start the first SCS train, he will go to start the second SCS train. The technical specifications requires that the train be available. It must be manually started. This event tree branch includes both operator failure to start the train and the probability of mechanical failure of the train to start and run. The CSS pump can be aligned to substitute for the SCS pump. This action is included in the OS1 failure rate. The mechanical failure for a single train is 2.0E-2.

Since the technical specifications require that a redundant train of the Shutdown cooling system be available, the operator may be able to start this manually if re-starting the original train was unsuccessful.

Assuming that the train is available and operable, the operator would need to identify the need to make use of the redundant train (task A), start the SDC pump (task B), ensure that the SDC flow control valve is controlling the SDC flow (task C), ensure that the SDC temperature controller is performing correctly (task D-see Figure 3.4-3).

Task	HEP	PSF(specific)	Source (NUREG- CR-1278)
A	0.0001 +0.01	/5 (control room ergonomics)	20-23 20-7
В	0.001 + 0.0005/5	/5 (control room ergonomics)	13-3
с	0.001 + 0.0005	/5 (control room ergonomics)	13-3
D	0.001 + 0.0005	/5 (control room ergonomics)	13-13

Quantification

HEP = 0.0101/5 + 0.0011/5 + 0.0015/5 + 0.0015/5= 2.8 x 10⁻³

The total failure rate is the sum of the mechanical and operator failure rates and is 2.3E-2. It is dominated by mechanical failure.

SGHR (Steam Generator Heat Removal)

The operator normally uses the SGs for heat removal in Modes 1 through 3 and will try to use them after failure to restore the SCS trains. By Technical Specification, he must have a SG available if 2nd SCS train is out of service. He only has electric feedwater pumps available. In many cases the SGs are filled with water when laid up, and in this case, the operator has additional time to start the feedwater pumps. The secondary side can be vented through the ADVs.

The operator normally uses the steam generators during modes 1 through 3 and will be required to attempt to use them after failure to restore the SDC trains. Technical specifications require that at least one steam generator be available if the 2nd train of SDC is out of service.

In order to utilize this method of decay heat removal, the operator needs to identify the need to use a steam generator (task A), start the electric emergency feedwater pumps (task B), when the steam generator pressure rises to 950 psia to vent the ADV (task C monitor, task D actuate ADVs), and monitor the decay heat removal (task E, see Figure 3.4-4).

The indications available to the operator to identify the need for this method include SDC alarms, component indications for the availability of the SDC system and critical function decay heat removal alarms. The actions associated with this method can all be achieved from the control room. These actions are adequately covered in the lower mode functional recovery procedures.

Task	EEP	PSF (specific)	Source (NUREG- CR-1278)
Α	0.0001 +0.01	x2 (stress)	20-23 20-7
В	(0.001 + 0.0005/5)x2	x2 (stress)	13-3
C	0.001	x2 (stress)	13-3
D	0.001 + 0.001 + 0.0005	x2 (stress)	13-13
E	0.001	x2 (stress)	20-10

Quantification

HEP = 0.0101*2 + 0.0022*2 + 0.001*2 + 0.0025*2 + 0.001*2= 3.4×10^{-2} The mechanical failure to start and run 1 of 2 electric feedwater pumps was estimated as 7.0E-4. For this brench point, the total failure rate is 3.5E-2 and is dominated by operator error.

SIFB2 (Safety Injection system Feed and Bleed, 1 of 2 trains)

The operator has the option of going to a feed and bleed operation. Two SI trains are available in the division with the Diesel also available. One of the two pumps will supply sufficient flow. In addition, one of the two charging pumps could also be used but was not modeled here. The primary system must be vented with the Safety Depression System (SDS). The failure rate for SIFB2 is the sum of the failure of the operator to initiate feed and bleed and mechanical failures.

While in mode 4 or lower, if the operator looses shutdown cooling he will attempt to satisfy decay heat removal. Assuming that all else fails, the operator would attempt to initiate feed and bleed cooling.

To initiate feed and bleed cooling the operator must first identify that feed and bleed cooling is required (task A). Then ensure that the RCPs are tripped (task B) and ensure that the line-up from the IRWST through the SI pumps to the DVI line is available (task C). Since RCS pressure is below the shutoff head of the Safety Injection Pumps already, safety injection initiation would need to be achieved manually (task D). Then they must open the Safety Depressurization Valves (task E) and the Safety Depressurization Block Valves (task F) to allow circulation. In Mode 5, reduced inventory operation, opening the SDS valves is not necessary because the pressurizer manway is removed. Finally the operator would monitor flow to ensure the correct operation of the system (task G - see Figure 3.4-5). To be successful, Feed and Bleed Cooling must be initiated at or before the time at which the primary safety valves lift.

All actions can be performed from the control room. Procedures would be available in the lower mode functional recovery procedure for loss of secondary side heat removal. Inclusion of this event represents failure to initiate feed and Bleed cooling in mode 4.

6 33.3	10. 81	Sec. 12 (Sec.	5	55 W S	10. 20.	
Nº U	CL 1 1	1.1.1	1.1.2	a c 1	OR.	
- Carlos and a second		and a substrated by	a film along	1000	and the first state	

Task	HEP	PSF (s	pecific)	Source (NUREG-CR-1278)
A	0.0001+0.01	/5 (control room ergonomics	20-23 20-7

B	0.05*0.05	/5	(control room ergonomics	20-22
С	0.05*0.05*0.05*0 .05	/5	(control room ergonomics	20-22
D	0.001*0.001	/5	(control room ergonomics	20-12
E	0.001*0.001	/5	(control room ergonomics	20-12
F	0.001*0.001	/5	(control room ergonomics	20-12
G	0.001	/5	(control room ergonomics	13-3

The mechanical failures consist of failure of both SI trains (7.0E-4) and the SDS (8.0E-4). Total failure for feed and bleed is: SIFB2 = 4.2E-3 and is dominated by operator error.

BOC (Boil-Off using CVCS)

Using either 1 of 2 charging pumps or the boric acid makeup pump as a backup, water can be added directly to the reactor vessel from the Boric Acid Storage Tank (BAST). There are approximately 80,000 gal. of borated water available and by matching the boil-off rate, the water in the BAST can be used to cool the core for approximately 12 hrs. There is an additional 20,000 gal of water available in two of the four SITs that could be added to the core through gravity feed and give the operators an additional 3 hours. It is assumed that if BOC is successful, DHR is restored in the approximate 12 hours during boil-off.

BOC failure rate consists of two failure rates, failure to feed with the CVCS and failure to restore some DHR capability during the approximate 12 hours.

During a Loss of Coolant Accident in the lower modes, making up inventory loss may be achieved via a different method than at

power. It is possible to makeup inventory from the Boric Acid Storage tank through the CVCS and the charging pumps.

This requires that a line up be established from the BAST to the suction line for a charging pump, then the charging pump can deliver the inventory to the RCS.

The actions on the part of the operator would be to identify the need for makeup (task A), open the gravity feed line from the BAST to the charging pumps i.e. valve CH-532 (task B) and CH-536 (task C), ensure that one of the two charging pumps is operating (task D), and monitor the makeup (task E - see Figure 3.4-6).

All actions can be performed from the control room. Procedures would be available in the lower mode functional recovery procedure for loss of secondary side heat removal. Inclusion of this event represents a failure to makeup inventory coolant from the CVCS.

all a set of	a sea des	3 10	St. march			
0112	1.73.70	3.1	3 634	5 M. S	1.000	S
Sec. 3-6, 54	A. J. J. Sec.	A. A.	al. 144 1	A 60.3	L 307 L	
Contract of the second s				and the second		

Task	HEP	PSF (specific)	Source (NUREG-CR-1278)	
A	0.0001+0.01	/5 (control room ergonomics	20-23 20-7	
В	0.001 + 0.0005	/5 (control room ergonomics	13-3	
c	0.001 + 0.0005	/5 (control room ergonomics	13-3	
D	0.001 + 0.0005/5	/5 (control room ergonomics	13-3	
E	0.001	/5 (control room ergonomics	13-3	

HEP = $0.0_01/5 + 0.0015/5 + 0.0015/5 + 0.0011/5 + 0.001/5$ = 3.0×10^{-3}

In this scoping study, we did not explicitly analyze the probability of restoring DHR in approximately 12 hours, but assumed a failure rate of 0.1. The operator has the options of restoring any of 4 SI trains, 2 SCS trains or 2 CSS trains. He could also find additional water for his BAST and continue boil-off. As an example of such repairs, after the severe Brown's Ferry cable fire (March, 1975), operators successfully restored DHR using temporary jumpers. BOC failure rate is dominated by the probability of repair and BOC = 0.1.

Figure 3.4-1 gives the event tree with the branch failure rates discussed above. In this event tree, one sequence leads to core damage with a frequency of 2.3E-9.

3.4.2 LOSS OF DHR, MODES 5,6, REFUELING CAVITY EMPTY

Figure 3.4-7 gives the event tree for loss of DHR during Mode 5 operation when the coolant level is not reduced and during Mode 6 operation while the IRWST is still full. During this time, two SIS trains and one CSS train is available. It is assumed that the SGS

3-19

are not available and the system is being cooled with one SCS, and one in standby.

This event tree is similar to the previous tree. The operator has approximately 2.2 hours to restore DHR assuming that the sequence occurs when the plant first enters Mode 5 (highest decay heat) and not in later Mode 5 or 6 operations. The following paragraphs discuss each branch point of the event tree.

DHRCE (Decay Heat Removal loss with refueling Cavity Empty)

The initiating event frequency for loss of SCS while in Mode 5 is 0.054/ry. In Mode 6, with the refueling cavity empty, the frequency is 0.032. The derivation for this frequencies is given in Section 3.3. The total frequency is the sum, DHRCE = 0.086.

OR (Operator Restores)

The failure rate for the operator to restore the SCS train is (see Section 3.4.1): OR = 1.6E-1.

OS1 (Operator Starts)

The failure rate for the operator to start the second SCS train is (see Section 3.4.1) : OS1 = 2.3E-2.

SIFB2 (Safety Injection train Feed and Bleed)

The failure rate for the operator to start the feed and bleed operation is 4.2E-3 and discussed in Section 3.4.1.

SCSFB (Shutdown Cooling System for Feed and Bleed)

If the SIS fails, the reactor inventory and DHR can be maintained by using the SCS in a feed and bleed mode. The SCS can be aligned to the IRWST and used to inject water. Heat removal is through the SCS coolers. If the SCS pumps fail, the CSS pumps can be aligned to substitute for the SCS pumps.

While in mode 4 or lower and the operator attempts to satisfy decay heat removal, assuming that all else fails, the operator would attempt to initiate feed and bleed cooling. Also given that the safety injection pumps may be unavailable it is possible to use the SCS pumps for this purpose. To initiate feed and bleed cooling using the SCS, the operator must first identify that feed and bleed cooling is required (task A). Then ensure that the RCPs are tripped (task B) and the crossconnect valve SI-110 is opened to ensure the line-up from the IRWST is available (task C). This must then be checked by another operator, who could fail in either of two ways; task D - the 2nd operator makes a checking error and then changes the valve position to reflect his incorrect belief of the correct position, and task D'- a simple checking error with a report to the control room that the valve change was successful. The operators would need to start the Shutdown Cooling Pumps manually (task E). Then they must open the Safety Depressurization Valves (task F) and the Safety Depressurization Block Valves (task g) to allow circulation. In Mode 5, reduced inventory operation, the pressurizer manway is removed and opening the SDS valves is not necessary. Finally the operator would monitor flow to ensure the correct operation of the system (task H - see Figure 3.4-8). To be successful Feed and Bleed Cooling must be initiated at or before the time at which the primary safety valves lift.

All actions, except one, can be performed from the control room. Procedures would be available in the lower mode functional recovery procedure for loss of secondary side heat removal. Inclusion of this event represents failure to initiate feed and bleed using the SCS and the IRWST.

Task	HEP PSF (specific)		Source (NUREG-CR-1278)	
A	0.0001+0.0)	/5 (control room ergonomics	20-23
В	(0.05) ²	/5 (control room ergonomics	20-22
с	0.001	/5 (control room ergonomics	20-6
D	0.1*0,001	/5 (control room ergonomics	
D'	0.1			

Quantification

E	0.001+0.001	/5 (control room ergonomics	
F	0.001*0.001	/5 (control room ergonomics	
G	0.001*0.001	/5 (cortrol room ergonomics	
н	0.001	/5 (control room ergonomics	20-12

HEP = a + b + cd' + ce + d + e + cf + f + cg + g + ch + h

= 0.0101/5 + 0.0025/5 + 0.0001/5 + 0.0001/5 + 0.000002/5 + 0.002/5 + 0.00000001/5 + 0.000001/5 + 0.000000001/5 + 0.000001/5 + 0.000001/5 + 0.001/5

 $= 3.1 \times 10^{-3}$

The mechanical failures for SCSFB are the mechanical failures for SCS injection (MUI) and the mechanical failure of the SDS (DP). SDS is needed to return the coolant to the IRWST tank and not to depressurize the system. The total failure rate is: SCSFB = 4.9E-3.

BOC (Boil-Off using the Cvcs system)

As discussed in Section 3.4.1, the failure rate to successfully cool the core and restore DHR is 0.1.

Figure 3.4-7 gives the event tree with the branch failure rates discussed above. In this event tree, one sequence leads to core damage with a frequency of 6.5E-10.

3.4.3 LOSS OF DHR, MODE 5, REDUCED INVENTORY

Reduced inventory is defined as having the primary coolant level 3 ft. below the flange or lower. Experience developed with operating plants and results of earlier Shutdown PRAs has shown that because of the short time for coolant boiling given loss of DHR (approximately 15 minutes assuming mid loop operation), the plant has technical specifications and operational guidance for operation in this mode. Figure 3.4-9 gives the event tree for loss of DHR during Mode 5 operation when the coolant level is reduced. This event tree is also used to evaluate LOCAs during reduced inventory because the operator would treat both events the same way. For this event, the operator has approximately 1.5 hours to restore DHR²¹ before core damage assuming that the loss of DHR occurs at the first time in the outage that the plant is operated at reduced inventory (a conservative assumption). The following paragraphs discuss each branch point of the event tree.

DHR5R (loss of Decay Heat Removal in mode 5, Reduced inventory)

The initiating event frequency for loss of SCS while in mode 5 is 0.047/ry. This frequency includes both loss of DHR (0.036) and the frequency for LOCAs in this mode (0.011). LOCAs were included here because the operator would perceive them as a loss of DHR. The derivation for this frequency is given in Section 3.3.

OR (Operator Restores)

The failure rate for the operator to restore the SCS train, discussed in Section 3.4.1, is: OR = 1.6E-1.

MUI (Make Up Inventory)

For reduced inventory events, it is assumed that the operator must diagnose and increase the coolant level before starting the second SCS train. We assumed the operator would use the SCS system for this task and align it to the IRWST. He could also use the CVCS system, but this was not explicitly modeled. Inclusion of the CVCS make up would not significantly reduce the failure rate because the failure rate is dominated by operator error.

While in the lower modes, making up inventory loss may be achieved via a different method than at power. It is possible to makeup inventory from IRWST through the SC system.

This requires that a line up be established from the IRWST to the suction line for a shutdown cooling pump, then the SC pump can deliver the inventory to the RCS.

The actions on the part of the operator would be to identify the need for makeup (task A), Ensure the RCPs are tripped (task B), locally open the cross-connect valve (task C), have this checked by another operator (task D - checking failure and reposition of valve to wrong position based on incorrect checking and task D' - checking failure and report to control room that change was successful), and start the pump (task E - see Figure 3.4-10).

Most actions can be performed from the control room. Procedures would be available in the lower mode functional recovery procedure for loss of secondary side heat removal. Inclusion of this event represents a failure to makeup inventory coolant from the IRWST through the SCS.

0° ...

Quantification

Task	HEP	PSF(spe	cific)	Source (NUREG- CR-1278)
A	0.01+0.0001	/5 (control room ergonomics)	20-23 20-2
В	0.05 x 0.05	/5 (control room ergonomics)	20-12
с	0.001	/5 (local ergonomics)	20-12
D	(0.1 x 0.001)			13-13
D'	0.1			13-13
E	0.001+0.001	/5 (control room ergonomics)	20-12

HEP = a + b + cd' + ce + d + e

0.0101/5 + 0.0025/5 + 0.0001/5 + 0.0000002/5 + 0.0001 + 0.002/5

 3.1×10^{-3}

The mechanical failure of the SCS to make up inventory is 1.0E-3. The total failure rate is: MUI = 4.1E-3, and is predominately operator error.

OS1 (Operator Starts)

The failure rate for the operator to start the second SCS train is 2.3E-2 and discussed in Section 3.4.1.

SIFB2 (Safety Injection train Teed and Bleed, 1 of 2 trains)

In reduced inventory mode, suggested new technical specifications require that two SIS train be available. As noted in Section

3.4.1, in this mode, the pressurizer manway has been removed and opening the SDS valves is not necessary. The fillure rate is: $S^{T}FB2 = 4.2E^{-3}$.

BOC (Boil-Off using the Cvcs system)

As discussed in Section 3.4.1, the failure rate to successfully cool the core using boil-off and restore DER is 0.1.

Figure 3.4-9 gives the event tree with the branch failure rates discussed above. In this event tree, two sequences lead to core damage and the total CDF is 8.5E-8.

3.4.4 LOSS OF DHR, MODE 6, REFUELING CANAL FLOODED

At this time in the refueling, the refueling cavity and canal is flooded. The IRWST is empty and in general, not available for any make up, safety injection, or feed and bleed operation. The steam generators also are not available for decay heat removal (assumed isolated with nozzle dams). Because of the lower decay heat and large volume of water in the refueling cavity, the operator has 81 hours to restore DHR before core damage²¹. Figure 3.4-11 gives the event tree for this event.

DHRCF (Decay Heat Removal with the Refueling Cavity Filled)

The initiating event Frequency for loss of SCS while in Mode 6, cavity flooded, is 0.026/ry. The derivation for this frequency is given in Section 3.3.

OR (Operator Restores)

The failure rate for the operator to restore the SCS train is OR = 1.6E-1 and is discussed in Section 3.4.1.

OS1 (Operator Starts)

The failure rate for the operator to start the second SCS train is 2.3E-2 and is discussed in Section 3.4.1.

REP (Repair)

Having failed to restore or start either SCS train in a timely manner, the operators now have approximately 81 hours to repair or restore DHR, or replenish the water in the refueling cavity. In this scoping study, we did not explicitly analyze the probability of restoring DHR in approximately 81 hours but assumed a failure rate of 0.01. The operator has the options of restoring any of the 2 SCS trains or 2 CSS trains. He could also refill the 1kWST and use any 1 of 4 SIS trains that he could make operational in a feed and bleed mode. Finally, he could also find additional water for his refueling cavity and continue boil-off.

BOC (Boil-Off using the CVCS)

As discussed in Section 3.4.1, the failure rate to successfully cool the core and restore DHR is 0.1.

Figure 3.4-11 gives the event tree with the branch failure rates discussed above. In this event tree, one sequence leads to core damage and the CDF is 1.3E-7. This is the second largest sequence in the study. It represents a sequence where there are only 2 trains for a DHR available (two SCS trains) and no capability to use any other paths (SIFB2, SCSFB).

3.4.5 LOCA, NODE 4, PRESSURE ABOVE 500 PSIG

Mode 4 is hot shutdown and includes the period of time that the primary system is being cooled and depressurized (or heated and pressurized). In mode 4, the coolant temperature varies from 350°F to 210°F. Above 500 psig, the SCS system is not available but the SIS, and SITs are. This plant configuration is very limitar to higher modes and are modeled here because it represents a mode where the SGs are being used and depressurization is required to get onto SCS or a feed and bleed mode. It is assumed that the temperature is above 317°F, and the pressure is above 500 psig. Two CSS trains, the IRWST, two RCS loops, and two SIS trains are available. The SCS trains are available only after depressurization. The low pressure and temperature part of Mode 4 is represented by the Mode 5 normal inventory event tree described in the next section. The Mode 4 tree is given in Figure 3.4-12.

M4LOCA

The frequency of a LOCA in Mode 4 is 0.013/ry (Section 3.3).

OI (Operator Isolates loak)

Most of the LOCAs in shutdown modes are caused by operator errors in aligning valves and other operations and are quickly corrected. Others are caused by mechanical failures that can not be easily isolated.

Given that most of errors are caused by operators inadvertently opening values or creating leak paths, it is assumed that most of these occurrences are caught either by the operator who caused it or by operator check work already performed.

With this assumption, it is reasonable to assume that the type of checking being performed would be associated with some kind of arerting factors associated with the type of situation i.e sound of rushing water when none should be present, visual indication of the leak etc.

For the purposes of this analysis, the quantification of this factor will make use of the value for special one-of-a-kind checking with alerting factors. This is because of the unique nature of most of the activities associated with the lower modes and the concept that this will be the type of error the operator will be attuned to.

Quantification

Task	HEP	PSF(specific)	Source (NUREG- CR-1278)
A	0.05		20-22

HEP = 0.05

For leaks caused by mechanical failures it was assumed that 4% could not be isolated and are sufficiently large to be true LOCAs. The total failure to isolate is the sum of the mechanical and operator failures and is : OI = 0.09.

SIP2 (Safety Injection Pumps)

In Mode 4, and at these pressures, two SI trains are available and are automatically actuated. Operator action to initiate SI is not required and the failure rate for both trains to start and run is 7.0E-4. The IRWST is available as a water supply.

CS2 (Containment Spray)

To remove decay heat from the containment, one of two containment spray trains must start and run. Failure of CS is 3.10E-3.

SGCOM (Steam Generator Operator Commission Error)

The operator was using the SGs before this event started and, after isolating the leak, would continue to use the SGs for heat removal. This branch element represents failure of the operator to continue using this DHR path.

Typically the method employed to cooldown the RCS using the steam generators in the lower modes is to fill the steam generator using the emergency feedwater system then switch off and let this boil away.

Since the operator action involves checking the steam generator level and initiating the emergency feed this operator error has two components. Task A - failure to initiate steam generator heat removal (SGHR) and task B - operator fails to check for the initiation (see Figure 3.4-13). 7.2.wp(9212)bh

Quantification

Task	HEP	PSF (specific)	
Α	3.4 × 10 ⁻³	see SGHR	see SGHR
В	0.01		20-22

HEP = $3.4 \times 10^{-3} \times 0.01$ SGCOM = HEP= 3.4×10^{-5}

DP (DePressurization)

If the SIS fails or heat can not be removed through the SGs, than the operator must depressurize the primary system in preparation to use the SCS or CSS to add inventory and remove decay heat. The operator must manually actuate the Safety Depressurization System (SDS).

In order to use the shutdown cooling pumps or containment spray pumps to add inventory the operator must depressurize the system. This is achieved by utilizing the safety depressurization system.

In order to achieve depressurization the operator must identify the need for depressurization using the SDS (task A), cpen the safety depressurization valves (task B), open the safety depressurization block valves (task C), and monitor the flow through the system to ensure that it is operating properly (task D - see Figure 3.4-14).

These actions can be performed from the control room and would be part of the lower mode functional recovery procedure.

3-29

Quantification

Task	HEP	PSF(specific)	Source (NUREG- CR-1278)
A	0.01+0.0001	/5 (control room ergonor_cs)	20-23 20-2
В	0.001 x 0.001	x2 (stress)	20-12
с	0.001x 0.001	x2 (stress)	20-12
D	0.001		13-13

HEP = 0.0101/5 + (0.000001)*2 + (0.000001)*2 + 0.001= 3.0×10^{-3}

Mechanical failure of 1 cf 2 MOVs in the SDS to open is 8.0E-4. The total failure to depressurize is: DP = 3.8E-3 and is dominated by operator error.

OS2 (Operator Starts one of 2 SCS trains)

Having depressurized the system, the operator can now start one of two SCS systems. Since the plant was pressurized, both trains are assumed available. The operator error in starting 1 of 2 SCS trains is assumed to be the same as that for starting 1 SCS train and is 2.8E-3 (Section 3.4.1). Mechanical failure of both SCS trains is 2.0E-3 and the total failure rate is: CS2 = 4.8E-3.

SIFB2 (Safety Injection to Feed and Bleed, 1 of 2 trains)

Assuming that the normal DHR paths can not be used and the operator has depressurized the system, the operator can now go to a feed and bleed operation. This operation is not available if the SIS had failed earlier in the event sequence. The failure rate for this branch is: SIFB2 = 4.2E-3 and is discussed in Section 3.4.1.

Figure 3.4-12 gives the event tree with the branch failure rates discussed above. In this event tree, eight sequences lead to core damage and the total CDF is 2.3E-8. Half of the sequences and almost all of the risk is caused by failure to depressurize so that the SCS or feed and bleed trains can be employed.

3.4.6 LOCA, MODES 5, 6 (IRWST FULL)

In Mode 5, and some part of Mode 6, the lRWST is available for safety injection and two SIS trains are available for both injection and feed and bleed operation. Figure 3.5-15 gives the event tree for this sequence.

The starting point for this event is that the plant is depressurized and one SCS train is in operation. The SGs are assumed to be isolated. Two SIS trains are available for manual actuation in Mode 5 and Mode 6 with the IRWST full. One CS pump is also available for injection or feed and bleed.

L56IR (LOCA Mode 5 and 6, IRWST Available)

The frequency for this event is C.025. It consists of Mode 5 LOCAs (0.016) and Mode 6 LOCAs with the IRWST full (0.009). LOCAs occurring during Mode 5 Mid loop operation would be diagnosed as loss of DHR by the operator and are addressed in Section 3.4.3.

OI (Operator Isolates leak)

The failure of the operator to isolate the leak is discussed in Section 3.4.5 and is: OI = 9.0E-2.

SIP2 (Safety Injection Pump, 1 of 2)

In Mode 5 and 6 with the IRWST full, two SI trains and the IPWST are available. SIS would be manually actuated.

While in lower modes, the system pressure would be lower than the Safety Injection setpoint, and since procedures require the override of the ESFAS actuation for normal progression below, it would be necessary for the operators to manually initiate Safety Injection. The actions required or this task are that the operator must identify the need fc. safety injection actuation (task A), ensure the correct line up for the required Safety Injection pump (task B), then start the Safety Injection pump (task C, see Figure 3.4-16).

All of these actions can be performed from the control room, and are governed by lower mode emergency procedures. Inclusion of this event in the fault tree represents the failure of the operator to manually initiate Safety Injection.

Quantification

Task	HEP	PSF (s	pecific)	Source (NUREG- CR-1278)
Α	0.0001+0.01	/5 (control room ergonomic s	20-23
В	(0.01)2			20-22
С	0.001 + 0.0005/5	/5 (control room ergonomic s	20-12

HEP = 0.0101/5 + 0.0002 + 0.0011/5= 2.3 x 10⁻³

The mechanical failure rate for one of two trains of SIS to start and run is $\therefore 0E-4$. The operator error rate is 2.3E-3 and the total failure rate is: SIP2 = 3.0E-3 and is dominated by operator error.

OS1 (Operator Starts one SCS train)

Having isolated the leak, the operator can now start one of the SCS systems. It is assumed that the leak caused the other train to be unavailable. The failure rate is: OS1 = 2.3E-2 and discussed in Section 3.4.1.

SIFB2 (Safety Injection to Feed and Bleed)

If the leak isolation was successful but starting the SCS train was not successful, then the operator can continue to cool the core by operating one of two SI pumps after having opened the SDS to return the fluid to the IRWST. This failure rate SIFB2 = 4.2E-3 and is discussed in Section 3.4.1.

SCSFB (Shutdown Cooling System for Feed and Bleed)

If the SIS fails, the reactor inventory and DHR can be maintained by using the SCS in a feed and bleed mode. The use of the SC pump is included in the SCS fault tree. This branch point failure rate was evaluated in Section 3.4.2 and is: SCSFB = 4.9E-3.

BOC (Boil-Off using the CVCS)

As discussed in Section 3.4.1, the failure rate to successfully cool the core and restore DHR is 0.1.

Figure 3.4-15 gives the event tree with the branch failure rates discussed above. In this event tree, three sequences lead to core damage and with total CDF being 8.5E-9. The dominant sequence is failure to isolate the leak, followed by failure of the SIS and SCS to feed and bleed.

3.4.7 LOCA OUTSIDE CONTAINMENT, MODE 6

The LOCA outside containment (Figure 3.4-17) is an interfacing type LOCA where the coolant inventory is lost and in Mode 6 with the refueling cavity filled, the IRWST is not available. This means that the SIS is not available and no feed and bleed path is available (SIFB2 or SCSFB).

L6OC (LOCA Mode 6 Outside Containment)

The frequency for this event is 0.0055 assuming that half of the Mode 6 LOCAs occurred inside containment and the other half occur outside.

OI (Operator isolates leak)

The failure of the operator to isolate the leak is discussed in Section 3.4.5 and is: OI = 9.0E-2.

OIC (Operator Isolates at Containment interface)

If the leak was not initially isolated locally (OI), the operator will try to isolate the leak at the containment penetrations using 1 of 2 MOVs that are in series.

During a Loss of Coolant Accident in the lower modes and once the makeup issue has been addressed, the operators' primary concern will be isolating the leak that is causing the flow outside containment. This does not necessarily mean identifying the leak only dealing with the flow path to outside containment. In order to mitigate this, the operator would need to actuate a close signal to either of the two containment isolation valves.

The actions performed to achieve isolation at the containment would be, to identify the need to isolate once core decay heat removal has been satisfied (task A), actuate the controls to close valve 1 (task B), and close valve 2 (task C), then verify the success of this approach by monitoring the RCS level indication (task D, see Figure 3.4-18).

All actions can be performed from the control room. Procedures would be available in the lower mode functional recovery procedure for loss of secondary side heat removal.

Task	HEP	PSF (specific)		Source (NUREG-CR-1278	
A	0,0001+0.01	/5 (control room ergonomics	20-23 20-7	
В	0.001 + 0.0005	x2 (stress)	13-3	
C	0.001 + 0.0005	×2 (stress)	13-3	
D	0.001	/5 (control room ergonomics	13-3	

Quantification

= 0.0101/5 + 0.0015x2 + 0.0015x2 + 0.001/5 $= 8.2 \times 10^{-3}$

The mechanical failure of 1 of 2 MOVs was estimated to be 8.0E-4 and the total failure rate is: OIC = 9.0E-3 and is dominated by operator error.

OS1 (Operator Starts one SCS train)

Having isolated the leak, the operator can now start one of the SCS systems. It is assumed that the leak caused the other train to be unavailable. The failure rate is: OS1 = 2.3E-2 and is discussed in Section 3.4.1.

REP (REPair)

At this point in the sequence, the operator has successfully isolated the leak but now has a loss of DHR event, with neither of the SCS trains working and no feed and bleed capability. As

discussed in Section 3.4.4, failure to restore either SCS train was estimated as: REP = 0.01.

BOC (Boil-Off using the CVCS system)

As discussed in Section 3.4.1, the failure rate to successfully cool the core and restore DHR is 0.1.

Lyure 3.4-17 gives the event tree with the branch failure rates discussed above. In this event tree, two sequences lead to core damage and the total CDF of 4.6E-7. The dominant sequence in this event (and this study) is the LOCA outside containment that is not isolated. The second sequence represents a loss of DHR and requires operator action to repair one of the two SCS trains.

3.4.8 LOCA INSIDE CONTAINMENT, MODE 6 (REFUELING CAVITY FULL)

In Mode 6, with the refueling cavity filled, if the LOCA occurs inside the containment, the coolant will drain back into the IRWST and be available for injection or feed and bleed using any of the SCS pumps or the CS pump in the make-up (MUI) mode. The coolant also becomes available for a feel and bleed mode using the SCS and CS trains. Figure 3.4-19 gives the event tree for this sequence.

The starting point for this event is that the plant depressurized and one SCS train in operation. The SGs are assumed to be isolated. No SIS trains are available. One CS pump and both SCS pumps are available for injection after the coolant drains back into the IRWST.

LGIC Loca Mode 6 Inside Containment

The frequency for this event is 0.0055. This assumes that half of the Mode 6 LOCAs occurs inside containment.

OI (Operator Isolates leak)

The failure of the operator to isolate the leak is discussed in Section 3.4.5 and is: OI = 9.0E-2.

MUI (Make Up Inventory)

In Mode 6 with the IRWST empty, SIS is not available but injection is possible with SCS and CS pumps (Section 3.4.3): MUI = 4.1E-3.

OS1 (Operator Starts one SCS train)

The failure rate is: OS1 = 2.3E-2 and is discussed in Section 3.4.1.

SCSFB (Shutdown Cooling System for Feed and Bleed)

If the SIS fails, the reactor inventory and DHR can be maintained by using the SCS or the CS pump in a feed and bleed mode. This branch point failure rate was evaluated in Section 3.4.2 and is: SCSFB = 4.9E-3.

BOC (Boil-Off using the Cvcs system)

As discussed in Section 3.4.1, the failure rate to successfully cool the core and restore DHR is 0.1.

Figure 3.4-19 gives the event tree with the branch failure rates discussed above. In this event tree, three sequences lead to core damage and with total CDF being 6.3E-8. The dominant sequence is failure to restore the second SCS train (including the CS pump) including feed and bleed.

3.4.9 LOSS OF OFFSITE POWER

In the previous Shutdown Risk Studies summarized in NUREG 14493. loss of offsite power (LOOP) was not a major contributor to risk. In the Seabrook Shutdown PRA (May, 1988), LOOP contributed 6% of the core damage risk. This sequence was only qualitatively studied in this analysis because of the low frequency for LOOP for the System 80+ plant. Section 2.4.3.2 describes AC power reliability. The System 80+ has the capability to draw from four different AC sources including a combustion turbine. With two independent incoming sources and switchyards, the EPRI ALWR Utility Requirements Document gives the probability of losing both switchyard as 7.7E-3. When this is coupled with failure of 1 Diesel and failure of the combustion turbine, the risk becomes very small. Additionally, in Section 3.3 it was estimated that the refueling cavity is flooded for 29% of the time the plant is in Modes 3 through 6. During this period, a long recovery time exists for restoration of AC power.

3.4.10 CRITICALITY EVENTS

In the previous Shutdown Risk Studies summarized in NUREG 1449³ reactivity events were not a major contributor to risk. Boron dilution could be either a rapid injection of deborated water or a slow dilution. Section 2.6 discusses rapid boron dilution events and concludes that the reactor would stay subcritical, even for the

maximum credible size slug. A slow boron dilution event would be detected and terminated resulting in acceptable consequences.

The critical function monitoring system has been incorporated into The System 80+ and applicable for shutdown modes. The probability of missing a violation of one of these critical functions, particularly, Reactivity Control is small. The increased concern of the operators with reactivity events, due to the possibility of personnel hazards associated with having operators in containment, causes their stress level to increase positively and increases their vigilance.

Because the risk of criticality events is believed to be small, compared to loss of DHR and LOCAS, These events were not quantified.

EVENT TREE FOR LOSS OF DHR, MODE 4



83

LOSS OF RHR EVENT TREE MODE =4





Operator Fails to Make Use of the Steam Generators as Emergency Decay Heat Removal



Operator Fails to Initiate Feed and Bleed Cooling



Operator Fails to Inject Inventory from the Boric Acid Storage Tank to the RCS



EVENT TREE FOR LOSS OF DHR, MODE 5,6 REFUELING CAVITY EMPTY




EVENT TREE FOR LOSS OF DHR, MODE 5 REDUCED INVENTORY





.

EVENT TREE FOR LOSS OF DHR, MODE 6 REFUELING CAVITY FULL



EVENT TREE FOR LOCA, MODE 4



Operator Fails to Continue to Make Use of the Steam Generators as Emergency Decay Heat Removal



Operator Fails to Initiate Safety Depressurization



EVENT TREE FOR LOCA MODES 5,6 (IRWST AVAILAPLE)



Operator Falls to Manually Initiate Safety injection



EVENT TREE FOR LOCA, MODE 6 LOCA OUTSIDE CONTAINMENT

CLASS		OK YO	Xo	OK	OK	CM	OK	ð	
SEG, PROB.		5.00E-03	4.79E-04	1.12E-05	1.02E-07	1.13E-08	4.01E-06	4.46E-07	
BOIL OFF WITH 2VCS	E DC					1.00E-01		1.00E-01	
81 HOUR REPAIR	REP				1.00E-02				
OPERATOR STARTS SECOND DHR THAIN	0S1				<				
OPERATOR ISOLATES AT CONTAIN MENT	OIC						9.005-03		
DPERATOR ISOLATES	21		9.00E-02						
	LGOC			5.505-03					



1

•

•

.

Operator Fails to Isolate the Leak at the Containment Boundary

Operator identifies the need to isolate the leak to outside by isolating the containment boundary

.

Operator falls to identify the need to isolate the leak to outside by isolating the containment boundary

> Operator actuates CI operator's module to close isolation valve 1

Operator fails to actuate Ct operator's module to close isolation valva 1

Operator actuates CI operator's module to close isolation valve 2

Operator vertifies that isolation worked by observing RCS vesses level

Operator fails to actuate Cl operator's module to close isolation valve 2

The state

þ.

A Starter

5. 6

-

6

Operator fails to verify that Isolation worked by observing RCS vessel level

EVENT TREE FOR LOCA, MODE 6, LOCA IN CONTAINMENT



1.

3.5 RADIOLOGICAL CONSEQUENCES

The analysis presented in Section 3.4 quantifies the probability of core damage during various shutdown modes. For a Level III PRA this analysic would be continued to quantify the risk to the offsite population from the release of the fission products. The fission product inventory is reduced during shutdown because of two effects. If the accident occurs after the fuel shuffling is completed, one third of the core contains fresh fuel with no fission products. Also, as the outage continues, the nuclear inventory is reduced by fission product decay.

To demonstrate the effects of fission product decay on the radiological consequences, an evaluation of the nuclide inventory was performed starting with the nuclide inventory associated with 4 days after shutdown. This time was selected because it represents the minimum time during which mid-loop operation can be initiated. Tables 3.5-1 and 3.5-2 illustrate the effects of the fission product decay process on the overall core radioactive nuclide inventory and the concentration of potentially volatile fission products. A review of the radionuclide decay chain indicates that the fission product inventory has decreased by 30% by the end of a 20 day refueling. Furthermore, the concentration of potentially volatile nuclide releases will be reduced by at least a factor of 5. If the shutdown is extended to 40 days, the nuclide inventories would be one half of the four-day level and the concentration of the volatile fission products would be only 6% of that potentially available at 4 days into shutdown.

TABLE 3.5-1

FRACTIONAL CORE RADIOACTIVITY FOLLOWING SHUTDOWN

FRACTIONAL RADIOACTIVITY
1.0
0.62
0.45

* VALUES NORMALIZED TO RADIONUCLIDE CONCENTRATIONS AT 4 DAYS

TABLE 3.5-2

FRACTIONAL DECAY OF SELECT RADIOACTIVE NUCLIDE GROUPS FOLLOWING SHUTDOWN

NUCLIDE GROUP	DAYS AFTER SHUTDOWN					
	4 DAYS	20 DAYS	40 DAYS			
NOBEL GASES	1.0	.13	.02			
IODINE	1.0	.25	.04			
CESIUM	1.0	.95	.90			

4.0 APPLICABILITY OF CHAPTER 15 ANALYSES

4.0.1 FORMAT AND CONTENT

The purpose of this section is to present evaluations which confirm the analyses presented in Chapter 15 of CESSAR-DC conservatively bound the outcome of transients and accidents initiated during shutdown modes (Mode 2 subcritical or Modes 3 through 6) for the System 80+ Standard Design.

The approach used in the documentation of the events in Chapter 15 of CESSAR-DC is to present the results for the events with the most adverse consequences. As a result, reference is most frequently to events postulated to occur in Mode 1 or Mode 2 critical. Only in certain cases which intrinsically involve shutdown modes (e.g. startup of an inactive reactor coolant pump) are shutdown modes stressed. The purpose of this section is to ensure that all operating Modes have been treated in the Gocumentation for Chapter 15 events. The analysis models used are the same as used in Chapter 15 and have been approved by the NRC.

Table 4.0-1, together with Figure 4.0-1 and Table 4.0-2 present the complete range of initial conditions used for the analyses of Chapter 15 events for all Modes: 1 through 6.

The following sections have been organized to parallel the sections of Chapter 15 of CESSAR-DC. For example, Section 4.1.1 treats the same group of initiating events that are documented in Section 15.1.1 of CESSAR-DC. 77a.wp(9212)bh

TABLE 4.0-1

INITIAL CONDITIONS

Core Power	% of 3800 MWt	0 to 102 ⁽¹⁾
Axial Shape Index		$-0.3 \le ASI \le +0.3^{(2)}$
Reactor Vessel Inlet Coolant Flow Rate	% of 445600 gpm	95 to 116 ⁽³⁾
Pressurizer Water Level	<pre>% distance between upper tap and lower tap</pre>	26 to 60 ⁽⁴⁾
Core Inlet Coolant Temperature		
<90% Power ≥90% Power	• F.	543 to 565 ⁽⁵⁾ 553 to 563
Pressurizer Pressure	psia	1905 to 2375 ⁽⁵⁾
Steam Generator Water		
TOM	% Wide Range ⁽⁶⁾	33.7 ^{(8),(9)} 40.7
High	% Narrow Range ⁽⁷⁾	95.0

(1) Core power is that due to decay heat in Modes 3 through 6.

- (2) ASI = (area under axial shape in lower half of core area under axial shape in upper half of core)/(total area under axial shape). A range of valves of ASI from -0.6 to +0.6 was considered for subcritical core conditions.
- (3) For Modes 3 through 6 see TABLE 4.0-2.
- (4) Between 0 and 100% in Modes 4 and 5. Pressurizer is ampty in Mode 6 when the refueling pool is empty.
- (5) For Modes 3 through 6 see Figure 4.0-1.
- (6) Percent of distance between the wide range instrument taps. See Chapter 5 of CESSAR-DC for details.
- (7) Percent of distance between the narrow range instrument taps. See Chapter 5 of CESSAR-DC for details.
- (8) For steam and feedwater line breaks only.
- (9) ≥ 25% SG water level in Modes 3 and 4 and in Mode 5 for loops filled and additional RCS loop/SDC division not OPERABLE. Not applicable for Mode 5 with loops not filled and Mode 6.

TABLE 4.0-2

MODE SYSTEM FLOW RATE (%) 3 RCP 33 to 116 4 RCP 33 to 116 SCS 0.94 to 2.24 5 RCP 33 to 116 (Loops SCS 0.94 to 2.24 Filled) 5 SCS 0.22 to 2.24 (Loops Not Filled) 6 0.94 to 2.24 SCS

REACTOR VESSEL INLET COOLANT FLOW RATE



4.1 INCREASE IN HEAT REMOVAL BY THE SECONDARY SYSTEM

4.1.0 INTRODUCTION

The purpose of this section is to present evaluations which confirm that all increase in heat removal by the secondary system events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 80+ Standard Design.

Section 15.1 of CESSAR-DC documents results which show that all increase in heat removal by the secondary system events have acceptable consequences if they are explicitly postulated to occur in Mode 1 or Mode 2 critical for the System 80+ Standard Design. This is demonstrated for steam system piping failures inside and outside containment (Section 15.1.5) by analyses for both Mode 1 and Mode 2 critical. The choice of initial conditions for the other analyses of Section 15.1 to minimize the transient DNBR for any operating condition ensures that the results presented bound those for all of Mode 1 and Mode 2 critical.

4.1.1 DECREASE IN FEEDWATER TEMPERATURE

The consequences of decrease in feedwater temperature events initiated in a shutdown mode are bounded by those of the increase in feedwater flow events presented in Section 4.1.2. The increase in feedwater flow events of Section 4.1.2 are initiated at the coldest possible feedwater temperature and the highest feedwater flow rate. This extreme combination of conditions results in the most severe reactor coolant system overcooling transient and, consequently, the greatest potential for a return to criticality with subsequent power generation and potential for DNB.

4.1.2 INCREASE IN FEEDWATER FLOW

Evaluation of the consequences of increase in feedwater flow events postulated to be initiated in a shutdown mode shows that the results are no more adverse than those of the full power event documented in Section 15.1.2 of CESSAR-DC. The minimum transient value of DNBR is predicted to be greater than 3 for the most adverse combination of conditions, thus fuel integrity is not challenged. All increase in heat removal events analyzed in this section are characterized by decreasing steam generator pressures and RCS pressure, thus there is no approach to 110% of the design pressure of the steam generators or RCS.

Increase in feedwater flow events are only applicable when the steam generators are being used as a heat sink. This could occur in Mode 2 subcritical, Mode 3, or Mode 4 when the shutdown cooling system (SCS) is not being used for decay heat removal. The turbine-generator would be off-line. Steam would be exiting the steam generators via the atmospheric dump valves or the steam bypass control valves. Normal feedwater flow would be entering the steam generators from the startup feedwater system via the main feedwater piping. An increase in feedwater flow could occur via either the main or emergency feedwater systems.

Administrative controls require main feedwater pumps to be unavailable for inadvertent delivery of feedwater to the steam generators in shutdown modes. (See Table 2.1.1.) There are, therefore, two increase in feedwater flow event scenarios which can be postulated as potentially bounding events for shutdown modes:

- 1. Inadvertent EFAS Actuation: The initiating event is assumed to be a spurious Emergency Feedwater Actuation Signal (EFAS) for one steam generator. Each steam generator is fed from a separate Emergency Feedwater (EFW) train and each EFW train is actuated by a separate EFAS. The postulated initiating event results in starting both the turbine-driven and motor-driven EFW pumps to the affected steam generator. The cavitating venturi in the EFW supply pipeline limits the flow to less than 800 gpm. The minimum temperature from the EFW storage tanks is greater than 40°F.
- 2. Maximum Startup Feedwater Flow: The initiating event is assumed to be one control valve failing open, diverting all flow from the startup feedwater (SUFW) pump to one steam generator. One SUFW pump supplies both steam generators and there is a separate control valve for each steam generator. A bounding value of the feedwater flow is 4000 gpm (more than twice the specified 5% of main feedwater flow and greater than the expected runout flow). The water source is assumed to be the condensate storage tank which is conservatively assumed to be at 32°F.

To cover these postula 4 scenarios with extreme, bounding assumptions, a spectrum of cases were investigated using feedwater flows rates from less than 800 to 4000 gpm, all at 32°F, and all initiated from the most adverse set of initial conditions.

The most adverse initial conditions for excess feedwater flow events initiated in a shutdown mode occur in Mode 3. A core inlet temperature of 565°F maximizes the positive moderator and Doppler reactivity addition caused by the RCS cooldown. (Parametric cases demonstrated that the most adverse results occurred with the maximum initial temperature.) Initiating at a lower core inlet temperature results in less positive moderator and Doppler reactivity coefficients and in a lower rate of RCS overcooling. Further, less than four reactor coclant pumps (RCPs) can be operating in Mode 3 (as opposed to Mode 2 subcritical). This parameter has the most significant effect on the minimum DNBR. All cases were, therefore, initiated in Mode 3. The maximum k_{eff} allowed by Technical Specifications in Mode 3 provides the greatest potential for reaching criticality and consequent power generation due to the cooldown event prior to encountering a reactor trip on low steam generator pressure or high steam generator level. The excess feedwater flow events were, therefore, initiated at $k_{eff} = 0.99$. It was assumed that one or more regulating CEAs was inserted and all remaining regulating CEAs and shutdown CEAs were fully withdrawn.

To maximize the cooldown rate and the time required to reach the high steam generator water level trip setpoint, the events were assumed to be initiated at the minimum Mode 3 technical specification steam generato, water level of 25% wide range. Further, at the start of each event it was assumed that the decay heat production was zero. Consequently, all supplied feedwater was "excess" and contributed to the RCS overcooling.

Decreasing core coolant flow reduces the minimum DNBR for a given level of power generation. However, the maximum rate of RCS overcooling -- and the consequent potential for . ver generation -for these events occurs with four RCPs operating. The reactor coolant flow low trip prevents operation with the reactor trip breakers closed and less than one RCP operating in each loop for Modes 3 or 4. Further, with less than all RCPs in operation a reactor trip would be generated two decades lower in power -- due to the automatic removal of the CPC bypass (see the discussion in Section 4.4.1) -- than when all RCPs are in operation. In addition, any postulated occurrence which would result in tripping of operating RCPs after event initiation -- e.g. a loss of offsite power -- would cause a reactor trip on low flow and terminate any approach to criticality earlier than that calculated without the loss of flow. It has been determined via parametric studies that the net effect of these factors is that the minimum transient DNBR occurs for postulated events for which all four RCPs are in operation.

These cases included the effect of the most reactive CEA stuck in the fully withdrawn position. There were no single failures found which could adversely impact these events with respect to reducing the transient minimum DNBR. The calculation of the maximum posttrip reactivity included the effect of a failure of one emergency generator to start, run, or load (resulting in unavailability of two safety injection pumps).

The rate of cooldown for the excess feedwater flow events is less dependent up is the steam generator temperature (and resultant pressure) than it is for the steam line break events. Further, the maximum cooldown rate attainable with the spectrum of potential flows is such that all of the excess feedwater flow events were mitigated at low core powers by a high logarithmic power trip or CPC trip. Consequently, the most adverse results were calculated to occur for the highest flow rate and the highest initial RCS temperature.

The investigation of this spectrum of cases with the assumed conservative initial conditions found that the minimum transient DNBR for increased feedwater flow events postulated to be initiated in a shutdown mode is greater than 3. Further, the maximum reactivity following reactor trip was more negative than $-3\&\Delta\rho$. There was, therefore, no post-trip power generation or approach to DNB.

Since the minimum transient DNBR for the events to which the events of Section 15.1.2 of CESSAR-DC were referenced was 1.24, the consequences of events in shutdown modes are no more adverse than those of the events presented in CESSAR-DC.

4.1.3 INCREASED MAIN STEAM FLOW

As noted in Section 15.1.3 of CESSAR-DC, the steam flow due to an increased main steam flow event is the same as (or less than) that due to an inadvertent opening of a steam generator relief or safety valve event. Further, there are no other differences between these events which affect their consequences. Therefore, the conclusions of Section 4.1.4 of this repair apply also to an increased main steam flow event.

4.1.4 INADVERTENT OPENING OF A STEAM GENERATOR RELIEF OR SAFETY VALVE

Evaluation of the inadvertent opening of a steam generator relief or safety valve events postulated to be initiated in a shutdown mode shows that the results are less adverse than those of the events documented in Section 15.1.4 of CESSAR-DC.

This evaluation was completed as integral to the study of steam system piping failures postulated to be initiated in a shutdown mode. The transient caused by an inadvertent opening of a steam generator reliaf or safety valve is identical to that caused by a steam line bigger of area equal to that of a relief or safety valve. The study presented in Section 4.1.5 covered a full spectrum of break sizes and initial conditions, including all and less than all reactor coolant pumps running. As documented in Section 4.1.5, the minimum transient DNBR for a steam system piping failure in a shutdown mode is greater than 2. The minimum transient DNBR for inadvertent opening of a steam generator relief or safety value events postulated to be initiated in a shutdown mode is, therefore, also greater than 2. Since the minimum transient DNBR for the events of Section 15.1.4 of CESSAR-DC was 1.24, the consequences of the events presented in CESSAR-DC bound those of events in shuldown modes.

4.1.5 STEAM SYSTEM PIPING FAILURES INSIDE AND OUTSIDE CONTAINMENT

Evaluation of steam system piping failures inside and outside containment, postulated to be initiated in a shutdown mode, shows that the results are less adverse than those of the events documented in Section 15.1.5 CESSAR-DC.

The evaluation presented in this section focuses on fuel performance as measured by the departure from nucleate boiling ratio (DNBR) for verification of fuel integrity. Steam system piping failures are characterized by decreasing reactor coolant system (RCS) and steam generator pressures. Therefore the RCS prefire remains well below 110% of design pressure and the pressuretemperature limits for brittle fracture, ensuring that the integrity of the RCS is maintained, and the steam generator pressure remains below 110% of design pressure ensuring that the integrity of the secondary system is maintained. The two-hour inhalation dose at the EAB (Exclusion Area Boundary) is also examined to confirm that, if such an event were postulated to be initiated in a shutdown mode, the value meets the Standard Review Plan Acceptance Criteria of Reference 6.

For shutdown modes the reactor core is subcritical with power being generated only by decay heat. Substantial margins to DNA exist at the time of a postulated event initiation. Steam system piping failures would cause a decrease in the temperature of the reactor coolant and in the RCS and steam generator pressures. The decrease in the reactor coolant temperature would result in an increase in core reactivity due to the negative moderator temperature coefficient. If the cooldown were sufficiently large, a return to criticality followed by an increase in core power might occur. This could create a potential for degrading the margins to DNB. There are two possib lities to be considered for a return to criticality for these events:

- a. cooldown sufficient to reach criticality prior to reactor trip and
- b. extended cooldown after reactor trip of sufficient magnitude to reach criticality if there were insufficient safety injection boration to prevent a return to criticality.

The initial reactor core and steam generator temperatures have particularly significant effects for steam system piping failures. The total potential magnitude of the possible RCS cooldown rises with increasing initial steam generator temperature. In addition, the initial rate of cooldown would be greater if the initial steam generator tomperature were higher. Further, both the potential rate and the total potential magnitude of the reactivity increase caused by cooldown rise with increasing reactor core temperature. Steam system piping failures would, therefore, be expected to have the most adverse consequences if they were assumed to be initiated at the highest possible initial temperatures, if event mitigation were not affected by the initial conditions.

Event mitigation could, however, be delayed by decreasing the rate and extent of cooldown. The evaluations presented here have, therefore, explored the entire range of conditions from the maximum possible initial temperatures to those for which the changes in temperature induced by the cooldown would be insufficient to result in a return to criticality. Further, a full range of break sizes have been explored, up to and including a double-ended guillotine break of a main steam line upstream of a MSIV. Table 4.0-1 presents a full list of the ranges of initial conditions considered.

As noted above, one of the possibilities to be considered for a return to criticality for these events is a cooldown sufficient to cause criticality prior to reactor trip (assuming the core is initially at the maximum allowed reactivity for a shutdown mode). Exploration of a spectrum of break sizes shows that for the more rapid transients. caused by larger breaks and higher initial temperatures, a reactor trip would be initiated by a low steam generator pressure signal prior to reaching criticality, thus precluding any fission power generation. For slower transients (smailer breaks and lower initial temperatures) the low steam generator trip pressure would not be reached before the core was calculated to reach criticality and some fission power generated. This would result in the generation of a reactor trip signal at a low power level by one of two reactor trip functions: a CPC trip or a high logarithmic power level trip.

Figure 4.1-1 illustr tes these effects for a spectrum of break sizes with an initial RCS average temperature of 500 °F. In this figure reactor trip is due to a high logarithmic power level signal for smaller break sizes and a low steam generator pressure trip signal for larger break sizes. For events initiated under conditions which would result in a CPC trip (e.g. less than four reactor coolant pumps operating), the trip signal would be generated when the reactor power was two orders of magnitude lower, resulting in significantly lower powers than those shown in Figure 4.1-1. (See the discussion of CPC trip in Section 4.4.1.) The core powers reached for the shutdown mode case are, however, all substantially lower than that of the limiting Mode 1 case documented in CESSAR-DC. Further, the minimum transient DNBR for the shutdown mode cases was greater than 2. There is, therefore, no approach to loss of fuel integrity for steam system piping failures postulated to be initiated in a shutdown mode. The minimum DNBR of the events documented in Section 15.1.5 of CESSAR-DC bound those of events in shutdown modes.

The second potential possibility for a return to criticality for steam system piping failures postulated to be initiated in a shutdown mode is extended cooldown after reactor trip of sufficient magnitude to reach criticality (if there is insufficient safety injection boration). Consideration of the maximum possible integral reactivity change due to cooldown shows that the Technical Specification Shutdown Margin (6.5% Ap) is sufficient to prevent a return to criticality following reactor trip for postulated steam system piping failures for which the initial RCS average temperature is less than 500 °F -- even if no safety injection boration were to occur. A series of cases was run to show that, for initial RCS average temperatures between 500 and 565 °F (the maximum temperature for shutdown modes) and with other initial conditions within the limits of Table 4.0-1, a safety injection actuation signal would be generated on low pressurizer pressure and sufficient boration would occur to prevent a recurn to criticality. These cases included the effect of the most reactive CEA stuck in the fully withdrawn position and of a single failure of either a failure of one emergency generator to start, run, or load (resulting in unavailability of two safety injection pumps) or of one MSIV failing to close. The maximum post-trip reactivity for these cases was $-1.2 \Delta \rho$, which is significantly more negative than the limiting case documented in Section 15.1.5 of CESSAR-DC. NO fission power was generated and there was no approach to DNB.

The mass of steam released to the environment for steam system piping failures postulated to be initiated in a shutdown mode is bounded by that for the Mode 2 analysis presented in CESSAR-DC. Further, no loss of fuel integrity is calculated to occur for the shutdown mode events. The two-hour inhalation dose at the EAB is, therefore, bounded by that presented in Section 15.1.5 of CESSAR-DC.



4.1-1

4.2 DECREASE IN HEAT REMOVAL BY SECONDARY SYSTEM

4.2.0 INTRODUCTION

The purpose of this section is to present evaluations which confirm that all decrease in heat removal by the secondary system events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 80+ Standard Design.

Section 15.2 of CESSAR-DC documents results which show that all decrease in heat removal by the secondary system events have acceptable consequences if they are explicitly postulated to occur in Mode 1 or, in general, in Mode 2 critical for the System 80+ Standard Design. However, if there is a question as to whether an event has already been considered in Mode 2 critical, this Mode is also included in the evaluation.

The focus of the evaluations presented in this section is on ensuring that the peak primary and secondary pressures are less than 110% of their design pressures, that the pressure-temperature limits for brittle fracture of the reactor coolant system (RCS) are not violated, and that fuel integrity is maintained for the events considered. Fuel performance, as measured by the departure from nucleate boiling ratio (DNBR), is used for verification of fuel integrity.

4.2.1 LOSS OF EXTERNAL LOAD

Since the turbine is not on line, a loss of external load is not possible in a shutdown Mode.

4.2.2 TURBINE TRIP

Since the turbine is not on line, a turbine trip is not possible in a shutdown Mode.

4.2.3 LOSS OF CONDENSER VACUUM (LOCV)

A LOCV event postulated to occur in a shutdown mode would result in less severe consequences than the event documented in Section 15.2.3 of CESSAR-DC, since only decay heat is being generated by the reactor core.

The LOCV event is only applicable when the condenser is being used as a heat sink. This could occur in Mode 2 subcritical, Mode 3, or Mode 4 when the shutdown cooling system (SCS) is not being used for decay heat removal. The turbine-generator would be off-line and steam would be entering the condenser via the steam bypass control valves. Loss of decay heat removal in Modes 4 through 6, when the SCS is being utilized, is addressed in Section 2.4 The dominant factor which determines the peak primary and secondary pressure following a LOCV event is the magnitude of the energy mismatch between the primary and secondary systems. This mismatch is very much less for events postulated to occur in a shutdown mode than for the event of Section 15.2.3 of CESSAR-DC, even when an instantaneous termination of steam flow and feedwater flow is assumed. The maximum peak primary and secondary pressure will, therefore, be less for a LOCV postulated to occur during a shutdown mode than for this event at full power. Consequently, there is no approach to 110% of the design pressure of the steam generators or RCS. The LOCV event would not challenge the pressure-temperature limits for brittle fracture of the RCS, since the condenser is not on line (shutdown cooling conditions have been established) when LTOP is active.

Likewise, fuel integrity is maintained for LOCV events initiated in a shutdown mode, due to the relatively low reactor core heat fluxes associated with decay heat. Decreased coolant flow is the only parameter which can significantly reduce the minimum DNBR, since reactor power does not increase and RCS pressure increases for this event. Loss of offsite power (LOP) is the only single failure which can impact coolant flow. Because the plant is in a shutdown mode, there would be no turbine trip after the LOCV event, consequently, there would be no perturbation to the electrical grid which could cause a LOP. However, analyses have shown that even when the LOCV event is initiated with no reactor coolant pumps (RCPs) operating at the highest decay heat flux value (just after reactor shutdown), the minimum value of DNBR for this event is greater than 9. For colder conditions the minimum DNBR is even higher. For example, when the LOCV event is initiated just prior to establishing shutdown cooling conditions in Mode 4 and natural circulation conditions exist (i.e. no KCPs operating), the minimum value of DNBR is greater than 100.

4.2.4 MAIN STEAM ISOLATION VALVE CLOSURE

The comparison between main steam isolation valve (MSIV) closure and LOCV events made in Section 15.2.4 of CLSSAR-DC applies for shutdown Modes, also. The evaluation of the LOCV event presented in Section 4.2.3 assumes a faster reduction in steam flow rate than would result from MSIV closure. The consequences of the MSIV closure event are, therefore, no more adverse in shutdown Modes than those for the LOCV presented in Section 4.2.3.

4.2.5 STEAM FRESSURE REGULATOR FAILURE

This event does not apply to the SYSTEM 80+ Standard Design and is, therefore, not evaluated here.

4.2.6 LOSS OF NUN-EMERGENCY AC POWER TO THE STATION AUXILIARIES

The results of the loss of non-emergency AC power to the station auxiliarizs (LOAC) event are the same as those for the loss of reactor coolant flow event presented in Section 4.3.1.

4.2.7 LOSS OF NORMAL FEEDWATER FLOW

A postulated loss of normal feedwater flow (startup feedwater flow) during a shutdown Mode would be less adverse than the LOCV event. The analysis assumptions for the LOCV event result in termination of steam flow, as well as termination of startup feedwater flow, causing a more severe decrease in heat removal by the secondary system. The consequences of the loss of normal feedwater flow are, therefore, bounded by the consequences of the LOCV event for shutdown Modes.

4.2.8 FEEDWATER SY.TEM PIPE BREAKS

Depending on the 'reak size and location and the response of the feedwater system, a 'fect of a postulated feedwater system pipe break can vary from stup to a cooldown of the RCS. Based on the same arguments give. In Section 15.2.8.1 of CESSAR-DC, the heatup event is considered in this section. The cooldown potential would be worse for a steam line break, which is discussed in Section 4.1.5. A heatup event is mitigated by the pressurizer safety valves or the shutdown cooling (SCS) relief valves when RCS temperatures are above or below, respectively, the LTOP enable/disable temperatures, the main steam safety valves, and the emergency feedwater (EFW) system.

A feedwater system pipe break postulated to occur in a shutdown Mode would result in less severe consequences than the event documented in Section 15.2.8 of CESSAR-DC due to the lower initial reactor power level. DNB is not of concern due to the low initial core power levels, in addition to the pressurization following event initiation. Further, the EFW system is capable of removing decay heat event with only one EFW pump in operation. The dominant factor which determines peak primary and secondary pressures is the magnitude of the energy mismatch between the primary and secondary systems. This mismatch is very much less for events postulated to occur in shutdown Modes than for the event of Section 15.2.8 of CESSAR-DC. There is, therefore, no approach to 110% of steam generator design pressure, since the maximum pressure will be much less than that for the full power event. For the same reason there is no approach to the criterion of 110% of RCS design pressure when RCS temperatures are above the LTOP enable disable temperatures. For RCS temperatures below the LTOP enable/d sable temperatures the design of the SCS relief valves ensures that the pressure"

77a.wp(9212)bh

temperature limits for brittle fracture of the RCS are not violated.

4.3 DECREASE IN REACTOR COULANT FLOW RATE

4.3.0 INTRODUCTION

The purpose of this section is to present evaluations which confirm that all decrease in reactor coolant flow rate events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 804 Standard Design.

Section 15.3 of CESSAP-DC documents results which show that all decrease in reactor coolant flow rate events have acceptable consequences if they are explicitly postulated to occur in Mode 1 or Mode 2 critical for the System 80+ Standard Design. This is demonstrated for total loss of reactor coolant flow (Section 15.3.1) by explicit analyses. The choice of initial conditions for the other analyses of Section 15.3 to minimize the transient DNBR for any operating condition ensures that the results presented bound those for all of Mode 1 and Mode 2 critical.

Decrease in reactor coolant flow rate in Modes 4 through 6 when the shutdown cooling system (SCS) is being used for decay heat removal is addressed in Section 2.4 as integral to the evaluation of loss of decay heat removal capability. Therefore this section addresses evants postulated to be initiated in Mode 2 subcritical, Mode 3 or Modes 4 and 5 when the SDS is not being used.

4.3.1 TOTAL LOSS OF REACTOR COOLANT FLOW

Evaluation of the factors affecting the consequences of the total loss of reactor coolant flow event shows that if this event is postulated to be initiated in shutdown Hodes, the results are less adverse than those of the CESSAR-DC Chapter 15.3 full power event.

Loss of offsite power is the postulated initiating event for the total loss of reactor coolant flow event in Modes 1 or 2. All systems available to mitigate the Node 1 transient are available in Node 2 subcritical. The initial conditions for Mode 2 subcritical include four pumps operating, temperature and pressure identical to Mode 1, a low power level, and total energy stored in the reactor core much less than at full power. Therefore, a very large margin to DNB exists at the initiation of the event and the heat to be temoved during the event is much less than for an event initiated at full power. The minimum ENBE for this event is substantially higher for Mode 2 than that for the full power case.

In addition, the initiating event could be postulated to be loss of power to any operating RCPs in Mode 1 or in Modes 4 or 5 when the SCS is not being used. Natural circulation is however, sufficient for the removal of decay heat in these Modes. Thus no approach to DNB would occur. The consequences of the Chapter 15.2 full power event are therefore more adverse than an event in these Modes.

The full power four pump loss of flow transient produces RCS and steam generator pressures which are less than 110% of their design values. Transients postulated to be initiated in Modes 2, 3 or 4 when RCS temperatures are above the LTOP enable/disable temperatures, and for which the rate of heat production is orders of magnitude below full power values, would therefore yield even greater margins to the design pressure values. For transients postulated to be initiated in Modes 4 or 5 with RCS temperatures below the LTOP enable/disable temperatures, the design of the SDC relief valves ensures that the pressure-temperature limits for brittle fracture are not violated.

4.3.2 FLOW CONTROLLER MALFUNCTION CAUSING FLOW COASTDOWN

This event is categorized as a Boiling Water Reactor event in SRP 15.3.2 and is, therefore, not evaluated here.

4.3.3 SINGLE REACTOR COOLANT PUMP ROTOR SEIZUPE WITH LOSS OF OFFSITE POWER

The major parameter of concern for the single reactor coolant pump rotor seizure with loss of offsite power event documented in Section 15.3.3 of CESSAR-DC is the minimum hot channel DNBR. This is minimized by higher power conditions. Lower power Modes would, therefore, not produce any conditions which are more adverse than those presented in CESSAR-DC.

The second parameter of concern is the peak RCS pressure attained. The full power single reactor coolant pump rotor seizure with loss of offsite power event produces RCS pressures which are less than 110% of their design values. Transients postulated to be initiated in Modes 2, 3, or 4 when RCS temperatures are above the LTOP enable/disable temperatures, and for which the rate of heat production is orders of magnitude below full power values, would therefore yield even greater margins to the design pressure values. For transients postulated to be initiated in Modes 4 or 5 with RCS temperatures below the LTOP enable/disable temperatures the design of the SCS relief valves ensures that the pressure-temperature limits for brittle fracture of the RCS are not violated.

In addition, at these low power levels a concurrent turbine trip which results in a loss of offsite power is not an issue since the turbine is not in operation below 5% power.

4.3.4 REACTOR COOLANT PUMP SHAFT BREAK WITH LOSS OF OFFSITE POWER

Since a postulated reactor coolant pump shaft break (SB) transient results in a less rapid flow coastdown than a rotor seizure (RS) event, the results of the SB event are bounded by those of the evaluation of Section 4.3.3.

4.3-2

4.4 REACTIVITY AF POWER DISTRIBUTION ANOMALIES

4.4.0 INTRODUCTION

The purpose of this section is to present evaluations which confirm that all reactivity and power distribution anomaly events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 80+ Standard Design.

4.4.1. UNCONTROLLED CONTROL ELEMENT ASSEMBLY WITHDRAWAL FROM SUBCRITICAL OR LOW POWER CONDITIONS

Evaluation of the consequences of an uncontrolled CEA withdrawal event postulated to be initiated from a shutdown mode shows that the results are less adverse than those of the event documented in Section 15.4.1 of CESSAR-DC.

The event documented in CESSAR-DC was initiated from low power, critical conditions. The Technical Specifications include requirements which ensure that the most adverse subcritical CEA withdrawal event scenario is the inadvertent withdrawal of a regulating CEA bank. For the evaluation presented here, therefore, events which postulate initia subcritical conditions and an uncontrolled withdrawal of a regulating CEA bank were analyzed to demonstrate that the SAFDL on DNBR is not violated and that no fuel melting is predicted to occur.

The essential reason that withdrawal of a regulating CEA bal. from subcritical is less adverse than the same event at low power, core critical conditions is that one of two reactor trip functions terminates the subcritical event at low power levels: a CPC trip or a high logarithmic power level trip. The setpoints for these trip functions are such that the maximum power levels reached for this event are much lower than those for the low power event.

The CPCs may be bypassed at power levels below 0.0001%. This bypass is automatically removed upon exceeding this power level. Although a reactor trip would not occur when the CPCs were bypassed, the CPCs continue to perform their calculations and can be in a tripped condition. The CPCs would be in a tripped condition if less than all four reactor coolant pumps were running or if the calculated radial peaking were out of range due to the presence of an undesired CEA bank in the core. As the core fission power increases during the CEA withdrawal event the 0.0001% power bypass is removed and, if the CPCs are in a tripped condition, a reactor trip is generated.

The high logarithmic power level trip is required to be operable in all shutdown modes when the reactor trip breakers are closed and the CEA drive mechanism is capable of CEA withdrawal. Although the high logarithmic power level trip can be bypassed during Modes 1 and 2 critical, it cannot be bypassed during shutdown conditions.

For those CEA withdrawal events which do not generate a CPC trip upon removal of the 0.0001% power bypass, the high logarithmic power level trip will produce a reactor trip for which the setpoint is 0.01% power. (The corresponding setpoint used in the analyses is 0.05%, as documented in Table 15.0-2 of CESSAR-DC.) If, however, the transient is the withdrawal of one of the early regulating banks, or if less than all reactor coolant pumps are in operation, a reactor trip would be generated two decades lower in power upon the removal of the CPC bypass at 0.0001% power.

CEA withdrawal events initiated from subcritical conditions were analyzed with assumptions which bound all shutdown conditions and determine the consequences when reactor trip is due to each of the above functions. The cases analyzed included:

- A. Regulating bank withdrawal, worst combination of burn-up dependent parameters, four pumps running, reactor trip on high logarithmic power level.
- B. Regulating bank withdrawal, worst combination of burn-up dependent parameters, less than four pumps running, reactor trip upon removal of the CPC 0.0001% power bypass due to CPC pump counting.

The case with full forced reactor coolant flow experienced a peak heat flux of approximately 5% of full power heat flux. This is well below the value of 34.8% calculated for the event documented in Section 15.4.1 of CESSAR-DC. The DNBR for this case remained we'l above the DNBR SAFDL.

Due to the earlier reactor trip, the case with fewer than all four reactor coolant pumps running experienced a peak heat flux increase to less than 2% of full power heat flux. This low value of heat flux would not cause DNB.

The deposited energy calculation assumed that all of the energy produced in the short power excursion is deposited in the fuel with no heat transfer from the fuel. The evaluation shows that the hot spot in the core does not experience melt for any withdrawal of a regulating CEA bank from subcritical.

In summary, the System 80+ Technical Specifications ensure that the reactor will not become critical due to the inadvertent withdrawal of a shutdown CEA bank. The CPC and high logarithmic power level trip functions ensure that the consequences of an inadvertent withdrawal of a regulating CEA bank from subcritical conditions are substantially less adverse than those documented in Section 15.4.1 of CESSAR-DC. This event will not result in the violation of the DNBR SAFDL or in fuel melt.

4.4.2 UNCONTROLLED CONTROL ELEMENT ASSEMBLY WITHDRAWAL AT POWER

This event is not an issue for shutdown Modes since its intent is to examine high power oper: ion only.

4.4.3 SINGLE CONTROL ELEMENT ASSEMBLY DROP

A postulated single control element assembly drop at power is analyzed for approach to the DNBR limit in CESSAR-DC, Section 15.4.3. For Mode 2 subcritical through the other subcritical Modes, a dropped rcd only adds more negative reactivity to an already subcritical core and is, therefore, much less adverse than the full power event documented in CESSAR-DC.

4.4.4 STARTUP OF AN INACTIVE REACTOR COCLANT PUMP

Chapter 15.4.4 of CESSAR-DC presents analysis of startup of an inactive reactor coolint pump events which show that events postilated to be initiated in Modes 3 through 6 have acceptable consequences for the System 80+ Standard Design. (Operation with less than 4 RCPs is not permitted in Modes 1 or 2.)

4.4.5 FLOW CONTROLLER MALFUNCTION CAUSING AN INCREASE IN BWR CORE FLOW RATE

This event is categorized as a Boiling Water Reactor event in SRP 15.4.5 and is, therefore, not evaluated here.

4.4.6 INADVERTENT DEBORATION

Analyses of inadvertent deboration events in shutdown Modes have been presented in Chapter 15 of CESSAR-DC. An additional evaluation which considers rapid deboration is presented in Section 2.6 of this report.

4.4.7 INADVERTENT LOADING OF A FUEL ASSEMBLY INTO THE IMPROPER POSITION

This event has been evaluated in Section 15.4.7 of CESSAR-DC and is not mode dependent.

4.4.8 CONTROL ELEMENT ASSEMBLY (CEA) EJECTION

The CEA ejection analysis documented in Section 15.4.8 of CESSAR-DC is for an ejection from full power initial conditions. This case was found to be more limiting than an ejection from Mode 2 critical initial conditions. The evaluation documented in Section 15.4.8

credited the technical specification limits on allowable CEA insertion given in Technical Specifications 3.1.6 and 3.1.7, which apply in Mode 1 and in Mode 2 critical.

Technical Specifications 3.1.5 and 3.1.7 do not apply directly for Mode 2 subcritical or for Modes 3 through 5. However, the requirements of Technical Specification 3.1.2, which do apply in these shutdown modes and which use the criteria of Technical Specifications 3.1.6 and 3.1.7 ensure that the full power CEA ejection case remains limiting.

For Mode 2 subcritical and for Modes 3 through 5, Technical Specification 3.1.2.a requires that the calculated critical position be within the limits of Technical Specifications 3.1.6 and 3.1.7 when the reactor trip breakers are closed. This requirement ensures that a CEA ejection postulated to be initiated at these conditions would result in less net positive reactivity insertion than for a case initiated from a critical position. The case described in Section 15.4.8 of CESSAR-DC, therefore, remains limiting for conditions under which the reactor trip breakers are closed.

When the reactor trip breakers are open in Nodes 3 through 5, Technical Specification 3.1.2.b requires that the value of k_{eff} must remain less than 1.0 when the highest worth CEA is excluded from the calculation. This requirement ensures that the reactor would not reach criticality for a CEA ejection event postulated to be initiated under these conditions.

The full power event of Section 15.4.8 of CESSAR-DC is, therefore, the limiting CEA ejection event.

4.5 INCREASE IN RCS INVENTORY

4.5.0 INTRODUCTION

The purpose of this section is to present evaluations which demonstrate that all increase in RCS inventory events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 80+ Standard Design.

Section 15.5 of CESSAR-DC documents results which show that all increase in RCS inventory events have acceptable consequences if they are explicitly postulated to occur in Mode 1 or, in general, in Mode 2 critical for the System 80+ Standard Design. If there is a question as to whether an event has already been considered in Mode 2 critical, however, this Mode is also included in the evaluation.

The focus of the evaluations presented in this section is on ensuring that the peak primary pressure is less than 110° of design pressure and that the pressure-temperature limits for brittle fracture of the RCS are not violated. Fuel performance, as measured by the doparture from nucleate boiling ratio (DNBR), is used for verificat on of fuel integrity. Peak secondary pressure is also evaluated, as necessary, to ensure it remains less than 110% of its design pressure.

4.5.1 INADVERTENT OPERATION OF THE ECCS

The evaluation presented in Section 15.5.1 of CESSAR-DC establishes that a postulated inadvertent operation of the ECCS would have acceptable consequences for any Mode for the System 80+ Standard Design.

4.5.2 CVCS MALFUNCTION-PRESSURIZER LEVEL CONTROL SYSTEM MALFUNCTION WITH LOSS OF OFFSITE POWER

Peak RCS pressure due to a postulated malfunction/actuation of a charging pump in Modes 2 through 4 (before LTOP is active), is well within 110% of design pressure. Further, the pressure-temperature limits for brittle fracture of the RCS are not challenged for a postulated malfunction/actuation of a charging pump in Modes 4 through 6 (when LTOP is active or when the reactor vessel head is off).

In Modes 2 subcritical through 4, before shutdown cooling and LTOP are placed in service, the peak RCS pressure is limited by the pressurizer safety valves. Since only decay heat exists under these conditions, the peak pressures during the CVCS malfunction event would be substantially less severe than the Mode 1 case described in CESSAR-DC. It should be noted that the centrifugal
charging pumps are protected from runout at low RCS pressure: by design.

In Modes 4, 5, and 6 when LTCP is active (or in Mode 6 with the reactor vessel head off and LTCP not active), the CVCS malfunction event would be less limiting than the inadvertent SIS actuation described in Section 4.5.1 above, due to the much lower flow from one charging pump versus the four SI pumps.

4.6 DECREASE IN REACTOR CCOLANT SYSTEM INVENTORY

4.6.0 INTRODUCTION

The purpose of this section is to present evaluations which demonstrate that all decrease in RCS inventory events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 80+ Standard Design.

Section 15.6 of CESSAR-DC documents results which show that all decrease in RCS inventory events have acceptable consequences if they are explicitly postulated to occur in Mode 1 or, in general, in Mode 2 critical for the System 80+ Standard Design. If there is a question as to whether an event has already been considered in Mode 2 critical, however, this Mode is also included in the evaluation.

Fuel performance, as measured by the departure from nucleate boiling ratio (DNBR), is used for verification of fuel integrity.

4.6.1 INADVERTENT OPENING OF A PRESSURIZER SAFETY/RELIEF VALVE

The LOCA evaluations presented in Section 4.6.5 of this report show that the inadvertent opening of a pressure safety/relief valve is at a non-limiting location for LOCAs postulated to occur in shutdown Modes. The inadvertent opening of a pressurizer safety valve as described in the Standard Review Plan 15.6.1 is, therefore, a non-limiting event in the Safety Injection System evaluations.

4.6.2 DOUBLE-ENDED BREAK OF A LETDOWN LINE OUTSIDE CONTAINMENT

The MODE 1 analysis presented in Section 15.6.2 of CESSAR-DC credited a number of non-safety grade alarms for the detection of a letdown line break. Credit for these alarms was rejected by the NRC in RAI 440.104, and ABB-CE was asked to reanalyze the event using only safety grade equipment.

While ABB-CE believes that a double ended letdown line break would be detected as described in CESSAR-DC Section 15.6. an alternate resolution to mitigating this event is being considered which will utilize only safety grade equipment. Thus a revised letdown line break analysis for MODE . along with the evaluation for shutdown modes will be provided in the fall of 1992.

4.6.3 STEAM GENERATOR TUBE RUPTURE

Evaluation of the concequences of a Steam Generator Tube Rupture (SGTR) accident postulated to be initiated in a shutdown mode shows

that the results are less adverse than those of the full power event documented in Section 15.6.3 of CESSAR-DC. Consequently, the radiological doses are well within the 10 CFR 100 guidelines and fuel integrity is not challenged.

The worst case initial conditions for a SGTR accident initiated in a shutdown mode would be in Mode 3, at the highest possible RCS temperature and pressure and lowest possible RCS flo/ (less than four reactor coolant pumps operating). This would maximize the initial leak rate and flashing fraction of the leaked fluid. Also, the worst case single failure would be the assumption that the atmospheric dump valve (ADV) on the affected steam generator (SG) sticks open after the operator attempts to close this valve, as in Section 15.6.3 of CESSAR-DC.

Since the turbine-generator is off-line in Mode 3, there would be no perturbation to the electrical grid which could cause a loss of offsite power (LOP). A consequence of a LOP would be a further reduction in coolant flow, which would increase the core exit temperature (hence increasing the flashing fraction of the leaked fluid). However, even if the very unlikely event of a LOP were to be assumed, fuel integrity would not be challenged, since only decay heat is being generated and natural circulation is sufficient for decay heat removal.

For a SGTR accident postulated to be initiated in a shutdown mode, there are a number of plant parameters and alarms available in the control room that will enable the operators to diagnose the event. These include:

- Indications of event:
 - Decreasing pressurizer pressure
 - Decreasing pressurizer liquid level
- Indications of event and of which SG is affected:
 - Increasing SG liquid level
 - Increasing SG liquid blowdown radiation level
 - Increasing main steam line radiation level
- Additional indications of event if the steam bypass system is being used for cooldown:
 - Increasing steam jet air ejector radiation level
 - Increasing stack radiation level

77a.wp(9212)bh

The sequence of events for a SGTR accident postulated to occur in Mode 3 would be:

- A. Prior to the SGTR accident, the reactor was shut down according to normal procedures, consequently, the turbine-generator is off line. A postulated iodine spike caused by a reactor trip or primary system depressurization (event generated iodine spike, GIS) would be unlikely to occur under these conditions. However, it is assumed that either a GIS has occurred or the RCS radioactivity concentration was at its technical specification limit (pre-accident iodine spike, PIS), since this resulted in the most adverse RCS radioactivity concentrations when the SGTR accident was assumed to be initiated in Mode 1 (as documented in 15.6.3 of CESSAR-DC). Also it was assumed that the secondary activity is at the Technical Specification limit of 0.1 microcurie per gram I-131 dose equivalent, since this assumption also maximizes the radiological doses. Although a LOP is very unlikely under these circumstances, a LOP is conservatively assumed to occur concurrent with event initiation. It is assumed that the ADVs are being used to cool down the plant, as opposed to the usual method of utilizing the steam bypass control valves. This results in the accumulation of radiological doses from the very beginning of the event. The maximum plant cooldown rate is achievable with the partial opening of one ADV on each SG. One ADV on each SG is, therefore, assumed to be open at the time of event initiation.
- B. The SG liquid blowdown radiation monitor would immediately provide the operator with the information which would enable him to determine which SG is damaged. The SG liquid blowdown radiation alarm would be activated within seconds of event initiation, since it would be set to alarm below the 0.1 microcurie/gram technical specification limit. Based on the adverse RCS and SG radioactive concentrations at event initiation, the special purpose main steam line area radiation monitors may also alarm.

For the double ended rupture of a SG tube the operator would also have immediate indication of event occurrence in the form of rapidly decreasing pressurizer pressure and level and increasing SG level. These parameter changes would then result in a high SG liquid level alarm, a low pressurizer pressure alarm, and a low pressurizer liquid level alarm after the high SG blowdown radiation alarm.

C. Acting upon these indications, within thirty minutes after event initiation i. was assumed that the operator has diagnosed the event and knows which SG has the damaged tube. At this point the operator isolates feedwater to the affected SG, closes the main steam isolation valves to both steam generators, and attempts to close the ADV on the affected SG.

- D. The operator determines that the ADV did not close (continued control room indication of steam flow) and closes the block valve upstream of the stuck open ADV within 30 minutes of the initial attempt to isolate the affected SG (i.e. within one hour of event initiation).
- E. The operator commences cooldown to shutdown cooling entry conditions, by using the unaffected SG ADV. The operator also uses the pressurizer gas vent and SI flow to decrease RCS pressure and control subcooling. The operator steams the affected SG as necessary to prevent overfilling.

A comparison of the above sequence of events with the worst case SGTR accident in 15.6.3 of CESSAR-DC (see Table 15.6.3-7), shows that a SGTR accident postulated to occur in Mode 3 would result in less integrated steam released to the envilonment. This is due to the fact that there would be no release by the main steam safety valves, since there is no initial transient imbalance between reactor power and turbine power. The integrated steam release through the stuck open ADV and the flashing fraction of the leaked primary fluid, while the affected ADV is discharging to the atmosphere, are essentially the same for the postulated Mode 3 and full power events. Thus the concentration of radioactive nuclides in steam released to atmosphere is no greater for the Mode 3 event than for the event documented in CESSAR-DC. The total radiological release for this event would, therefore, be less than that of the full power events.

If the SGTR were postulated to occur at lower RCS temperature and pressure than that assumed above, the radiological doses would be even less than the above event. This would be due to less integrated steam released to the environment and the lower concentration of radioactive nuclides in this released steam, because of the lower fraction of primary fluid flashed to steam in the both the affected and unaffected SGs.

The indications of event occurence and the event consequences are somewhat different if a SGTR is postulated to be intiated when the shutdown cooling system is in operation in Mode 4 or Mode 5 (when the RCS loops are filled). (A SGTR accident would not have any consequences in Mode 6 since primary and secondary prossures are both at atmospheric pressure.) The plant parameters and alarms available in the control room which would enable the operators to diagnose an event under these circumstances include:

- Indication of event for all leak sizes and of which SG is affected:
 - Increasing SG liquid level

- indications of event for larger leak sizes (the cooldown rate may mask the leak rate for these indications when the SGTR leak rate is small):
 - Decreasing pressurizer pressure
 - Decreasing pressurizer liquid level
 - Imbalance between charging and letdown flow rates

The leakage rate could range from 0.0 to approximately 315 gallons per minute (assuming a double ended break of one steam generator tube) under these circumstances. At the maximum leak rate the operator would have more than three-quarters of an hour in which to diagnose the event and take appropriate mitigating actions. (This conservatively assumes that the initial SG level is at 100% wide range, which is much greater than the normal water level when the SG is isolated or in wet layup). The mitigating actions could include utilizing the SG blowdown system to prevent overfilling the SG and minimizing the pressure difference between the primary and secondary systems. In the unlikely case the SG was overfilled, the main steam line pipe supports are designed to accomodate the water loads. At lower leak rates the operator would have even more time to take corrective action. The SCS relief valves ensure the pressure-temperature limits for brittle fracture of the RCS would not be violated, although they would not be actuated for this event. There would be no approach to the criterion of 110% of steam generator design pressure. Also, fuel integrity would not be challenged, since no approach to DNB would occur. The resulting radiological doses would be much less than for the SGTR accident initiated in Modes 3 and 4 and, consequently, would be well within 10 CFR 100 limits.

4.6.4 RADIOLOGICAL CONSEQUENCES OF MAIN STEAM LINE FAILURE OUTSIDE CONTAINMENT (BWR)

The radiological consequences of main steam line failure outside containment (BWR) do not apply to the System 80+ Standard Design and are, therefore, not evaluated here.

4.6.5 LOSS-OF-COOLANT ACCIDENT

Consequences of a LOCA initiated from Modes 2 through 4 are bounded by the consequences reported for a LOCA from Mode 1 since the containment spray and annulus ventilation systems which are available in Mode 1 are also available in Modes 2 through 4.

For a LOCA initiated from Mode 5, the evaluation in Section 5 concludes that the top of the core will not uncover. Fuel clad damage is prevented when the core remains covered so the source term for radiological dose calculations is based on only the radioactivity in the coolant. Prior to entry into Mode 5 from Mode 4, the radioactive concentration in the coolant is verified to be less than the limits in Technical Specification 3.4.15. For conservatism, the radioactive source for the Mode 5 LOCA is taken as the total activity from the limit concentration in a volume of water equal to the normal full RCS and all of the radioactivity is postulated to be released to the atmosphere through an open hatch within two hours. The calculated cffsite doses are smaller than the doses calculated for the Mode 1 LOCA.

4.7 <u>KADIOA'TIVE MATERIAL RELEASE FROM A SUBSYSTEM OR</u> COMPOL ST

4.7.0 INTRODUCTION

The purpose of this section is to present evaluations which confirm that all radioactive material release from a subsystem or component events postulated to be initiated in a shutdown Mode have acceptable consequences for the System 80+ Standard Design.

4.7.1 RADICACTIVE GAS WASTE SYSTEM FAILURE

This section of the Standard Review Plan has been deleted (Reference 26 of Section 15.0)

4.7.2 RADIOACTIVE LIQUID WASTE SYSTEM LEAK OR FAILURE

This section of the Standard Review Plan has been deleted (Reference 26 of Section 15 0)

4.7.3 POSTULATED RADIOACTIVE RELEASES DUE TO LIQUID CONTAINING TANK FAILURES

This event has been evaluated in Section 15.7.3 of CESSAR-DC and is not Mode dependent.

4.7.4 FUEL HANDLING ACCIDENT

This event has been evaluated in Section 15.7.4 of CESSAR-DC and is not Mode dependent.

4.7 5 SPENT FUEL CASK DROP ACCIDENTS

This event has been evaluated in Section 15.7.4 of CESSAR-DC and is not Mode dependent.

THIS PAGE INTENTIONALLY BLANK

5.0 APPLICABILITY OF CHAPTER 6 LOCA ANALYSES TO LOWER MODES OF OPERATION

5.1 ISSUE

A majority of the analysis of the Loss-Of-Coolant-Accident and all criteria associated with the accident have focused on scenarios starting from 102% of rated thermal power as prescribed in 10CFR50, Appendix K. In this section, scenarios initiated from all other modes of operation will be addressed to demonstrate that the analyses performed in CESSAR-DC are the bounding cases for all modes of operation. This section will identify the scenarios anticipated for non-Mode 1 operation, and where required the acceptance criteria for the different scenarios.

5.2 ACCEPTANCE CRITERIA

The acceptance criteria for lower operating mode LOCAs is established to be the same as those for higher operating mode LOCAs (Reference 19):

- 1) Peak clad temperature < 2200°F.
- 2) < 17% peak local clad oxidation.
- < 1% core wide oxidation.
- 4) Maintain coolable geometry.
- 5) Maintain long-term cooling.

5.3 DISCUSSION

The types of LOCAs and corresponding break sizes in lower modes are as follows: the large break loss-of-coolant accident (i.e., double-ended guillotine pipe breaks, etc.) analyzed in Chapter 6 of CESSAR-DC is not considered appropriate for lower modes of operation (Reference 2). As such only small break LOCAs are considered. The following lower mode scenarios are considered: most adverse misoperation of valves, the likelihood of cross train maintenance errors, and the possibility of steam generator nozzle dam blowout. The limiting size and location for a "small" break LOCA for Modes 3 and 4 is a direct vessel injection (DVI) line discharge leg (0.4 square feet). A DVI line break at this location would minimize injection flow into the RCS. Modec 5 and 6 consider a loss of decay heat removal (DHR) resulting from a break in the bottom of the hot leg or a lower head instrument line (.003 square feet).

A discussion of the equipment which would be available to mitigate the consequences of a LOCA in each mode of operation, the effectiveness of that equipment under the conditions for that mode, and the applicability of analyses to that mode are as follows: Mode 1: In Mode 1, RCS temperature is >350°F and power is >5%. The CESSAR-DC analysis for LOCAs in this mode concludes that all criteria are met.

Mode 2: In Mode 2, RCS temperature is >350°F and power is <5%. The equipment available in Mode 1 is also available in Mode 2. Mode 2 LOCA results are bounded by Mode 1 because the decay heat and stored energy, which are of concern to meeting LOCA criteria, are proportional to the lower power level in Mode 2.

Mode 3: In Mode 3, RCS temperature is >350°F and k_{eff} is <.99. For the case where RCS pressure is >900 psia, the equipment available in Mode 1 (SIT, 4 SI pumps) is available. Because the stored energy and decay heat is lower, this parameter space in Mode 3 is bounded by Mode 1 for LOCA. For the case where pressure is <900 psia, 4 SI pumps are available. Although at a higher temperature than Mode 4, the parameter space in Mode 3 is bounded by the Mode 4 analysis because Mode 4 will not credit automatic SI actuation. Furthermore, despite the fact that SITs would not be available when the pressure is <900 psia, this Mode 3 space is bounded by the Mode 1 analysis because there is less RCS pressure to drive the LOCA leak flow.

Mode 4: RCS temperature is <350°F. Four subspaces are considered:

1) Conditions:

pressure >900 psia and temperature >317°F. SITs and 2 SI pumps on automatic.

Conclusions:

Space 1) is bounded by a Mode 4, space 4, analysis included in this report (which credits neither SITs nor SI pumps on automatic) because the SITs are available to accommodate the higher pressure and 2 SI pumps are available to accommodate higher temperatures.

2) Conditions:

Pressure < 900 psia and temperature >317°F: 2 SI pumps on automatic.

Conclusions:

Space 2) is bounded by the Mode 4, space 4, analysis because 2 SI pumps are available to accommodate the higher temperature.

3) Cor stions:

Pressure > 900 psia and temperature < 317°F.

Conclusions:

Space 3) is bounded by the Mode 4, space 4, analysis because SITs are available to accommodat the higher source.

c. tonst

Lassure < 900 psia and temperature < 317°F.

a sions:

. analysis (space 4 described in Section 5.3.1) which credited neither automatic SITs nor SI pumps on automatic was performed for this report.

Mode 5: RUS temperature <210°F. An analysis was performed which assumed the maximum Technical Specifications cooldown rate of 100°F/hr. At the time of entry into Mode 5, a loss of shutdown cooling was assumed to result from a LUCA in the bottom of the hot leg. Although Technical Specifications require an RCS level at mid-loop (due to the SCS being required to operate), the LOCA was assumed to star it' an RCS level at the bottom of the hot leg. Approximately the nutes was required for the RCS level to drop below the top of Loc core. The manual start of one SI pump within ten minutes after the LOCA would be sufficient to cool the core after a LOCA in ode 5. The manual start of three SI pumps at ten minutes after the LOCA would be sufficient to prevent boiling after a LOCA in Mode 5. The break of a lower head instrument line (0.003 square feet) was considered in Mode 2 and required approximately one your for the RCS level to drop to the bottom of the hot leg initially from the mid-elevation height of the RCS piping. In Mode 5, the break of a lower head instrument line was, therefore, bounded by a break in the bottom of a hot leg.

Mode 5: Temperature is <135°F. The equipment available in Mode 5 is available in Mode 6. Because the temperature is lower in Mode 6, the Mode 6 analysis will yield acceptable consequences with one SI pump operable within ten minutes.

Subsection 5.3.1 describes the primary system boundary conditions and event scenario for the postulated LOCA used in this study. Subsection 5.3.1 also compares and contrasts this lower operating mode event scenario to the conservative design basis licensing LOCA event normally associated with ECCS evaluation analyses. Subsections 5.3.2 and 5.3.3 describe the System 80+ plant parameters and conditions for the analysis and the computer codes and analysis methods used in the LOCA calculations. Subsection 5.3.4 details the objectives and bases of the LOCA analysis.

Subsection 5.3.5 describes the results of an analysis of a limiting LOCA during Mode 4. These calculations show that hot fuel rod conditions remain in compliance with ECCS Acceptance Criteria during an assumed 10 minute time delay for oper tor action without SI pump availability. Furthermore, the analysis for the postulated LOCA shows that availability of 1 SI pump at the 10 minute mark maintains the hot rod cladding temperatures in compliance with the ECCS Acceptance Criteria.

5.3.1 DESCRIPTION OF LOCA SCENARIO

Following powered operation of the NSSS, cooldown proceeds at the Technical Specifications maximum rate of 100°F/hour (the maximum cooldown rate shown in keference 18). Therefore, a primary coolant temperature of 317°F could be reached as early as 2.4 hours after shutdown.

Safety injection tanks (SIT) and safety injection (SI) pumps are available to mitigate LOCAs from lower modes. SITs are available for pressures >900 psia. SI pumps are automatically actuated for temperatures >317°F.

1. a postulated LOCA transient were to occur at pressures and temperatures slightly below these conditions (i.e., pressure <900 psia, temperature <317°F, and time >2.4 hours), the LOCA would be significantly less dynamic than a design-basis LOCA transient from full power operating conditions (i.e., 2250 psia, ~600°F and full fission power and associated decay heat). Factors which would significantly mitigate the potential and consequences of a low r operating mode LOCA compared to a full power LOCA are (1) lower initial primary system pressure which would limit the internal forces on the piping and the duration of the blowdown,(2) lr er coolant flow rate out of a postulated break and slower depressurization rate which would reduce inventory loss and flashing rate, and (3) lower decay heat levels which would lessen the core boiloff rate.

Based on these factors, a postulated lower operating mode LOCA followed by, if necessary, timely operator action to initiate safety injection flow would be expected to be much less severe than a LOCA from full pc er conditions. The most severe lower operating mode LOCA scenario would occur for (1) the largest potential pipe break, (2) after the most rapid possible cooldown from full power, (3) after reducing temperature slightly below which no SI pumps are required to be on automatic, (4 after reducing pressure slightly below which SITs are not available, and (5) the longest expected

time for mitigating operator action. The largest and most harmful potential pipe break would be a significant leak in one of the direct vessel injection (DVI) lines of the reactor coolant system corresponding to the flow area of the DVI line. This DVI break size of 0.4 ft2 was chosen even though LBB technology would eliminate 14 from consideration which envelopes the size and limiting location of all traditional small break LOCAs. Decay heat levels based on a time period of 2.4 hours after shutdown is No operator action for 10 minutes is also assumed. assumed Forced circulation through the core during low mode operation would tend to prolong the time of ade wate core . bling during a postulated LOCA; therefore, for conservatism, the RCDs are tripped before the start of the event. An aggressive cooldown of the secondary wide of the steam generators would considerably benefit the RCS heat removal process for a postulated LOCA; therefore, for conservatism, the steam generator secondary sides are isolated for the event.

5.3.2 SELECTION OF REFERENCE PLANT PARAMETERS AND CONDITIONS FOR MODE 4 ANALYSIS

2 limiting set of values is selected from among the plant parameters and conditions. RCS temperature was selected at the maximum at which no SI pumps would automatically actuate due to SIAS. RCS pressure was selected at the maximum at which no SITs would be available. Table 5-1 is a list of initial conditions.

5.3.3 ANALYSIS COMPUTER CODES

This tas) mequired the selection of realistic inputs such as decay heat to provide as much realism in representation of the system transient response luring a LOCA as possible. An adaptation of the Realistic Evaluation Model (REM) for small break LOCA was selected for this reason. This model is a second-generation small break LOCA evaluation model intended to replace the 1974 EM currently used for ECCS licensing calculations. Topical reports describing the REM were submitted to the NRC starting in 1988 and are currently under review (see References 12 through 16).

For design basis LOCA calculations from plant initial primary pressures of 2250 psia, the largest break size analyzed using the CEFLASH-4AS code has historically been a 0.5 ft[°] break. For the 0.4 square foot DVI line lower mode LOCA analysis, the REM version of the CEFLASH-4AS code was used. A realistic model for decay heat is used, which is based on the 1979 ANS Standard 5.1 plus two sigma uncertainty plus actinide decay.

The REM version of the PARCH code was used for calculating the hot rod heatup (References 12 through 16). The PARCH base deck included a realistic decay heat.

5.3.4 LOCA ANALYSIS FOR MODE 4

The LOCA analysis examines the hot rod heatup response during a LOCA with respect to the ECCS Acceptance Criteria of peak cladding temperature and peak local cladding oxidation. The objective of this analysis is to determine if the calculated response of the hot rod remains in compliance with the ECCS Acceptance Criteria crediting operator action in a 10 minute time frame.

A bounding analysis, starting at 2.4 hours after shutdown, allowed the LOCA to proceed without safety injection. Primary coolant inventory is assumed to be lost through an opening in the direct vessel injection line at the vessel penetration to the upper annulus primary piping. When the two-phase level falls to the top of the active core, there is a reduction in core cooling and the fuel rods begin a heatup driven by the core decay heat power and at higher temperatures by the heat added from cladding oxidation. As cladding temperatures increase, fuel rod swelling and rupture may occur.

For this analysis, the PARCH code is used for calculating hot rod heatup. The hot rod in the core is initialized by the PARCH code with 900 psia and 317°F primary coolant conditions. A limiting axial power shape is assumed for the LOCA. At 8500 seconds (~2.4 hrs), the coolant inventory and, consequently, core cooling is reduced. Figure 5-1 shows the resulting hot rod heatup calculation with safety injection initiated ten minutes after break.

5.3.5 RESULTS OF LOCA ANALYSIS FOR MODE 4

This realistic model produces a cladding heatup rate of about 3.5°F/sec tollowing core uncovery and causes the cladding to exceed 2200°F in approximately 16 minutes from time of break.

The LOCA event is analyzed for realistic or credible thermalhydraulic blowdown conditions. The objective of this analysis was to determine requirements for Technical Specifications to assure adequate cooling flow in lower modes of operation. The realistic analysis examined the RCS primary coolant and hot fuel rod transient response during the blowdown phase of the LOCA.

For this analysis, a break in one of the Direct Vessel Injection (DVI) lines is postulated. For traditional "small" break scenarios, this is the largest and most limiting location for a break in the primary system because a break in the DVI line dictates that safety injection flow delivered through this line be lost to the containment. The flow area of the DVI line is 0.4 ft and, therefore, the largest effective break area.

For this analysis, modified input for the REM version of the CEFLASH-4AS code for calculating the thermal-hydraulic system

response is used. With special features and model options in the REM version, a cooldown from full power is simulated to achieve a set of initial conditions at 2.4 hours. The calculation is then restarted at this time with the break size indicated above to study the system response during the LOCA. The thermal-hydraulic boundary conditions from CEFLASH-4AS are then input into the PARCH code to determine the hot rod heatup during core uncovery.

The primary system at the start of the LOCA is at 900 psia and 317°F. The reactor coolant pumps have been stopped and the steam generators (steam dump and bypass system) have been isolated. The operation of the SCS is also not simulated.

Because it is not safety grade, no charging flow is credited in the calculations. The first set of calculations are made with no SI delivery. The results of these calculations are given in Subsection 5.3.5.1 below. A second calculation for the limiting break size is then made where SI delivery is initiated within 10 minutes. The results of this calculation are given in Subsection 5.3.5.2

5.3.5.1 Results of JOCA Case With No SI Delivery

The following discussion presents thermal-hydraulic results for a LOCA break size of 0.4 ft². The break is postulated to occur in a DVI line at the vessel penetration. LOCA transient results are shown up until time peak clad temperature (PCT) exceeds 2200°F approximately 16 minutes from initiation of break.

The purpose of the analysis is to define the RCS inventory makeup requirements. The 0.4 ft² break size is the cross-sectional area of the DVI line. For traditional "small" break scenarios (full power, mode 1), this is the most severe break location and is also the limiting location for lower modes of operation as shown in following section.

Primary Pressure

The primary system pressure transient for the 0.4 ft² break size at the DVI line location is shown in Figure 5-2. The subcooled decompression from 900 to 100 psia occurs within the first minute. The depressurization proceeds until the steam production in the core and its volume expansion from superheating due to the combination of secondary-to-primary steam generator heat transfer, core hes, transfer from uncovered fuel rods, and from wall heat transfer exceeds the steam volumetric break flow. The primary system then begins a gradual repressurization.

Reactor Vesse, Two-Phase Mixture Level

Figure 5-3 shows the two-phase mixture level in the reactor vessel for the DVI line break. Inventory is lost above the level of the hot leg elbow within the first several minutes of the LOCA. The mixture level remains at this level near the hot leg centerline (25.87 ft) until the 3team generator U-tubes and hot legs empty. During this draining period, steam produced by heat transfer from the fuel rods swells the mixture level in the core and is released at the surface to make its way down the hot leg countercurrent to the draining liquid and around the U-tubes through the loop seal and pump to the break in the DVI line. Figure 5.3-3 shows that the mixture level drops below the top of the core within approximately 7 minutes from time of break due to boiloff.

Hot Spot Cladding Temperature

The cladding temperature results using the PARCH code for the hot spot on the hot fuel rod is shown in Figure 5-4. For the limiting DVI line break location, before the mixture level drops below the top of core, the cladding temperature follows the saturation temperature of the primary coolant. After the two-phase mixture level drops below the top of the core, cladding heatup begins first at elevations near the top of the core and then progressively to lower elevations in the core which have more local power. Figure 5-4 shows the hot spot elevation temperature which is located roughly 10% of the active core height below the top of the core. There is an increase in cladding heatup rate beginning at roughly 1600°F due to the exothermic Zircaloy oxidation reaction. The time at which the hot spot cladding temperature exceeds 2200°F, the ECCS acceptance criterion for PCT, is 16 minutes.

5.3.5.2 <u>Initiation of SI Delivery Within Ten Minutes After</u> Break

The results presented below show that for break sizes of 0.4 ft² or less, operator action within 10 minutes to initiate SI injection would mitigate the postulated LOCA, i.e., prevent the cladding temperature from exceeding 2200°F.

To determine the effectiveness of restoring cooling flow to the reactor cooling system (RCS) the 0.4 ft² break size was postulated to occur at several locations, in a hot leg and a cold leg of the RCS as well as the DVI line. Ten minutes after the initiation of the LOCA, a cooling flow of 980 gpm is credited (equivalent to minimum run-out flow of one SI pump).

For the DVI line break, Figure 5-5 shows that injected cooling flow restores a positive core flow and causes the two-phase mixture level in the core to swell and recover the fuel rods. The fuel rods are recovered within 25 minutes from time of injection. The two-

phase mixture levels for the hot and cold legs as well as for the DVI line break are shown in Figure 5-6 for the ten minutes of nondelivery after the break. For the hot and cold leg break locations the level does not drop below the top of the core, clearly showing the DVI line break location to be limiting.

Figure 5-7 shows the corresponding fuel rod hot spot cladding temperature transient for the postulated DVI line break. The PCT of the cladding hot spot with cooling flow restored in 10 minutes after break is compared to the case without coolant flow restoration which resulted in a runaway heatup driven by oxidation. The restoration of cooling flow within 10 minutes prevented the cladding temperature from reaching the point of significant Zircaloy oxidation and a runaway cladding heatup.

A lack of significant 7 incaloy oxidation results in the maintenance of core geometry after LOCAs in lower modes of operation. The post-LOCA long-term cooling analysis for Mode 1 operation given in CESSAR-DC bounds long-term cooling analyses for LOCAs in lower modes because the decay heat levels are significantly lower.

The results for this Mode 4 LOCA analysis show that restoration of a cooling flow of 980 gpm (equivalent to minimum runout flow of a SI pump) within 10 minutus of the start of a LOCA will mitigate the consequences for all broad sizes of at least 0.4 ft² or less.

5.4 RESOLUTION

Analysis of small break LOCAs which are considered possible in operating Modes 2 through 6 were either limited by the Mode 1 analysis in CESSAR-DC or the Mode 4 analysis presented in this report. For the lower mode LOCAs not bounded by Mode 1 analyses, this Mode 4 analysis is bounding.

The analysis presented in this section demonstrated that a LOCA event from $t_{RCS} \leq 317$ °F and $P_{RCS} \leq 900$ psia can be mitigated provided that the equivalent of at least one SI pump is injecting to the RCS no later than 10 minutes after the start of the LOCA. For lower mode LOCAs not bounded by Mode 1 analyses, the break size and location analyzed is limiting for small breaks which could occur in lower modes with neither SI pumps on automatic nor SITs available.

For the case where SI pumps are not on automatic, the transient response for the postulated break in the DVI line nozzle of 0.4 ft shows that in approximately seven minutes from time of break the two-phase level drops below the top of the core and without cooling flow injection the cladding exceeds 2200°F sixteen minutes from time of break. The time interval for exceeding ECCS acceptance criteria for PCT is considerably larger than the 10 minute delay time assumed for operator action. Due to the 10 minute requirement for deliveries of injection flow, ABB-CE has added a technical specification to require operability of two Safety Injection Trains in Modes 4 with RCS temperature < 317°F and Modes 5 and 6. (See Table 2.2-1)

Provided that cooling flow equivalent to one SI pump can be provided within a ten minute period after a lower mode LOCA, it is concluded from the results of the analysis that the acceptance criteria for LOCAs detailed in Section 5.2 are met. 77a.wp(9212)bh

TI.BLE 5-1

SYSTEM 80+ LOCA ANALYSIS MODE 4 INITIAL CONDITIONS

Parameter

System BO+

RATED CORE POWER, MWT	3800
TIME TO ENTER MODE 4, HRS AFTER SHUTDOWN	>2.4
INITIAL PRESSURE, PSIA	900
INITIAL TEMPERATURE, 'F	317
SI PUMPS	NOT ON AUTOMATIC
SIIS	NOT AVAILABLE
STEAM GENERATOR	ISOLATED
REACTOR COOLANT PUMPS	TRIPPED BEFORE START OF



CLAD SURFACE TEMPERATURE SAFETY IMJECTION INITIATED AT 10 MINUTES DVI LINE BREAK 5-1



5Ë SESTER

PRESSURIZER PREDSURE NO SAFETY INJECTION DVI LINE BREAK Figure



SYSTEM 30 55

REACTOR VESSEL MIXTURE HEIGHT NO SAFETY INJECTION DVI LINE BREAK Figure

5-3



STERIO SUFFACE TEMPERATURE Figure NO SAFETY INJECTION DY" LINE BREAK 5-4



SYSTEM 30 5 REACTOR VESSEL MIXTURE HEIGHT SAFETY INJECTION INITIATED AT 10 WINUTES DVI LINE BREAK

5-5

Figure



SYSTEMIO C

REACTOR VESSEL MIXTURE HEIGHT THREE BREAM LOCATIONS COMPARED Figure

5 6





COMPARISON OF HOT SPOT CLADDING TEMPERATURE DVI LINE BREAK WITH AND WITHOUT SAFETY INJECTION

6.0 APPLICABILITY OF CHAPTER 6 CONTAINMENT ANALYSES

6.1 INTRODUCTION

CESSAR-DC Section 6.2.1.1 discusses containment functional design. A series of loss of coolant accidents (LOCAs) and main steam line breaks (MSLBs) were analyzed to determine the resulting containment pressure and temperature for comparison with the containment design pressure and the equipment environmental qualification envelope. The highest containment pressures and temperatures occur when the NSSS stored energy is maximized and the containment heat removal capability is minimized. Maximum .CS stored energy occurs when the plant is at 102 percent power. Maximum steam generator stored energy occurs at 0 percent power (hot standby). Consistent with the Standard Review Plan (SRP), a series of LOCAs were analyzed at 102 percent power and a series of MSLBs were analyzed at powers from 102 percent to 0 percent. The 0 percent MSLB with the failure of a containment spray train is the design basis event (DBE) for System 80+. These cases are presented in CESSAR-DC.

In going from Mode 2 tc 6, stored energy is removed from the NSSS. If the safeguards features available in Modes 1 and 2 were available through Mode 6, there would be no question that the DBE identified in CESSAR-DC was limiting. However, some safeguards equipment is removed from service at lower modes. Table 6-1 lists the availability of safeguards equipment credited in the containment analyses as a function of operating mode. Since the Main Steam Isolation Signal (MSIS) and Containment Spray Actuation Signal (CSAS) may be removed from service in Modes 5 and 6, an evaluation of LOCAs and MSLBs in these modes must be made. In addition, a LOCA initiated from zero power (Node 2) was analyzed. This report discusses the results of these evaluations. The results show that the events presented in CESSAR-DC remain limiting for both containment pressure and equipment environmental qualification. Table 6-2 presents a list of cases that were considered.

6.2 LOSS OF COOLANT ACCIDENTS (LOCAS)

Section 6.2.1.1 of CESSAR-DC presents the results of the containment pressure and temperature analysis for a series of hot leg, suction leg, and discharge leg LOCAs. Consistent with SRP 6.2.1.3, the cases were based on an initial power level of 102 percent. The limiting LOCA in terms of containment peak pressure is the Double Ended Hot Leg Slot (DEHLS) break.

As power level decreases from 102 percent, the hot leg coolant and core temperatures decrease. At the same time, the mass of coolant and pressure on the steam generator secondary sides increases. As a result, primary side stored energy decreases and secondary side energy increases. To show that the cases presented in CESSAR-DC are limiting, a 0 percent power LOCA has been analyzed. Although the case in CESSAR-DC which produced the highest LOCA containment peak pressure was the DEHLS, the Double Ended Suction Leg Slot (DESLS) break with minimum safety injection (most limiting cold leg break) was selected for this analysis because the effect of the increased secondary inventory at no load has more impact on cold leg breaks. Table 6-3 lists the assumptions and initial conditions for this case. This case is actually a Mode 2 case. Wi h the RCS and SG coolant at 556 F, this represents the case with the most NSSS energy of any Mode 2 case. Table 6-4 provides the chronology of events table. The containment peak pressure for this case is 42.0 psig compared to 44.96 psig for the equivalent case at 102 percent power. Results are shown in Figures 6-1 and 6-2.

For Modes 3 and 4, the NSSS stored energy is less than for the case analyzed above. Containment spray is still available so that LOCAs for these modes would produce lower containment pressures and temperatures. For Modes 5 and 6, the Containment Spray Actuation Signal (CSAS) may not be activated. As a result, should a LOCA occur, the containment sprays would not be available; however, ince the RCS coolant temperature for Mode 5 is less than 210 F, a LOCA in Modes 5 and 6 would not result in any significant containment pressurization.

Since the LOCAs described in Chapter 6.2.1 of CESSAR-DC are more severe than a LOCA during shutdown modes as far as containment pressurization is concerned, the annulus transient described in Chapter 6.2.1.8 of CESSAR-DC bounds all modes.

6.3 MAIN STEAM LINE BREAKL (MSLBS)

In CESSAR-DC, MSLBs were analyzed at 102, 50, 20, and 0 percent power, representing Modes 1 and 2. The cases were analyzed with either the failure of an MSIV or the loss of a containment spray train. The 0 percent power case with the loss of a containment spray train produced a containment peak pressure of 48.34 psig. This pressure was the highest of any LOCA or MSLB and this case is the containment DBE. The 0 percent cases analyzed in CESSAR-DC are Mode 2 cases. Since they were based on SG pressures of 1100 psia and RCS and SG temperatures of 556 F, they represent the cases with the most stored energy for Mode 2. Mode 2 cases with less stored energy would be less limiting.

In Mode 3 and 4, the NSSS stored energy is less than Modes 1 and 2. As shown in Table 6-1, main steam isolation and containment spray are still available in these modes. Therefore, a MSLB during these modes would not be more limiting than a MSLB during Mode 1 or Mode 2.

In Mode 5 or 6, main steam isolation and containment spray may not be available. On the other hand, the RCS coolant temperature in

77a.wp(9212)bh

Mode 5 will be less than 210 F. If the SG coolant temperature was also at 210 F, no containment pressurization would occur following a MSLB. An analysis has been performed conservatively assuming that the SG coolant was still at 350 F following shutdown cooling of the RCS to 210 F. Table 6-5 lists the assumptions and initial conditions. Table 6-6 lists the chronology of events. The containment peak pressure for this case is 13.4 psig, well below the CESSAR-DC DBE result. Results for this Mode 5 MSLB are shown on Figures 6-3 and 6-4.

6.4 INADVERTENT OPERATION OF CONTAINMENT HEAT REMOVAL SYSTEMS

During shutdown the containment is purged using either the low of high volume containment purge. An inadvertent actuation of the spray system with the containment purge valves open will result in an insignificant decrease in the containment internal pressure. The parameters affecting the negative containment pressure are the containment atmosphere initial conditions and the spray water temperature. The source of the spray water for the System 80+ design is the IRWST. Figure 3.5.4.1 of Chapter 16 of CESSAR-DC which specifies acceptable IRWST temperatures for a range of containment atmosphere temperatures is based on the properties of steam and is not mode dependent.

6.5 CONCLUSION

CESSAR-DC Section 6.2.1 provides containment analyses to support the establishment of the containment design pressure and temperature and an envelope for equipment environmental qualification. A spectrum of primary and secondary line breaks were analyzed. With the exception of the 0 percent power MSLB cases (Mode 2), all of the cases analyzed were for Mode 1. As discussed in the sections above, NSSS stored energy decreases in going from Mode 2 to Mode 6. Safeguards equipment important to containment analyses (containment spray and main steam isolation) are available in all modes with the possible exception of Modes 5 and 6. The analyses above show that by the time the plant is in Modes 5 or 6, the NSSS energy has been reduced to the point where if a postulated primary or secondary line break were to occur, the resulting containment pressure and temperature would be less severe than those presented in CESSAR-DC even with containment spray and main steam isolation unavailable. This result ensures also that the annulus transient presented in CESSAR-DC bounds all Modes.

The inadvertent containment spray actuation event presented in CESSAR-DC Section 6.2.1 is used to determine the maximum external containment design pressure. The analysis presented is not mode dependent.

TABLE 6-1

ESFAS INSTRUMENTATION

SIGNAL

APPLICABLE MODES

CSAS 1,2,3, (Note 1)

MSIS 1,2,3,4

Note 1: Table 3.3.10-1 of CESSAR-DC Chapter 16 does not show CSAS applicable for Mode 4. Section 2.2 of this report expands the applicability of CSAS to include Mode 4. 77a.wp(9212)bh

TABLE 6-2

CASES ANALYZED

MODE	LOCA	MSLB
1	102% Power DEHLS, DESLS, DEDLS	102%, 50%, 20%
	CESSAR-DC	CESSAR-DC
2	0% DESLS	0,8
	Current Report	CESSAR-DC
3	Analysis Not Required	Analysis Not Required
	Note 1	Note 1
4	Analysis Not Required	Analysis Not Required
	Note 1	Note 1
5	Analysis Not Required Since RCS T < 210 F	Case Analyzed With RCS T At 210 F and SG T At 350 F
		Current Report
6	Analysis Not Required Since RCS T < 135 F	Analysis Not Required Since Less Limiting Than Mode 5

Note 1: Less NSSS stored energy than Modes 1 and 2. Same ESF available as in Modes 1 and 2.

TABLE 6-3

INITIAL CONDITIONS FOR LOCA INITIATED FROM ZERO POLEK

Parameter

Value

Reactor Coolant System

Average Coolant Temperature, F 556

Containment

Initial conditions are consistent with CESSAR-DC Chapter 5, Table 6.2.1-18.

TABLE 6-4

ACCIDENT CHRONOLOGY FOR LOCA INITIATED FROM ZERO POWER

Time, sec	Event
0.00	Break occurs
18.00	Start Safety Injection Tank Injection
21.98	Peak Containment Pressure before End cf Blowdown
22.00	End of Blowdown
25.90	Downcomer Full
71.00	Containment Spray Injection
71.80	Peak Containment Pressure Subsequent to End of Blowdown
100.00	Safety Injection Tank Empty
108.30	End of Reflood
202.57	End of Post Reflood








TABLE 6-5

INITIAL CONDITIONS FOR MELB INITIATED FROM MODE 5

Parameter

Value

Reactor Coolant System

Average Coolant Temperature, F210Steam Generator Secondary Pressure, psia132.8Steam Generator Secondary Temperature, F350

Containment

Initial conditions are consistent with CESSAR-DC Chapter 6, Table 6.2.1-18.

TABLE 6-6

ACCIDENT CHRONOLOGY FOR MSLB INITIATED FROM MODE 5

Time, sec

Event

0.00

Break cccurs

148.71

Peak Containment Temperature Peak Containment Pressure

Note: MSIS and Containment Sprays are assumed not to be available in this mode







-



7.0 SYSTEM 80+ DESIGN FEATURES FOR SIMPLICITY OF SHUTDOWN OPERATIONS

7.1 INTRODUCTION

The System 80+ evolutionary ALWR design takes maximum benefit from prior design and operating experience. It is an objective that this benefit be evident in design features that aid outage planning, reduce operator stress and simplify operator training. Previous sections of this report describe and evaluate features of System 80+ that will improve the overall shutdown operations. In the following discussion, the outage benefits accruing from these design features are presented.

7.2 DISCUSSION

There are several features of the System 80+ design which will aid in the management of an outage. Many of these same features relieve some of the stresses and pressures placed on the licensed operators. These features are presented in Sections 7.2.1 through 7.2.9.

7.2.1 TECHNICAL SPECIFICATIONS FOR REDUCED INVENTORY

Technical Specifications for the System 80+ design specify restrictions on operation in Reduced Inventory. Reduced Inventory is determined by water level in the Reactor Coolant System (RCS). This plant operational condition is defined as "RCS level greater than 3 feet below the reactor vessel flange". These specifications provide guidance to the operations staff and the outage management team. This guidance ensures that equipment assumed to be available for accident mitigation is operable. This aids the outage planners by identifying equipment on which maintenance cannot be accomplished during Reduced Inventory. This planning process reduces the chance for removal of components relied upon for detection and response to accidents. These specifications provide the Senior Licensed Operator a clear standard for determining minimal equipment availability. This will alleviate some of the stress placed on the operator to make the final judgement as to whether equipment is required to be operable for Reduced Inventory operations.

7.2.2 SHUTDOWN COOLING SYSTEM

The Shutdown Cooling System (SCS) design provides two divisions for decay heat removal capability. These two divisions are completely redundant, that is, they share no components or equipment. This redundancy provides the operator with a standby decay heat removal system if any component fails to perform its function. This assurance of a standby system availability reduces some of the

7-1

pressures and stresses placed on the operator when no standby system is available.

Another feature of the SCS is that it is not a subsystem of the Emergency Core Cooling System. Therefore, the SCS is not required to be operable in Modes 1 through 4. This feature of System 80+ design increases the availability of SCS during Mode 5, Mode, 6 and Reduced Inventory. In addition, this feature aids the outage planning team by allowing maintenance and repair of SCS components and equipment to be accomplished during power operations. This feature eliminates the necessity of finding a window during the outage to allow work on decay heat removal equipment.

Besides the features discussed above, the SCS is designed to provide faster venting of the pumps. If SCS pumps become vapor bound due to misoperation, venting is required. The vent piping for the pumps is hard piped and directed to the floor drain sumps. This allows the plant equipment operator to quickly vent the pump without attaching vent rigs. These vent rigs waste valuable time if recovery from a loss of decay heat removal is required.

7.2.3 CONTAINMENT SPRAY SYSTEM

The Containment Spray System (CSS) design provides two divisions of equipment which can be utilized as an alternate decay heat removal flow path. The CSS pumps are interchangeable with the SCS pumps. This feature provides the operations staff with increased flexibility in the area of forced circulation. Therefore, this alt nate alignment for forced coolant flow during shutdown conditions reduces stresses placed on the operators since it increases the redundancy and therefore reliability of the System 80+ decay heat removal capability. With these alternate pumps, the operators have assurances that redundant equipment is available.

7.2.4 COMPONENT COOLING WATER SYSTEM

The Component Cooling Water System (CCW) design has two redundant divisions. Each of the two divisions contains two pumps and two heat exchangers.

This interdivisional redundancy of system components provides flexibility for the management of the maintenance outage. Therefore, major components (e.g., pumps and heat exchangers) requiring maintenance can be removed from service without affecting the availability or reliability of the interdivisional equipment. This enhancement of system design provides the outage planner with options to facilitate easier outage scheduling.

7-2

7.2.5 STATION SERVICE WATER SYCTEM

The Station Service Water System (SSW) design has two redundant divisions. Each of the two divisions contains two pumps. This interdivisional redundancy of the pumps provide flexibility for outage planning. The outage planner can schedule a pump for maintenance without affecting the availability or reliability of the interdivisional equipment nor the redundant division equipment.

7.2.6 ELECTRICAL DISTRIBUTION SYSTEM

The Electrical Distribution System (EDS) design has two redundant safety divisions. (Refer to Figure 2.4-5.) Each division is capable of being powered from four separate and diverse sources. These sources include:

- 1. Switchyard Interface I,
- 2. Switchyard Interface II,
- . Diesel Generators, and
- 4. Combustion Turbine.

The EDS provides the outage planner with the flexibility to remove a source of power for maintenance and still maintain other reliable sources to the safety buses. Therefore, required maintenance activities are scheduled without reducing the reliable sources of power to unacceptable levels. These same features provide the licensed operator with alternate sources to which safety buses can be aligned. The operator is aware of these approved alternate alignments through procedures and training. Therefore, the stress placed upon the operator to align to any available source regardless of guidance is reduced. In addition, operator training is facilitated by the procedural guidance.

Another feature of the EDS design is the use of 4 safety buses, 2 per Division. The 1E loads are evenly distributed on the buses to ensure redundancy of system components. For example, each bus powers 1 of Component Cooling Water pumps. This feature provides flexibil. For the outage planner. One bus can be removed from service fo. na mtenance and redundant components still have a power supply available.

7.2.7 NUPLEX 80+ ADVANCED CONTROL COMPLEX

The NUPLEX 80+ Advanced Control Complex (ACC) is an integral part of the System 80+ design. The design goals of NUPLEX 80+ include the integration of NSSS and balance of plant systems into a unified control complex, reduction of human errors that affect plant safety and improving the reliability of the man-machine interface through redundancy, segmentation and diversity. Control room information provided by NUPLEX 80+ is consistent with operator information requirements when performing operational tasks during plant evolutions or responding to unexpected conditions. The operator can obtain information from a number of sources in the NUPLEX 80+ ACC. These sources include:

- A large plant overview status board known as the Integrated Process Status Overview (IPSO).
- Alarm tiles and associated message windows.
- Discrete indicators provide frequently used and important information.
- CRT displays (ontaining essentially all power plant information.

More detailed information on the NUPLEX 80+ ACC can be found in Chapter 18 of System 80+ CESSAR-DC.

The NUPLEX 80+ ACC utilizes the same parameter conventions for the indicators and alarms for shutdown operations as required for power operations. The features of each which simplify operator training, aid in outage planning and reduce operator stress are described below.

- IPSO IPSO provides the operators, especially Senior Reactor Operators with an overview of plant status during shutdown conditions. This overview allows the operators to view system status during outage activities. Having an overview of the plant will reduce uncertainty of the availability of required safety systems. This knowledge of the availability reduces the stress placed on the operators by uncertainties.
- Alarms Mode and equipment dependent alarms are a special feature of NUPLEX 80+ ACC. This feature eliminates the alarming of alarms not applicable to the current mode, operating conditions, or equipment status. A large amount of maintenance activities involved with the outage affect control room alarms. The mode and equipment dependent alarms eliminate operator response to nuisance alarms caused by authorized work being performed. Outage work may be planned, alarms disabled and unnecessary investigation by operators into these alarms eliminated.
- Discrete Indicators Discrete Indicators provide several simplifying attributes for operators. Automatic ranging scales on Eiscrete Indicators allow accurate indication over the entire range of system design with the same indicator. Using the same indicator for all conditions of operation, including shutdown, avoids confusion for the operators. This feature

allows training to utilize the same indicators on a simulator. It also eliminates the utilization of indicators solely for shutdown. Discrete Indicators receive multiple channel input signals to be displayed on one indicator. These signals are validated and provided the operator with reliable indication even if some channels are removed from service. Individual channel inputs to the Discrete Indicators alarm to alert the operator when one has been removed from service. This allows the operator to check the status of information provided. Using validated displays reduces stress on the operator by eliminating doubt of instrument availability and accuracy.

CRT Displays - CRT displays for the NUPLEX 804 ACC are arranged in a structured information hierarchy. This structure provides the operator information consistent with operational needs. Levels of display information start with IPSO and continue through detailed plant information. A feature of the CRT display is graphic representation of systems. This reinforces system layout training and leads to better understanding by the operator. It provides consistency for the operators which reduces stress. Color representation of valves to indicate operable/inoperable status gives the operator the information to determine flowpath status. This feature also provides status of maintenance in progress on important valves in the plant.

7.2.8 REDUCED INVENTORY INSTRUMENTATION

The System 80+ design provides the instrumentation to insure the Control Room Operator (CRO) is informed of the decay heat removal system performance and the reactor coolant system level and temperature. The instrumentation includes:

- 1. Reactor Coolant System Level,
 - Heated Junction Thermocouple
 dP
- 2. Reactor Coolant Temperature,
 - Core Exit Thermocouple
 - · Heated Junction Thermocouple
- 3. Shutdown Cooling System Flow,
- 4. Shutdown Cooling System Pressure,
- 5. Shutdown Cooling System Temperature, and
- 6. Shutdown Cooling Pump Motor Current.

See Section 2.8, Instrumentation, of this report for a discussion of this instrumentation. The instrumentation is coupled with the Nuplex 80+ Advanced Control Complex (ACC) to provide the CRO with indications and alarms to monitor reduced inventory operations. (See Section 7.2.7 above for a description of Nuplex 80+ ACC features). The CRO has the responsibility to keep the reactor core cooled and covered during reduced inventory operations. Although the burden of responsibility placed on the CRO cannot be relieved, the CRO's tasks are made simpler to perform. With the System 80+ instrumentation, the operator is not required to rely on local gauges and/or temporary instruments installed only for shutdown operation. This design increases the operator's confidence level of parameter indications and therefore decreases the stress operators are placed under when in reduced inventory operations.

7.2.9 CONTAINMENT

The containment for the System 80+ is a 200 foot diameter, spherical design steel vessel with a free volume of 3.3 million cubic feet. The spherical design incorporates a large operating deck inside containment with 21,500 square feet of floor space.

The containment structure utilizes internal boundaries for ventilation systems and therefore eliminates a significant portion of ductwork in containment. The elimination of ductwork reduces the amount of outage activities required in containment for ventilation system maintenance and repair.

The design of the containment operating deck includes open floor space allocated for storage and maintenance during outage periods. Portions of the operating deck are suitable for pre-staging and laydown of equipment in support of maintenance activities. Examples of items that may be prestaged are:

Valve maintenance tools and test equipment, Reactor vessel and head equipment, and Steam Generator inspection and repair equipment.

Pre-staging equipment required for containment work activities in advance eliminates having to open the equipment hatch unexpectedly due to work schedule and scope changes.

The layout of the operating deck allows for multiple maintenance activities during the outage and reduces the number of times the equipment hatch must be opened. Valve maintenance, reactor vessel stud cleaning and reactor coolant pump work is performed in shielded areas on the perimeter of the operating dick. The space provided for these maintenance activities combined with crane support and pre-staged tools eliminates the need to transfer components through the equipment hatch to work spaces outside containment.

The elimination of equipment hatch openings during critical plant evolutions such as RCS reduced inventory operations helps to ensure containment integrity is maintained. The design simplifies outage planning by providing space for prestaging and work activities to be accomplished in parallel without reliance on equipment hatch opening.

7.3 CONCLUSION

The System 80+ design has incorporated features which are consistent with the objective outlined above in Section 7.1. The features discussed provide the outage management team with redundancy, diversity, and guidance to simplify outage planning. In addition, these features provide the operator with reliable and redundant equipment and indications. This also provides the operator with some relief from the stresses experienced when no redundancy or parameter indication is available. Operator training is simplified due to equipment redundancy and diversity available in shutdown operations. THIS PAGE INTENTIONALLY BLANK

3

8.0 <u>CONCLUSIONS</u>

This report has provided responses to issues on shutdown risk raised by NRC RAIS. The report shows that the System 80+ design features along with Technical Specifications and operational guidance will result in acceptable consequences from accidents during shutdown.

System 80+ is engineered with features that enhance shutdown safety: 1) by deliberate system engineering, equipment specification and plant arrangements for shutdown operation, 2) by mode dependent control logic that assists and limits operations, 3) by instrumentation, displays and alarms that clearly portray plant status in each mode and 4) by thorough procedural guidance and Technical Specifications that address important shutdown evolutions. This report shows that these features successfully prevent the occurrence of shutdown risk events or mitigate the consequences of the events.

The PRA study performed for this report shows that the core damage frequency for internal events while the plant is in shutdown modes is about the same as the frequency while the plant is at power operation. Evaluations of SAR Chapter 6 and 15 events while in shutdown modes show that all acceptance criteria are met. In addition to the Chapter 6 and 15 analyses, the System 80+ design was reviewed with regard to other issues such as flooding and spills, fires, the dropping of heavy loads, and containment closure times. Again, all acceptance criteria were met.

As a result of the shutdown risk study, several changes to the Technical Specifications will be made. In addition, two design changes will be made to provide improved RC3 water level indication during midloop operation. One of these changes will be the addition of two narrow range delta P cells between the DVI nozzle and the SDC nozzle at the hot leg. The other change will be the addition of two heated junction thermocouple strings to cover the hot leg region. These changes will provide a more accurate measurement of the water level in the vicinity of the hot leg elevation. THIS PAGE INTENTIONALLY BLANK

77a.wp(9212)bh

9.0 REFERENCES

- Letter LD-92-038, C. B. Brinkman (ABB-CE) to D. M. Crutchfield (NRC) dated March 25, 1992.
- Memo Letter, "Summary of Meeting Held on December 18, 1991 Regarding Shutdown Risk", T. V. Wambach (NRC), dated January 30, 1992.
- NUREG-1449, "Shutdown and Low-Power Operation at Commercial Nuclear Power Plants in the United States", Draft Report dated February, 1992.
- USNRC Generic Letter No. 88-17, "Loss of Decay Heat Removal", dated October 17, 1988.
- USNRC AEOD Special Report, "Review of Operating Events Occurring During Hot and Cold Shutdown and Refueling", dated December 4, 1990.
- 6. NUREG-0800, USNRC Standard Review Plan, Revision 1, July, 1981.
- Jaquith, R.E., et.al., "Probabilistic Risk Assessment for the System 80+ Standard Design", Combustion Engineering Inc., DCTR-RS-02, January, 1991.
- Letter from James H. Wilson (NRC) to E. E. Kintner (EPRI) dated September 5, 1991.
- Brockhold, G. (EPRI), Memo to ALWR Utility Steering Committee, ALWR-92-18, January 15, 1992.
- 10. This Reference was deleted.
- 11. ANSI/ANS-58.8-1984.
- CEN-373-P, Volume 1, "Realistic Small Break LOCA Evaluation Model, Calculational Models," April 1988.

Volume 2, "Application of Evaluation Model," December 1988.

Volume 2 Supplement 1-P, "Application of Evaluation Model to Calvert Cliffs Units 1&2, September 1989.

Volume 3, "Computer Program Input and Output Description," December 1988.

 Letter LD-88-030, "Submittal of Realistic SBLOCA Evaluation Model," A. E. Scherer (ABB CE) to J. A. Norberg (NRC), April 27, 1988.

- Letter LD-88-155, "Submittal of Volumes 2 and 3 of Combustion Engineering's Realistic Small Break Loss-Of-Coolant-Accident Evaluation Model." A. E. Scherer (ABB CE) to J. A. Norberg (NRC), December 9, 1988.
- 15. Letter LD-89-001 "Addandum to Volume 3 of Combustion Engineering's Realistic Small Break Loss-of-Coolant Accident Evaluation Model," A. E. Scherer (ABB CE) to J. A. Norberg (NRC), January 11, 1989.
- Letter LD-89-099, "Supplement 1 to Volume 2 of Combustion Engineering's Realistic Small Break LOCA Evaluation Model," A. E. Scherer (ABB CE) to J. A. Norberg (NRC), August 28, 1989.
- 17. ANSI-N18.2-1973.
- 18. "System 80+, CESSAR-DC," Technical Specification, 3.4.3.

19. 10CFR50.46.

- 20. CENPD-138, Supplement 2-F, "PARCH, A FORT: N-IV Digital Program to Evaluate Pool Boiling, Axial Rod and Coolant Heatup," January, 1977.
- 21. Chu, T-L, et. al., "PWR Low Power and Shutdown Accident Frequencies Program, Phase 1A - Course Screening Analysis," Brookhaven National Laboratory, Letter Report FIN L-1344, November 13, 1991.

APPENDIX A

RESPONSES TO

REQUESTS FOR ADDITIONAL INFORMATION

APPENDIX A

TABLE OF CONTENTS

Section

A1.0 INTRODUCTION

A2.0 REQUESTS FOR ADDITIONAL INFORMATION

Question	440.14*
Question	440.16a
Question	440.16b
Question	440.16c
Question	440.16d
Question	440.16e*
Question	440.16f
Question	440.16g
Question	440.16h
Question	440.161
Question	440.16j
Question	440.35
Question	440.36*
Question	440.37*
Question	440.49*
Question	440.54*
Question	440.70*
Question	440.86f
Question	440.91
Question	440.109*
and the second se	
Question	440.120
Question Question	440.120
Question Question Question	440.120 440.130 440.131
Question Question Question Question	440.120 440.130 440.131 440.132
Question Question Question Question	440.120 440.130 440.131 440.132 440.133
Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.134
Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.134 440.135
Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.133 440.135 440.136
Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.134 440.135 440.136 440.137
Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.133 440.135 440.135 440.137 440.139*
Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 440.133 440.133 440.135 440.135 440.137 440.139* 440.140
Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.135 440.135 440.136 440.137 440.139* 440.140
Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142
Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142 440.143
Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142 440.143
Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 440.133 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142 440.143
Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 440.133 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142 440.143 440.145 440.146
Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 440.133 440.133 440.135 440.135 440.136 440.139* 440.140 440.141 440.141 440.142 440.143 440.145 440.145
Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142 440.143 440.144 440.145 440.145 440.145
Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question Question	440.129 440.130 440.131 449.132 440.133 440.133 440.135 440.135 440.137 440.139* 440.140 440.141 440.142 440.143 440.144 440.145 440.145 440.145 440.145

APPENDIX A (Continued)

TABLE OF CONTENTS

Section

A2.0 REQUESTS FOR ADDITIONAL INFORMATION (CONTINUED)

Question 440.151 Question 410.54* Question 410.65* Question 410.66* Question 410.72* Question 410.73* Question 410.103* Question 410.107* Question 720.98 Question 722.94 Question 280.1*

* Response was previously submitted and is reproduced in Section A2.

A3.0 REFERENCES

A1.0 INTRODUCTION

Letters from the NRC that transmitted Requests for Additional Information (RAIs) on shutdown related topics are listed in Section 3.0 of this Appendix as References A-1 through A-6. ABB-CE provided individual responses to some of these RAIs in References A-7 through A-13 and A-15. These responses are reproduced in this Appendix. The remaining RAIs related to shutdown risk were listed in the ABB-CE letter, Reference A-14 along with a commitment to provide responses via this report. These responses are also given in this Appendix, where the response may refer to the content of applicable sections of this report. The combination of the specific responses provided in the Appendix along with the overall report content fulfills all outstanding commitments to provide shutdown risk information in support of the CESSAR-DC review process.

A2.0 RAIS

The following pages provide the usual format for question and response to the RAIs. Responses to all but one of the RAI's listed in the Table of Contents are provided. The one remaining response, to Question 720.98, will be submitted in the July 31, 1992 final submittal of this report.

Question 440.14:

Table B1.3-1 of Appendix B provides comparisons of severe accident frequency for System 80+ versus System 80. Recent scoping studies have indicated that events when not at power (i.e., shutdown) can provide a significant risk contribution.

Have these events been analyzed and incorporated into the PRA and if so provide relevant information in the summary table? If not evaluated, provide a schedule for submitting this additional analysic.

*Response 440.14:

A shutdown risk assessment is being performed for the System 80+ ALWR and will be provided to the NRC by July 31, 1992. This risk assessment includes a Shutdown PRA which is being performed using the methodology described in the revised EPRI PRA Groundrules and Key Assumptions Document. A comparison of the results for System 80+ and System 80 will not be possible because a Shutdown PRA is not available for the earlier plant.

*This response, and the mentioned risk assessment, supplement the previous response to this RAI in Reference A-15.

77c.wp(9213)mls

Question 440.16(a):

Generic Letter (GL) 87-12 requested information regarding lower RCS inventory operation. Please provide a response to the generic letter with respect to the System 80+.

*Response 440.16(a):

Generic Letter 97-12 requested each licensee of operating PWRs and holder of construction permits for PWRs to "provide the NRC with a description of the operation of [the] plant during the approach to a partially filled RCS condition and during operation with a partially filled RCS to ensure [the plant] meet[s] the licensing basis." The description provided in response to Generic Letter 87-12 was to include each of the following nine (9) items, which are in turn responded to.

Generic Letter Item 1:

A detailed description of the circumstances and conditions under which [the] plant would be entered into and brought through a draindown process and operated with the RCS partially filled, including any interlocks that could cause a disturbance to the system. Examples of the type of information required are the time between full-power operation and reaching a partially filled condition (used to determine decay heat loads); requirements for minimum steam generator (SG) levels; changes in the status of equipment for maintenance and testing and coordination of such operations while the RCS is partially filled; restrictions regarding testing, operations, and maintenance that could perturb the nuclear steam supply system (NSSS); ability of the RCS to withstand pressurization if the reactor vessel head and steam generator manway are in place; requirements pertaining to isolation of containment; the time required to replace the equipment hatch should replacement be necessary; and requirements pertinent to reestablishing the integrity of the RCS pressure boundary.

Response to Item 1:

Responses these items are provided in the Shutdown Risk Report; refer to Sections 2.1, 2.2. 2.3, and 2.5.

Generic Letter Item 2:

A detailed description of the instrumentation and alarms provided to the operators for controlling thermal and hydraulic aspects of the NSSS during operation with the RCS partially filled. [D]escribed temporary connections, piping, and instrumentation used for this RCS condition and the quality control process to ensure proper functioning of such connections, piping and instrumentation, including assurance that they do not contribute to loss of RCS inventory or otherwise lead to perturbation of the NSSS while the RCS is partially filled. [A]lso provide a description of [the] ability to monitor RCS pressure, temperature, and level after the RHR function may be lost.

Response to Item 2:

The response to this item is provided in Section 2.8 of the Shutdown Risk Report.

Generic Letter Item 3:

Identification of all pumps that can be used to control NSSS inventory. Include: (a) pumps require[d] operable or capable of operation (include information about such pumps that may be temporarily removed from service for testing or maintenance); (b) other pumps not included in item a (above); and (c) an evaluation of items a and b (above) with respect to applicable TS requirements.

Response to Item 3:

The response to this item is provided in Sections 2.1, 2.2, 2.3 and 2.4 of the Shutdown Risk Report.

Generic Letter Item 4:

A description of the containment closure condition require[d] for the conduct of operations while the RCS is partially filled. Examples of areas of consideration are the equipment hatch, personnel hatch containment purge valves, SG secondary-side condition stream of the isolation valves (including the valves), piping penetrations, and electrical penetrations.

Response to Item 4:

The response to this time is provided in Section 2.5 of the Shutdown Risk Report.

Generic Letter Item 5:

Reference to and a summary description of procedures in the control room of [the] plant which describe operation while the RCS is partially filled. [The] response should include the analytic basis used for procedures development. [The staff is] particularly interested in treatment of draindown to the condition where the RCS is partially filled, treatment of minor variations from expected behavior such as caused by air entrainment and de-entrainment, treatment of boiling in the core with and without RCS pressure boundary integrity, calculations of approximate time from loss of RHR to core damage, level differences in the RCS and the effect upon instrumentation indications, treatment of air in the RCS/RHR system, including the impact of air upon NSSS and instrumentation response, and treatment of vortexing at the connectics of the RHR suction line(s) to the RCS.

Example how [the] analytic basis supports the following pertaining to [the] facility: (a) procedural guidance pertinent to timing of op. ations, required instrumentation, cautions, and critical parameters; (b) operations control and communications requirements regarding operations that may perturb the NSSS, including restrictions upon testing, maintenance and coordination of operations that could upset the condition of the NSSS; and (c) response to loss of RHR, including regaining control of RCS heat removal, operations involving the NSSS if RHR cannot be restored, control of effluent from the containment if containment was not in an isolated condition at the time of loss of RHR, and operations to provide containment isolation if containment was not isolated at the time of loss of RHR (guidance pertinent to timing of operations, cautions and warnings, critical parameters, and notifications is to be clearly described).

Response to Item 5:

The response to this item is provided in Sections 2.1, 2.2, 2.3, 2.4, 2.5, 2.8, 2.9, 2.10 and 2.12.

Generic Letter Item 6:

A brief description of training provided to operators and other affected personnel that is specific to the issue of operation while the RCS is particlely filled. [The NRC staff is] particularly interested in such areas as maintenance personnel training regarding avoidance of perturbing the NSSS and response to loss of decay heat removal while the RCS is partially filled.

Response to Item 6:

Operator training is the responsibility of the utility, since the material is plant specific and requires information concerning personnel organization and plant procedures. However, by information on avoidance of, and response to, a loss of doay heat removal that results from C-E's efforts preparing the Shutdown Risk Report will be included in procedure guidelines for the System 80+ plant. Refer to Section 2.1 of the Shutdown Risk Report for detailed information.

Generic Letter Item 7:

Identification of additional resources provided to operators while the RCS is partially filled, such as assignment of additional personnel with specialized knowledge involving the phenomena and instrumentation.

Response to Item 7:

Per NUREG-0737, the Shift Technical Adviser shall be knowledgeable of, and thoroughly trained in, all aspects of abnormal operation. Since the Shift Technical Adviser will have access to all available information resulting from C-E's Shutdown Risk Assessment program, as well as thorough training, it is anticipated the Shift Technical Adviser will be a key person during partially-filled RCS operation. The identification of any additional resources beyond the normal complement of control room personnel is not the responsibility of the NSSS designer or Architect Engineer. C-E will, however, make any recommendations deemed appropriate to System 80+ licensees.

Generic Letter Item 8:

Comparison of the requirements implemented while the RCS is partially filled and requirements used in other Mode 5 operations. Some requirements and procedures followed while the RCS is partially filled may not appear in the other modes. An example of such differences is operation with a reduced RHR flow rate to minimize the likelihood of vortexing and air ingestion.

Response to Item 8:

The response to this item is provided in Sections 2.1, 2.2, 2.3 and 2.4 of the Shutdown Risk Report.

Generic Letter Item 9:

As a result of consideration of these issues, [licensees of operating PWRs and holders of construction permits for PWRs] may have made changes to [the] current program related to these issues. If such changes have strengthened [the] ability to operate safely during a partially filled situation, describe those changes...

Response to Item 9:

The response to this item is provided throughout the Shutdown Risk Report.

Question 440.16(b):

Please describe instrumentation provided to the operator during shutdown operations which characterize the state of the reactor coolant system (RCS). Include RCS lev-1, RCS temperature, and shutdown cooling system (SCS) performance and provide a description of the appropriateness and accuracy of each instrument with respect to its intended function. Also, include identification of audible and visual alarms used to delineate out-of-range conditions, including the values which constitute those conditions.

*Response 440.16(b):

The response to this item is provided in Section 2.8 of the Shutdown Risk Report.

77c.wp(9213)mls

Question 440,16(c):

NUREG-1269 identified that containment was open throughout the April 10, 1987 event at Diablo Canyon, and there were no procedures to reasonable assure containment closure in the event of progression of the accident to a core damage condition. Address this situation with respect to the System 80+ design and the anticipated methods that will be used to operate the plant. Include such design considerations as the need for removal of the equipment hatch and improvements in the System 80+ design which facilitate rapid replacement of the hatch should the need arise. Similarly address other containment penetrations and potential bypass paths.

*Response 440.16(c):

The response to this item is provided in Sections 2.1 and 2.5 of the Shutdown Risk Report.

Question 440.16(d):

NUREG-1269 contains the statement "Design of the nuclear steam supply system (NSSS) did not appear to provide detailed provisions for mid-loop operation." Please address this identified deficiency in PWR design with respect to the System 80+ design. Include identification of and discussion of each of the design changes in the System 80+ which represents an improvement over existing designs and establish the adequacy of the System 80+ design for lowered RCS inventory operation.

*Response 440.16(d):

The response to this item is provided in Sections 2.3 and 2.4 of the Shutdown Risk Report.

Question 440.16(e) (Shutdown Operations):

NUREG-1269 identified that containment was open throughout the April 10, 1987 event at Diablo Canyon, and there were no procedures to reasonably assure containment closure in the event of progression of the accident to a core damage condition. Address this situation with respect to the System 80+ design and the anticipated methods that will be used to operate the plant. Include such design considerations as the need for removal of the equipment hatch and improvements in the System 80+ design which facilitate rapid replacement of the hatch should the need arise. Similarly address other containment penetrations and potential bypass paths.

*Response 440.16(e):

The NRC position on the subject of containment closure and plant procedural and design measures required to prevent and to mitigate the consequences of a Reactor Coolant System (RCS) overheating event during Cold Shutdown in preparation of and during Refueling was promulgated by Generic Letter No. 88-17, "Loss of Decay Heat Removal," dated October 17, 1988. In this guidance, the NRC set a limit of 2 hours for Combustion Engineering NSSS plants to establish containment closure, which constitutes isolation of all penetrations which could provide a release path for radioactive material, should the RCS overheat and boil. In a shutdown, refueling outage plant condition, release paths are the equipment hatch, the personnel locks, and any penetrations being tested or maintained. The guidance of the NRC Generic Letter 88-17 allows each plant to establish procedures and conduct closure drills to demonstrate that the 2-hour closure requirement is achievable. For purposes of design certification, release paths such as the equipment hatch and personnel locks, will be designed to be closable within 2 hours. The owner/operator will be responsible for employing the guidance of Generic Letter 88-17 to assure via procedures, drills, and administrative means that all containment release paths are closable within the prescribed 2 hours.

*This response was previously transmitted by Reference A-15.

Question 440.16(f):

The Diablo Canyon event and subsequently obtained information have shown operating procedures to be inadequate for lowered RCS inventory operation. What plans exist for recommending improved procedures and administrative controls to System 80+ owners/ operators so that this situation is eliminated. What changes will be made to the Emergency Procedure Guidelines (EPGs) (CEN-182) to accommodate this need?

*Response 440.16(f):

Sections 2.1 and 2.2 of the Shutdown Risk Report address Reduced Inventory Operating Guidelines and Technical Specifications. Emergency operating guidance for Reduced Inventory Operations are discussed throughout the Shutdown Risk Report. At present, there are no provisions to revise the EPG's (CEN-152). (See Response to RAI 440.141)
Question 440,16(g):

What equipment exists in the System 80+ that can be used to assure adequate core cooling in the event of a complete loss of SCS?

*Response 440.16(q):

Section 2.4.3.1 of the Shutdown Risk Evaluation Report addresses shutdown event initiation and analyses. The discussion in this section examines event initiators, the design features which improve the SCS's resistance to initiators, the ability to recover from initiators and the equipment, operator actions, operating procedures and technical specifications to cope with a prolonged loss of the SCS's.

*This response and the mentioned report sections fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.16(h):

The Vogtle loss of all safety-related AC power event occurred in March 1990 and the NRC incident investigation team (IIT) identified many areas of concern involving shutdown operation and they are documented in NUREG-1410. Please address these staff concerns with respect to the System 80+ design.

*Response 440.16(h):

The major concerns raised in NUREG-1410 are:

- (a) risk-management concepts applied to outage planning and scheduling during shutdown;
- (b) power availability during shutdown modes;
- (c) diesel generator instrumentation and control system availability;
- (d) coping with the loss of DHR;
- (e) technical specifications for reduced inventory operations.

Item (a) is addressed in general by most of the sections in the Shutdown Risk Evaluation Report. Item (b) is addressed in sections 2.4.3.1 and 2.4.3.2. Item (c) is addressed in section 2.4.3.3. Item (d) is addressed in section 2.4.3.1 and item (e) in sections 2.2 and 2.3.

*This response and the mentioned report sections fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.16(i):

Evidence exists that certain Technical Specifications (TSs) may not be optimum when consideration is given to operation during nonpower conditions. For example, requirements for SCS suction valve interlocks impact upon SCS reliability, SCS flow rate requirements may overly restrict flow rate range and increase the likelihood of loss of SCS due to vortexing, and TSs written on the basis of the state of the NSSS and/or containment may be impacted. Please address this topic with respect to the System 80+ design and provide recommendations for improvement, particularly with respect to the unique design aspects of System 80+.

*Response 440.16(i):

Technical Specifications for shutdown modes are discussed in section 2.2 of the Shutdown Risk Evaluation Report.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.16(j):

Safety analysis reports (SARs) typically concentrate on power operation when consideration is given to many of the potential operational transients. The recent experience from the events in operating reactors indicated that further evaluation for plant operation at lower modes may be required. Hence, it may be prudent to address non-power operation in more depth than has been traditional. What plans exist, if any, with respect to this topic and the System 80+ program?

*Response 440.16(j):

The System 804 program encompassed an extensive evaluation to assess the vulnerability of the System 804 design to various transients. The results of this evaluation are provided in the System 804 Shutdown Risk Evaluation Report. Sections 2.4, 2.5, 2.6, 4.0, 5.0, and 6.0 of this report present transient analyses covering the extent of events of concern for shutdown operations. Sections 2.4 through 2.6 concentrate on the loss of decay heat removal and rapid boron dilution events. Section 4.0 represents the evaluation of the impact of CESSAR-DC Chapter 15 events occurring during shutdown modes of operation. Section 5.0 represents the evaluation of the impact of the CESSAR-DC Section 6.3 loss of coolant accidents occurring during shutdown modes of operation. Section 6.0 evaluates the impact of the CESSAR-DC Section 6.2 events on the containment response when these events are initiated from shutdown modes of operation.

In addition, to supplement the above evaluations Section 3.0 of the System 80+ Shutdown Risk Evaluation Report presents a Probabilistic Risk Assessment which covers various event sequences which can occur during shutdown operations.

This response and the mentioned report sections fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.35:

Since operation with less than all 4 RC rumps is allowed during shutdown modes, discuss the effects of this initial condition on the CEA withdrawal event from these modes, particularly with respect to the pressure transient.

*Response 440.35:

The effect of an initial condition of less than all four reactor coolant pumps operating on the consequences of the CEA withdrawal is included in the evaluation of this event contained in Section 4.4.1 of the Shutdown Risk Evaluation Report.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAT.

Question 440.36:

Discuss the adequacy of a high neutron flux alarm to indicate a boron dilution event in sufficient time during Modes 3, 4 or 5.

*Response 440.36:

The high neutron flux alarm is activated when the SRM (Source Range Monitoring) ratio exceeds its setpoint. The SRM ratio is defined as follows:

Source Range Signal (t) Source range signal at start of dilution

For Mode 3, 4 and 5 operation, time is calculated from event initiation to loss of shutdown margin. From this time, a selected operator action time which exceeds SRP criteria is subtracted to determine the latest allowable time for alarm actuation. The analysis ensures that at this time the TRM ratio will have exceeded its setpoint.

*This response was previously transmitted by Reference A-9. Changes have been made as noted by the bars in the margin.

Question 440.37:

Standard Review Plan 15.4.6 requires redundancy of alarms that alert the operator to an unplanned boron dilution event. Describe the redundant alarms available in each operating mode.

*Response 440.37:

The following pre-trip alarms are available for operational Modes 1 and 2: a high power or, under certain conditions, a high pressurizer pressure pre-trip alarm in Mode 1 or a high logarithmic power pre-trip alarm in Mode 2. Furthermore, a high RCS temperature alarm may also occur prior to trip. In operational Modes 3 through 6, a single failure proof boron dilution alarm will alert the operator to an unplanned boron dilution event.

In addition to the above mentioned alarms, there are also sampling and boronometer indications which would provide information in the case of a boron dilution event.

*This response was previously transmitted by Reference A-9. Changes have been made as noted by the bars in the margin.

Question 440.49

Previde a discussion of the procedures and plant systems used to take the plant from normal operating conditions to cold shutdown conditions. This discussion should include, heat removal, depressurization, flow circulation, and reactivity control.

*Response 440.49

The principal systems utilized in taking the plant from Mode 1, Power Operation, to Mode 5, Cold Shutdown are:

Reactor Coolant System Feedwater Control System Reactivity Control System Boron Control System Chemical & Volume Control System Shutdown Cooling System Pressurizer Level Control System Steam Bypass Control System Pressurizer Pressure Control System Liquid and Gaseous Waste Management Systems Main Steam System Condensate System

Reactivity control capability is discussed in Sections 7.7.1.1.1; 7.7.1.1.7; 9.3.4.1.3.3; 9.3.4.2.1 (last paragraph), and 9.3.4.2.3C. Power is reduced by increasing the boron concentration in the RCS to reduce k-effective to ≤ 0.99 . At low power the rods are inserted. The operator borates to the cold shutdown boron concentration consistent with the Technical Specifications prior to the beginning of cooldown. This margin is maintained throughout cooldown by making up shrinkage volume by means of the CVCS with water at the cold shutdown margin boron concentration.

Cooldown is effected by the systems described, and techniques discussed in Sections 5.4.7.2.6.A; 9.3.4.2.3C; 7.7.1.1.2.1; 7.7.1.1.2.2; 7.7.1.1.4; 7.7.1.1.5 and 10.4.7.2.4. Additionally, the following precautions, limits and techniques are utilized during cooldown:

- The reactor coolant pumps continue to run until they are manually tripped
 - o Four RCPs shall not be operated below approximately 500°F
 - O During cooldown RCP 1A and 1B shall be running to maintain pressurizer spray capability until they are required to be shutdown for some other reason

Response 440.49 (continued)

- The RCPs shall not be operated when the system pressure is below cavitation or scal operation limits
- 2. The RCS pressure is maintained at 2250 psia until cooldown is initiated.
- Pressurizer pressure and level controls are placed in manual mode at the beginning of cooldown, and power to heaters is reduced.
- Core flow is maintained throughout cooldown by RCPs and/or shutdown cooling system pumps.
- The bubble in the pressurizer is maintained as long as possible.
- Volume Control Tank (VCT) gas space is vented to reduce fission gas and hydrogen gas prior to cooldown.
- 7. Letdown flow is directed, as required, to the gas stripper to remove dissolved gas.
- 8. Initially, heat is removed from RCS by dumping steam:
 - Steam may be dumped to the condensers through the Steam Bypass System, or to the atmosphere through the Atmospheric Dump Valves (ADV).
 - Feed control is in manual during cooldown using the startup pump and manual control valves.
 - The MSIS setpoint is adjusted to 200 psi below existing steam pressure as cooldown progresses.
- 9. As RCS water cools, pressure is decreased by manually adjusting pressurizer spray to cool the vapor space. Pressurizer pressure is controlled such that saturation margin limit is not exceeded, and such as to comply with the pressure-temperature curves specified for the plant.
- As pressurizer pressure decreases the SIAS & CIAS setpoints are decreased to 400 psi below existing pressurizer pressure.
- 11. RCS cooldown rate shall be maintained within Technical Specifications (TS) at all times during cooldown.
- 12. Pressurizer water temperature should exceed RCS water temperature by no more than 350°F and no less than 50°F whenever there is a bubble in the pressurizer.

Response 440.49 (continued)

- Auxiliary pressurizer spray is utilized to reduce pressurizer pressure whenever normal spray is inadequate or not available.
- 14. When the pressurizer pressure is approximately 400 psia and the RCS temperature decreases to 350°F, cooldown is transferred to the Shutdown Cooling System (SCS). Cooldown from this point is fully described in Section 5.4.7.2.6A. Steaming and feed may be terminated.

*This response was previously transmitted by Reference A-12

Question 440.54:

Discuss the alarms and indications which would inform the operators that the 'S suction line isolation valve has closed while the plant is in shutdown cooling? Is there any common mode failure which would result in isolation valves in both trains being closed while in shutdown cooling? Are there any manual maintenance valves whose closure could isolate the SCS suction, if so, describe procedures and controls to restrict this possibility?

*Response 440.54:

The SCS suction line isolation valves are SI-651, SI-652, SI-653, SI-654, SI-655, and SI-656, as shown on CESSAR-DC Figure 6.3.2-1C. Valves SI-651, SI-653, and SI-655 are in line with shutdown cooling pump 1. Valves SI-652, SI-654, and SI-656 are in line with shutdown cooling pump 2. There are several alarms and indications to inform the operators that a SCS suction valve has closed while the plant is in shutdown cooling:

- Valve position indications is provided in the main control room.
- If a suction valve were to close, a drop in SCS pump flow for the affected train would actuate a low shutdown cooling flow alarm.
- SCS pump current and pressure indicators in the control room would indicate a loss of flow.
- Valves SI-651, SI-652, SI-653, and SI-654 are alarmed when not fully open with concurrent low RCS temperature (below the LTOP enable temperature).

There is no interlock to automatically close the SCS suction valves if RCS pressure increases during shutdown cooling operation. The suction valve interlock with pressurizer pressure described in Section 5.4.7.2.3.A.2 only provides a permissive open signal to allow the operator to open the valves when aligning the SCS.

Electrical power assignments are as follows (all valves can be powered from the emergency diesel generators or the alternate AC power source):

Valves SI-651 and SI-655 are on electrical train A. Valve SI-653 is on electrical train C. Valves SI-652 and SI-656 are on electrical train B. Valve SI-654 is on electrical train D. Because of the redundancy and diversity of power supplies, no postulated single failure would result in loss of shutdown cooling capability as one shutdown cooling train would remain operable.

Manually operated isolation valves SI-106 and SI-107 in the shutdown cooling pump suction lines are normally locked open and administratively controlled.

*This response was previously transmitted by Reference A-10.

Question 440.70

Describe the means provided for ECCS pump protection including instrumentation and alarms available to indicate degradation of ECCS pump performance. The staff's position is that suitable means should be provided to alert the operator promptly to possible degradation of ECCS pump performance. All instrumentation associated with monitoring the ECCS pump performance should be operable without offsite power, and should be able to detect conditions of low discharge flow.

*Response 440.70

The following instrumentation is used to determine Safety Injection and Shutdown Cooling pump performance. This instrumentation is shown in Figures 6.3.2-1A, B, and C and described in Section 6.3.5.3 of CESSAR-DC.

Channe	*	Function	Control Room Features
P-302,	305	3CS Pump Discharye Pressure	Indication
P-306, 368,	307,	SI Prmp Discharge Pressure	Indication
P-319, 339,	329, 349	SI Line Pressure	Indication Alarm (High)
P-390,	391	SI Hot Leg Injection Pressure	Indication Alarm (High)
F-302,	305	SCS Line Flow	Indication Alarm (Low)
F-306, 308,	307, 309	SI Pump Discharge Flow	Indication Alarm (Low)
F-311, 331,	321, 341	DVI Nozzle Injection Line Flow	Indicut:on
F-390,	391	SI Hot Leg Injection Flow	Indication

The normal power supplies for the above instrumentation are the 120 VAC vital I&C buses, which are powered from either the Class 1E 480 VAC buses or the 125 VDC buses through inverters (See CESSAR-DC Figure 8.3.2-2). If offsite power is lost, the Class 1E 480 VAC buses may be powered via the 4.16 KV safety buses by the emergency diesel generators (See CESSAR-DC Figure 8.3.1-1) or by the alternate AC power source (gas turbine). Testing to confirm that SI and SCS pump performance is within spacification is included in the Safety Injection System Test sections of Preoperational Tests, Section 14.2.12.1 of CESSAR-DC. In addition, Technical Specification 3.5.2 (CESSAR-DC Section 16.8.2) provides requirements for testing safety injection flowrates.

*This response was previously transmitted by Reference A-12

Question 440.86(f):

(1) Expand the table of assumed initial conditions to include the worst initial conditions for each event considering events occurring at all modes or plant operation.

*hasponse 440,86(f):

(f) The table of assumed initial conditions (of CESSAR-DC, Section 15.0) has been supanded to include the worst initial conditions for each event considering events occurring at all modes of plant operation. This expended table is presented in Section 4.0 of the Shutdown Risk Evaluation Report.

*This response and the mantioned raport section and CESSAK-DC change fulfill the commitment in Reference A-14 relevant to this RA1.

Question 440.91:

Most of the Chapter 15 and 6.3.3 (LOCA) analyses are performed based on the event being initiated at full power operation. The staff requires that C-E provide an assessment on the consequences of the transients and accidents initiated at low power levels or lower modes of plant operation such as shutdown operations. This is required to demonstrate that the analyses performed in CESSAR-DC are the bounding cases for all modes of plant operation.

*Response 440,91:

This assessment is provided in the System 80+ Shutdown Risk Exclusion Report. Section 4.0 of the report contains the assessment of CESSAR-DC Chapter 15 events and Section 5.0 contains the assessment of CESSAR-DC Section 6.3.3 events. Also see the response to RAI 440.15(1).

*The response and the mentioned report sections fulfill the commitment in Reference A-14 relevant to this RAI.

77a.wp(9075)bh

Question 440.109 (15.6.3)

Provide the results of an analysis for the potential boron dilution event during the recovering phase following a SGTR when backfill from the secondary system through the ruptured steam generator occurred.

*Response 440.109

The System 80+ Emergency Procedure Guides will include steps to prevent backfill from the secondary system through the ruptured steam generator by maintaining a positive pressure difference between the primary and secondary systems. (See Section 2.1 of the System 80+ Shutdown Risk Evaluation Report.) Therefore, boron dilution should not occur and has not been analyzed. A further note is made that backfill is not necessary to prevent overfilling of the larger System 80+ steam generator as the result of a SGTR event.

*This response was previously transmitted by Reference A-11. Changes have been made as noted by the bars in the margin.

Question 440.129 (Shutdown and Low Power Operations: USI A-45, USI A-31, GSI-99, GSI-22):

For the purpose of advanced reactor design reviews, the term shutdown risk encompasses operation when the reactor is subcritical or in transition between subcriticality and power operations up to five percent RATED THERMAL POWER; i.e., for the System 80+ design between MODE 6 and MODE 2 where the modes of operation are defined in CESSAR-DC Chapter 16, Table 1.1-1.

It should be noted that the NRC risk evaluation program is an ongoing study where the projected completion date is scheduled for the end of the 1991 calendar year. At such time, additional concerns not addressed in Generic Letter (GL) 88-17 will be emerging and additional measures may be required of the System 80+ design in order to ensure there is an adequate level of protection for the public health and safety during shutdown and low power operations. The following shutdown risk RAIs are supplemental to RAI 440.16 issued in December of 1990, and the corresponding responses should be specifically addressed in CESSAR-DC Appendix A and the associated sections of CESSAR-DC.

(GSI-99: Loss of RHR Capability in PWRs)

The System 80% resolution of GSI-99 states that midloop operation heatup analyses are performed to provide a basis for operating procedure guidelines. In order to assure that NSSS behavior is fully understood under loss of decay heat removal (DHR) events in MODE 5 and MODE 6, please provide an assessment including the following analysis of the relationship between time after shutdown, decay heat, RCS heatup rate, and boil-off rate for the System 80+ design for shutdown conditions:

- (1) Profile of total shutdown heat generation (fission and fission product decay heat) in terms of percent of RATED THERMAL POWER versus the time after shutdown from entry into MODE 2 up to 1 year following shutdown using the appropriate bounding reactor operating history parameters (i.e., cycle length, cycle burnup, atc...).
- (2) Fuel rod heatup rate (F/minute) versus time after shutdown with the steam generator nozzle dams in place (with and without the pressurizers steam vent path (SDS and PSVs) operable).
- (3) Time to reach a peak clad temperature of 2200 F versus time after shutdown (with appropriate assumption for initial rod temperature, hot rod, PLHGR, etc...)
- (4) Time to core damage versus time after shutdown with the S/G nozzle dams in place (...e. no reflux cooling), SDS vents open, and vessel head on/off. Also include the corresponding times

when bulk boiling is fully developed in the core for the above conditions.

Also provide a simplified RCS elevation reference diagram indicating minimum levels for RCS inventory for SCS use in mic oppoperation so as to preclude air binding/vortexing of the RHR pupe illustrating water levels and major systems and components including the reactor vessel, S/Gs, and pressurizer, top of core, etc...

*Response 440.129:

Analyses have been performed in order to understand the loss of decay heat removal events in Modes 5 and 6. For Mode 5 the RCS loops need not be filled and the maximum RCS temperature is 210F. For Mode 6 the RCS loops are not completely filled and the maximum RCS temperature is 135F.

Denay Heat

The earliest time for entry into Mode 5 operation, based on the maximum cooldown rate, is approximately 3.5 hours after shutdown. Long before that time the fission contribution to the total shutdown heat generation rate becomes negligible and the decay heat (fission products and actinides) constitutes the source of power from the core. Thus, this response addresses the time dependent decay heat generation rate.

The ABB-CE decay heat function is based on the model described in the ANS Standard 5.1 (Reference 440.129 -1). In addition, for decay times beyond 100,000.0 seconds the actinide portion of the decay heat is based on the actinide correction factor given in Reference 440.129 -2.

The decay heat profile as a function of time after shutdown is given in Figure 440.129-1 This profile has been used for the heatup analysis of fuel and coolant described below.

Heatup of Fuel and Coolant

The fuel rod and coolant heatup analysis has been performed with the assumption that the steam generator nozzle dams are in place (i.e. no reflux cooling). Also, the RCS pressure has been assumed to be one atmosphere. This assumption is appropriate for Mode 6 (vessel head closure studs less than fully tensioned or head removed). Also, this assumption is conservative for Mode 5 as a higher RCS pressure is associated with a higher fluid saturation temperature which results in a longer time to reach boiling after loss of shutdown cooling.

Heatup to Boiling

For Mode 5 the maximum fluid temperature cited in the Technical Specifications is 210F. This is close to the fluid saturation temperature (212F) at the assumed 1 atmosphere pressure. Hence, for Mode 5, the time to heatup to the fluid boiling point is approximately zero following loss of shutdown cooling. For Mode 6 the maximum fluid temperature cited in the Technical Specifications is 135F. For this initial condition the time to heatup to boiling has been analyzed for various assumed times of loss of shutdown cooling (referenced to the time of reactor shutdown). The assumed times of loss of shutdown cooling range from the earliest entry into shutdown cooling (3.5 hrs. after reactor shutdown) out to one year after reactor shutdown.

The heatup analysis accounted for the heatup of the core (UO2 and clad) mass and the mass of liquid in the vessel up to the bottom of the CSB nozzle (hot leg bottom). Credit was not taken for heatup of any liquid that may be in the RCS piping or for the mass of the vessel and internals up to mid-loop height. The time when bulk boiling is developed in the core as a function of time of loss of shutdown cooling is given in Table 440.129-1.

Boileff of Liquid Above Top of Active Core

For both Modes 5 and 6 net boiling will deplete the liquid inventory. All portions of the core remain at the liquid saturation temperature until the top of the core is uncovered. Thereafter, the upper portions of the core receive some cooling from the flow of steam evolved in the lower portions. For the present analysis it is conservatively assumed that the core hot spot is uncovered at the time that the liquid level drops to the top of the active core.

The time span for the liquid, between the bottom elevation of the CSB nozzle (bottom of hot leg) and the top ele ation of the active core, to boil off is given in Table 440.129-1 as a function of time of loss of shutdown cooling.

Hot Spot Heatup

Heatup of the core hotspot is assumed to occur in an adiabatic fashion. Heat input to the hotspot is consistent with the long term axial power shape which is flatter than the worst axial shape that can occur under full power conditions.

The hotspot heatup rate as a function of time of loss of shutdown cooling is given in Table 440.129-2. This heatup spans the fuel rod temperature range from 212F (liquid saturation) to 1800F. Beyond 1800F the exothermic reaction between zirconium and steam adds significant amounts of heat to the clad and fuel. This causes the clad temperature to reach a solue of 2200F in a time which is short relative to the time to rise to 1800F based only on long term decay heat. For this reason the time to reach a peak clad temperature of 2200F is taken to be consistent with the time to heatup to 1800F. Also this same time represents the time for core damage to pacer.

Table 440.129-2 also gives the time span for the fuel rod to heatup to a temperature of 1800F as a function of time of loss of shutdown cooling. This time span includes the times for fluid and core heatup and liquid boiloff as given in Table 440.129-1.

Reactor Coolant System Elevations

Figure 440.129-2 shows the various elevations within the RCS related to midloop operation and boiloff to the top of the core.

*This response fulfills the commitment in Reference A-14 relevant to this RAI.

TABLE 440.129-1

TIMES TO BUILK BOILING OF CORE LIQUID AND TO COMPLETE BOILOFF OF LIQUID ABOVE ACTIVE CORE

Time of Loss of Shutdown Cooling	Time Span to Bulk Boiling in Core	Time Span to Boiloff of Liguid Above Active	Total Time Span from Loss of	
	(Mode 6)*	(Nodes 5 and 6)	(Mode 5)	(Mode 6)
12,500. (3.5 hrs)	413.	714.	714.	1127.
50,000. (14 hrs)	558.	964 .	964.	1522.
100,000. (1.2 days)	676.	1167.	1167.	1843.
500,000. (5.8 days)	957.	1653.	1653.	2610.
1,000,000. (11.6 days)	1460.	2522.	2522.	3982.
5,000,000. (8.3 days)	3150.	5461.	5441.	8591.
10,000,000. (16.5 weeks)	4546.	7852.	7852.	12,398
31,536,000. (1 yr.;	10,256.	17,718.	17,718.	27,974

(All times in seconds)

*For Mode 5 the coolant is essentially at the boiling point for the assumed 1 atmosphere pressure.

TABLE 440.129-2

NOTSPOT HEATUP TO 1800F

(Times in seconds, rate in degrees F/min.)

Time of Loss of Shutdown Cooling	Hot Spot Heatup Rate	<u>Time Span to Reach</u> <u>1800F</u>	Total Time Span from Loss of Shutdown	
	(Modes 5 and 5)	(Nodes 5 and 6)	(Mode 5)	(Mode 61
12,500. (3.5 hrs)	138.	688.	1402.	1815.
50,000. (14 hrs)	103.	929.	1893.	2451,
100,000. (1.2 days)	85.	1125.	2292.	2968.
500,000. (5.8 days)	60.	1596.	3247.	4204.
1,000,000. (11.6 days)	39.	2431.	4953.	6413.
5,000,000. (8.3 days)	18.	5235.	10,676.	13,826
10,000,000. (16.5 weeks)	13.	7562.	7852.	19,960
31,536,000. (1 yr.)	5.6	17014.	17 718.	44,989





Question 440.130:

One of the actions provided in the System 80+ GSI-99 resolution is that containment isolation will be initiated if the RCS temperature reaches 200 F. What are the bases and analyses for determining this as the appropriate set point for containment isolation for the System 80+ design. Include consideration of personal hazards which may exist if boiling occurs in the RCS during closure operations. What alternate shutdown events, such as loss of coolant, might result in fuel damage and containment bypass prior to your assumed two hour closure period.

*Response 440.130:

The 200 F set point for containment isolation has been changed in CESSAR-DC to the initiation of containment isolation upon detection of the loss of decay heat removal capability. The methods used to detect the loss of decay heat removal capability are reviewed in Section 2.8 of the Shutdown Risk Report.

This change was based on the analyses reported on in Section 2.5 of the Shutdown Risk Evaluation Report that addressed the issue of the consequences of the loss of decay heat removal capability in Modes 5 and 6 and potential exposure of the public to radiation from an open containment.

The objective of this study was to compute the potential exposure to the public from an open containment following loss of core heat removal events in Modes 5 and 6 that could result in the coolant reaching saturation and boiling. The analyses assumed no operator intervention to recover core cooling. The analyses also assumed no credit for detection of the loss of decay heat removal before the coolant reached saturation.

Events were considered in Mode 5, at reduced inventory (loss of shutdown cooling at the earliest recommended time, 2 days, to achieve reduced inventory), Mode 5 at full inventory (a LOCA occurring within 2.4 hours of shutdown, resulting in earliest time to reach saturation), and Mode 6 in refueling configuration (loss of core cooling at the earliest recommended time, 3.5 days, to remove the head).

The time at which the containment must be closed was established by the limits on radiological exposure to the public, set by 10CFR100 as;

 For pipe breaks, at full inventory, limiting site boundary dose to < 25 rem whole body dose and < 300 rem thyroid dose at the EAB after a two hour period following the start of the release, For the reactor with loss of core cooling, limiting site boundary dose to < 2.5 rems, whole body, and <30 rem thyroid dose at the EAB after a two hour period following the start of the release.

The System 80+ containment has two personnel locks and one equipment hatch. Technical Specifications require that one of the two doors on each of the personnel locks and the equipment hatch be closed during Mode 5, at reduced inventory, and Mode 6 during refueling. The doors of the personnel locks require 10 minutes to close and seal. The equipment hatch is designed to close, without power, in an hour.

The time available for utility personnel to close the containment is determined considering radiation exposure and temperature limits within the containment for the events considered. These limits were based on;

- Limiting whole body dose to utility personnel to < 2.5 rems.
- Limiting internal dose to utility personnel to < 520 MPC per guarter,
- 3. Temperature limits inside containment based on a minimum work time of 10 minutes needed to close personnel hatches in Mode 5, at reduced inventory, and Mode 6, and 60 minutes to close the equipment hatch in Mode 5 at full inventory.

For Mode 5, at reduced inventory, the integrated two hour whole body and thyroid doses are below the acceptance limits of 2.5 rem (whole body) and 30 rem (thyroid). In addition, radiation and temperature within the containment are within limits to give utility personnel up to 30 minutes to evacuate and close personnel locks in the containment.

In Mode 5, for full inventory, the integrated two hour whole body and thyroid doses are below the acceptance limits of 25 rem (whole body) and 300 rem (thyroid). Internal radiation levels for utility personnel are within acceptable limits provided either air supplied or self-contained breathing units are used. Whole body dose is also within acceptance limits. Temperature within the containment are at levels that could limit the time available to utility personnel to close the containment equipment hatch within the assumed one hour period. Ways to decrease the time needed to close the equipment hatch and lengthen the work time within containment need to be evaluated: e.g., the use of fan coolers to reduce temperature, improved equipment hatch design to reduce closure time, review of procedures and training to reduce time to evacuate and close the containment. In Mode 6, the integrated two hour whole body and thyroid doses are below the acceptance limits of 2.5 rem (whole body) and 30 rem (thyroid). Radiation and temperature levels in the containment should provide utility personnel up to 30 minutes to evacuate and close the personnel hatches in the containment.

100

*This response and the mentioned report sections fulfill the commitment in Reference A-14 relevant to this RAI.

(uestion 440.131

Since the System 80+ no longer has RWST gravity feed capability due to the location of the In-containment RWST, discuss other methods of restoring RCS water during midloop operations via gravity feed. Also, CESSAR-DC page A-45i states that a safety injection pump and "another means would be available during cold shutdown" to add water to the RCS. What is the other means and has this system/method been accounted for in the Technical Specifications?

*Response 440.131

The response to this item is contained in Sections 2.3 and 2.4 of the Shutdown Risk Report.

*This response and the mentioned report sections fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.132

It would appear necessary to modify the proposed ECCS Technical Specifications (CESSAR-DC Chapter 16) to reflect all shutdown operational modes with respect to the RCS makeup capability including midloop operations as recommended in Generic Letter 88-17 and adopted in the System 80+ proposed resolution of GSI-99. System 80+ GSI-99 resolution states that at least one safety injection train could be available during midloop and the System 80+ LTOP proposed T/S 3.4.11 states that a maximum of one SI pump shall be operable in MODES 4, 5, and 6. Why is only a single SI train desirable for this condition? (If the limitation of 1 SI pump for MODES 4, 5, and 6 is necessary for LTOP consideration, one other alternative available means would offer greater RCS makeup capability.) An appropriate T/S for SI train(s) operability for MODE 5 and MODE 6, reflective of LTOP concerns, should be proposed. Also, the proposed T/S 3.5.4 for the IRWST should specify a minimum IRWST level and boron concentration for MODE 5 and MODE 6 so that adequate net positive suction head (NPSH) is available should SI pump(s) actuation be necessary during MODE 5 and MODE 6.

*Response 440.132

Technical Specifications to address shutdown risk concerns are discussed in Section 2.2 of the Shutdown Risk Evaluation Report.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.133:

During midloop operations, if a SI pump is actuated due to loss of DHR, what System 80+ provisions are made to accommodate the added RCS mass and resulting pressure transient for MODE 5, and MODE 6? Is the low temperature overpressure protection (LTOP) system adequately sized to handle the load?

*Response 440.133

The low temperature overpressure protection (LTOP) valves, SI-179 and SI-189 shown in CESSAR-DC Figure 6.3.2-1C, are specifically designed to limit RCS pressure in response to a mass addition or energy addition transient during Modes 5 and 6. The analysis supporting the selection of the LTOP valve set pressure and rated flow rate is described in section 5.2.2.10 of CESSAR-DC. As described in CESSAR-DC Section 5.2.2.10.2.1 a mass addition transient is defined to include the inadvertent actuation of all four safety injection pumps and one charging pump when the RCS is in a water solid condition. Such considerations are conservative and bound the situation where the RCS is not water solid (as would be the case during midloop) and only one safety injection pump is actuated.

*This response fulfills the commitment in Reference A-14 relevant to this RAI.

Question 440.134 (16.0, GSI-99):

Since the safety despressurization system is a safety grade system, propose technical specifications for SDS operability for all appropriate modes of plant operation (include full power conditions).

Response 440.134

The safety depressurization system is not relied on for mitigation of Design Basis Events and, consequently, does not require a Technical Specification for operability.

Question 440.135:

What System 80+ provisions are there to make the on-site alternate AC power supply available during shutdown operations?

*Response 440.135

Since the System 80+ on-site alternate AC (AAC) power supply is a non-safety source of power, it is not considered applicable to be included in the plant Technical Specifications.

Under the guidance of NRC Generic Letter 88-17, "LOSS OF DECAY HEAT REMOVAL," however, the AAC power supply could be used to provide power supply flexibility needed to enhance availability of safe shutdown-maintaining equipment. Procedural and administrative control will be utilized in the comprehensive shutdown/refueling procedures/scheduling to ensure that adequate power supplies, including the AAC power supply, are readily available to power equipment required to maintain safe shutdown and refueling conditions. Such administrative directives and procedures developed by the owner/operator will base their instructions on a shutdown risk analysis for applicable conditions and incidents.

*This response fulfills the commitment in Reference A-14 relevant to this RAI.

Question 440.136:

In the event a SCS or DVI component is taken out of service for maintenance and the line downstream of the respective component is drained, are there any SCS or DVI line interfaces that have the potential for inadvertently refilling these lines with deborated water?

*Response 440.136

Issues involving boron dilution due to deborated water introduced into the RCS from the Shutdown Cooling System (SCS) or Safety Injection System (SIS) during maintenance of inline components, or due to the introduction of a deborated water slug during startup or refueling are addressed in section 2.6 of the Shutdown Risk Evaluation Report.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440,137:

What analyses, if any, are performed in CESSAR-DC to address the potential for a loss of shutdown margin and consequences of a resulting power excursion caused by injection of a water slug with reduced boron concentration via an RCS loop, SCS line, or safety injection line during startup and refueling operations?

*Pesponse 440.137

Section 2.6 of the System 80+ Shutdown Risk Evaluation Report addresses the potential for a consequences of the injection of an unborated water slug into the core. The results, the evaluation indicates that considering the injection of the largest credible unborated water slug, which was determined to be via the DVI lines, the core will remain subcritical.

*This response and the mentioned report section fulfill the commitment in Reference A-11 relevant to this RAI.

Question 440.139

As mentioned in RAI 440.109, discuss the potential for boron dilution during the recovering phase following a SGTR when backfill from the secondary system through the ruptured S/G occurs. This analysis should also be provided in support of GSI-22, CESSAR-DC Section 15.4.6, etc. ...

* Response 440.139

The issue of a "potential" boron dilution resulting from a SGTR accident was addressed in the response to RAI 440.109.

*This response was previously transmitted by Reference A-11
Question 440.140:

How will plant instrumentation be designed to operate properly during shutdown operations, and what key parameters will be monitored? The answer to this question should address particularly the issues of instrument availability and appropriate ranges for shutdown measurements, including:

- a. A description for any deviations, bases and justifications of the deviations, from the Generic Letter 83-17 recommendations that each plant provide two independent RCS level indications, two independent core exit temperature measurements, the capability of continuously monitoring decay heat removal (DHR) system performance whenever a DHR system is being used for cooling the RCS, and visible and audible indications of abnormal conditions in temperature, level, and DHR system performance.
- b. Indicate what instruments identified in resolution of GSI-99 (CESSAR-DC Appendix A) and CESSAR-DC Section 5.4.7 are classified as safety related and therefore within the scope of environmental qualification and qualicy assurance criteria. For those that are not so classified, provide a description of the quality assurance program that will be used to provide reliable instruments with accurate information in the expected ranges of shutdown measurements that will be enhance operator confidence in the instrument, and the training program for operators to understand and interpret and the data provided by the instruments.

*Response 440.140

The response to this item is provided in Sections 2.1, 2.2, 2.3 and 2.8 of the Shutdown Risk Report.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.141:

Emergency operating procedures (EOP) are generally developed for use in normal plant operation assuming the availability of the plant systems. During shutdown, however, many systems will be out for maintenance and the plant is in a different configuration. What is the CE System 80+ emergency procedure guidelines for the development of enhanced EOPs to provide better operator guidance for shutdown and low power operation? Are there any additional entry conditions to address shutdown and lower power operations? What will the scope of EOP enhancement be, and will they be symptom- or event-oriented?

Response 440.141:

A comprehensive assessment of risks associated with shutdown operations was performed for the System 80- design and the results are summarized in the Shutdown Risk Report. Abnormal and accident events postulated to be initiated during shutdown operation were evaluated and means to prevent, detect and monitor them and to mitigate consequences were developed.

Based on this analyses, guidance was developed to respond to abnormal events initiated from shutdown conditions. The guidance is intended to be used for development by the plant owner/operator of appropriate procedures to govern shutdown operations. It is not intended to formally incorporate guidance for shutdown accident response into the EOPs, since the number of possible initiators and contributing occurrences is extremely large. The appropriate approach for reducing shutdown risks is to adequately plant outage activities and sequences, have sufficient means for monitoring the availability and performance of systems, and to use the diverse and redundant means incorporated in the design to mitigate consequences. These are addressed in the Shutdown Risk Report. Abnormal operating procedure or contingency actions produced by the plant owner/operator based on the plant designer's guidance are expected to be symptom oriented. Once the problem is identified, available success paths are evaluated and appropriate actions are taken to utilize available means for mitigation.

Question 440.142:

How has the plant design been enhanced to provide operators with adequate control of activities and plant conditions during shutdown? What design features reduce the likelihood of cross train maintenance errors (e.g., inadvertent isolation of the redundant train required to be operable) or the necessity of isolating multiple trains for component repairs?

*Response 440.142:

The System 80+ design features to reduce inadvertent errors and improve early operator evaluation of any failures resulting in a loss of coolant are:

- Integrated Process Status Overview (IPSO) display with critical function and system status specific to the shutdown modes;
- 2. CRT displays with system lineups and component status;
- 3. Alarms that are dependent on plant mode and equipment status.

A more detailed description of System 80+ instrumentation that addresses shutdown conditions is presented in Section 2.8 of the System 80+ Shutdown Risk Evaluation Report.

In addition to design features, System 80+ has specific technical specifications and procedural guidance that are provided to reduce the likelihood of personnel errors. A detailed description is provided in Sections 2.1 and 2.2 of the System 80+ Shutdown Risk Evaluation Report.

The System 80+ plant arrangement, including the separation of redundant divisions tends to reduce the likelihood of cross train maintenance errors and the necessity of isolating multiple trains for component repairs. There is no communication between divisions, including piping.

A significant System 80+ design enhancement to the Shutdown Cooling System (SCS) is its dedication to the Decay Heat Removal (DHR) function. The system is comprised to two identical, redundant, and totally separate trains, each capable of performing the required DHR function. The System 80+ SCS components do not double for components in other safety systems during Modes 1 through 4. In particular, the system is not credited in the safety analysis as part of the safety injection system.

Dedicated heat exchangers have been provided in each SCS train. Previous designs required the SCS heat exchanger to double as the containment spray heat exchanger during Modes 1 through 4 and part of Mode 5. The System 80+ design eliminates the need for system realignment when entering into or securing from shutdown cooling. The elimination of these manual actions decreases the opportunity for operator error and uncertainty during Modes 5 and 6 when maintenance is performed.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.143:

Describe in detail what changes have been incorporated into CESSAR-DC Chapter 16 Technical Specifications (TSs) to deal with shutdown operations. Have these changes been considered:

- a. TS conditions of risk-significant shutdown operation conditions.
- b. Establishment of minimum requirements during shutdown operations including those that control containment, DHR system flow rate, equipment operability, operation and availability, and instrumentation.

In support of this response, you should provide supporting analyses that have been performed, for System 80+ to develop the technical bases for procedures, instrumentation installation and response, equipment/NSSS interaction and response, and Limiting Conditions for Operation of TSs relative to shutdown and low power operations.

Response 440.143:

Changes to the CESSAR-DC Chapter 16 Technical Specifications required as a result of shutdown risk assessment were developed to insure availability and operability of systems to prevent and detect abnormal and accident events, monitor system performance and mitigate consequences. A new set of TS were developed for reduced RCS operations in Modes 5 and 6. Technical bases for the TS modifications and additions were developed. Details of the TS changes are provided in Section 2.2 and technical bases in the remainder 2 of the Shutdown Risk Report.

Question 440.144:

What system design features have been incorporated in the System 80+ to increase the allowable water level operating band during mid-loop operation to prevent a loss of decay heat removal capability and water flooding the steam generator and containment?

*Response 440.144:

The response to this item is provided in Section 2.8 of the Shutdown Risk Report.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.145:

High steam velocity may prevent water from falling because of counter-current flow limit. In the eve t of a loss of reactor vessel inventory, can the configuration of upper internals in the System 80+ result in high steam velocity and prevent water in the refueling cavity from reaching the core?

*Response 440,145:

The response is to the shutdown risk evaluation for the effects of PWR upper internals are discussed in section 2.10 of the Shutdown Risk Evaluation Report. The analysis indicates that a sufficiently long period of time will exist before any steam is generated and that plant operators will have sufficient time to restore shutdown cooling. This is a consequence of water in the refueling cavity reaching the core by natural circulation through the upper internals thereby cooling the core and extending the time to boiling from several minutes to over 30 minutes for the highest anticipated initial coolant temperature. For lower initial coolant temperatures several hours are required to reach boiling.

*This response and the mentioned report section fulfill the commitment in Reference A-14 relevant to this RAI.

Question 440.146:

How did the plant design incorporate the consideration that outage and maintenance activities require only minimal isolation of important systems and components?

*Response 440.146:

The System 80+[™] Design utilizes divisional separation of important systems and components to ensure the ability of the design to comply with the following recommendations of NRC Generic Letter 88-17, "LOSS OF DECAY HEAT REMOVAL:"

- "(3) Equipment
 - (a) Assure that adequate operating, operable, and/or available equipment of high reliability is provided for cooling the RCS and for avoiding a loss of RCS cooling.
 - (b) Maintain sufficient existing equipment in an operable or available status so as to mitigate loss of DHR or loss of RCS inventory should they occur. This should include at least one high pressure injection pump and one other system. The water addition rate capable of being provided by each equipment item should be at least sufficient to keep the core covered.

"Reliable equipment is equipment that can be reasonably expected to perform the intended function."

To further enhance divisional separation, the System $80+^{\text{M}}$ Design utilizes four safety injection pumps to ensure the availability of at least one high pressure injection pump, as specified in (b) above. Additionally, the SIS design allows full fl & pump testing without injecting water to the core, which minimizes RCS perturbations. The System $80+^{\text{TM}}$ Design also allows shutdown cooling system pumps to be interchangeable with containment spray pumps, thus allowing further redundancy and flexibility in shutdown operations.

Because of the complete divisional separation of important systems and equipment in the System 80+[™] Design, proper administrative control of maintenance activities together with the plant technical specifications will ensure that sufficient equipment and systems remain operational and available as required to maintain safe shutdown and refueling conditions.

*This response fulfills the commitment in Reference A-14 relevant to this RAI.

Question 440.147:

What design measures have been taken to ensure that demands on equipment during shutdown operations are consistent with equipment design operating range? You should consider the possibility of additional equipment which could be potentially useful during shutdown conditions .f its control monitoring and operating parameters were appropriately selected during design.

*Response 440.147:

The System 80+[™] Design fulfills the regulatory requirements of NRC Generic Letter 88-17, "LOSS OF DECAY HEAT REMOVAL," regarding level and temperature instrumentation compatibility with shutdown and refueling conditions and parameters. Instrument ranges will be such as to ensure sufficient accuracy for plant shutdown/refueling modes. The following statement will be added (part added underlined) to CESSAR-DC, Section 5.4.7.2.2.B, to clarify "a" on page 5.4-25:

"a. Two independent, highly reliable instruments are provided for RCS level management. These instruments function to monitor RCS level, to preclude SCS suction line vortexing and air entrainment. Level instrument ranges are optimized to encompass all reduced RCS inventory conditions."

One aim of the System $80+^{\mathbb{M}}$ Design is to utilize equipment for specific and unique purposes. An example of this is the use of the Shutdown Cooling Pumps for shutdown cooling alone, instead of using the pumps for shutdown cooling and low head safety injection. However, in instances where equipment serves both plant operating and non-operating functions, the equipment will be specified for procurement so that the full range of design requirements is met. Additionally, operating guidance will be provided to the plant owners to ensure cognizance of the operating envelope of this equipment.

In addition, the CESSAR-DC design includes the following features to facilitate continued SCS operations during reduced RCS inventory:

a. Two independent, highly reliable instrument systems (dP based and HJTC based) are provided for RCS level measurement. These instruments function to monitor RCS level, to preclude SCS suction line vortexing and air entrainment. Level instrument ranges are optimized to encompass all reduced RCS inventory conditions.

- b. Two independent thermocouples are provided to measure core exit temperature, with a large range optimized for SCS and refueling modes.
- c. Instruments which will monitor the state of SCS performance (such as pump suction pressure, vortexing monitoring equipment, flow instrumentation and/or pump motor current) are provided. These instruments function to sufficiently eliminate SCS pump loss events by monitoring the formation of vortexing and subsequent air entrainment.
- d. SCS suction isolation values are not automatically closed in the event of an RCS pressurization during shutdown cooling. This precludes a loss of shutdown cooling by automatic closure of the isolation values.
- e. The plant design provides other means of initiating alternate cooling for loss of SCS events. The plant design also ensures that a vent pathway is available to prevent pressurization.

*This response fulfills the commitment in Reference A-14 relevant to this RAI.

21.1

Question 440.148:

Will there be any maintenance activities for the System 80+ that will require isolation of IRWST pump suction inlets (or allow foreign material in the sump with potential for blockage)? If s., this would preclude operation of safety systems. What guidance can be provided to minimize this potential risk? Have TSs been provided limiting such maintenance activities?

*Response 440.148:

No maintenance activities that will require isolation of the IRWST pump suction inlets are possible because the inlets will be submerged during all modes of operation. Maintenance in the IRWST is only possible during mode 6, when IRWST inventory has been transferred to the refueling pool. During refueling operations, the Shutdown Cooling System pumps utilize the IRWST ECCS suction connections to fill the refueling cavity. Due to NPSH and vortexing considerations, the suction inlets are sufficiently submerged to protect the SCS pumps while the pumps are in operation.

While maintenance is being performed in the IRWST, the possibility exists for foreign material to accumulate in the tank. The System 80+ design includes provisions to prevent this debris from entering or blocking the ECCS suction lines. The suction lines are isolated from areas of high maintenance (i.e., away from the IRWST spargers) to decrease the possibility that debris will reach the inlets. Large vertical screens capable of filtering particles greater than 0.09 inches diameter are located within the IRWST to effectively block debris. Should maintenance be required in areas where maintenance-generated trash would not be filtered by the debris screens, a mesh "cage" that completely surrounds the suction inlets would prevent debris from entering the lines. A complete discussion of this issue is provided in section 2.9.3 of the shutdown Risk Evaluation Program Report.

Using procedural guidance provided by the plant designer in Section 2.1 of the final submittal of the System 80+ Shutdown Risk Evaluation Report, the owner-operator will develop plant specific procedures that require that maintenance-generated trash be removed from the IRWST before refilling the tank.

* This response supplements this report and together they fulfill the commitment to respond in Reference A-14.

Question 440.149:

Will there be procedures to prevent conducting those activities during lowered inventory operation that could perturb the RCS inventory or could lead to a loss of decay heat removal given a single malfunction?

*Response 440.149

The Shutdown Risk Evaluation Report examines ways in which losses of the SCS can occur in sections 2.4 and 2.12. Procedural recommendations to prevent, detect and mitigate such events are identified in Section 2.1.

* This response and the mentioned report sections fulfills the commitment to Reference A-14 relevant to this RAI.

<u>Cuestion 440.150</u>:

For shutdown operations, what design modifications have been made to minimize the likelihood and consequences of loss of AC power during outage activities?

*Response 440.150:

AC Power reliability is discussed in the "System 80+ Shutdown Risk Evaluation Report". Please refer to Sections 2.4.3.2 and 2.4.3.3 of the report for a complete response to the question.

* This response and the mentioned report sections fulfill the commitment to Reference A-14 relevant to this RAI.

Question 440.151:

Several probabilistic safety assessment studies performed for low power and shutdown conditions for the existing plants indicated that losses of decay heat removal during mid-loop operation of PWRs and operator errors dominate the risk. Your response to RAI 440.14 has qualitatively described a number of System 80+ design enhancements to decrease shutdown risk. However, to demonstrate adequate treatment of the issue, CE should conduct an analysis that provides a structured evaluation specifically identifying the System 80+ design's vulnerabilities to shutdown and low power operations. This evaluation can take the form of a quantitative probabilistic assessment or an appropriately detailed engineering assessment. It is essential that the assessment be of sufficient depth to identify remaining significant shutdown and low power vulnerabilities and identify any potential corrective actions.

*Response 440.151:

As noted in the revised response to RAI 440.14, ABB-CE is performing a detailed Shutdown Risk Assessment including both a detailed engineering assessment and a Shutdown PRA. This assessment is included in the Shutdown Risk Evaluation Report and addresses both operator actions and reduced inventory operation.

* This response and the mentioned report sections fulfill the commitment to Reference A-14 relevant to this RAI.

Question 410.54:

The safety evaluation of both the new and spent fuel storage area: includes an evaluation of the effects of dropping a fuel assembly and its handling tool from a height of two feet above the storage rack. Provide the following additional information in accordance with SRP 9.1.2. Item III.2.e quidance: Verify that the drop of any allowed lighter loads at a greater height does not result in a higher potential energy than a fuel assembly and its handling tool dropped from its normal operating elevation. Perform an evaluation of this in accordance with SRP 9.1.4 guidance.

* Response 410.54:

The spent fuel racks have been evaluated and the results show that the rack $k_{\rm ev}$, will be less than .95 under the following postulated accident conditions:

- Drop of a fuel assembly handling tool from its maximum lift height over the fuel racks.
- (2) Drop of a fuel assembly and the handling tool from their maximum lift height over the fuel racks.
- (3) Drop of other items, such as a failed fuel canister with a fuel assembly, from their maximum lift height over the fuel racks.

The response to RAI 410.54 is incomplete. Please provide the values for the maximum lifting height a sumed for each case analyzed.

*Response 410.54

Lift heights and grapple weights are not known prior to final procurement of the refueling machine used in the fuel building. System 80° spent fuel racks can absorb an impact energy of 93,100 inch-1b° without exceeding the rack K_{eff} criteria. The refueling equipment will be designed so that the maximum impact energy resulting from a dropped fuel assembly and handling tool, a dropped handling tool or any other dropped fuel handling related load from their maximum respective lift heights will not exceed the energy absorbing capacity of the fuel rack while maintaining K_{eff} criteria. In addition, the owner-operator, using procedural guidance provided by the plant designer in Section 2.1 of the final submittal of the System 80°. Shutdown Bisk Evaluation Report, will develop administrative controls to limit the size and lift height of any other non-fuel handling loads that are carried over the fuel racks such that this maximum impact energy is not exceeded.

*This response was previously transmitted by Reference A-13. Changes have been made as noted by the bars in the margin.

You have stated in Section 9.1.2.3.1.3 that one of the accidents considered in the design of the spent fuel pool storage racks is a fuel assembly and its handling tool "falling into a blocked-off fuel storage cavity." Supply additional information concerning the mechanical blocking assemblies to allow determination of the extent of penetration of a fuel assembly into a blocked cavity.

* Response 410.65

The spent fuel racks provide storage for 363 fuel assemblies in Region I (50% density) and 544 fuel assemblies in Region II (75% density). The restricted rack cells contain cell blockers which prevent the placement of a fuel assembly into the restricted cells. The racks have been analyzed based on a postulated accident condition of a fuel assembly fully inserted into a restricted cell. Taking pool boron concentrations into consideration, the results show that the rack $k_{\rm eff}$ is less than .95.

The cell blockers cannot be inadvertently removed once installed as special tooling is required to unlock and remove them from the spent fuel storage racks.

Your submittal does not provide information concerning the handling of heavy loads in the vicinity of the spent fuel pool. Provide an evaluation of the capability of the spent fuel loading pit to withstand a dropped heavy load. The evaluation should include a shipping cask drop without breach of the pit area or loss of spent fuel pool water.

*Response 410.66

The spent fuel cask laydown area is separated from the spent fuel pool by a gate and a structurally reinforced concrete wall. The gate is closed, sealed, and locked during all cask handling operations. The floor in the laydown area has been designed to withstand the impact of a shipping cask dropped from a height of 30 feet without breaching the integrity of the floor plate.

Any small water loss as a result of local damage to the laydown area wall liner cannot be communicated to the spent fuel pool due to the closed gate and the integrity of the independent spent fuel pool liner. Damage to the gate is prevented during cask handling by stops on the bridge crane rail that limit cask travel and by the recessed gate design.

Design features to address the spent fuel cask drop accident are summarized in CESSAR-DC, Section 15.7.5, Amendment H.

Provide the fuel building layout drawings which show the (1) overhead heary load paths and (2) safety-related equipment locations in the vicinity of those paths susceptible to tamage by failure of electrical interlocks, swinging of the load, or other mechanisms for causing damage.

* <u>Response 410.72</u>

The containment polar crane, the cask handling crane, and the fuel handling crane are designed to prevent the drop of a heavy load such as the reactor vessel head and the spant fuel shipping cask. In addition, predetermined load paths for major lifts (see Figures 9.1-19 and 9.1-20), operator training, and regular crane maintenance minimize the possibility of load mishandling.

Limit switches, electrical interlocks and mechanical interlocks prevent improper crane operation which might result in a fuel handling accident. This is also discussed in Section 9.1.4.2.1.1. The spent fuel cask handling hoist is restricted from movement over the new and spent fuel storage areas when the fuel racks contain fuel assemblies. The new fuel handling hoist is restricted from movement over the spent fuel storage area when the spent fuel racks contain fuel assemblies.

In accordance with the regulatory position of Regulatory Guide 1.13 and General Design criteria 61 of Appendix A to 10 CFR 50, the hoists are also restricted from passing over the spent fuel pool cooling system or ESF systems which could be damaged by dropping the load.

Set points for the hoist interlocks are set to preclude falling or tipping of the loads into the fuel storage areas.

Typically, administrative controls prepared by the Owner-Operator preclude movement of heavy loads within the containment building pool when the refueling machine contains a fuel assembly. During heavy load movement, the fuel transfer tube valve is closed to avoid water level changes in the fuel building during postulated accident conditions such as dropping the heavy load on the reactor vessel pool seal.

The first sentence of the last paragraph of Section 9.1.4.3.1 has been modified to state: "Administrative controls prepared by the Owner-Operator..."

Provide containment layout drawings showing the react vessel head storage location, the upper guide structure storage stand, the load paths from the reactor to those locations, and safety-related equipment in the vicinity of the load paths susceptible to damage by load handling accidents.

* Response 410.73

Figure 9.1-19 depicts the load paths of the reactor vessel closure head, the core support barrel (CSB), and the upper guide structure (UGS) from the reactor vessel to their respective storage areas during the refueling outage. The designated load path for each component passes over the reactor vessel flange. An analysis has shown that in the event the reactor vessel head or the internals are dropped on the reactor vessel, the core will be maintained in a coolable condition.

Typically, operating procedures prepared by the Owner-Operator control the lift height of the UGS to minimize its clearance with the pool floor and the polar crane is positioned to insure direct travel from the reactor vessel to the UGS storage area. Additionally, the ICI holding frame is installed over the seal table at the operating floor level during fuel handling. This prevents the UGS from moving over the seal table. Therefore, operating procedures preclude the UGS from being lifted above the seal table or being closer than approximately five feet to the seal table, thereby making seal table damage a remote possibility. However, under a postulated load drop on the seal table, the seal table would fail resulting in containment pool draindown to the reactor vessel flange area. Since ICI tubes are only restrained laterally and not vertically, the tubes would be bent down to the level of the reactor vessel flange. Any tube failure, therefore, would in all likelihood be at or near the reactor vessel flange level which would result in a water level within the reactor vessel similar to that prior to reactor vessel head removal. The accident condition would not be any more severe than that analyzed for the reactor vessel head drop on the reactor vessel flange.

There are no other unprotected safety-related components within the load paths of the reactor vessel head, UGS and CSB. An unprotected component is defined as a component that is not protected by the pool walls and/or operating floor.

The transfer tube valve will be closed during these handling evolutions to preclude water level changes in the fuel building.

The following sentence will be added to the end of the first paragraph of Section 9.1.4.3.1: "The Owner-Operator's operating procedures will control the load paths and height of the reactor vessel closure head, the core support barrel and the upper guide structure above the pool flcer.

77a.wp(9075)bh

Question 410.103

- a. Section 9.1.1.1 states compliance with the "intent" of Regulatory Guide 1.13 as a design basis. Considering that Regulatory Guide 1.13 pertains to spent fuel storage, explain what parts of the Guide, and to what extent, are met by the new fuel storage design.
- b. Section 9.1.1.3.3 states that "new fuel storage racks and facilities are qualified as Seismic Category I." Identify the "facilities" which are so qualified.
- c. Section 9.1.1.2 does not provide sufficient descriptive information on features illustrated in the figures. For instance, what is the function of "L" insert slots and boxes? How are the "cell blockers" attached to the structure? What is the equipment in the "new fuel inspection area"? What is their seisric classification?
- d. The new fuel storage capacity changed from 166 in Amendment E to 121 in Amendment I. What is the design basis for the storage capacity of the system?
- e. According to SRP Section 9.1.1, the design of the new fuel storage facility is acceptable if the integrated design is in accordance with, among other criteria, General Design Criteria 61 and 62 of 10 CFR 50, Appendix A. Specific criteria necessary to meet the requirements of GDC 61 and 62 are ANS 57.1 and ANS 57.3 as they relate to the prevention of criticality and to the aspects of 13diological design. Provide information on the extent of compliance of the design to ANS 57 1 and ANS 57.3.
- f. According to SRP Section 9.1.1, design calculations should show that the storage racks and the anchorages can withstand the maximum uplift forces available from the lifting devices without an increase in k_{eff}. A statement in the Safety Analysis that excessive forces cannot be applied due to the design is acceptable if justification is provided.
- g. It is the position of the Plant Systems Branch that the vaults and racks of the new fuel storage facility are to be designed to preclude damage from dropped heavy objects. Provide the design features included in the design which either preclude the fall of heavy objects onto the racks or preclude damage from a drop of the load with the maximum potential energy.
- h. Reference to Section 9.1.1.3.1.2.D in Section 9.1.1.3.1.1, regarding potential moderators such as fire extinguishing aerosols, appears to be in error. Should it be 9.1.1.3.1.2.C?

i. According to SRP Section 9.1.1, the failure of non-seismic Category I systems or structures located in the vicinity of the new fuel storage racks should not cause an increase in k_{eff} beyond the maximum allowable. Provide analysis that this condition is met or include in your application a commitment to the above condition as a design criterion.

*Response 410.103

- a. Although Regulatory Guide 1.13 pertains to the design of spent fuel storage racks, it is also used for the design of the new fuel racks. The applicable portions of the Regulatory Guide that are met are defined by Paragraphs 9.1.1.1.A and 9.1.1.1.C.
- b. The "facilities" associated with new fuel storage consist of che storage vault and the rack restraint system. The seismic category of other building components associated with handling fuel assemblies is noted in Table 3.2-1. (see response to NRC RAI 210.1)
- c. The L-insert slots are provided in the wall of the fucl rack cavity (box) to permit the L-insert to be locked to the fuel cavity by its locking tab after it has been installed. The design of the locking tab and slot is such that the L-inserts can be remotely removed from the fuel racks, if required.

The cell blockers are installed in the fuel racks before the fuel assemblies are placed in the fuel rack and before the pool is flooded. The design is basically two concentric tubes with end restraints that limit the engagement of the tubes in the rack cavity wall (to avoid protrusion into an adjacent fuel rack cavity). The tubes are collapsed, installed into the fuel rack cavity, expanded into the holes in the fuel rack cavity wall, then locked together with a captured pin. In this manner the cell blockers are positively locked to the fuel racks but can be remotely removed if desired.

*This response was previously transmitted by Reference A-13. Changes have been made as noted by the bars in the margin. The new fuel inspection area is provided for the inspection of new fuel assemblies after they have been removed from their shipping container and before they have been placed in the fuel racks. It will contain a Seismic Category II inspection device to ascertain if the fuel assemblies meet the dimensional requirements for installation into the reactor vessel. Visual inspections will also be performed to check for shipping damage and to ensure that protective wrapping material has been removed.

- d. The number of new fuel assemblies required for a 12 month refueling cycle, an 18 month refueling cycle, and a 24 month refueling cycle was evaluated. This evaluation disclosed that a 24 month cycle is controlling from the standpoint of the maximum number of new fuel assemblies required, i.e., 108. Since the rack structure is square, the minimum array to accommodate 108 fuel assemblies at a density of 50% is two 11 x 11 fuel rack modules or 121 fuel assemblies.
- e. The fuel handling equipment located in the new fuel storage area meets the requirements of ANS 57.1. The new fuel racks meet the requirements of ANS 57.3.
- f. The lifting capacity of the overhead crane that is used to remove new fuel assemblies from the new fuel rack is restricted by either adjusting the motor stail torque or using load limiting devices. (See paragraphs 9.1.1.3.1.1.D and 9.1.4.2.1.7.B).
- g. The new fuel racks are located at the opposite end of the fuel building from the spent fuel pool to eliminate the possibility of moving heavy loads near the new fuel storage area. (See response to Question i). Using procedural guidance provided by the plant designer in Section 2.1 of the final submittal of the System 80+ Shutdown Risk Evaluation Report, the owneroperator will develop administrative controls to limit the size of the load that can be carried over the new fuel racks so that the design impact energy that the racks can absorb without affecting k_{eff} will not be exceeded.
- h. The reference section of Section 9.1.1.3.1.1 that discusses potential moderators should be 9.1.1.3.1.2.B instead of 9.1.1.3.1.2.D. This change will be incorporated in the next submittal.
- i. The new fuel storage racks are located in a concrete vault at the opposite end of the fuel building from the spent fuel pool area to preclude passage of the spent fuel shipping cask overhead crane over the racks during the handling operations associated with spent fuel inspection, handling, and shipping.

This location minimizes the number of systems or structures located in the vicinity of the new fuel storage facility. All systems or structures in the vicinity will be designated as Seismic Category II to preclude their failure and entry into the new fuel storage area.

The new fuel storage racks will be designed to limit the rack $k_{\rm eff}$ to 0.98 based on the postulated accident conditions and assumptions of Paragraph 9.1.1.3.1. Using the procedural guidance provided by the plant designer in Section 2.1 of the final submittal of the System 80+ Shutdown Evaluation Report, the owner-operator will develop administrative controls to prohibit carrying any load over the loaded fuel racks whose impact energy, if dropped, will exceed the impact energy of the postulated dropped fuel handling tool or the combination of the dropped fuel handling tool or the combination of the dropped fuel handling tool and fuel assembly. The maximum impact energy shall be limited such that dropped loads do not change the $K_{\rm eff}$ of the fuel array to more than 0.98.

Question 410.104(d):

(d) According to SRP Section 9.1.2, the design of the spent fuel storage facility is acceptable if the integrated design is in accordance with General Design Criterion 63, as it relates to monitoring systems provided to detect conditions that could result in the loss of decay heat removal capabilities, to detect excessive radiation levels, and to initiate appropriate safety functions. Acceptance for meeting this criterion is based on conformance with Paragraph 5.4 of ANS 57.2. Provide the design features which satisfy GDC 63 and discuss compliance with Paragraph 5.4 of ANS 57.2.

*Response 410.104(d):

(d) The design of the spent fuel storage facility meets the intent of Paragraph 5.4 of ANS 57.2. As an example, the facility incorporates monitoring systems to verify pool water temperature to ensure adequate fuel assembly cooling, radiation detectors to determine if radiation levels exceed predetermined setpoints and alarms to notify plant personnel of abnormal conditions.

Question 410.107

- a. Evaluate the structural design features of the refueling cavity water seal that would preclude a leak or failure from occu.ring. Include the possibility of a fuel assembly or other structure dropping on the seal.
- b. If a seal failure/leak occurred, determine the time to lower a fuel assembly below the reactor vessel flange level before unacceptable dose rates from a lowered water level above spent fuel in the reactor core.
- c. For a postulated seal failure/leak, evaluate containment dose rates from a lowered level above spent fuel in reactor core.
- d. For a postulated seal failure/leak, evaluate the following parameters: makeup capacity, emergency procedures, fully loaded spent fuel pool thermal-hydraulic and dose effects including dose rate to someone trying to manually close the transfer tube valve to hydraulically isolate the spent fuel pool from the leak, time to cladding damage without operator action. Specifically provide the maximum allowable time to isolate the spent fuel pool from the transfer tube and refueling pool before there are unacceptably high dose rates in the spent fuel pool area and inadequate spent fuel pool cooling due to the level dropping below the minimum NPSH requirement above the elevation of the pool cooling suction inlet piping.

*Response 410.107

a. The refueling pool seal is designed to be installed in one piece between the reactor vessel flange and the pool floor. All fabrication welds will be liquid penetrant inspected prior to installation to ensure adequacy. After the seal assembly has been set in place, it will be permanently attached to the reactor vessel flange and to an embedment plate in the pool floor. Penetrations in the seal plate for ventilation and access to the ex-core instrumentation will be covered by bolted access hatches equipped with double seals when the pool is flooded. The annulus between the seals will be pressure tested after the hatches have been installed to determine the sealing adequacy.

*This response was previously transmitted by Reference A-13. Changes have been made as noted by the bars in the margin. The pool seal is designed to withstand OBE displacements without leakage. The pool seal is designed to limit potential leakage resulting from SSE displacements. Pool seal inspection will be required as part of the post seismic recovery procedure. The pool seal is also designed to accommodate, without leakage, relative displacements between the pool floor and the reactor vessel due to normal plant operation.

During refueling operations with the pool flooded, the heavy lift components that pass over the pool seal are the reactor vessel head, the upper guide structure with its lift rig, and the upper guide structure lift rig. Administrative controls as provided in CESSAR-DC Section 9.1.4 require that prior to transfer of heavy loads over the pool seal, the fuel transfer tube valve or the gate between the fuel building transfer system canal and the spent fuel pool shall be closed. This is done to preclude any change to the spent fuel pool water level during a postulat i heavy load drop on the pool seal which may result in containment pool draindown. In addition, administrative controls as provided in CESSAR-DC Section 9.1.4 preclude the movement f heavy loads over the pool seal if the refueling machine contains a fuel assembly. The refueling machine is designated seismic category II so that it will not fall on the pool seal during the seismic accelerations. The maximum clearance between the bottom of the refueling machine and the top of the pool seal is less than two inches to minimize the impact energy for the postulated accident condition of a dropped fuel assembly on the pool seal. The pool seal has been designed so that it will not leak as a result of this impact load.

- b. It has been determined that a 24 square inch opening in the pool seal will result in pool draindown to the reactor vessel flange level in approximately 4 hours without additional water being added to the pool. It has also been determined that present plant systems are capable of maintaining the pool water level in the event there is a 24 square inch opening in the pool seal. A fuel assembly can be lowered below the reactor vessel flange level from the fully withdrawn position within 3 minutes.
- c. With the water at the reactor vessel flange level, the radiation level at the pool seal area as a result of the fuel assemblies within the core will not be significantly greater than that with the reactor vessel head in place. The exposed CEA extension shafts will result in an increase in the overall radiation level. However, if it is necessary to do maintenance on the reactor vessel pool seal, temporary shielding can be placed around the extension chafts to reduce the radiation levels in the work area.

d. The responses to parts a, b, and c of this RAI address the concerns of this question.

Question 720.98 (Tasks 2.1.5, 2.1.8, 2.1.10, 2.1.11)

The bottom line core damage frequency estimate for the System 80+ design appears to be very low. Given these low estimates, address how the System 80+ PRA evaluated initiating events that have a lower frequency than those normally postulated, but may have more serious consequences. Examples include the question, "What does a 1E-5/yr steam generator tube rupture initiator look like and how is it handled in the PRA?", "What do various 1E-5 common cause failures look like, and how were they looked for and evaluated in the PRA?", and "What is the effect of multiple failures/equipment outages during modes other than full power?"

Response 720.98

The System 80+ PRA does not address initiating events with very low frequencies, in part because there is little information on what these initiating events might be. ABB-CE is currently updating the System 8C+ PRA to reflect some system design changes. ABB-CE is willing to address this type of initiating event in the updated PRA. However, ABB-CE does request that the NRC provide some guidance as to how to identify a reasonable set of such initiators and the level and type of analysis expected. ABB-CE has evaluated the Shutdown Risk potential of System 80+ in response to other NRC requests for additional information (RAI). In that evaluation, equipment not specified to be available in the Technical Specifications was assumed to be out of service. Multiple failures of equipment are addressed in the event trees and fault trees. The results of the shutdown risk analysis show that the risk during shutdown is equal to the risk in normal operation.

Question 722.94:

Provide a description of the design of the equipment hatch and its ability to be rapidly reclosed during shutdown, if necessary. Include a discussion of the need for AC power or any other support systems in order to effect closure, and the pressure seal arrangement, ie., whether the hatch is pressure-seating as opposed to pressure-opening (which would require full bolting to accomplish sealing under pressure). Discuss any strategies/procedures for rapidly closing major penetrations during shutdown.

*Response 722.94:

The equipment hatch is described in the "System 80+ Shutdown Risk Evaluation Report". Please refer to Section 2.5.3.2.3 of the report for a complete response to the question.

* This response and the mentioned report sections fulfill the commitment to Reference A-14 relevant to this RAI.

Question 280.1

Provide the fire protection analysis and/or interface requirements to ensure that safe shutdown can be achieved, assuming that all equipment in any one fire area will be rendered inoperable by fire and that re-entry into the fire area for repairs and operator actions is not possible with exception of the control room. For the control room, provide the fire protection analyses and/or interface requirements having an independent alternative shutdown capability that is physically and electrically independent of the control room. Also, provide the fire protection requirements for rodundant shutdown systems in the reactor containment building that will ensure, as much as practicable, that one shutdown division will be free of fire damage. Additionally, also ensure that smoke, hot gases, or the fire suppressant will not migrate into other fire areas to the extent that they could adversely affect safe shutdown capabilities, including operator actions.

*Response 280 1

A fire protection analysis of each fire area is conducted as part of the Fire Hazards Analysis. The System 80+ design basis, as stated in CESSAR-DC Section 9.5.1 (as revised in Amendment I), is to assure the ability to achieve Safe Shutdown following fire in any fire area outside of containment. This includes loss of all equipment in any given area and effects of electrical interaction which may disable equipment outside of the immediate area. The plant is arranged so that Safe Cold Shutdown can be achieved following fire in any area outside of containment without need for repairs or extraordinary operator action. Emergency shutdown from outside the control room is described in CESSAR-DC Sections 7.4.1.1.10 and 7.4.2.5. Outside of containment, redundant divisions of safety related equipment are separated by three hour fire rated boundaries. In the control complex (and most locations in the Nuclear Annex) redundant safety related divisions of protective electrical channels are separated by three hour fire rated barriers so that loss of all equipment in these areas would not affect either division of safety related equipment required to achieve cold shutdown. Inside containment Engineering Analysis conducted as part of the Fire Hazaro Analysis assure that fire at any location which can disable more than one channel of cold shutdown equipment will not affect the ability to achieve cold shutdown using equipment which would not be affected by fire at that location.

Smoke control is recognized as an important element of the Plant Fire Protection design features. In the subsphere area, containment, Fuel Pool building, Reactor Annex and Diesel Generator building the HVAC System has smoke control capability by allowing any area to be purged with 100% outside air. In the control complex, dedicated smoke exhaust fans are provided for the control room and TSC. In addition, a smoke exhaust system is provided for each channel of safety related equipment. A connection to the normal HVAC system intake is used for fresh air supply. Smoke detectors are installed in return air ducts to alarm and annunciate in the control room. Smoke dampers are arranged for remote operation from the control room.

The System 80+ design does not have connections (door or ventilation openings) between redundant safety-related divisions. This further mitigates the possibility that smoke and products of combustion of fire suppression agents will affect redundant safety-related equipment.

A3.0 <u>REFERENCES</u>

- A-1 Letter, Reactor Systems Branch RAIs, T. V. Wambach (NRC) to E. H. Kennedy (C-E), dated December 24, 1990
- A-2 Letter, Reactor Systems Branch RAIs, T. V. Wambach (NRC) to E. H. Kennedy (C-E), dated January 31, 1991
- A-3 Letter, Reactor Systems Branch RAIs, T. V. Wambach (NRC) to E. H. Kennedy (C-E), dated May 13, 1991
- A-4 Letter, Reactor Systems Branch RAIs, T. V. Wambach (NRC) to E. H. Kennedy (C-E), dated August 21, 1991
- A-5 Letter, Plant Systems Branch RAIs, T. V. Wambach (NRC) to E. H. Kennedy (C-E), dated October 10, 1991
- A-6 Letter, Risk Assessment Branch RAIs, T. V. Wambach (NRC) to E. H. Kennedy (C-E), dated October 30, 1991
- A-7 Letter LD-91-013, E. H. Kennedy (C-E) to USNRC, dated March 15, 1991.
- A-8 Letter LD-91-014, E. H. Kennedy (C-E) to USNRC, dated March 26, 1991.
- A-9 Letter LD-91-019, E. H. Kennedy (C-E) to USNRC, dated May 6, 1991.
- A-10 Letter LD-91-024, E. H. Kennedy (C-E) to USNRC, dated May 16, 1991.
- A-11 Letter LD-91-062, E. H. Kennedy (C-E) to USNRC, dated November 27, 1991.
- A-12 Letter LD-91-071, E. H. Kennedy (C-E) to USNRC, dated December 24, 1991.
- A-13 Letter LD-92-017, C. B. Brinkman (C-E) to USNRC, dated February 12, 1992.
- A-14 Letter LD-92-008, E. H. Kennedy (C-E) to USNRC, dated January 29, 1992.
- A-15 Letter LD-91-018, E. H. Kennedy (C-E) to USNRC, dated April 26, 1991.

APPENDIX B

PROPOSED REVISIONS TO THE ABB-COMBUSTION ENGINEERING STANDARD SAFETY ANALYSIS REPORT-DESIGN CERTIFICATION

B1.0 INTRODUCTION

The evaluations in the Shutdown Risk Evaluation Report and the responses provided in Appendix A to Requests for Additional Information (RAI's) have resulted in revisions to the ABB-CE CESSAR-DC. These proposed revisions are provided in this Appendix and will be incorporated into CESSAR-DC by a later Amendment.

B2.0 CESSAR-DC PROPOSED REVISIONS

The following pages from CESSAR-DC are marked with the proposed revisions.

77a.wp(9212)bh

APPENDIX B

TABLE OF CONTENTS

CF3SAR-DC SECTION	TOPIC	PAGE NO.
5.4.7.2.2	Component Description	5,4-25
15.0	Organization and Methodology	15.0-1
App. A; 099	Loss of RHR Capability in PWR's	A-451
15.0	Additional Figure 15.0-2	
15.0	Additional Table 15.0-3	
	Additional Table 15.03a	
4. Shutdown Cooling Flow

A chutdown cooling flow indicator in each train of the SCS measures shutdown cooling flow, and indicates the flowrate in the control room. A low flow alarm is provided in the control room.

In addition, the CESSAR-DC design includes the following features to facilitate continued SCS operations during reduced RCS inventory:

- a. Two independent, highly reliable instruments are provided for RCS level measurement. These instruments function to monitor RCS level, to preclude SCS suction line vortexing and air entrainment. Level instrument types and corresponding instrument ranges are optimized to encompass all reduced RCS inventory b. Two independent thermocouples are provided to measure core
 - exit temperature, with a range optimized for SCS and refueling modes.
- c. Instruments which will monitor the state of SCS performance (such as pump suction pressure, vortexing monitoring equipment, flow instrumentation and/or pump motor current) are provided. These instruments function to sufficiently eliminate SCS pump loss events by monitoring the formation of vortexing and subsequent air entrainment.
- d. SCS suction isolation valves are not automatic. ly closed in the event of an RCS pressurization during shutdown cooling. This precludes a loss of shutdown cooling by automatic closure of the isolation valves.
- e. The plant design provides other means of initiating alternate cooling for loss of SCS events. The plant design also ensures that a vent pathway is available to prevent pressurization (see Section 6.7).

Although the features described above do not exclusively describe SCS system instrumentation, they are provided in this section there they focus on precluding RHR system failures due to loss of the SCS pumps.

Piping

All SCS piping is austenitic stainless steel. All piping joints and connections are welded, except for a minimum number of flanged connections that are used to facilitate equipment maintenance or accommodate component design. SCS piping is designed to accommodate venting the SCS pumps to RCS, if necessary, with as few high points as practical.

> Amendment I December 21, 1990

1

15. ACCIDENT ANALYSES

15.0 ORGANIZATION AND METHODOLOGY

This chapter presents analytical evaluations of the Nuclear Steam Supply System (NSSS) response to postulated disturbances in process variables and to postulated malfunctions or failures of equipment. Such incidents (or events) are postulated and their consequences analyzed despite the many precautions which are taken in the design, construction, quality assurance, and plant operation to prevent their occurrence. The effects of these incidents are examined to determine their consequences and to evaluate the capability built into the plant to control or accommodate such failures and situations.

15.0.1 CLASSIFICATION OF TRANSIENTS AND ACCIDENTS

15.0.1.1 Format and Content

This charter is structured according to the format and content suggested by Reference 1 and required by Reference 26.

15.0.1.2 Event Categories

Each postulated initiating event has been assigned to one of the following categories:

- A. Increased Heat Removal by Secondary System
- B. Decreased Heat Removal by Secondary System
- C. Decreased Reactor Coolant Flow
- D. Reactivity and Power Distribution Anomalies

E. Increase in RCS Inventory

- E. Decrease in RCS Inventory
- G. Radioactive Release from a Subsystem or Commonent

The assignment of an initiating event to one of these seven categories is made according to Reference 26.

All Modes have been considered.

Amendment H August 31, 1990 Ы

15.0-1

initiation of containment isolation upon detection of the loss of RHR.

- O Plant communication systems are described in CESSAR-DC Section 9.5.2. The system normally used for plant shutdown operation and maintenance is the Intraplant Sound-Fowered Telephone System. Phone jacks connect specific areas of the plant and the control room, and the system is powered from diesel-backed power sources. The communications panel in the control room is described in CESSAR-DC Section 18.7.4.13.
- Redundant vent lines are provided between the pressurizer and the in-containment refucling water storage tank to prevent significant pressurization of the RCS if boiling occurs with the steam generator nozzle dams installed.
- The SCS suction isolation valves do not have an auto-closure interlock. As described in CESSAR-DC Section 7.6.1.1.1, the valves are interlocked to prevent them from being opened if the RCS pressure has not decreased to an acceptable value. The interlocks are redundant so that no single failure can prevent the operator from aligning the valves in at least one SCS inlet line after RCS pressure requirements have been satisfied.
- The plant design is such that both a high pressure safety injection pump and another means could be available during cold shutdown to add water to the RCS to mitigate loss of RHR capability or RCS inventory if needed.

In addition to the design features previously described, midloop operation heatup analyses are performed to provide a basis for operating procedure guidelines. These include the relationships between time after shutdown and decay heat, RCS heatup rate and boil-off rate. Guidelines are provided for reduced inventory operating and administrative procedures, including verifying availability of equipment, avoiding concurrent operations that perturb the RCS, and closing contained if the field temperature freduced RCS inventory is described in CESSAR-DC Section

Since the foregoing design features and guidelines for operations with reduced RCS inventory meet the intent of the recommendations in GL 88-17, this issue is resolved for the System 80+ Standard Design.

REFERENCES

 NUREG-0933, "A Status Report on Unresolved Safety Issues", U.S. Nuclear Regulatory Commission, December 1989.

Amendment I December 21, 1990

I



15.0-2

TABLE 15.0-3

INITIAL CONDITIONS

Core Power	% of 3800 MWt	0 to 102 ⁽¹⁾
Axial Shape Index		-0.3 < ASI < +0.3 (2)
Reactor Vessel Inlet Coolant Flow Rate	% of 445600 gpm	95 to 116 ⁽³⁾
Pressurizer Water Le 91	<pre>% distance between upper tap and lower tap</pre>	26 to 60 ⁽⁴⁾
Core Inlet Coolant Temperature		
<90% Power ≥90% Power	* F * F	543 to 565 ⁽⁵⁾ 553 to 563
Pressurizer Pressure	psia	1905 to 2375 ⁽⁵⁾
Steam Generator Water		
Low	% Wide Range ⁽⁶⁾	33.7 ^{(8),(9)} 40.7
High	% Narrow Range ⁽⁷⁾	95.0

 $\begin{pmatrix} 1 \\ 2 \end{pmatrix}$

Core power is that due to decay heat in Modes 3 through 6. ASI = (area under axial shape in lower half of core - area under axial shape in upper half of core)/(total area under axial shape). A range of values of ASI from +0.6 to +0.6 was considered for subcritical core conditions

- (3)
- (4)(5)
- (6)
- For Modes 3 through 6 see TABLE 15.0-3a. Between 0 and 100 % in Modes 4 and 5. Pressurizer is empty in Mode 6. For Modes 3 through 6 see Figure 15.0-2. Percent of distance between the wide range instrument taps. See Chapter 5 of CESSAR-DC for details.

(7) Percent of distance between the narrow range instrument taps. See Chapter 5 of CESSAR-DC for details. For steam and feedwater line breaks only. ≥ 25% SG water level in Modes 3 and 4 and in Mode 5 for loops filled and

(8)

(9) additional RCS loop/SDC division not OPERABLE. Not applicable for Mode 5 with loops not filled and Mode 6.

AdditIONAL TABLE,

TABLE 15.0-3a

Reactor Vessel Inlet Coolant Flow Rate

MODE	SYSTEM	FLOW RATE (%)
3	RCP	33 to 116
4	RCP	33 to 116
	SCS	0.94 to 2.24
5 (Loops Filled)	RCP	33 to 116
	SCS	0.94 to 2.24
5 (Loops Not Filled)	SCS	0.22 to 2.24
6	scs	0.94 to 2.24

EAdditionAL TABLES

THIS PAGE INTENTIONALLY BLANK

77c.wp(9213)mls

APPENDIX C

PROCEDURAL GUIDANCE TO SUPPORT REDUCED RCS INVENTORY OPERATIONS

REDUCED INVENTORY OPERATIONAL GUIDANCE

1.0 OBJECTIVE

Append: C provides guidance to develop reduced inventory operati _ guidelines and procedures. It contains information provided by the plant designer based on analysis and review of reduced inventory operations.

2.0 INITIAL CONDITIONS

- 2.1 The reactor is subcritical, [K_{eff} < .99] for greater than (96 hrs).</p>
- 2.2 RCS core exit temperature [< 150°F].
- 2.3 RCS leve? > El. [117'-0"].
- 2.4 Technical specification surveillance requirements for reduced inventory are met.
- 2.5 Mintenance activities are not being performed on the shutdown cooling system or the operable containment spray pump.

3.0 PRECAUTIONS

- 3.1 Reduced inventory operations duration should be minimized due to reduce risk of core uncovery due to the loss of decay heat : smoval.
- 3.2 Perturbations affecting RCS level should be minimized during reduced inventory operations to minimize the possibility of loss of decay heat removal capabilities.
- 3.3 Isolation (closure of a containment isolation value) in the non-operating loop can reduce the possibility of an inadvertent draindown to the RCS.
- 3.4 Operations directly affecting the reactor vesse! pressure boundary, i.e. In-core Instrumentation Seal Table evolutions, shall be minimized during mid-loop operations.

4.0 OPERATIONAL CUIDANCE

- 4.1 Verify RCS vent path established per Technical Specification (3.10.3).
- 4.2 Verify that the shutdown cooling/containment spray cross connection isolation valves are administratively closed.

4.3 Perform the RCS drain procedure to lower RCS level to the desired reduced inventory elevation identified below:

Scheduled Maintenance Activity	RCS_Elevation
S/G cold leg nozzle dams	[]
S/G hot leg nozzle dams RCP seal housing removal	[]
DVI nozzle 2A or 2B valve maintenance	[]

4.4 Monitor the following RCS/SDC system parameters during reduced inventory operations.

RCS	core exit temperature	[List instruments]
SDC	system flow rate	[]
RCS	boron concentration	ι 1
SDC	system temperature	[]
RCS	pressure	[]
RCS	level	[]

NOTE

Decay heat production decreases steadily with time after shutdowr. Shutdown couling system flow rate should be throttled to match heat removal requirements to reduce the possibility of vortexing.

- 4.5 Adjust SDC flow rate to match decay heat removal requirements. Minimum flow must be maintained > (3000 gpm).
- 4.6 Perform the scheduled maintenance activities while in the reduced inventory mode.

NOTE

Should reduced inventory maintenance require the installation of S/G nozzle dams, the cold leg dams shall be installed first, prior to the hot leg dams and removed last, after hot leg nozzle dam removal.

4.7 After the completion of the desired maintenance activities, restore RCS level to greater than elevation [117'-0"] per the applicable RCS make-up procedure.

77c.wp(9213)mls

5.0 ABNORMAL OPERATING CONDITIONS

5.1 Loss of shutdown cooling flow.

NOTE

There are a number of potential initiators that lead to the loss of shutdown cooling flow. The more probable initiators and the immediate actions to restore decay heat removal are discussed below.

A. Pump failure, i.e., bearing failure, motor failure, shaft breakage, etc.

Actions

- 1. Verify RCS level > minimum RCS level [E1 -xxx]
- Align the alternate 3DC division, if required, for decay heat removal.
- Start alternate division SDC system pump and verify decay heat removal capability.
- Align the containment spray pump in the failed division for operation; hold system in standby.
- 5. Determine cause of SDC pump failure and determine most reliable means (division) of heat decay removal. Realign plant systems, if required, to support decay heat removal operation. If technical specification surveiliance requirements/LCOs cannot be met, actions should be taken to raise RCS level to > elevation [117'-0"] as soon as possible.
- B. SDC flow degradation due to vortexing
 - 1. Secure the operating SDC pump.
 - Restore RCS level using one or more of the systems identified below. The methods of level restoration are specified in the order of preference:
 - a. Operable safety injection system
 - Alternate SDC system via IRWST (requires manual valve realignment)
 - c. Operable containment spray pump

- d. Charging pump alignment to the BAST or designated alternate borated water source (verify boron concentration and level before use)
- e. Safety injection tanks (verify level before use)
- Start alternate division SDC system pump and verify decay heat removal capability.
- Vent (if necessary) and verify containment spray pump operability as backup to SDC pump.
- 5. Vent failed loop SDC system pump.
- 6. Determine most reliable means (division) of decay heat removal. Realign plant systems, if required, to support decay heat removal operation. If technical specification surveillance requirements/LCOs cannot be met, actions should be taken to raise RCS level to > elevation [117'-0"]- as soon as possible.
- C. Inadvertent SDC pump suction isolation valve closure
 - Align, if necessary, and start the alternate SDC division pump to restore decay heat removal.
 - Realign the failed division flow path. If the SDC pump cannot be aligned, align the failed division containment spray pump.
 - 3. Determine most reliable means (division) of decay heat removal. Realign plant systems, if required, to support decay heat removal operation. If technical specification surveillance requirements/LOCs cannot be met, actions should be taken to raise RCS level to > elevation [117'-0"] as soon as possible.
- D. Loss of offsite power/station blackout
 - Align, if necessary, and start the alternate division SDC pump if power is available to the alternate pump.
 - 2. If no power is available, restore power immediately.
 - Verify diesel generator operation and align/start the applicable division SDC pump to restore decay heat removal.
 - Start and align the combustion turbine, if available, to the Class 1E buss.

- 5. Determine most reliable means (division) of decay heat removal. Realign plant systems, if required, to support decay heat removal operation. If technical specification surveillance requirem nts/LOCs cannot be met, actions should be taken to raise RCS level to > elevation [117'-0"] as soon as possible.
- 5.2 Loss of coolant inventory
 - 1. Stop/isolate leak.
 - 2. Secure the operating RHR pump if vortexing is indicated.

NOTE

In the event of decay heat removal interruption due to the loss of forced shutdown cooling flow, alternate methods of decay heat removal, i.e., S/Gs (if available) should be considered.

- Restore RCS inventory as described in Section 5.1.B.2.a through e.
- 4. Restore SCS flow after inventory recovery and associated venting operations are completed.

APPENDIX D

TECHNICAL SPECIFICATION MARKUPS OF LCO AND APPLICABILITY FOR SHUTDOWN OPERATIONS

APPENDIX D

TABLE OF CONTENTS

PAGE NO. (CESSAR-DC)	Tech Spec No.	Topic.
16.1-4		Definition
16.4.1	3.1.1	Shuldown Margin
16.4.2	3.1.2	Shutdown Margin
16.4.8	3.1.10	CEDMS Testing
16.4.9	3.1.X	Boron Dilution
		Alarms
16.6-1	3.3.1	RPS Instiu-
		mentation
16.6-4	3.3.1	Table 3.3.1-1
16.6.5	3.3.5	Core Protection
		Calculations
16.6-42	3.3.12	ESFAS Instru-
		mentation
16.6-43	3.3.12	Table 3, 3, 12-1
16.6.14	3.3.14	AMT
16.6-50	3.3.14	Table 3.3.14-1
		page 1 of 2
16.6-51	3.3.14	Table 3, 3, 14-1
		page 2 of 2
16.6-35	3.3.35	Table 3.3.10-1
16.6-41	3.3-41	Table 3.3.11-
16.7-4	3.4.3	RCS Proceuro
		and Temperature
		Limite
16.7-6	3.4.6	Figure 3 4 3-18
16.7-7	3.4.6	Figure 3 A 3-1B
16.7-21	3.4.11	LTOD ST DUMDE
16.8.3	3.5.3	STS Divisions -
		Shutdown
16.8.4	3.5.4	TRWGT
16.8-8	3.5.4	Figure 3 5 -1
16.11.2	3.8.2	AC Sources
		Shutdown
16.11-25	3.8.5	DC Sources
	0.0.0	Shutdown
16,11-33	388	Distribution
	5.0.0	C v c t o m c
		chutdeur
16.12-6	3.0.4	Silucuowi
	21214	Operability

Definitions 1.1

a.1 DEFINITIONS (continued)

Term

CONTAINMENT SHIELD BUILDING INTEGRITY (continued)

CONTROLLED LEAKAGE

CORE ALTERATION

DIVISION DOSE EQUIVALENT 1-131

É - AVERAGE DISINTEGRATION ENERGY

ENGINEERED SAFETY

Definition

The scaling mechanism associated with each penetration (e.g., welds, bellows, or O-rings) is OPERABLE.

See LEAKAGE-CONTROLLED.

CORE ALTERATION shall be the movement or manipulation of any fuel, sources, or reactivity control components [excluding CEAs withdrawn into upper guide structure] within the reactor vessel with the vessel head removed and fuel in the vessel. Suspension of CORE ALTERATIONS shall not preclude completion of movement of a component to a safe conservative position.

(see attached)

0.

DOSE EQUIVALENT I-131 shall be that concentration of I-131 (microcuries/gram) which alone would produce the same thyroid dose as the quantity and isotropic mixture of I-131, I-132, I-133, I-134, and I-135 actually present. The thyroid dose conversion factors used for this calculation shall be those listed in Table III of TID-14044, "Calculation of Distance Factors for Power and Test Reactor Sites".

 \tilde{E} shall be the average (weighted in proportion to the concentration of each radionuclide in the reactor coolant at the time of sampling) of the sum of the average beta and gamma energies per disintegration (in MeV) for isotopes, other than iodines, with half lives greater than 15 minutes, making up at least 95% of the total noniodine activity in the coolant.

The ENGINEERED SAFETY FEATURE RESPONSE TIME shall be that time interval from when the monitored parameter exceeds its actuation setpoint at the sensor until the ESF equipment is capable of performing its safety function (i.e., the valves travel to their required

(continued)

SYSTEM 80+

1.1-4

Amendment I December 21, 1990

16.1-4

Definitions 1.1 1.1 DEFINITIONS (continued) Term Definition ENGINEERED SAFETY positions, pump discharge pressures reach their required FEATURE RESPONSE TIME values, etc.) Tunes shall include diesel generator starting (continued) and sequence loading delays where applicable. The response time may be measured by any sequence of sequential, overlapping, or total steps such that the entire response time is measured. Estimated Critical Position (see attached) IDENTIFIED LEAKAGE See LEAKAGE-IDENTIFIED. KN-1 (see attached) LEAKAGE-CONTROLLED CONTROLLED LEAKAGE shall be the seal water flow from the reactor coclant pump seals. LEAKAGE-IDENTIFIED IDENTIFIED LEAKAGE shall be: Leakage, except CONTROLLED LEAKAGE, . into closed systems, such as pump seal or valve packing leak. that are captured and conducted to a sump or collecting tank, or Leaf age into the containment atmosphere from b. sources that are both specifically located and known either not to interfere with the operation of leakage detection systems, or not to be PRESSURE BOUNDARY LEAKAGE, or Reactor Coolant System leakage through a steam c. generator to the secondary system. LEAKAGE-PRESSURE PRESSURE BOUNDARY LEAKAGE shall be leakage, BOUNDARY except steam genera'r tube leakage, through a nonisolable fault in a Rea for Ceplant System component body, pipe wall, or vessel wail. LEAKAGE-UNIDENTIFIED UNIDENTIFIFO LEAKAGE shall be all leakage which is not IDENTIFIED LEAKAGE or CONTROLLED LEAKAGE. MID-LOOP (see attached) (continued) SYSTEM 80+ 1.1-5

> Amendment I December 21, 1990

	Definition	
1.1 DEFINITIONS (continued)	1.	
Term	Definition	
PRESSURE BOUNDARY LEARAGE	See LEAKAGE-PRESSURE BOUNDARY.	
PROCESS CONTROL PROGRAM	PROCESS CONTROL PROGRAM (PCP) shall contain the current formulas, sampling, analyses, tests, and determinations to be made to ensure that processing and packaging of solid radioactive wastes (based on demonstrated processing of actual or simulated wet solid wastes) will be accomplished in such a way as to assure compliance with 10 CFR Parts 20, 61, and 71, State regulations, burial ground requirements, and other requirements governing the disposal of solid radioactive waste.	
RATED THERMAL POWER	RATED THERMAL POWER (RTP) shall be a total reactor core heat transfer rate to the reactor coolant of 3800 MWt.	
REACTOR PROTECTION SYSTEM RESPONSE TIME	The REACTOR PROTECTION SYSTEM RESPONSE TIME shall be that time interval from when the monitored parameter exceeds its typ setpoint at the channel sensor until electrical power is interrupted to the CEA drive mechanism.	
REPORTABLE EVENT RCS REDACED INVERTORY SHUTDOWN MARGIN	A REPORTABLE EVENT shall be any of those conditions specified in 10 CFR 50.73. (See attrached) SHUTDOWN MARGIN shall be the instantaneous amount of reactivity by which the reactor is subcritical or would be subcritical from its present condition assuming: a. No change in part strength CEA position, and	
	b. All full length CEAs (shutdown and regulating) are fully inserted except for the single CEA of highest reactivity worth which is assumed to be fully withdrawn.	
	(continued)	
YSTEM 80+	1.1-7	
	Amendment I 16.1-7 December 21, 15	

	Definicion 1.
1.1 DEFINITIONS (continued)	
Term	Definition
SITE BOUNDARY	The SITE BOUNDARY shall be that line beyond which the land is neither owned, nor leased, nor controlled by the licensee.
SOFTWARE	The digital computer SOFTWARE for the Reactor Protection System shall be the program codes and their associated data, documentation and procedures.
THERMAL POWER	THERMAL POWER shall be the total reactor core
Train	heat transfer rate to the reactor coolant.
UNIDENTIFIED LEAKAGE	See LEAKAGE-UNIDENTIFIED.
UNRODDED INTEGRATED PEAKING FACTOR - F _r	The UNRODDED INTEGRATED RADIAL PEAKING FACTOR (F_r) is the ratio of the peak pin power to the average pin power in the unrodded core, excluding tilt.
UNRODDED PLANAR RADIAL PEAKING FACTOR - F _{XY}	The UNRODDED PLANAR RADIAL PEAKING FACTOR (F_{xy}) is the maximum ratio of the peak to the average power density of the individual fuel rods in any of the unrodded horizontal planes, excluding tilt

SYSTEM 80+

1.1-8

Amendment I December 21, 1990

DIVISION

ONE OR MORE TRAINS THAT SHARE A COMMON COMPONENT, e.g. AC POWER. DIVISIONS ARE THE HIGHEST LEVEL OF SEPERATION AND INDEPENDENCE.

ESTIMATED CRITICAL POSITION (ECP)

A CALCULATED SET OF REACTOR CONDITIONS AND/OR PARAMETERS THAT DEFINE A CRITICAL REACTOR STATE. (Keff=1.0)

KN-1

KN-1 IS THE K EFFECTIVE CALCULATED BY CONSIDERING THE ACTUAL CEA CONFIGURATION AND ASSUMING THAT THE FULLY OR PARTIALLY INSERTED FULL-LENGTH CEA OF THE HIGHEST INSERTED WORTH IS FULLY WITHDRAWN.

MID-LOOP

PWR CONDITION WITH FUEL IN THE REACTOR VESSEL AND LEVEL BELOW THE TOP OF THE HOT LEGS AT THEIR JUNCTION TO THE REACTOR VESSEL.

REDUCED INVENTORY

PWR CONDITION WITH FUEL IN THE REACTOR VESSEL AND LEVEL LOWER THAN THREE FEET BELOW THE REACTOR VESSEL FLANGE.

TRAIN

A SET OF SAFETY RELATED COMPONENTS THAT PERFORM A SAFETY FUNCTION. TRAINS PERFORMING REDUNDANT FUNCTIONS ARE PHYSICALLY, ELECTRICALLY AND MECHANICALLY SEPERATED TO THE EXTENT NECESSARY TO INSURE INDEPENDENT PERFORMANCE OF ITS SAFETY FUNCTION.



\$



16.4.8 3.1.10 SPECIAL TEST EXCEPTIONS - CEDMS TESTING

3.1.10 STE- CEOMS TESTINE

3.1 PEACTOR

3.1.10 Spaced Test Exception - CEOMS TESTING

LCO 3.1.10

The SHUTDOWN MARGIN requirement of Specification 3.1.1 and the GHUTDOWN MARGIN and KN-1 requirements of Specification 3.1.1.2 may be suspended for pre-startup tests to demonstrate the OPERABILITY of the control element drive mechanism system provided:

- a. No more than one CEA is withdrawn at any time.
- b. No CEA is withdrawn more than 7 inches.
- c. The K_{N-1} requirement of Specification 3.1.1. ➡ is met prior to the start of testing.
- All other operations involving positive reactivity changes are suspended during the testing.

APPLICABILITY: MODES 4 and 5.

16.4.9 3.1.11 BORON DILUTION ALARMS

3.1.11 BORDN DILUTION ALARM

3.1 REACTOR

3.1.11 BORON DILUTION ALARME

LCO 3.1. 11 BOTH STRETUP CHANNEL WIGH NEUTEON FLUX ALARMS SHALL BE OPERABLE

APPLICAEILITY: MODES 3, 4, 5, AND 6

* WITHIN I HOUR AFTER THE NEUTRON FLUX IS WITHIN THE STARTUP RANGE FOLLOWING A REACTOR SHUTDOWN.

16.6 **3.3 INSTRUMENTATION**

3.3.1 RPS INSTRUMENTATION: PRESSURIZER, CONTAINMENT, STEAM 16.6.1 GENERATOR, REACTOR COOLANT FLOW, LOSS OF LOAD

RPS Process Inst. - PZR, Containment, SGs, RC Flow, Loss of Load 3.3.1

3.3 INSTRUMENTATION

3.3.1 Reactor Protective System (RPS) Instrumentation: Pressurizer, Containment, Steam Generator, Reactor Coolant Flow, Loss of Load

The RPS Instrumentation channels and the associated operating bypasses, shall be LCO 3.3.1 OPERABLE with the Limiting Safety System Settings of Table 3.3.1-1. and the second s

AS SHOWN IN TABLE 3.3.1-1.

MODES I AND Z:

APPLICABILITY

SYSTEM 80+

			RPS Process Inst PZR, Containment, SC RC Flow, Loss of Lo 3.3
	REACTOR P	TABLE 3.3	.1-1 ISTRUMENTATION
	LIMITIN	G SAFETY SYS	TEM SETTINGS
	and and a second se	ten den die die Arte Areko alle die Ar	
FL	INCTIONAL UNIT	TOTAL NUMBER OF CHANNELS	MINIMUM APRICALE CHANNELS OPERABLE ALLOWABLE VALUE
١.	Pressurizer Pressure - High	4	3 [1, 2 [∞]] ≤ []psia
1	Pressurizer Pressure - Low	4	3 1 j 2 1 ≥ [] psia
	Containment Pressure - High	4	3 1_1 $2 \leq [] psig$
	Stea:n Generator Pressure - Low	4/SG	3/SG 1, 2, 3 [™] ≥ [] psia
	Steam Generator Level - Low	4/SG	3/SG 1, Z 1/2 ()%
	Steam Generator Level - High	4/SG	3/SG - 1,2 \$ 5 [.]%
	Reactor Coolant Flow - Low	4/SG	3/SG 1, 2, 3 , Kamp: ≤ [] psid/sec > Floor: ≥ [] psid
	The Pressurizer Pressure - Low	NOTES	manually decreased as cressurizer

1. The Pressu pressure is maintained at ≤ 400 psi. Below 400 psia, the trip may be manually bypassed. The bypass shall be automatically removed whenever pressurizer pressure is ≥ 500 psia, and the setpoint shall be automatically incre-sed as pressurizer pressure is increased.

- 2. The Steam Generator Pressure Low trip setpoint may be manually decreased as steam generator pressure is reduced, provided the margin between steam generator pressure and the setpoint is maintained at ≤ 200 psi. The setpoint shall be increased automatically as steam generator pressure is increased.
- 3. The Reactor Coolant Flow Low trip setpoints may be manually adjusted when THERMAL POWEK is $< 10^{-4}$ %.
- The Steam Generator Level-Low trip setpoint varies with reactor power and is rate limited with a p. set low power-value

F WITH THE RTCBS IN THE CLOSED POSITION, THE CEA DRIVE SYSTEM CAPABLE OF CEA WITHDRAWAL, AND FUEL IN THE REACTUR VESEL SYSTEM-804

16.6.5 3.3.5 CORE PROTECTION CALCULATORS

CFCs 3.2.5 3.3 INSTRUMENTATION 3....5 Core Protection Calculators (CPCs) LCO 3.3.5 Four Core Protection Calculator (CPC) channels, and the associated operating bypass, shall be OPERABLE. APPLICABILITY: MODES 1 3, 4 " AND 5 " -NOTES-1. The DNBR and LPD trips may be manually bype sed when THERMAL POWER is less than 10⁻⁴% RTP. The bypass shall be automatically removed when THERMAL POWER is greater than or equal to 10⁻⁴% RTP. 2. During special testing pursuant to LCO 3.1.9 and 3.4.16, the trips may be manually bypassed below 5% RTP. The bypass shall be automatically removed when THERMAL POWER is \geq 5% RTP. * WITH THE RTCBS IN THE CLOSED POSITION, THE CER DEIVE SYSTEM ? CAPABLE OF CER INITHDRAWAL, AND FUEL IN THE REACTOR VESSEL. The second state of the se

"Museum un a state and the many of the state of the

16.6.12 3.3.12 ENGINEERED SAFETY FEATURES ACTUATION SYSTEM (ESFAS) INSTRUMENTATION - MANUAL ACTUATION

ESFAS Instrumentation - Manual Actuation 3.3.12

3.3 INSTRUMENTATION

3.3.12 Engineered Safety Features Actuation System (ESFAS) Instrumentation - Manual Actuation

LCO 3.3.12 The ESFAS Manual Actuation Channels of Table 3.3.12-1 shall be OPERABLE.

APPLICABILITY: As specified in Table 3.3.12-1.

ACTIONS

CONDITION		RE	QUIRED ACTION	COMPLETION TIME	
Α.	Number of channels OPERABLE one less than the Total Number of Channels	A.1 <u>CR</u>	Restore the inopenable channel to OPERABLE status.	48 hours	
	of Table 5.5.12-1.	A.2.1	Be in MODE 3.	54 hours	
			AND		
		A.2.2	Be in MODE 4.	60 hours	
		A.2.2	Be in MODE 5.	84 hours	

SURVEILLANCE REQUIREMENTS

SURVEILLANCE	FREQUENCY
SR 3.3.12.1 Perform a CHANNEL FUNCTIONAL TEST.	[18 months]

CROSS-REFERENCES - None

SYSTEM 80+

3.3-42

Amendment I December 21, 1990

	ESFAS	Instrumentation -	- Manual Actuati 3.3.
T <u>ESFAS INSTRUMEN</u>	TABLE 3.3.12-1 TATION - MANUAL	ACTUATION	
FUNCTIONAL UNIT	TOTAL NUMBER OF CHANNELS	MINIMUM CHANNELS OPERABLE	APPLICABLE MODES
1. Sufety Injection (SIAS)	2 sets of 2	2 sets of 2	1, 2, 3, 4.
2. Containment Spray (CSAS)	2 sets of 2	2 sets of 2	1.2.3,4
3. Containment Isolation	2 sets of 2	2 sets of 2	1, 2; 3, AP
Main Steam bine Isolation (MSIS)	2 sets of 2	2 sets of 2	1, 2, 3, 4
5 Gomainment Gooling (CCAS)	2-sets of 2	-2 acts of 2-	
5 Emergency Feudwater (ERAS)	2 sets of 2 per SG	2 sets of 2 per SG	1.2.3.4
7. Safety Depressurization (SDAS)		2 sols of 2	-1.2.3.4

25AS 1

SYSTEM 80+

3.3-43

Amendment I December 21, 1990

16.6-43

16.6.14 3.3.14 ACCIDENT MONITORING INSTRUMENTATION

	AMI 3.3.14
3.3 INSTRUMENTATION	
3.3.14 Accident Monitoring Instrumentation (AMI)	
LCO 3.3.14 The Accident Monitoring Instrumentation specified in Table 3.3.14- OPERABLE.	l sha'i be
APPLICABILITY: MODES 1, 2 and 3 AND AS SPECIFIED IN TABLE 3.3.1	4-1

			A 3.3
	TABLE 3.3.14-1		
	(Sheet 1 of 2)		
	ACCIDENT MONITORING INSTRUME	NTATION	
INS	STRUMENT	TOTAL NUMBER OF CHANNELS	MINIMUM CHANNELS OPERABLE
1.	Containment Pressure (WR) (NR)	2 4	1 [2]
2.	Reactor Coolant Outlet Temperature (T-hot) Wide Range (WR)	4	1
З,	Reactor Coolant Inlet Temperature (T-cold) - WR	4	1
4.	Reactor Coolant Pressure - WR	2	1
5.	Pressurizer Water Level	2	1
6.	Steam Line Pressure	2/SG	1/SG
7.	Steam Generator (SG) Water Level - WR	2/SG	1/SG
8.	In containment Refueling Water Storage Tank Wat., Level	2	1
9.	Emergency Feedwater Flow Rate	2/SG	1
tt.,	Reactor Coolant System Subcooled Margin Monitor	2	1
11.	Pressurizer Safety Valve Status	1/valve	1/valve
12.	Reactor Vessel Water Level Narrow Range	2	1
13.	Core Exit Thermocouples	15/core quadrant	2/core quadrant
14.	Emergency Feedwater Storage Tank Water Level	2/tank	1/tank
15.	Wide Range Neutron Flux	2	1
16.	Reactivity Cavity Level	2	1
17.	Containment Area Radiation	2	1
18.	Containment Hydrogen Concentration	2	1
19.	Containment Isolation Valve Position	1 pair/valve	1 pair/valve

Amendment I December 21, 1990

Constantial and a second strain and an inclusion of the second strain second strain second strain second strain		3.3
TABLE 3.3.14-1 (Cont'd)		
(Sheet 2 of 2)		
ACCIDENT MONITORING INSTRUM	ENTATION	
INSTRUMENT	TOTAL NUMBER OF CHANNELS	MINIMUM CHANNELS OPERABLE
20. RCS Radiation Level	2	1
21. Containment Spray Flow	1	1
22. Containment Atmosphere Temperature	2	1
23. Safety Injection Flow	4	2
24. Safety Injection Tank Level	1/tank	1/tank
25. Safety Injection Tank Pressure	1/tank	1/tank
26. Shutdown Cooling Flow	2	2
27. Shutdown Cooling Hx Outlet Temperature	2	2
28. Steam Generator Safety Valve and (ADV) Position	1 pair/valve	1 pair/valve
29. Emergency Ventilation Damper Position	1 pair/damper	1 pair/damper
30. Component Cooling Water Flow to ESF System	1	1
31. Component Cooling Water Temperature to ESF System	1	1
32. DC Bus Voltage	2	2
33. Diesel Generator Voltage	2	2
34. Diesel Generator Current	2	2
35. Diesel Generator Status	2	2
36. 4.16 kV Switchgear Voltage	2	2
37. 480 V Switchgear Voltage	2	2
18. 4.16 kV Switchgear Current	[]	[]
19. 480 V Switchgear Current	[]	[]

Amendment I December 21, 1990

16.6-51

ROD TO TABLE 3.3.14-1

40. SE LIQUID BLOWDOWN RADIATION MONITOR

41. MAIN STEAM LINE RADIATION MONITOR

42. STEAM JET ALL EJECTOR ZADIATION MONITOR

43. STACK BADIATION MONITOR

e

.5

* ALCO RECOTECO TO BE CALENDE TO MODEE A NO OF

	Republication of			Andrah Ala Italia ang barra			3.3
			TAE	BLE 3.3.10-1			
			(5)	need 1 of 2)			
			ESFAS INS	TFUMENTATI	ON		
FU	NCTIO	NAL UNIT	TOTAL NUMBER OF CHANNELS	MINIMUM CHANNELS OPERABLE	APPLICABLE MODES	ALL	OWABLI ALUES
1.	SLAS					and a state of the second s	
	8.	Containment Pressure - High	4	3	1, 2, 3, 4	ĺ] psig
	b.	Pressurizer Pressure - Low	4	3	1, 2, 3, 4	ſ] psia
2.	CSAS				\sim		
	A .	Containment Pressure - High-High	4	3	1, 2, 3, 4) (] psig
3.	CLAS						
	A.	Containment Pressure - High	4	3	1, 2, 3	l] psig
	b.	Pressurizer Pressure - Low	4	3	1, 2, 3	(] psia
4.	MSIS						
	۵.	Steam Generator (SG) Pressure	4/SG	3/SG	1, 2, 3, 4	l] psia
	b.	- Low Containment Pressure -	4	3	1, 2, 3, 4	I] psig
	c.	High Steam Generator Level - High	4/SG	3/SG	1, 2, 3, 4	{] %

Amendment I December 21, 1990

16.6-35

		TABL	E 3.3.11-1				
		ESFA	S LOGIC				
FUI	NCTIO	NAL UNIT	TOTAL NUMBER OF CHANNELS	MINIMUM CHANNELS OPERABLE	APPLICABLE MODES		
1.	Safe	ty Injection Actuation Signal (SIAS)		And a state of the second	C Manuf Staff Sciences and an other states and		
	8.	Local Coinc dence Logic	4	4	1. 2. 3. 4		
	b	Initiation Logic	4	4	1, 2, 3, 4		
	¢.	Actuation Logic	4	4	1, 2, 3, 4		
2.	Cont	tainment Spray Actuation Signal					
	(CS/	AS)	-4	4	1224		
	8.	Local Coincidence Logic	4	4	1234		
	b.	Initiation Logic	4	4	1, 2, 3, 4		
	c.	Actuation Logic	J.				
3	Cont	ainment Isolation Actuation 6 anal	W				
1	(CIA	(c) and the solution were and the second	1.				
	8	Local Coincidence Logic		4	1, 2, 3		
	b.	Initiation Logic	2	4	1, 2, 3, 4		
	с.	Actuation Logic		4	1, 4, 3, 4		
4.	Mair	a Steam Isolation Signal (MSIS)					
	a. h	Local Conscidence Logic	4	4	1, 2, 3, 4		
	0.	Actuation Logic	4	4	1, 2, 3, 4		
		Newarion Logic	4	2	1, 2, 3, 4		
5.	Eme	rgency Feedwater Actuation					
	Sign	al - 1 (EFAS-1)					
	8.	Local Coincidence Logic	4/SG	4/SG	1, 2, 3		
	ð.	Initiation Logic	4/SG	4/SG	1, 2, 3, 4		
	C.	Actuation Logic	4/SG	4/SG	1, 2, 3, 4		
б.	Eme	rgency Feedwater Actuation					
	Signal - 2 (EFAS-2)						
	8.	Local Coincidence Logic	4/SG	4/SG	1, 2, 3		
	ь.	Initiation Logic	4/SG	4/SG	1, 2, 3, 4		
	C.	Actuation Logic	4/SG	4/SG	1, 2, 3, 4		

16.6-41

Amendment I December 21, 1990

16.7.3 3.4.3 RESSURE AND TEMPERATURE (P/T) LIMITS

				RCS P/T Limit 3.4.
3.4	REACTOR COOLANT SYS	TEM (R	CS)	
3.4.	3 RCS Pressure and Tempera	ature (P/	D. Limits	
LCO	3.4.3 The combination of rates shall be maint	f RCS pr aineo wi	essure, RCS temperature and RC thin the limits specified in Figur	C5 heatup and cooldown re 3.4.3-1.
APP	LICABILITY: At all times			
	CONDITION	REQUIRED ACTION		COMPLETION TIME
All con Cor	NOTE Required Actions must be apleted whenever this adition is entered.			
А.	Requirements of the LCO not met.	A.1	Restore parameter(s) to within limits.	30 minutes
		A.2	Determine RCS is acceptable for continued operation.	72 hours
B.	Required Actions and associated Completion Times not met.	B.1	Be in MODE 3.	6 hours
		B.2	Be in MODE 5 with RCS	36 bours

SYSTEM 80+

3.4-4

Amendment I December 21, 1990


10



16.7-7

16.7.11 3.4.11 LOW TEMPERATURE OVERPRESSURE PROTECTION (LTOP) SYSTEM

		LTOP 3.4.11
3.4 REACTO	R CO	OLANT SYSTEM (RC?)
3.4.11 Low T	emper	sture Overpressure Protection (LTOP) System
LCO 3.4.11	LTOP	55 - Lewis System shall be OPERABLE with a maximum of oes before injection pump-
		Two SCS Relief Valves with lift settings \leq [550] psig and associated block valves open, or
,	b.	The RCS depressurized with both divisions of Rapid Depressurization valves open.
APPLICABILI	TY:	MODE 4, with any RCS cold leg temperature $\leq [317^*F]$, MODE 5, MODE 6, with the reactor vessel head on
		LCO 3.0.4 is not applicable.

- SIS Divisions Shutdown 3.5.3 3.5 SAFETY INJECTION SYSTEM (SIS) 3.5.3 <u>SIS Divisions - Shutdown</u> LCO 3.5.3 Two Safety Injection (SI) divisions, shall be OPERABLE. APPLICABILITY: (MODES 4, S, AND 6 MODES 4, S, AND 6 WITH RCS Level > R, Vessel Glange elevation [120'-0"]
- 16.8.3 3.5.3 SIS DIVISIONS SHUTDOWN

-79

16.8.4 3.5.4 IN-CONTAINMENT REFUELING WATEF STORAGE TANK (IRWST)

In-containmest Refueling Water Storage Tank 3.5.4

3.5 SAFETY INJECTION SYSTEM (SIS)

3.5.4 In-containment Refueling Water Storage Tank (IRWST)

LCO 3.5.4 The IRWST shall be OPERABLE.

APPLICABILITY: "HODES MODES EXCEPT MODES WITH

3



Amendment I December 21, 1990

16.8-8

.

16.11.2 3.8.2 AC SOURCES - SHUTDOWN

AC Sources - Shutdown 3.8.2 3.8 ELECTRICAL POWER SYSTEMS 3.8.2 <u>AC Sources - Shutdown</u> LCO 3.8.2 The following AC Electrical Power Sources shall be OPERABLE: a. One circuit between the offsite transmission network and the onsite Class IE distribution system, and of each division, and b. One diesel generator. APPLICABILITY: MODES 5 and 6, when bendling irrediated fuel.

16.11.5 3.8.5 DC SOURCES - SHUTDOWN

Acres

DC Sources - Shutdown 3.8.5 3.8 ELECTRICAL POWER SYSTEMS 3.8.5 <u>DC Sources - Shutdown</u> LCO 3.8.5 <u>DC Power Source Division & sectDivision & shall be OPERABLE</u>. CpinCiding with the associated operable APPLICABILITY: MODES 5, and 6, Wher bandling irrediated fuel.

ACTIONS

.

4

CONDITION			REQUIRED ACTION	COMPLETION TIME
Α.	Required DC Power Source (Division) inoperable.	A.I AND	Suspend CORE ALTERATIONS.	Immediately
		A.2	Suspend handling of irradiated fuel.	Immediately
		AND		
		A.3	[Initiate action to] suspend operations with a potential for draining the reactor vessel.	Immediately
		AND		37.614.146.1
		A.4	Suspend operations involving positive reactivity additions.	Immediately
		AND		
		A.5	Initiate action to restore required DC Power Source [Division] to OPERABLE status.	Immediately

16.11-25

Amendment I December 21, 1950 0

16.11.8 3.8.8 DISTRIBUTION SYSTEMS - SHUTDOWN

3.8 3.8.8 1.CO 3. APP, JO	ELECTRICAL POWER Distribution Systems - 3 5.8 The Division 1 or I CABILITY: MODE When	SYSTE <u>Enuidown</u> Division 2 ES 5 and handling	MS 2 Power Distribution System Shall 95500 6. irradiated fuel. 9 en e	be OPERABLE. Lated with th ble diesel rates
	CONDITION		REQUIRED ACTION	COMPLETION TIME
Α.	Required [Division] inoperable.	A.1	Suspend CORE ALTERATIONS.	Immediately
		A.2	Suspend handling of irradiated fuel.	Immediately
		AND		
		A.3	operations with a potential for draining the reactor vessel.	Immediately
		AND		
		A.4	Suspend operations involving positive reactivity additions.	Immediately
		AND		
		A.5	Initiate action to restore required (Division) to OPERABLE status.	Immediately
SYSTE	M 80+		3.8-33	

16.11-33

Amendment I December 21, 1990

16.12.4 3.9.4 SHUTDOWN COOLING (SDC) AND COOLANT CIRCULTTION - HIGH WATER LEVEL

颾

SDC - High Water Level 3.9.4 3.9 REFUELING OFERATIONS 3.9.4 Shutdown Cooling (SDC) and Coolant Circulation - High Water Level at least ove sivision shall be Twodivisions LCO 3.9.4 One SDC loop shall be OPERABLE and in operation. -NOTE-The required SDC loop may be removed from operation for s one hour per [2] hour period, provided: No operations are permitted that would cause dilution of the RCS boron ٤. concentration. APPLICABILITY: MODE 6 with the water level ≥ 24 feet above top of the reactor vessel flange. ACTIONS CONDITION REQUIRED ACTION COMPLETION TIME Α. SDC loop A.1 Suspend operations involving Immediately requirements not met. a reduction in reactor coolant boron concentration. AND A.2 Suspend operations involving Immediately an increase in reactor decay heat load. AND A.3 Initiate actions to satisfy SDC 15 minutes loop requirements.

SYSTEM 80+

.

3.9-6

Amendment I December 21, 1990

10

16.12-6

THIS PAGE INTENTIONALLY BLANK

.

.

...

APPENDIX E

TECHNICAL SPECIFICATIONS FOR REDUCED RCS INVENTORY OPERATIONS

Reduced RCS Inventory Operations - RTCB's 3.10.1

3.10 REDUCED RCS INVENTORY OPERATIONS

3.10.1 Reactor Trip Circuit Breakers

LCO 3.10.1 The RTCB's shall be open

APPLICABILITY: MODE 5 REDUCED RCS INVENTORY

ACTIONS

CONDITION	REQUIRED ACTION	COMPLETION TIME
RTCB's Closed	Open RTCB's	[Immediately]
PROPERTY OF COMPANY AND ADDRESS OF ADDRESS OF COMPANY ADDRESS OF		

Reduced RCS Inventory Operations - RTCB's 3.10.1

SURVEILLANCE REQUIREMENTS

		SURVEILLANCE	FREQUENCY
SR 3.10.1.1	Verify	RTCB'S OPEN	[12 hours]

SYSTEM 80+

3.10-2

3.10 REDUCED RSC INVENTORY OPERATIONS

3.10.2 Reduced RCS Inventory Operations - Instrumentation

LCO 3.10.2 The following reactor coolant system instrumentation shall be operable.

- a. Two independent means of monitoring RCS level indication; one narrow range and one wide range instrument. And,
- b. I wo independent and diverse means of monitoring RCS temperature. And,
- c. Two independent indications available to monitor SCS performance in the loop on service for decay heat removal.

AFPLICABILITY: MODE 5 REDUCED RCS INVENTORY and MODE 6 REDUCED RCS INVENTORY

ACTIONS

CONDITION	REQ	UIRED ACTION	COMPLETION TIME	
A. WR RCS Level Indication INOPERABLE	A.1.	Initiate action to restore Instrument to OPERABLE Status. And,	[Immediately]	
	A. B.	Monitor RCS Temp Monitor SCS Performance	[Every 30 minutes] [Every 30 minutes]	
	C.	Monitor NR Level	[Every 30 minutes]	

SYSTEM 80+

Reduced RCS Inventory Operations - Instrumentation 3.10.2 (Continued)

ACTIONS (Continued)

CONDITION	REQUIRED ACTION	COMPLETION TIME
B. NR RCS Level Indication INOPERABLE	B.1. Initiate Action to restore Instrument to OPERABLE status. And,	[Immediately]
	A. Monitor RCS Temp B. Monitor SCS Per- formance	[Every 30 minutes] [Every 30 minutes]
	C. Monitor WR level	[Every 10 minutes]
	AND	[Immediately]
	B.2. Initiate action to restore RCS level to >[EL-117'0"]	
C. One RCS Temperature Indication INOPERABLE	C.1. Initiate action to restore instrument indication to OPERABLE status. And,	[Immediately]
	A. Monitor RCS Level B. Monitor SCS Perfor-	[Every 30 matutes]
	mance	(with) so minutes
	C. Monitor OPERABLE Temperature Instru- ment	[Every 30 minutes]

Reduced RCS Inventory Operations - Instrumentation 3.10.2 (Continued)

ACTIONS (Continued)

CONDITION	REQUIRED ACTION	COMPLETION TIME
D. Two RCS Temperature Indication INOPERABLE	D.1. Initiate Action to restore one instru- ment to OPERABLE status. And,	[6 hours]
	A. Monitor RCS level B. Monitor SCS perfor-	[Every 10 minutes] [Every 10 minutes]
	AND	[Immediately]
	D.2. Initiate action to restore RCS level to >[EL-11 ^m 0 ⁿ]	
E. SCS Performance Indications INOPERABLE	E.1. Initiate action to restore instrument to OPERABLE status.	[Immediately]
	AND	
	A. Monitor RCS Temp B. Monitor RCS Level	[Every 10 minutes] [Every 10 minutes]
	OR	
	E.2. Align decay heat removal systems to the alternate loop.	[2 hours]

Reduced RCS Inventory Operations - Instrumentation 3.10.2

SURVEILLANCE REQUIREMENTS

-	FREQUENCY	
SR 3.10.2.1	Perform a CHANNEL CHECK of RCS Level; One WR and One NR	[6 hours]
SR 3.10.2.2	Perform a CHANNEL CHECK of RCS Temperature	[6 hours]
SR 3.10.2.3	Perform a CHANNEL CHECK of SCS performance in the loop removing decay heat.	[6 hours]
SR 3.10.2.4	Perform a CHANNEL CALIBRATION of RCS level, temperature and SCS performance.	[60 days]

Reduced RCS Inventory Operations - Vent Paths 3.10.3

3.10 REDUCED RCS INVENTORY OPERATIONS

3.10.3 Peduced RCS Inventory Operations - Vent Paths

LCO 3.10.3 A RCS Vent Path of ≥[Pressurizer Manway Removal] is established and maintained

APPLICABILITY: MODE 5 REDUCED RCS INVENTORY and MODE 6 REDUCED RCS INVENTORY WITH RX VESSEL HEAD IN PLACE*

ACTIONS

CONDITION	REQUIRED ACTION	COMPLETION TIME	
A. RCS Vent Path Isolated	A.1. Initiate action to restore Vent Path.	[Immediately]	
	AND		
	A.2. Complete restoration of vent path.	[6 hours]	
	AND		
	A.3. Monitor RCS temperature, level and SCS performance.	[Hourly]	

* 1 or More bolts tensioned

SYSTEM 80+

Reduced RCS Inventory Operations - "."ent Patha 3.10.3

0

٠

ACTIONS (Continued)

CONDITION	REQUIRED ACTION	COMPLETION TIME	
B. Required Action and completion time not met.	B.1 Restore RCS Level to > [E1 117'0*]	[6 hours]	

SURVEILLANCE REQUIREMENTS

	FREQUENCY	
SR 3.10,3.1	Verify Pressurizer Manway is removed and unobstructed or an equivalent vent path is established.	[12 hours]

3.10-8

3.10 REDUCED RCS INVENTORY OPERATIONS

3.10.4 Reduced RCS Inventory Operations - Heat Removal

LCO 3.10.4

- a. Two stut Down Cooling (SDC) divisions shall be OPERABLE, and at least one division shall be in operation.
- b. The Containment Spray Pump shall be CPERABLE in the operating division.

APPLICABILITY: MODE 5 REDUCED RCS INVENTORY and MODE 6 REDUCED RCS INVENTORY

ACTIONS

CONDITION		REQUIRED ACTION		COMPLETION TIME
A.	One SDC division INOPERAGLE.	A.1	Restore division to OPERABLE status,	[Immediatey]
B.	Required Action and associated Completion Time not met.	B.1	Raise RCS level to > [EL-117'0'].	[6 hours]

9

a were

ACTIONS (Continued)

CONDITION	REQUIRED ACTION	COMPLETION TEXE		
C. No SDC division in operation.	C.1 Suspend all operations involving reduction in RCS boron concentration.	[Immediately]		
	AND			
	C.2 Initiate action to restore one SDC division to OPERABLE status and place in operation.	[Immediately]		
	AND			
	C.3 Initiate action to raise RCS level to > [EL-117'0"]	[Immediately]		

ø

3.10-10

.

al annual second	A 200 B	Sec. 1	S			
A (11	11.24	1342	S. S. Carrow		10 1 10 miles - 10	n
PAL 1	R	N	1	111 3 8		1.1
5 B. St. B.	4.000	7.50	1. 1. 1. 1. 1. 1.	121-22	1.42.5	8.2

•

 \mathcal{J}_{f}

CONDITION	REQUIRED ACTION	COMPLETION TIME
D. Containment Spray Pump in operating division INOPERABLE	D.1 Initiate action to place the alternate division in operation if the containment spray pump in that division is OPERABLE.	[6 hours]
	AND	
	D.2 Monitor SCS performance.	[Every 30 minutes]
	AND	
	D.3 Restore INOPERABLE Containment Spray Pump.	[48 hours]
E. Required Action and Completion time of Item D.3 not met.	S.1 Raise RCS Level > [EL-117'0*]	[6 hours]

SURVEILLANCE REOUIREMENTS

-	SURVEILLANCE	FREQUENCY
SR 3.10.4.1	Verify at least one SDC division operating	[12 hours]
SR 3.10.4.2	V Ty correct breaker alignment and indicated power available to the SDC pump that is not in operation and the OPERABLE Containment Spray Pump.	[24 hours]

Reduced RCS Inventory Operations - Contai ment Integrity 3.10.5

3.10 REDUCED RCS INVENTORY OPERATIONS

3.10.5 <u>Reduced RCS Inventory Operations - Containment Integrity</u>

LCO 3.10.5 The containment building penetrations shall be in the following status:

a. The equipment hatch closed and held in place by [a minimum of four bolts,]

- b. One door in each airlock closed,
- c. Each penetration providing direct access from the containment atmosphere to the outside atmosphere is either:
 - 1. Closed by an isolation valve, blind flange, manual valve, or equivalent, or
 - 2. Capable of being closed by an OPERABLE Containment Purge and Exhaust Isolation System.
- APPLICABILITY: MODE 5 REDUCED INVENTORY and MODE 6 REDUCED INVENTORY

ACTIONS

CONDITION		REQUIRED ACTION	COMPLETION TIME	
Α.	One or more containment penetrations not in the required status.	A.1. Restore Contrinment penetration to required status.	[6 hours]	

Reduced RCS Inveniory Operations - Containment Integrity 3.10.5

ACTIONS (Continued)

CONDITION		R	EQUIKED ACTION	COMPLETION TIME		
В.	Required Action and Completion time not met.	B.1	Restore RCS level to >[EL-117'0"]	[6 hours]		

SURVEILLANCE FEQUIREMENTS

•

	FREQUENCY	
SR 3.10.5.1	Verify each required containment building penetration is in its required status.	[12 hours]
SR 3.10.5.2	Verify the surveillance requirements of 3.9.3.2 are met.	[18 months]

Reduced RCS Inventory Operations - AC Power Availability 3.10.6

3.10 REDUCED RCS INVENTORY OPERATIONS

3.10.6 Reduced RCS Inventory Operations - AC Power Availability

LCO 3.10.6 The following AC Electrical Power Sources shall be OPERAELE.

a. Two independent sources of AC power to each division supplying the Class 1E Distribution System,

AND

b. A Diesel Generator in either division.

APPLICABILITY: MODE 5 REDUCED RCS INVENTORY and MODE 6 REDUCED RCS INVENTORY

ACTIONS

	CONDITION	REQUIRED ACTION	COMPLETION TIME
A.	One source of A.C. Fower to either division INOPERABLE.	A.1. Perform S.R. 3.8.1.1 AND	[1 hour] then every 12 hours
		A.2. Restore division to operable status	[30 hours]

Reduced RCS Inventory Operations - AC Power Availability 3.10.6

ACT	2532	362	10 mars	=1	(bears
mus.	IUL	13 1	1.00	un	uca)

	CONDITION	R!	UIRED ACTION	COMPLETION TIME
В.	One Source of A.C. Power to each division operable	B.1	Restore either division to two operable sources	[12 hours]
C.	Required Diesel Generator INOPERABLE	C.1 <u>AhaD</u>	Perform S.R. 3.8.1.1	[1 hour] then every 12 hours
		C.2	Restore required diesel generator to operable status	[12 hours]
D.	Required Action A, B or C not met within required Completion Time.	D.1	Raise RCS level > [EL-1170"]	[6 hours]

SURVEILLANCE REQUIREMENTS

	SURVEILLANCE					FREQUENCY	
SR 3.10.6.1	Verify 3.8.1.9	diesei generator and 3.8.1.18	operability	per 3.8.1.2,	3.8.1.4,	3.8.1.5,	As specified by applicable SR's

SYSTEM 80+

Reduced RCS Inventory Operations - DC Distribution 3.10.7

3.10 REDUCED RCS INVENTORY OPERATIONS

3.10.7 Reduced RCS Inventory Operations - DC Distribution System

- LCO 3.10.7 The following DC Electric Power Sources shall be OPERABLE
 - a. One Division of the DC Distribution System coinciding with the operable Diesel Generator. And,

Power is available to the opposite division 125 vdc & 120 VAC Distribution Centers.

APPLICABILITY: MODE 5 REDUCED RCS INVENTORY and MODE 6 REDUCED RCS INVENTORY

ACTIONS

	CONDITION		EQUIRED ACTION	COMPLETION TIME
A.	Required DC Power Division INGPERABLE	A.1	Restore to Operable Status	[4 hours]
В.	Opposite division distribution Centers INOPERABLE	B.1	Restore to OPERABLE status	[12 hours]
C.	Required Action A or B not met within required Completion Time.	C.1	Raise RCS level to > [EL-117'0"]	[6 hours]

SYSTEM 80+

3.10-17

Reduced RCS Inventory Operations - DC Distribution 3.10.7

SURVEILLANCE REQUIREMENTS

.

	FREQUENCY	
SR 3.10.7.1	Perform S.R. 3.8.4.1 through 3.8.4.8 on the OPERABLE division of the DC Distribution System	E Per 3.8.4.1 through 3.8.4.8 frequency

3.10-18